_

UNITED STATES

SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Fiscal Year Ended December 31, 2023

or

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Transition Period from to

Commission File Number 001-41063

JOURNEY MEDICAL CORPORATION

(Exact name of registrant as specified in its charter)

Delaware (State or Other Jurisdiction of Incorporation or Organization) 47-1879539

(I.R.S. Employer Identification No.)

9237 E Via de Ventura Blvd., Suite 105 Scottsdale, AZ

(Address of Principal Executive Offices)

85258 (Zip Code)

1 /

Registrant's telephone number, including area code: (480) 434-6670

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, par value \$0.0001 per share	DERM	NASDAQ Capital Market

Securities registered pursuant to section 12(g) of the Act: None.

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Y 🗟 No 🗵

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes No 🛛 Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 🖾 No 🗆

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes 🛛 No 🗆

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, smaller reporting company, or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act:

Large accelerated filer		Accelerated filer	
Non-accelerated filer	\boxtimes	Smaller reporting company	\times
Emerging growth company	\mathbf{X}		

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant has filed a report on and attestation to its management's assessment of the effectiveness of its internal control over financial reporting under Section 404(b) of the Sarbanes-Oxley Act (15 U.S.C. 7262(b)) by the registered public accounting firm that prepared or issued its audit report. Yes 🗆 No 🖂

If securities are registered pursuant to Section 12(b) of the Act, indicate by check mark whether the financial statements of the registrant included in the filing reflect the correction of an error to previously issued financial statements.

Indicate by check mark whether any of those error corrections are restatements that required a recovery analysis of incentive-based compensation received by any of the registrant's executive officers during the relevant recovery period pursuant to §240.10D-1(b).

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Act). Yes No 🗵

The aggregate market value of voting stock held by non-affiliates of the registrant on June 30, 2023, the last business day of the registrant's most recently completed second quarter, wak2252,052 based on the last reported sale price of the registrant's Common Stock on the Nasdaq Capital Market on that date of \$1.59.

Indicate the number of shares outstanding of each of the registrant's classes of common stock, as of the latest practicable date.

Class of Common Stock	Outstanding Shares as of March 28, 2024
Class A Common Stock, \$0.0001 par value	6,000,000
Common Stock, \$0.0001 par value	13,932,310

DOCUMENTS INCORPORATED BY REFERENCE

Portions of the registrant's Proxy Statement for its 2024 Annual Meeting of Stockholders are incorporated by reference in Part III of this Annual Report on Form 10-K.

JOURNEY MEDICAL CORPORATION ANNUAL REPORT ON FORM 10-K TABLE OF CONTENTS

		Page
PART I		6
<u>Item 1.</u>	Business	6
Item 1A.	<u>Risk Factors</u>	26
<u>Item 1B.</u>	Unresolved Staff Comments	61
<u>Item 1C</u> .	Cybersecurity	61
<u>Item 2.</u>	Properties	61
Item 3.	Legal Proceedings	62
Item 4.	Mine Safety Disclosures	62
PART II		63
Item 5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	63
<u>Item 6.</u>	Reserved	63
<u>Item 7.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	64
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	71
Item 8.	Financial Statements and Supplementary Data	71
<u>Item 9.</u>	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	71
Item 9A.	Controls and Procedures	72
Item 9B.	Other Information	73
Item 9C.	Disclosure Regarding Foreign Jurisdictions that Prevent Inspections.	73
PART III		74
Item 10.	Directors, Executive Officers and Corporate Governance	74
Item 11.	Executive Compensation	74
Item 12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	74
Item 13.	Certain Relationships and Related Transactions, and Director Independence	74
Item 14.	Principal Accountant Fees and Services	74
PART IV		75
Item 15.	Exhibits, Financial Statement Schedules	75
Item 16.	Form 10-K Summary	77

SPECIAL CAUTIONARY NOTICE REGARDING FORWARD-LOOKING STATEMENTS

Certain matters discussed in this report may constitute forward-looking statements for purposes of the Securities Act of 1933, as amended (the "Securities Act"), and the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and involve known and unknown risks, uncertainties and other factors that may cause our actual results, performance or achievements to be materially different from the future results, performance or achievements expressed or implied by such forward-looking statements. The words "anticipate," "believe," "estimate," "may," "expect," "will," "could," "project," "should," "intend" and similar expressions are generally intended to identify forward-looking statements. Our actual results may differ materially from the results anticipated in or implied by these forward-looking statements due to a variety of factors, including, without limitation:

- the fact that our products and product candidates are subject to time and cost intensive regulation and clinical testing and as a result may never be successfully developed or commercialized;
- a substantial portion of our sales derive from products that are without patent protection and/or are or may become subject to third-party generic competition, the
 introduction of new competitor products, or an increase in market share of existing competitor products, any of which could have a significant adverse impact on
 our operating income;
- we operate in a heavily regulated industry, and we cannot predict the impact that any future legislation or administrative or executive action may have on our operations;
- our revenue is dependent mainly upon sales of our dermatology products and any setback relating to the sale of such products could impair our operating results;
- competition could limit our products' commercial opportunity and profitability, including competition from manufacturers of generic versions of our products;
- the risk that our products do not achieve broad market acceptance, including by government and third-party payors;
- our reliance on third parties for several aspects of our operations;
- our dependence on our ability to identify, develop, and acquire or in-license products and integrate them into our operations, at which we may be unsuccessful;
- the dependence of the success of our business, including our ability to finance our company and generate additional revenue, on the successful development and
 regulatory approval of the DFD-29 product candidate and any future product candidates that we may develop, in-license or acquire;
- clinical drug development is very expensive, time consuming, and uncertain and our clinical trials may fail to adequately demonstrate the safety and efficacy of our current or any future product candidates;
- our competitors could develop and commercialize products similar or identical to ours;
- risks related to the protection of our intellectual property and our potential inability to maintain sufficient patent protection for our technology and products;
- our business and operations would suffer in the event of computer system failures, cyber-attacks, or deficiencies in our our third parties' cybersecurity;
- the effects of major public health issues, epidemics or pandemics on our product revenues and any future clinical trials;
- our potential need to raise additional capital;
- the substantial doubt expressed about our ability to continue as a going concern;
- · Fortress controls a voting majority of our common stock, which could be detrimental to our other shareholders; and
- the risks described under the section titled "Risk Factors" in this Annual Report.

The forward-looking statements contained in this report reflect our views and assumptions as of the effective date of this report. New risks and uncertainties arise from time to time, and it is impossible for us to predict these events or how they may affect us. Except as required by law, we assume no responsibility for updating any forward-looking statements.

We qualify all of our forward-looking statements by these cautionary statements. In addition, with respect to all of our forward-looking statements, we claim the protection of the safe harbor for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995.

SUMMARY OF RISK FACTORS

Our business is subject to a number of risks which you should be aware of before making an investment decision. The risks described below are a summary of the principal risks associated with an investment in our common stock and are not the only risks we face. These risks are more fully described in the section titled "Risk Factors" of this report on Form 10-K and include the following:

Risks Related to Our Business, Industry and Existing Operating Revenue Stream

- Our products and product candidates are subject to time and cost intensive regulation and clinical testing. As a result, they may never be successfully developed or commercialized. Further, any approved product may be subject to post-marketing requirements, including studies or clinical trials, the results of which could cause such product to be withdrawn from the market.
- A substantial portion of our sales derive from products that are without patent protection and/or are or may become subject to third-party generic competition, the introduction of new competitor products, or an increase in market share of existing competitor products, any of which could have a significant adverse impact on our operating income.
- We operate in a heavily regulated industry, and we cannot predict the impact that any future legislation or administrative or executive action may have on our operations.
- Our revenue is dependent mainly upon sales of our dermatology products and any setback relating to the sale of such products could impair our operating results.
- Our competitors may develop treatments for our products' target indications, which could limit our products' commercial opportunity and profitability.

Risks Related to Our Reliance on Third Parties

• We rely on third parties for several aspects of our operations, which limits our control over product development, marketing, manufacturing, and sale processes and may hinder our ability to develop and commercialize our products in a cost-effective and timely manner.

Risks Related to Our Growth

- Our future growth may depend on our ability to identify, develop, and acquire or in-license products and integrate them into our operations, at which we may be unsuccessful.
- We may expend resources on unsuccessful product candidates or indications and may fail to capitalize on more profitable or successful product candidates or indications.
- There is substantial doubt regarding our ability to continue as a going concern. We may need to raise additional funding (which may not be available on acceptable terms to the Company, or at all) and/or to delay, limit or terminate certain of our product development and commercialization efforts or other operations.

Risks Related to Development and Regulatory Approval of Our Product Candidates (DFD-29)

- The success of our business, including our ability to finance our company and generate additional revenue, may depend on the successful development and regulatory approval of the DFD-29 product candidate and any future product candidates that we may develop, in-license or acquire.
- Clinical drug development is very expensive, time consuming, and uncertain. Our clinical trials may fail to adequately demonstrate the safety and efficacy of our current or any future product candidates, which could prevent or delay regulatory approval and commercialization.

• We expect to rely on third-party contract research organizations ("CROs") and other third parties to conduct and oversee our clinical trials, other aspects of our product development and our regulatory submission process for our product candidates. If these third parties do not meet our requirements, conduct the trials as required or otherwise provide services as anticipated, we may not be able to satisfy our contractual obligations or obtain regulatory approval for, or successfully commercialize, our current or any future product candidates when expected or at all.

Risks Pertaining to Intellectual Property, Generic Competition and Paragraph IV Litigation

- If we are unable to maintain sufficient patent protection for our technology and products, our competitors could develop and commercialize products similar or identical to ours.
- Any dispute with our licensors may affect our ability to develop or commercialize our product candidates.
- Generic drug companies may submit applications seeking approval to market generic versions of our products.
- In connection with these applications, generic drug companies may seek to challenge the validity and enforceability of our patents through litigation and/or with the United States Patent and Trademark Office ("USPTO"). Such challenges may subject us to costly and time-consuming litigation and/or USPTO proceedings.
- As a result of the loss of any patent protection from such litigation or USPTO proceedings, or the "at-risk" launch by a generic competitor of our
 products, our products could be sold at significantly lower prices, and we could lose a significant portion of sales of that product in a short period of
 time, which could adversely affect our business, financial condition, operating results and prospects.

Risks Related to our Platform and Data

• Our business and operations would suffer in the event of computer system failures, cyber-attacks, or deficiencies in our or our third parties' cybersecurity.

Risks Related to Our Finances and Capital Requirements

- Due to the numerous risks and uncertainties associated with pharmaceutical product development, we may incur losses and may be unable to maintain profitability.
- If we are unable to raise capital as needed, we may be forced to delay, reduce, or eliminate our operations.

Risks Relating to Owning our Common Stock

• Our operating results have fluctuated in the past and we expect them to continue to do so. Any such fluctuation may cause our performance to fall below expectations, and our stock price may suffer.

Risks Related to our Relationship with Fortress Biotech, Inc. ("Fortress")

- Fortress controls a voting majority of our common stock, which could be detrimental to our other shareholders. Further, Fortress' ownership qualifies us as a "controlled company" under the Nasdaq listing standards.
- Fortress' financial obligations and any potential risk of default may adversely affect the Company or constrain our ability to take certain actions.

PART I

Item 1. Business

OVERVIEW

We are a commercial-stage pharmaceutical company founded in October 2014 that focuses on the development and commercialization of pharmaceutical products for the treatment of dermatological conditions. Our current portfolio includes seven branded and two authorized generic prescription drugs for dermatological conditions that are marketed in the U.S. We are managed by experienced life science executives with a track record of creating value for their stakeholders and bringing novel medicines to the market, enabling patients to experience increased quality of life, and enabling physicians and other licensed medical professionals to provide better care for their patients. We aim to acquire rights to future products by licensing or otherwise acquiring an ownership interest in, funding the research and development of, and eventually commercializing, these products through our field sales organization. Since inception, we have made significant investments to build out our commercial product portfolios, which we believe, coupled with our experienced dermatology sales leadership team and our seasoned field sales force, will position our business for growth. We are a majority-owned subsidiary of Fortress.

2023 Highlights and Events

On December 27, 2023, we entered into a Credit Agreement (the "Credit Agreement") with SWK Funding LLC ("SWK"). The Credit Agreement provides for a term loan facility (the "Credit Facility") in the original principal amount of up to \$20.0 million. On the closing date, we drew \$15.0 million. The remaining \$5.0 million may be drawn upon our request within 12 months after the closing date. Loans under the Credit Facility (the "Term Loans") mature on December 27, 2027 unless the Credit Facility is otherwise terminated pursuant to the terms of the Credit Agreement and bear interest at a rate per annum equal to the three-month term Secured Overnight Financing Rate ("SOFR") (subject to a SOFR floor of 5%) plus 7.75%. The interest rate resets quarterly. Interest payments begin in February 2024 and are paid quarterly. Beginning in February 2026, we are required to repay the outstanding principal of the Term Loans of the December 31, 2025, principal repayment is not required until February 2027, at which point we are required to repay a portion of the outstanding principal of the Term Loans quarterly in an amount of the Term Loans quarterly in an amount of funded Term Loans.

On August 31, 2023, we entered into a license agreement (the "New License Agreement") with Maruho Co., Ltd., a Japanese company specializing in dermatology ("Maruho"), whereby we granted an exclusive license to Maruho to develop and commercialize Qbrexza® for the treatment of primary axillary hyperhidrosis in South Korea, Taiwan, Hong Kong, Macau, Thailand, Indonesia, Malaysia, Philippines, Singapore, Vietnam, Brunei, Cambodia, Myanmar and Laos (the "Territory"). Under the terms of the New License Agreement, Maruho paid us \$19.0 million as a non-refundable upfront payment. Maruho is also obligated to make royalty payments to us related to sales of the product in the Territory equal to the corresponding rate payable by us to Dermira under the asset purchase agreement between us and Dermira.

In July 2023, we announced positive topline data from our two DFD-29 Phase 3 clinical trials for the treatment of papulopustular rosacea. The Phase 3 clinical trials achieved the co-primary and all secondary endpoints, the subjects completed the 16-week treatment and the drug was well-tolerated. DFD-29 demonstrated statistical superiority over both the standard of care, Oracea® capsules, and placebo for Investigator's Global Assessment treatment success and the reduction in the total inflammatory lesion count in both studies. We summitted a New Drug Application ("NDA") under Section 505(b)(2) of the United States Federal Food, Drug and Cosmetic Act ("FDCA") with the U.S. Food and Drug Administration (the "FDA") for DFD-29 on January 4, 2024, paying a \$4.0 million filing fee, and expect potential approval from the FDA in the second half of 2024. On March 18, 2024, we announced the FDA accepted the Company's NDA with a Prescription Drug User Fee Act goal date of November 4, 2024.

CORPORATE INFORMATION

Journey Medical Corporation was incorporated in Delaware in 2014. Our executive offices are located at 9237 E Via de Ventura Blvd. Suite 105, Scottsdale, AZ 85258. Our telephone number is 480-434-6670, and our e-mail address is info@jmcderm.com or ir@jmcderm.com.

We maintain a website with the address www.jmcderm.com. We make available free of charge through our website our annual reports on Form 10-K, quarterly reports on Form 10-Q and current reports on Form 8-K, and any amendments to these reports, as soon as reasonably practicable after we electronically file such material with, or furnish such material to, the SEC. We are not including the information on our website as a part of, nor incorporating it by reference into, this report. Additionally, the SEC maintains a website that contains annual, quarterly, and current reports, proxy statements, and other information that issuers (including us) file electronically with the SEC. The SEC's website address is http://www.sec.gov.

Our Market, Products and Relevant Disease States

As of December 31, 2023, our major actively marketed products, which have been approved by the FDA for sale in the United States, include:

- Qbrexza® (a medicated cloth towelette for the treatment of primary axillary hyperhidrosis), acquired and launched in May 2021;
- Accutane® (an oral isotretinoin drug for the treatment of severe recalcitrant nodular acne), licensed in July 2020 and launched in March 2021;
- Amzeeq® (minocycline) topical foam, 4% (a topical formulation of minocycline for the treatment of inflammatory lesions of non-nodular moderate to severe acne vulgaris in adults and children nine years and older), acquired and launched in January 2022;
- Zilxi® (minocycline) topical foam, 1.5% (a topical minocycline treatment for inflammatory lesions of rosacea in adults), acquired and launched in January 2022;
- Exelderm® Cream and Solution (a broad-spectrum antifungal intended for topical use), acquired and launched in October 2018;
- Targadox® (an oral doxycycline drug for adjunctive therapy for severe acne), licensed in March 2015 and launched in October 2016; and
- Luxamend® (a water-based emulsion formulated to provide an optimally moist healing environment for superficial wounds; minor cuts or scrapes; dermal ulcers; donor sites; first- and second-degree burns, including sunburns; and radiation dermatitis), acquired in 2021 and launched in 2023.

Additionally, we sell two authorized generic products:

- sulconazole nitrate cream and solution, 1% antifungal agents indicated for the treatment of *tinea cruris* and *tinea corporis* caused by *Trichophyton rubrum*, *Trichophyton mentagrophytes*, *Epidermophyton floccosum*, and *Microsporum canis*,* and for the treatment of *tinea versicolor*. *Efficacy for this organism in the organ system was studied in fewer than 10 infections. EXELDERM[®] Cream is also indicated for the treatment of *tinea pedis* (athlete's foot). Effectiveness of EXELDERM[®] Solution has not been proven in *tinea pedis*. These products were launched in January 2020; and
- doxycycline hyclate immediate release 50mg tablets, indicated as adjunctive therapy for severe acne to reduce the development of drug-resistant bacteria as well as to maintain the effectiveness of doxycycline hyclate and other antibacterial drugs, launched in May 2018.

Our Products and Relevant Disease States

Excessive Underarm Sweating and the Current Standard of Care

Excessive underarm sweating, commonly referred to as primary axillary hyperhidrosis ("PAH"), is a rare disorder characterized by excessive sweating in the armpits. The exact cause of PAH is not known, and the disorder affects males and females equally. When excessive sweating occurs as part of some other disorder, it is said to be secondary hyperhidrosis, which is a more commonly encountered condition than is primary hyperhidrosis. According to a 2016 article published in the Archives of Dermatological Research, there are approximately 10 million people who suffer from PAH in the United States. The symptoms of PAH typically begin during childhood or puberty and may often, although not always, persist throughout a person's life. Affected individuals may experience a heightened reaction to certain stimuli that can cause sweating such as anxiety, pain, exercise, tension, caffeine, and/or nicotine. The symptoms of this disorder develop due to overactivity of certain sweat glands, and incidences may be precipitated by social and/or physical stress. Some people with PAH experience relief from the symptoms during adulthood without treatment or obvious reason for the remission.

Pharmacological treatment options for PAH include topical, oral and iontophoretic treatments.

Qbrexza® (glycopyrronium 2.4% cloth) for the Treatment of Primary Axillary Hyperhidrosis

Our Qbrexza® (glycopyrronium 2.4%) product is a topical, once-daily anticholinergic cloth that was approved by the FDA in June 2018 for the treatment of PAH in adult and pediatric patients nine years of age and older. Anticholinergics are a class of pharmaceutical products that exert their effect by blocking the action of acetylcholine, a neurotransmitter that transmits signals within the nervous system that are responsible for the activation of sweat glands. Qbrexza is applied directly to the skin and is designed to block underarm sweat production by inhibiting sweat gland activation. Qbrexza has Orange Booklisted patents that extend through February of 2033.

The PAH market had approximately 450,000 prescriptions in 2023 according to Symphony Health, excluding over-the-counter ("OTC") clinical strength anti-perspirants.

Acne and the Current Standard of Care

Acne, also known as acne vulgaris, is a common skin disorder characterized by a blockage of hair follicles, which are clogged with oil and dead skin cells. According to the American Academy of Dermatology ("AAD"), acne is the most common skin condition in the US, affecting up to 50 million individuals annually.

Approximately 85% of people between the ages of 12 and 24 experience at least a minor form of acne. The disease is classified as mild, moderate or severe based on the severity of the disease progression, which is useful in identifying an appropriate treatment regimen. Mild acne is characterized by clogged hair follicles (known as comedones) that are either exposed to air (blackheads) or closed (whiteheads), with occasional inflammatory lesions which occur primarily on the face. Moderate acne is characterized by a higher presence of inflammatory lesions known as papules and pustules across the face and extending to the trunk. Severe acne is characterized by painful, deep lesions called nodules across the face, with extensive involvement of the trunk frequently.

Treatment options are based on the severity of disease, with certain drugs being reserved for more severe forms of the disease. Mild acne is addressed with dietary and lifestyle changes, along with OTC and prescription topical agents. Other therapies with varying degrees of success include dermabrasion and chemical peels, light therapy and hormonal therapy such as birth control pills or spironolactone. Moderate acne is treated with more aggressive therapy including topical and oral antibiotics such as tetracyclines, which are particularly effective due to their antibacterial and anti-inflammatory properties, and other topical agents including benzoyl peroxide and retinoids. Severe acne is treated with combination therapies, often including oral antibiotics. For resistant cases, physicians may use a potent drug known as isotretinoin (a vitamin A analog), which requires Risk Evaluation and Mitigation Strategy ("REMS") (safety) monitoring with regard to pregnancy. The current U.S. market size for treatment of acne is considerable and estimated at approximately \$3 billion annually, according to the American Medical Association.

Accutane® for the Treatment of Severe Recalcitrant Nodular Acne

Accutane® (isotretinoin 10mg, 20mg, 30mg, and 40mg capsules USP) is indicated for the treatment of severe recalcitrant nodular acne. Accutane is used to treat a type of severe recalcitrant nodular acne that has not been helped by other treatments, including antibiotics. Severe recalcitrant nodular acne occurs when many red, swollen, tender lumps form in the skin. Patients with severe nodular acne are at higher risk of scarring. Accutane belongs to a class of drugs that affects all four major pathogenic processes in acne: increased sebum production, irregular follicular desquamation, propionibacterium acnes proliferation and inflammation. Accutane has achieved a strong market position and is well known in the dermatology community.

The oral isotretinoin market had over 2 million prescriptions in 2023 according to Symphony Health.

Targadox® for the Treatment of Severe Acne

Targadox® (doxycycline hyclate immediate release 50mg tablets) is indicated as adjunctive therapy for severe acne, which is part of a class of oral antibiotics known as tetracyclines. The tetracycline class, which includes minocycline, doxycycline, sarecycline and tetracycline, is particularly effective in treatment for more severe forms of acne due to its antibiacterial and anti-inflammatory properties. Targadox is gluten-free, lactose-free, animal byproduct-free, and GMO-free.

The oral doxycycline market had more than 27 million prescriptions in 2023 according to Symphony Health.

Amzeeq® for the Treatment of Moderate-to-Severe Acne

Amzeeq® (4% minocycline foam), formerly known as FMX101, was approved by the FDA in October 2019 and became available in pharmacies nationwide in January 2020. Amzeeq is a once-daily novel topical antibiotic foam formulation of minocycline for the treatment of inflammatory lesions of non-nodular moderate-to-severe acne vulgaris in patients nine years of age and older. Amzeeq utilizes proprietary MSTTM technology and is the first topical minocycline to be approved by the FDA for any condition. We believe that the combination of a well-established antibiotic in a well-tolerated, easy to use foam makes Amzeeq a very attractive treatment option for patients.

The topical acne market had almost 21 million prescriptions in 2023 according to Symphony Health, presenting significant unmet needs of patients and healthcare providers to be addressed. As the first topical minocycline to be approved by the FDA for any condition, we believe that Amzeeq may provide a new treatment alternative for patients and healthcare providers who are unsatisfied with their current therapies. Amzeeq has Orange Book-listed patents that extend through September of 2037.

Ximino® for the Treatment of Inflammatory Lesions of Non-Nodular Moderate to Severe Acne

During fiscal 2022 and part of fiscal 2023, we marketed and sold Ximino® (minocycline hydrochloride extended-release 45mg, 90mg, and 135mg capsules), which is indicated for the treatment of inflammatory non-nodular lesions (pimples and red bumps) associated with moderate to severe acne. We discontinued selling Ximino in September 2023.

The oral minocycline market had just under 2.7 million prescriptions in 2023 according to Symphony Health.

Fungal Infections of the Skin and the Current Standard of Care

Fungal skin infections, collectively referred to as dermatomycoses, are common infections caused by ringworms (tinea) and include conditions such as athlete's foot, jock itch and ringworm of the body. Tinea pedis, commonly known as athlete's foot, is a form of ringworm that usually develops between the toes. Symptoms include peeling, cracking and scaly feet, blisters, and skin that is red, softened, itching, or burning. Tinea cruris, commonly known as jock itch, is a form of ringworm that affects the groin. Tinea corporis, commonly known as ringworm of the body, is a fungal infection that appears on the body in which the outer part of the sore might be raised while the skin in the middle appears normal. Fungal infections caused by ringworm cause skin rashes that present as itchy, red, raised and scaly rings. These infections are easily transmissible between people, pets or contaminated objects or surfaces but are usually not serious in nature.



Treatment options typically involve topical OTC and prescription antifungal medications. Where difficult to administer topically, oral options (such as for toenail fungus or oral thrush) or suppositories (such as for vaginal yeast infections) have proven to be more effective. OTC products typically include known antifungal ingredients such as clotrimazole, miconazole, terbinafine or ketoconazole. Prescription treatments are often reserved for more serious infection or for those in hard-to-treat areas. In conjunction with OTC or prescription medications, lifestyle adjustments, including daily washing of bedding and clothing during an infection, drying thoroughly after bathing, wearing loose clothing in affected areas and actively treating infected areas, can all contribute to disinfecting your surroundings and preventing a prolongation or recurrence of infection.

Exelderm® for the Treatment of Fungal Skin Infections

Exelderm® (sulconazole nitrate 1%, cream and solution) is a broad-spectrum antifungal agent indicated for the treatment of ringworm-caused fungal infections including tinea pedis, tinea cruris, tinea corporis and tinea versicolor. The active pharmaceutical ingredient (sulconazole) acts by inhibiting fungal cell division and growth and has been shown to have broad activity against candida species, aspergillus species and dermatophytes. Exelderm cream or solution is administered externally only, whereby a small amount of cream or solution is gently massaged into the affected and surrounding areas and only requires a convenient once or twice daily application. However, when used to treat tinea pedis, for which Exelderm cream is also indicated, twice daily application is required.

The topical antifungal market had more than 11 million prescriptions in 2023 according to Symphony Health.

Pruritus (Itch) and the Current Standard of Care

Pruritus or itch is defined as an unpleasant sensation of the skin that provokes the urge to scratch. It is a characteristic feature of many skin diseases and an unusual sign of some systemic diseases. Pruritus may be localized or generalized and can occur as an acute or chronic condition. Itch can be caused by a number of conditions, including skin conditions such as dry skin, eczema, psoriasis, scabies, parasites, burns, scars, insect bites and hives. Depending on the cause of itchiness, skin may appear normal, red, rough or bumpy. Repeated scratching can cause raised thick areas of skin that might bleed or become infected.

Treatment for itch may include moisturizing daily, using gentle cleansers, and bathing with lukewarm water. Long-term relief requires identifying and treating the underlying cause of itchy skin. Common treatments are prescription medicated creams and lotions, moist dressings, and oral anti-itch medicines.

Anti-Itch Product for the Treatment of Pruritus

Our acquired anti-itch product is indicated to treat pruritis, scabies, and other skin itch conditions ("Anti-itch Product"). Our Anti-itch Product delivers prescription relief and is non-steroidal and antihistamine free. Topical steroids are effective against itch because they reduce inflammation that can cause itch. However, they are not recommended for long-term use. Antihistamines are also effective in treating some types of itch, but they too have drawbacks with continued use. We plan on launching our Anti-itch Product through our field sales force during the second half of 2024 or first half of 2025.

Rosacea and the Current Standard of Care

Rosacea is a chronic, relapsing, inflammatory skin condition that most commonly presents with symptoms such as deep facial redness, acne-like inflammatory lesions (papules and pustules) and spider veins (telangiectasia). According to The National Rosacea Society, it is estimated that rosacea affects well over 16 million Americans (F1000Research 2018, 7(F1000 Faculty Rev):1885) and as many as 415 million people worldwide. Rosacea is most frequently seen in adults between 30 and 50 years of age. Surveys conducted by The National Rosacea Society report more than 90% of rosacea patients said their condition had lowered their self-confidence and self-esteem, and 41% reported that it had caused them to avoid public contact or cancel social engagements. Among rosacea patients with severe symptoms, 88% said the disorder had adversely affected their professional interactions, and 51% said they had missed work because of their condition. The rosacea market had 3.8 million prescriptions in 2023 according to Symphony Health.

The tetracycline class of antibiotics (minocycline and doxycycline) are considered to be effective options for the treatment of papulopustular rosacea, likely due to anti-inflammatory activities that are usually manifested at doses much lower than those prescribed for treatment of bacterial infections. A low dose of doxycycline (i.e., 40 mg taken once daily) as oral formulation has been approved for the treatment of only inflammatory lesions (papules and pustules) of rosacea and is available under the proprietary name Oracea® in the US. Oracea is generally considered to be the current standard of care. Minocycline, first introduced in 1971, is widely believed to be the most effective tetracycline agent due to its high lipophilicity, which is anticipated to permit greater permeation into, and accumulation in, the sebaceous follicles and layers of the epidermis. We offer two products, Zilxi and, if approved, DFD-29, that we believe provide a new treatment alternative for patients and healthcare providers who are unsatisfied with their current rosacea therapies.

Zilxi® for the Treatment of Papulopustular Rosacea

Zilxi® (1.5% minocycline foam), was approved by the FDA in May 2020 and became available in pharmacies nationwide in October 2020. Zilxi is a oncedaily novel antibiotic foam formulation of minocycline for the treatment of inflammatory lesions of rosacea in adults. Similar to Amzeeq, Zilxi leverages MSTTM technology and is the first minocycline product of any form to be approved by the FDA for use in rosacea. We believe the anti-inflammatory properties of minocycline delivered in our innovative foam technology make Zilxi a highly appealing treatment option for rosacea patients. Zilxi has Orange Book-listed patents that extend through October of 2030.

DFD-29 for the Treatment of Rosacea

DFD-29 is a low-dose minocycline (40 mg) extended release capsule formulation for oral use for the treatment of papulopustular rosacea. The rationale of selecting DFD-29 doses lower than the approved minocycline dose is based on the lower protein binding and higher lipophilicity of minocycline. In a Phase 1 PK study (DFD-29-CD-001) in 24 healthy subjects, the systemic exposure of minocycline from DFD-29 (minocycline HCl) ER capsules 40 mg was much lower than that seen with the approved antibiotic dose of minocycline, although this was not a head-to-head study. A Phase 2 study (DFD-29-CD-002) in 205 subjects with papulopustular rosacea, demonstrated that DFD-29 (40 mg) was significantly superior to placebo and Oracea®, on the co-primary endpoints of IGA treatment success and absolute inflammatory lesion count reduction. The study also showed DFD-29 was well-tolerated. DFD-29 has also shown superiority to Oracea and Placebo on the co-primary endpoints and all secondary endpoints in two phase 3 studies and was well-tolerated. The NDA was filed under Section 505(b)(2) of the FDCA, in January 2024 and is under review by the FDA.

Luxamend® for Wound Healing

Luxamend® is a water-based wound cream formulated for the dressing and management of superficial wounds; minor abrasions; dermal ulcers; donor sites; 1st and 2nd degree burns, including sunburns; and radiation dermatitis. Luxamend contains purified water, white mineral oil, ethylene glycol monostearate, stearic acid, propylene glycol, paraffin wax, squalane, avocado oil, trolamine/sodium alginate, triethanolamine, cetyl palmitate, sodium sulfate (anhydrous), potassium sorbate, methylparaben sodium, propylparaben sodium, sodium hexametaphosphate, sulfamic acid, and allergen-free fragrance. When applied properly to a wound, Luxamend provides an optimum moist environment for the healing process. It is approved as a prescription medical device and is supplied in a 114 gram tube.

Our Strategy

We are a highly focused, pharmaceutical company dedicated to developing and commercializing therapies for the treatment of dermatologic conditions that seeks to deliver value to patients, physicians and the healthcare system, as well as to our stakeholders. Our strategic priorities include continuing to augment and grow our product portfolio and organization in order to maximize long-term value creation. This will consist of both commercial execution on our existing product portfolio, including lifecycle management, out-licensing of our current branded products and/or technologies in global markets, as well as investing in additional growth strategies through product and company acquisitions, licensing, or developing new products.

An important part of our growth strategy is to identify new business development opportunities, including development stage and commercial drugs that we may acquire from other pharmaceutical companies. We are in various stages of discussion for other opportunities, both commercial and development stage, that could drive additional growth in the business. Successful development and commercialization of any future in-licensed development stage or commercial drugs will require us to navigate the many laws and regulations of governmental authorities and regulatory agencies around the world, including the FDA, relating to the manufacture, development, approval and commercialization of investigational drugs. For development stage drugs, we may require financial resources significantly in excess of our current cash on hand and amounts that we may borrow under our Credit Facility, and it may take many years for us to receive marketing approval, if ever, for any in-licensed or acquired product candidate.



Competitive Strengths

To successfully execute our strategy, we must continue to capitalize on our following core strengths:

- Commercial leadership of our management team with a track record of commercial execution. We have a highly skilled and customer-focused management team in critical leadership positions across our Company. Our senior management team has over 135 years of collective sales and marketing experience in the pharmaceutical industry and a proven track record of developing businesses and creating value. Members of our management team have developed, launched, commercialized, and managed brands generating over \$3 billion in aggregate peak sales, collectively, at leading dermatology organizations. This experience includes improving business performance through organic revenue growth, maximizing operational efficiencies and through the identification, consummation and integration of licensing and acquisition opportunities. Our senior management team has extensive roots in the dermatology industry, with many of them having worked at and held senior positions with Medicis Pharmaceutical, Inc. leading up to the company's acquisition by Valeant Pharmaceuticals, Inc. (now Bausch Health Pharmaceuticals, Inc.) for \$2.6 billion in 2012. Our strategic approach leverages our management team's experience with the capabilities of our field sales force to drive performance based on prescribing habits, brand preferences, promotional strategies and profit optimization while focusing on customer service excellence for our providers and their patients.
- Performance and experience of our accomplished field sales force. Our current seasoned field sales force has deep-rooted and longstanding customer relationships in their respective territories. We have strategically optimized our sales outreach to cover over 80% of dermatologists in the top 50 U.S. metropolitan statistical areas and over 70% of the overall dermatology prescribing market. We are able to leverage the experience of our field sales force to create a tailored and entrepreneurial compensation plan that incentivizes our field sales force and aligns their activities with our corporate performance and growth objectives. We intend to continue to build a team of committed, experienced employees and to engage with patients and members of the dermatology community. Additionally, we believe that consolidation in the medical dermatology industry has resulted in an enhanced opportunity for a medical dermatology-focused company to build relationships with these stakeholders and has made available a large and growing talent pool of experienced individuals who can make significant contributions to our Company.
- Specialized and differentiated access and distribution model. We have a specialized and differentiated access and distribution network of over 600
 specialty pharmacies and wholesalers, where we directly sell our products, with limited distribution through traditional national wholesalers. This
 decentralized approach allows us to maximize our brand equity across our product portfolio through strategic relationships directly with pharmacies
 and allows us to provide exceptional customer service and access to patients and physicians.
- Active business development initiative. Business development plays a vital role in our growth strategy as we look to build scale. We consistently evaluate both synergistic acquisitions that leverage our existing infrastructure, as well as more transformative assets that would require building out or restructuring our field sales force. We have extensive relationships in the industry that help us stay abreast of developments in our space and continually monitor new opportunities. We believe that we are an ideal partner for development stage companies with limited or no commercial capabilities, as well as established pharmaceutical companies looking to deprioritize their dermatology portfolio. We regularly engage in discussions with an array of companies, including traditional large pharma, mid-size specialty pharma companies and smaller companies that focus on research and development, although we have not entered into any definitive agreements or arrangements. Another important part of our business development strategy is to continue to out-license our branded products and/or proprietary technologies in global markets.

Major Customers

We primarily sell our prescription products to specialty pharmacies, independent wholesalers, and distributors with limited sales through the traditional national wholesaler channels. Our wholesalers and distributors purchase products from us and, in turn, supply products to retail drug store chains, independent pharmacies and managed care organizations. Customers in the managed care market include health maintenance organizations, group purchasing organizations, nursing homes, clinics, pharmacy benefit management companies and mail order customers.

License & Collaboration Agreements and Acquisitions

We continue to seek to enhance our product line and develop a balanced portfolio of differentiated products through product acquisitions and in-licensing or acquiring rights to products and technologies from third parties. We intend to enter into strategic alliances and collaborative arrangements with third parties, which will give us rights to develop, manufacture, market and/or commercialize pharmaceutical products, the rights to which are primarily owned by these third parties. These alliances and arrangements can take many forms, including licensing arrangements, co-development and co-marketing agreements, co-promotion arrangements, research collaborative arrangements. Such alliances and arrangements will potentially enable us to share the risk of incurring all research and development expenses that do not lead to revenue-generating products. However, because profits from alliance products are shared with the counterparties to the collaborative arrangement, the gross margins on alliance products are generally lower, sometimes substantially so, than the gross margins that could be achieved had we not opted for a development partner. From time to time, we may also seek to grant licenses of rights to develop, sell and distribute our products to third parties in exchange for the payment of license fees and/or royalty payments.

Environmental Matters

Our operations are subject to substantial federal, state and local environmental laws and regulations concerning, among other matters, the generation, handling, storage, transportation, treatment and disposal of, and exposure to, hazardous substances. Violation of these laws and regulations, which may change, can lead to substantial fines and penalties. Many of our third-party operations require environmental permits and controls to prevent and limit pollution of the environment. We believe that the facilities of our third-party service providers are in substantial compliance with applicable environmental laws and regulations, and we do not believe that future compliance costs will have a material adverse effect on our business, financial condition, results of operations or cash flows.

Employees and Human Capital Management

We currently employ 58 individuals, all of whom are full-time employees. We have 41 employees in sales and marketing, 14 employees in general and administrative positions, and 3 employees in research and development positions. Additionally, we have retained a number of expert advisors and consultants that help us navigate through different aspects of our business. We consider our relations with our employees to be good and have not experienced any work stoppages, slowdowns or other serious labor problems that have materially impeded our business operations.

Our human capital management objectives include, as applicable, identifying, recruiting, retaining, incentivizing, and integrating our new and existing employees. The principal purpose of our equity incentive plan is to attract, retain, and motivate selected employees, consultants, and directors through the granting of share-based compensation awards and cash-based bonus awards.

Geographic Areas

All of our product revenues are generated from operations or otherwise earned within the U.S. We are entitled to receive commercial milestones payments from Maruho, our exclusive licensing partner in Japan, based on certain net sales achievements for Rapifort® Wipes 2.5% (Qbrexza®), for the treatment of primary axillary hyperhidrosis. We also received a one-time upfront license payment totaling \$19.0 million from Maruho during 2023 under the New License Agreement in which we granted Maruho exclusive rights to Qbrexza in Korea and other Asian countries.

Seasonality of Business

Our business is affected by the standard annual insurance deductible resets, as well as the purchasing patterns and concentration of our customers; however, our business is not materially impacted by seasonality. There are no assurances that these historical trends will continue in the future.

Relationship with Fortress

General

Fortress is a biopharmaceutical company dedicated to acquiring, developing and commercializing pharmaceutical and biotechnology products and product candidates at its majority-owned and majority-controlled subsidiaries and joint ventures, and at entities founded by Fortress and in which it maintains significant minority ownership positions. Fortress has a talented and experienced business development team, comprised of scientists, doctors, and finance professionals, who identify, evaluate, and propose for our consideration promising products and product candidates. We have a nine-year operating history and we are a majority owned subsidiary of Fortress.

Product Licensing Agreements and Asset Acquisitions

Rapifort® Wipes 2.5% (Qbrexza)

On August 31, 2023, we entered into the New License Agreement with Maruho, whereby we granted an exclusive license to Maruho to develop and commercialize Qbrexza® for the treatment of primary axillary hyperhidrosis in the Territory. Under the terms of the New License Agreement, in exchange for the exclusive rights to Qbrexza in the Territory, Maruho paid \$19.0 million as a non-refundable upfront payment.

On February 11, 2022, we announced that Maruho received marketing and manufacturing approval for Rapifort Wipes 2.5% (Qbrexza), for the treatment of primary axillary hyperhidrosis, triggering a net \$2.5 million milestone payment to us. The net payment reflected a milestone payment of \$10.0 million to us from our exclusive licensing partner in Maruho, offset by a \$7.5 million payment to Dermira, pursuant to the terms of the asset purchase agreement between us and Dermira. In conjunction with the terms of the licensing agreement with Maruho, the milestone payment was due from Maruho within 30 days of the approval. We acquired global rights to Qbrexza from Dermira in 2021.

Amzeeq, Zilxi, FCD105 and the Molecule Stabilizing Technology Platform

On January 12, 2022, we entered into an Asset Purchase Agreement (the "APA") with Vyne Therapeutics Inc. ("Vyne") to acquire Vyne's Molecule Stabilizing TechnologyTM franchise (the "Acquisition") for an upfront payment of \$20.0 million, with an additional \$5.0 million payment due on the one-year anniversary of the closing of the Acquisition. The APA also provides for contingent net sales milestone payments: in the first calendar year in which annual sales reach each of \$100 million, \$200 million, \$400 million, \$400 million, we will be required to make a one-time payment of \$10 million, \$20 million, \$30 million, and \$50 million, per product, totaling up to \$450.0 million. In addition, Journey will pay Vyne 10% of any upfront payment received by Journey from a licensee of the products in any territory outside of the United States, subject to exceptions for certain jurisdictions as detailed in the APA. There are no subsequent milestone payments or royalties beyond the aforementioned payments. The Acquisition included two FDA-approved products (Amzeeq® and Zilxi®), and a development-stage dermatology program (FCD105), along with the Molecule Stabilizing Technology proprietary platform.



DFD-29 Agreement

On June 29, 2021, we entered into a license, collaboration, and assignment agreement with Dr. Reddy's Laboratories, Ltd. ("DRL") to obtain the global rights for the development and commercialization of DFD-29, a late-stage development modified release oral minocycline that is being evaluated for the treatment of inflammatory lesions of rosacea (the "DFD-29 Agreement"). We acquired global rights to DFD-29, including in the U.S. and Europe, except that DRL has retained certain rights to the program in select markets including Brazil, Russia, India, China and the Commonwealth of Independent States ("CIS") countries. Pursuant to the DFD-29 Agreement, we agreed to make an upfront payment of \$10.0 million, comprised of a \$2.0 million payment upon execution and \$8.0 million which was paid on September 29, 2021, 90 days following execution, with additional contingent regulatory, commercial, and corporate-based milestone payments, totaling up to \$158.0 million. Royalties ranging from ten percent to twenty percent are payable on net sales of the product. Royalties are payable in each country until the last-to-expire patent in such country expires. Royalties are subject to a 50% reduction in the event that a generic competitor launches in an applicable country where we market and sell the product. We are responsible for the prosecution and enforcement of patents licensed under the DFD-29 Agreement. The DFD-29 Agreement contains customary representations, warranties, and indemnities, and title transfers to us on the date of achievement of certain regulatory milestones set forth in the agreement, after which our licenses become our acquired assets. Each party may also terminate the DFD-29 Agreement for material breach by the other party or for certain bankruptcy or insolvency related events. Additionally, we agreed to fund and oversee the Phase III clinical trials. From inception to date we have incurred approximately \$2.3.8 million in costs associated with the development of DFD-29.

The DFD-29 Agreement will remain in effect on a country-by-county basis until the expiration of the revenue percentage term in the relevant country, which period begins on the first commercial sale of a product in that country and ends upon the expiration or invalidation date of the last revenue generating patent in such country. The DFD-29 Agreement terminates in its entirety upon the expiry of the revenue percentage term in the last country covered under the DFD-29 Agreement.

Qbrexza Agreement

On March 31, 2021, we executed an asset purchase agreement for Qbrexza® (the "Qbrexza APA") with Dermira Inc. ("Dermira"), pursuant to which we acquired global ownership to Qbrexza (glycopyrronium), a prescription cloth towelette approved to treat primary axillary hyperhidrosis in people nine years of age and older. The transaction closed on May 14, 2021, and pursuant to the Qbrexza APA, we made an upfront \$12.5 million cash payment to Dermira. We are obligated to make payments to Dermira of up to \$144.0 million in the aggregate upon the achievement of certain milestones. For the first two years, we were required to pay royalties on sales ranging from the mid-thirty to the mid-twenty percent. Thereafter, we are required to pay royalties on Qbrexza and the upper teen digits, which are payable for a period of eight years ending in 2029, subject to certain reductions. The Qbrexza APA contains customary representations, warranties, and indemnities. Each party may also terminate the Qbrexza APA for material breach by the other party.

As part of the Qbrexza APA, we were assigned an exclusive license agreement with Rose University ("Rose U") pursuant to which we obtained a worldwide exclusive license within a field of use including hyperhidrosis to practice, enforce and otherwise exploit certain patent rights, know-how and data related to Qbrexza. The license agreement with Rose U includes a sublicense of certain data and an assignment of certain regulatory filings which Rose U had obtained from Stiefel Laboratories ("Stiefel"). In connection with the license agreement, we assumed Rose U's obligations to Stiefel to use commercially reasonable efforts to develop and commercialize products using the licensed patent rights, know-how and data.

Pursuant to these agreements with Rose U and the related agreement with Stiefel with respect to Qbrexza, we are obligated to pay Rose U low-to-mid single-digit royalties on net product sales and low double-digit royalties on sublicense fees and certain milestone, royalty and other contingent payments received from sublicensees, to the extent such amounts are in excess of the milestone and royalty payments we are obligated to pay Rose U directly upon the events or sales triggering such payments.



We are permitted to grant sublicenses to the licensed rights and may assign the agreements upon our acquisition or that of our assets that relate to the license agreement. We may terminate the license agreement if Rose U experiences certain insolvency events or if Rose U commits a material breach of the license agreement, subject to applicable cure provisions. Rose U may terminate the license in certain circumstances if we experience certain insolvency events or if we commit a material breach of the license agreement or if we cause Rose U to be in material breach of its license agreement with Stiefel, subject in each case to applicable cure provisions. Subject to earlier termination, the license agreement remains in effect until 15 years following the first commercial sale of a licensed product have elapsed or, if later, the date that the last patent or patent application in the license under the license agreement with Rose U experience or abandoned. As of December 31, 2023, the last-to-expire issued patent relating to Qbrexza that we license under the license agreement with Rose U expires in 2029.

Accutane Agreement

On July 29, 2020, we entered into a license and supply agreement for Accutane® (the "Accutane Agreement") with DRL. Pursuant to the Accutane Agreement, we agreed to pay \$5.0 million, comprised of an upfront payment of \$1.0 million paid upon execution, with additional milestone payments totaling \$4.0 million. To date, we have paid all of the additional milestone payments. Three additional milestone payments totaling \$17.0 million are contingent upon the achievement of certain net sales milestones. Royalties in the low-double digits based on net sales, subject to specified reductions, are also due.

The term of the Accutane Agreement is ten years and renewable upon mutual agreement. The agreement contains customary representations, warranties, and indemnities. Each party may also terminate the Accutane Agreement for material breach by the other party or for certain bankruptcy or insolvency related events and we may terminate for upon 180 days written notice to the other party. We commenced sales of this product in April 2021.

Anti-Itch Product Agreement

On December 18, 2020, we entered into an asset purchase agreement for our Anti-itch Product (the "Anti-itch APA") with Sun Pharmaceutical Industries, Inc. ("Sun"). Pursuant to the Anti-itch APA, total consideration is \$4.0 million, comprised of an upfront payment of \$2.0 million, payable upon execution. Through December 31, 2023, we have paid \$4.0 million and have no additional payment obligations. The Anti-itch APA contains customary representations, warranties, and indemnities. There are no subsequent milestone payments or royalties beyond the aforementioned payments. We intend to launch this product during the second half of 2024 or first half of 2025.

Ximino Agreement

On July 22, 2019, we entered into an asset purchase agreement for Ximino® (the "Ximino APA") with Sun. Pursuant to the Ximino APA, total consideration is \$9.4 million, with an upfront payment of \$2.4 million, which was payable within 60 days after execution on September 22, 2019.

Pursuant to the terms of the Ximino APA, the remaining \$7.0 million is due on the second anniversary and for the next four anniversaries of the Ximino APA thereafter. In addition, we are obligated to pay royalties in the mid-single digits based on net sales of Ximino, subject to specified reductions until the end of 2022. The Ximino APA contains customary representations, warranties, and indemnities. Each party may also terminate the Ximino APA for material breach by the other party or for certain bankruptcy or insolvency related events. No additional licensing or milestone payments are required. We commenced sales of this product in August 2019. We discontinued selling Ximino in September 2023.

Exelderm Agreement

On August 31, 2018, we entered into an asset purchase agreement for Exelderm® (the "Exelderm APA") with Sun. Pursuant to the Exelderm APA, total consideration is \$1.6 million, comprised of an upfront payment of \$1.2 million, which was payable within 60 days after execution on October 31, 2018. The remaining milestone payment was contingent upon net sales reaching a certain threshold, at which point a \$0.4 million payment became due. This threshold was achieved in 2020 and paid in early 2021. We were obligated to pay royalties in the low-double digits based on net sales of Exelderm until the end of 2023, and no additional licensing or milestone payments are required. We commenced sales of this product in August 2018.

Targadox Agreement

On March 10, 2015, we entered into a license and supply agreement (as amended) for Targadox® (the "Targadox Agreement") with PuraCap International LLC n/k/a Caribe Holdings, Inc. ("Caribe"). We made an upfront payment of \$1.3 million. Further payments will be made based on a revenue sharing arrangement and no additional licensing or milestone payments are required. The term of the Targadox Agreement is ten years and automatically renews for three-year periods unless either party provides notice of its intent not to renew at least 180 days prior to the expiration of the applicable term. Under our revenue sharing arrangement, we are entitled to retain a majority of the net profits and pay Caribe a portion of the net profits after deducting certain commercial, marketing and sales expenses during the term of the Targadox Agreement. The Targadox Agreement contains customary representations, warranties, and indemnities. Each party may also terminate the Targadox Agreement for material breach by the other party or for certain bankruptcy or insolvency related events. We commenced sales of this product in October 2016.

Research and Development

As discussed above, on June 29, 2021, we obtained the global rights from DRL for the development and commercialization of DFD-29, a late-stage development modified release oral minocycline that is being developed for the treatment of inflammatory lesions of rosacea. Through this collaboration, the parties were required to work together to complete the development of DFD-29, which included conducting two Phase III studies to assess the efficacy, safety and tolerability of oral DFD-29 for the treatment of rosacea and the January 4, 2024 regulatory submission of an NDA under Section 505(b)(2) of the FDCA. DRL provided development support, including responding to any requests for information or clarification from FDA regarding the NDA.

On March 17, 2022, we dosed the first patient in our Phase III clinical trials evaluating DFD-29 (Minocycline Modified Release Capsules 40 mg) for the Treatment of Rosacea. We filed the NDA with the FDA on January 4, 2024. The two Phase III studies conducted in the US and Germany demonstrated that DFD-29 was statistically superior to Oracea[®] and Placebo on multiple efficacy endpoints relevant to the treatment of rosacea. DFD-29 also demonstrated statistical superiority to Placebo in reducing erythema of rosacea. The safety profile of DFD-29 in the two Phase III studies was similar to Placebo. Based on the Phase III data, we are seeking an indication to treat inflammatory lesions and erythema of rosacea in patients 18 years of age and older for DFD-29.

Intellectual Property

General

We rely on a combination of contractual provisions, confidentiality policies and procedures and patent, trademark, copyright and trade secrecy laws to protect the proprietary aspects of our technology and business. Three of our marketed products, Accutane, Targadox, and Exelderm, do not have patent protection and/or are not otherwise eligible for patent protection. As part of our development and acquisition strategy, we place a strong emphasis on the patent protection for potential products.

Three of our marketed products, Qbrexza, Amzeeq, and Zilxi, as well as DFD-29, currently have patent protection.

Qbrexza Patents

We own or have an exclusive license to twenty two issued U.S. patents and forty one issued foreign patents, which include granted European patent rights that have been validated in selected European Patent Organization ("EPO") member states (Switzerland, Germany, Spain, France, Great Britain, Ireland, and Italy), Australia, Canada, Mexico, Israel, Japan, Hong Kong, Korea, and New Zealand, Singapore, and South Africa, and six pending U.S. patent applications, one pending Patent Cooperation Treaty application, and sixteen pending foreign patent applications. Of these patents and patent applications:

There are eighteen issued U.S. patents, thirty seven issued foreign patents (AU, CA, selected EP member states, Mexico, Japan, Hong Kong, Korea, New Zealand, Singapore, and South Africa), two pending U.S. patent applications and three pending foreign applications (in Israel and Hong Kong), all relating to Qbrexza. We own fourteen of the issued U.S. patents, both of the pending U.S. patent applications, twenty of the issued foreign patents, and two of the pending foreign applications, and have exclusively licensed from Rose U worldwide rights to four of the issued U.S. patents, seventeen issued foreign patents, and no pending foreign patent applications. The issued Qbrexza patents contain claims directed to individually packaged wipes for the treatment of hyperhidrosis where the wipes contain a composition comprising Qbrexza or other related compounds, and methods of alleviating hyperhidrosis using such compositions, absorbent pads comprising Qbrexza or other related compounds, individually packaged wipes comprising such compositions and contain claims directed to compositions comprising Qbrexza or other related compounds, individually packaged wipes comprising such compositions, absorbent pads comprising Qbrexza pharmaceutical compositions and methods of treating hyperhidrosis with topical administration of Qbrexza or other related compounds. The issued U.S. and foreign patents relating to Qbrexza will expire between 2028 and 2033 and the pending U.S. and foreign patent applications relating to Qbrexza, if issued, will expire between 2028 and 2034.

Amzeeq, Zilxi & the Molecular Stabilizing Technology Platform Patents

We own thirty nine issued U.S. patents and twenty issued foreign patents, and nine pending U.S. patent applications, and two pending foreign patent applications. Of these patents and patent applications:

- There are twenty one issued U.S. patents, fifteen issued foreign patents (Australia, Canada, Europe, Israel, Mexico, United Kingdom, South Africa), six pending U.S. patent applications and one pending foreign application (Canada), all relating to Amzeeq. The issued Amzeeq patents contain claims directed to compositions and use of the compositions (method claims). The issued U.S. and foreign patents relating to Amzeeq will expire between 2030 and 2037 and the pending U.S. and foreign patent applications relating to Amzeeq will expire between 2030 and 2037.
- There are fourteen issued U.S. patents, fifteen issued foreign patents (Australia, Canada, Europe, Israel, Mexico, United Kingdom, South Africa), four pending U.S. patent applications and one pending foreign application (Canada), all relating to Zilxi. The issued Zilxi patents contain claims directed to compositions and use of the compositions (method claims). The issued U.S. and foreign patents relating to Zilxi will expire between 2030 and 2037 and the pending U.S. and foreign patent applications relating to Zilxi will expire between 2030 and 2037.
- The other patents related to molecular stabilizing platform but not products directly are sixteen issued U.S. patents, three pending U.S. patent applications, and five issued foreign patents (Canada, Israel, and Mexico).

DFD-29 Patents

With regard to DFD-29, we have an exclusive license to one U.S. patent family including three issued U.S. patents and one U.S. continuation application, as well as one issued foreign patent (Mexico) and eight foreign pending patent applications (one in each of Australia, Canada, Europe, Japan, Korea, and South Africa; and two in New Zealand) covering methods of treating an inflammatory skin condition by selecting and administering an oral composition comprising reduced dose of minocycline and the relevant pharmacokinetic parameters, and we intend to pursue composition-of-matter patents, where possible, and dosage and formulation patents, as well as method-of-use patents on novel indications for known compounds. The three issued U.S. patents will expire in 2039.

Additional Intellectual Property and Proprietary Right Protection

We also use other forms of protection, such as trademark, copyright, and trade secret protection, to protect our intellectual property, particularly where we do not believe patent protection is appropriate or obtainable. We aim to take advantage of all of the intellectual property rights that are available to us and believe that this comprehensive approach will provide us with proprietary positions for our product candidates, where available. Patents extend for varying periods according to the date of patent filing or grant and the legal term of patents in various countries where patent protection is obtained. The actual protection afforded by a patent, which can vary from country to country, depends on the type of patent, the scope of its coverage and the availability of legal remedies in the country.

Our goal is to obtain, maintain and enforce patent protection for our products, formulations, processes, methods and other proprietary technologies, to preserve our trade secrets, and to operate without infringing on the proprietary rights of other parties, both in the United States and in other countries. Our policy is to actively seek to obtain, where appropriate, the broadest intellectual property protection possible for any product candidates, proprietary information and proprietary technology through a combination of contractual arrangements and patents, both in the U.S. and elsewhere in the world.

Patents and other proprietary rights are crucial to the development of our business. We will be able to protect our proprietary technologies from unauthorized use by third parties only to the extent that our proprietary rights are covered by valid and enforceable patents, supported by regulatory exclusivity, or are effectively maintained as trade secrets.

Generally, patent applications in the U.S. are maintained in secrecy for a period of 18 months or more. The patent positions of biotechnology and pharmaceutical companies are highly uncertain and involve complex legal and factual questions. Therefore, we cannot predict the breadth of claims allowed in biotechnology and pharmaceutical patents, or their enforceability. To date, there has been no consistent policy regarding the breadth of claims allowed in biotechnology patents. Third parties or competitors may challenge or circumvent our patents or patent applications, if issued. If our competitors prepare and file patent applications in the U.S. that claim technology also claimed by us, we may have to participate in derivation proceedings declared by the USPTO to determine proper inventorship of a claimed of invention, which could result in substantial cost, even if the eventual outcome is favorable to us. Because of the extensive time required for development, testing and regulatory review of a potential product, it is possible that before we commercialize any of our products, any related patent may expire or remain in existence for only a short period following commercialization, thus reducing any advantage of the patent. However, the life of a patent covering a product that has been subject to regulatory approval may be extended through the patent term restoration program, although any such extension could still be minimal.

If a patent is issued to a third party containing one or more preclusive or conflicting claims, and those claims are ultimately determined to be valid and enforceable, we may be required to obtain a license under such patent or to develop or obtain alternative technology, neither of which may be possible. In the event of litigation involving a third-party claim, an adverse outcome in the litigation could subject us to significant liabilities to such third party, require us to seek a license under the disputed rights of such third party, and/or require us to cease use of the technology. Moreover, our breach of an existing license or failure to obtain a license to technology required to commercialize our products may seriously harm our business. We also may need to commence litigation to enforce any patents issued to us or to determine the scope and validity of third-party proprietary rights. Litigation could involve substantial costs.

Other Intellectual Property Rights

We depend upon trademarks, trade secrets, and continuing technological advances to develop and maintain our competitive position. We also depend upon the skills, knowledge and experience of our scientific and technical personnel, as well as that of our advisors, consultants and other contractors. This knowledge and experience we call "know-how." To help protect our proprietary know-how that is not patentable, and for inventions for which patents may be difficult to enforce, we rely on trade secret protection and confidentiality agreements to protect our interests. To this end, we require all employees, scientific advisors, consultants, collaborators and other contractors, upon commencement of a relationship with us, to enter into confidentiality agreements, which prohibit the disclosure of confidential information and, in the case of parties other than our research and development collaborators, require disclosure and assignment to us of the ideas, developments, discoveries and inventions important to our business. These agreements are designed to protect our proprietary information and to grant us ownership of technologies that are developed in connection with their relationship with us. These agreements may not, however, provide protection for our trade secrets in the event of unauthorized disclosure of such information.

There can be no assurance that any of our patents, licenses or other intellectual property rights will afford us any protection from competition, that our confidentiality agreements will not be breached, that we will have adequate remedies for any breach, that others will not independently develop equivalent proprietary information or that other third parties will not otherwise gain access to our trade secrets and other intellectual property.

We may find it necessary to initiate litigation to enforce our patent rights, to protect our intellectual property or trade secrets or to determine the scope and validity of the proprietary rights of others. Litigation is costly and time-consuming and there can be no assurance that our litigation expenses will not be significant in the future or that we will prevail in any such litigation.

Competition

Pharmaceutical Industry

Our competitors include pharmaceutical companies and biotechnology companies, as well as universities and public and private research institutions. In addition, companies that are active in different but related fields represent substantial competition for us. Many of our competitors have significantly greater capital resources, larger research and development staffs and facilities and greater experience in drug development, regulation, manufacturing and marketing than we do. These organizations also compete with us to recruit qualified personnel, attract partners for joint ventures or other collaborations, and license technologies that are competitive with ours. To compete successfully in this industry, we must identify novel and unique drugs or methods of treatment and then complete the development of those drugs as treatments in advance of our competitors.

Dermatology Sector

The dermatology competitive landscape is highly fragmented, with a large number of midsize and smaller companies competing in both the prescription sector and the OTC sector. Our competitors are pursuing the development and/or acquisition of pharmaceuticals, medical devices and OTC products that target the same diseases and conditions that we are targeting in dermatology. Competitive factors vary by product line and geographic area in which our products are sold. The principal methods of competition for our products include quality, efficacy, market acceptance, price, and marketing and promotional efforts.

Branded products often must compete with therapeutically similar branded or generic products or with generic equivalents. Such competition frequently increases over time. For example, if competitors introduce new products, delivery systems or processes with therapeutic or cost advantages, our products could be subject to progressive price reductions and/or decreased volume of sales. To successfully compete for business, we must often demonstrate that our products offer not only medical benefits, but also cost advantages as compared with other forms of care. Accordingly, we face pressure to continually seek out technological innovations and to market our products effectively.

Our major competitors, including Galderma Laboratories, Almirall, Novan Therapeutics, Leo Pharma, Mayne Pharma, Botanix Pharmaceuticals, and Ortho Dermatologics, among others, vary depending on therapeutic and product category, dosage strength and drug-delivery systems, among other factors.

Generic Competition

We face increased competition from manufacturers of generic pharmaceutical products, who may submit applications to the FDA seeking to market generic versions of our products. In connection with these applications, the generic drug companies may seek to challenge the validity and enforceability of our patents through litigation. When patents covering certain of our products (if applicable) expire or are successfully challenged through litigation or in USPTO proceedings, if a generic company launches a competing product "at risk," or when the regulatory or licensed exclusivity for our products (if applicable) expires or is otherwise lost, we may face generic competition as a result. Generic versions are generally significantly less expensive than branded versions, and, where available, may be required to be utilized before or in preference to the branded version under third-party reimbursement programs, or substituted by pharmacies. Accordingly, when a branded product loses its market exclusivity, it normally faces intense price competition from generic forms of the product. To successfully compete for business with managed care and pharmacy benefits management organizations, we must often demonstrate that our products offer not only medical benefits, but also cost advantages as compared with other forms of care. Generic products generally face intense competition form other generic equivalents (including authorized generics) and therapeutically similar branded or generic products.

Supply and Manufacturing

We have limited experience in manufacturing products for clinical or commercial purposes, and we currently do not have any internal manufacturing capabilities. We currently rely upon multiple contract manufacturers to produce our products and clinical supply of product candidates and will continue to rely upon contract manufacturers for any current or future product candidates under current Good Manufacturing Practice ("cGMP") regulations for use in pre-clinical and clinical activities. Due to the risks associated with reliance on third-party manufacturing, as part of our current and future strategy of licensing, acquiring, or the future development of assets, we currently, and will continue to, secure manufacturing agreements with either a counterparty to a transaction, with one or more of our contract manufacturers or additional contract manufacturers. As with any supply program, obtaining raw materials of the correct quality cannot be guaranteed, and we cannot ensure that we will be successful. Our third-party manufacturers have a limited number of facilities in which our product candidates can be produced and may have limited experience in manufacturing our product candidates in quantities sufficient for commercialization. Our third-party manufacturers will have other clients and may have other priorities that could affect their ability to perform the work satisfactorily and/or on a timely basis. Both of these occurrences would be beyond our control. We expect to similarly rely on contract manufacturing relationships for any products that we may in-license or acquire in the future. However, there can be no assurance that we will be able to successfully contract with such manufacturers on terms acceptable to us, or at all.

Contract manufacturers are subject to ongoing periodic and unannounced inspections by the FDA, the Drug Enforcement Administration and corresponding state and European agencies to ensure strict compliance with cGMPs and other state and federal regulations. We do not have control over third-party manufacturers' compliance with these regulations and standards, other than through contractual obligations. If they are deemed out of compliance with cGMPs, product recalls could result, inventory could be destroyed, production could be stopped, and supplies could be delayed or otherwise disrupted.

If we need to change manufacturers during the clinical or development stage for product candidates or after commercialization for our approved products, the FDA and corresponding foreign regulatory agencies must approve these new manufacturers in advance, which will involve testing and additional inspections to ensure compliance with FDA regulations and standards and may require significant lead times and delay. Furthermore, switching manufacturers may be difficult because the number of potential manufacturers is limited. It may be difficult or impossible for us to find a replacement manufacturer quickly or on terms acceptable to us, or at all.

Government and Industry Regulations - Overview

FDA Regulations

Numerous governmental authorities, principally the FDA and corresponding state and foreign regulatory agencies, impose substantial regulations upon any potential clinical development and the manufacture and marketing of our products. Before marketing in the U.S., any drug that we may develop must undergo rigorous pre-clinical testing and clinical trials and an extensive regulatory approval process implemented by the FDA under the FDCA. The FDA regulates, among other things, the pre-clinical and clinical testing, safety, efficacy, approval, manufacturing, record keeping, lot traceability, individual serialization, adverse event reporting, packaging, labeling, storage, advertising, promotion, export, sale and distribution of biopharmaceutical products.

The regulatory review and approval process is lengthy, expensive and uncertain. In the event that we acquire or develop a clinical stage asset, we will be required to submit extensive pre-clinical and clinical data and supporting information to the FDA for each indication or use to establish a product candidate's safety and efficacy before we can secure FDA approval to market or sell a product in the U.S. The approval process may take many years, depending on the stage of development of a target asset, and requires the expenditure of substantial resources and may involve ongoing requirements for post-marketing studies or surveillance.

Clinical testing must meet requirements for institutional review board oversight, informed consent and good clinical practices, and must be conducted pursuant to an Investigational New Drug ("IND") Application unless exempted.

For purposes of NDA approval, clinical trials are typically conducted in the following sequential phases:

Phase 1:

>The drug is administered to a small group of humans, either healthy volunteers or patients, to test for safety, dosage tolerance, absorption, metabolism, excretion and clinical pharmacology.

Phase 2:

> Studies are conducted on a larger number of patients to assess the efficacy of the product, to ascertain dose tolerance and the optimal dose range, and to gather additional data relating to safety and potential adverse events.

Phase 3:

>Studies establish safety and efficacy in an expanded patient population.

Phase 4:

> The FDA may require Phase 4 post-marketing studies to find out more about the drug's long-term risks, benefits, and optimal use, or to test the drug in different populations.

The length of time necessary to complete clinical trials varies significantly and may be difficult to predict. Clinical results are frequently susceptible to varying interpretations that may delay, limit or prevent regulatory approvals. Additional factors that can cause delay or termination in future clinical trials, or that may increase the costs of these trials, include:

- slow patient enrollment due to the nature of the clinical trial plan, the proximity of patients to clinical sites, the eligibility criteria for participation in the study or other factors;
- inadequately trained or insufficient personnel at the study site to assist in overseeing and monitoring clinical trials or delays in approvals from a study site's review board;
- longer treatment time required to demonstrate efficacy or determine the appropriate product dose;
- insufficient supply of the drug candidates;
- adverse medical events or side effects in treated patients; and
- ineffectiveness of the drug candidates.

In addition, the FDA, equivalent foreign regulatory authority, or a data safety monitoring committee for a trial may place a clinical trial on hold or terminate it if it concludes that subjects are being exposed to an unacceptable health risk, or for futility. Any drug is likely to produce some toxicity or undesirable side effects in animals and in humans when administered at sufficiently high doses and/or for a sufficiently long period of time. Unacceptable toxicity or side effects may occur at any dose level at any time in the course of studies in animals designed to identify unacceptable effects of a drug candidate, known as toxicological studies, or clinical trials of drug candidates. The appearance of any unacceptable toxicity or side effect could cause us or regulatory authorities to interrupt, limit, delay or abort the development of any of our drug candidates and could ultimately prevent approval by the FDA or foreign regulatory authorities for any or all targeted indications.

Sponsors of drugs may apply for a special protocol assessment ("SPA") from the FDA. The SPA process is a procedure by which the FDA provides official evaluation and written guidance on the design and size of proposed protocols that are intended to form the basis for an NDA. However, final marketing approval depends on the results of efficacy, the adverse event profile and an evaluation of the benefit/risk of treatment demonstrated in the Phase 3 trial. The SPA agreement may only be changed through a written agreement between the sponsor and the FDA, or if the FDA becomes aware of a substantial scientific issue essential to product safety or efficacy.



Before receiving FDA approval to market a product, we must demonstrate that the product is safe and effective for its intended use by submitting to the FDA an NDA, Abbreviated NDA ("ANDA"), 510(K) or Biologics License Application ("BLA") containing the pre-clinical and clinical data that have been accumulated, together with chemistry and manufacturing and controls specifications and information, and proposed labeling, among other things. The FDA may refuse to accept an NDA, ANDA, 510(K) or BLA for filing if certain content criteria are not met and, even after accepting an NDA, ANDA, 510(K) or BLA, the FDA may often require additional information, including clinical data, before approval of marketing a product.

Section 505(b)(2) NDAs may provide an alternate path to FDA approval for new or improved formulations or new uses of previously approved products. Section 505(b)(2) permits the filing of an NDA where at least some of the information required for approval comes from clinical trials not conducted by, or for, the applicant and for which the applicant has not obtained a right of reference. The FDA may then approve the new product candidate for all, or some, of the label indications for which the referenced product has been approved, as well as for any new indication sought by the Section 505(b)(2) applicant.

To the extent that the Section 505(b)(2) applicant is relying on the FDA's findings of safety and effectiveness for an already approved product, the applicant is required to certify to the FDA concerning any patents listed for the approved product in the Orange Book to the same extent that an ANDA applicant would. Thus approval of a Section 505(b)(2) NDA can be stalled until all the listed patents claiming the referenced product have expired; until any non-patent exclusivity, such as exclusivity for obtaining approval of a New Chemical Entity ("NCE"), listed in its publication "Approved Drug Products with Therapeutic Equivalence Evaluations," also referred to as the "Orange Book," for the referenced product has expired; and, in the case of a Paragraph IV certification and subsequent patent infringement suit, until the earlier of 30 months, settlement of the lawsuit or a decision in the infringement case that is favorable to the Section 505(b)(2) applicant. In the interim period, the FDA may grant tentative approval. Tentative approval indicates that the FDA has granted that the applicant meets the standards for approval as of the date that the tentative approval is granted. Final regulatory approval can only be granted if the FDA is assured that there is no new information that would affect final regulatory/ approval.

The FDA may request a Risk Evaluation and Mitigation Strategy ("REMS"), as part of an NDA, ANDA, 510(K) or BLA. The REMS typically contains some combination of post-marketing obligations of the sponsor to train prescribing physicians, monitor drug use, including off-label use, and conduct sufficient Phase 4 follow-up studies and registries to ensure the continued safe use of the drug.

As part of the approval process, the FDA must inspect and approve each manufacturing facility. Among the conditions of approval is the requirement that a manufacturer's quality control and manufacturing procedures conform to cGMP. Manufacturers must expend significant time, money and effort to ensure continued compliance, and the FDA conducts periodic inspections to certify compliance. It may be difficult for our manufacturers or us to comply with the applicable cGMPs, as interpreted by the FDA, and other FDA regulatory requirements. If we, or our contract manufacturers, fail to comply, then the FDA may not allow us to market products that have been affected by the failure.

If the FDA grants approval, the approval will be limited to those conditions and patient populations for which the product is safe and effective, as demonstrated through clinical studies. Further, a product may be marketed only in those dosage forms and for those indications approved in the NDA, ANDA, 510(K), or BLA. Certain changes to an approved BLA, including, with certain exceptions, any significant changes to labeling, require approval of a supplemental application before the drug may be marketed as changed. Any products that we manufacture or distribute pursuant to FDA approvals are subject to continuing monitoring and regulation by the FDA, including compliance with cGMP and the reporting of adverse experiences with the drugs. The nature of marketing claims that the FDA will permit us to make in the labeling and advertising of our products will generally be limited to those specified in FDA approved labeling, and the advertising of our products will be subject to comprehensive monitoring and regulation by the FDA. Drugs whose review was accelerated may carry additional restrictions on marketing activities, including the requirement that all promotional materials are pre-submitted to the FDA. Claims exceeding those contained in approved labeling will constitute a violation of the FDCA. Violations of the FDCA or regulatory requirements at any time during the product development process, approval process, or marketing and sale following approval may result in agency enforcement actions, including withdrawal of approval, recall, seizure of products, warning letters, untitled letters, Form 483s, injunctions, fines and/or civil or criminal penalties. Any agency enforcement action could have a material adverse effect on our business.

Pharmaceutical Coverage, Pricing and Reimbursement

Sales of any pharmaceutical product depend, in part, on the extent to which such product will be covered by third-party payors, such as federal, state, and foreign government healthcare programs, commercial insurance, and managed healthcare organizations, and the level of reimbursement for such product by third-party payors. Significant uncertainty exists as to the coverage and reimbursement status of any newly approved product. Decisions regarding the extent of coverage and amount of reimbursement to be provided are made on a plan-by-plan basis. One third-party payor's decision to cover a particular product does not ensure that other payors will also provide coverage for the product. As a result, the coverage determination process can require manufacturers to provide scientific details, information on cost-effectiveness, and clinical support for the use of a product to each payor separately. This can be a time-consuming process, with no assurance that coverage and adequate reimbursement will be applied consistently or obtained in the first instance.

In addition, third-party payors are increasingly reducing reimbursements for pharmaceutical products and related services. The U.S. government and state legislatures have continued implementing cost-containment programs, including price controls, restrictions on coverage and reimbursement and requirements for substitution of generic products. Third-party payors are increasingly challenging the prices charged, examining the medical necessity and reviewing the cost effectiveness of pharmaceutical products, in addition to questioning their safety and efficacy. Adoption of price controls and cost-containment measures, and adoption of more restrictive policies in jurisdictions with existing controls and measures, could further limit sales of any product. Decreases in third-party reimbursement for any product or a decision by a third-party payor not to cover a product could reduce physician usage and patient demand for the product.

At the state level, there are also new laws and ongoing ballot initiatives that create additional pressure on drug pricing and may affect how pharmaceutical products are covered and reimbursed. A number of states have adopted or are considering various pricing actions, such as those requiring pharmaceutical manufacturers to publicly report proprietary pricing information, limit price increases or to place a maximum price ceiling or cap on certain products. Existing and proposed state pricing laws have added complexity to the pricing of pharmaceutical drug products.

In international markets, reimbursement and healthcare payment systems vary significantly by country, and many countries have instituted price ceilings on specific products and therapies. For example, the European Union provides options for its member states to restrict the range of medicinal products for which their national health insurance systems provide reimbursement and to control the prices of medicinal products for human use. A member state may approve a specific price for the medicinal product, or it may instead adopt a system of direct or indirect controls on the profitability of the company placing the medicinal product on the market. Pharmaceutical products may face competition from lower-priced products in foreign countries that have placed price controls on pharmaceutical products and may also compete with imported foreign products. Furthermore, there is no assurance that a product will be considered medically reasonable and necessary for a specific indication, that it will be considered cost-effective by third-party payors, that an adequate level of reimbursement will be established even if coverage is available, or that the third-party payors' reimbursement policies will not adversely affect the ability for manufacturers to sell products profitably.

International Regulations

In addition to regulations in the United States, there are a variety of foreign regulations governing clinical trials and commercial sales and distribution of any product candidates. The approval process varies from country to country, and the time may be longer or shorter than that required for FDA approval.

Failure to comply with applicable federal, state and foreign laws and regulations would likely have a material adverse effect on our business. In addition, federal, state and foreign laws and regulations regarding the manufacture and sale of new drugs are subject to future changes.

PHRMA Code and April 3, 2003 Department of Health and Human Services Office of Inspector General, OIG Compliance Program for Pharmaceutical Manufacturers

We have established and implemented a corporate compliance program designed to prevent, detect and correct violations of state and federal healthcare laws, including laws related to advertising and promotion of our products that are in compliance with the PHRMA Code and the Health and Human Services Office of Inspector General ("OIG") Compliance Program requirements for Pharmaceutical Manufacturers.

Healthcare Fraud, Waste and Abuse

We are subject to various federal, state and local laws targeting fraud and abuse in the healthcare industry, violations of which can lead to civil and criminal penalties, including fines, imprisonment and exclusion from participation in federal healthcare programs.

These laws are applicable to us as both a manufacturer and a supplier of products reimbursed by federal healthcare programs, and they also apply to physicians and other potential purchasers of our products.

The federal Anti-Kickback Statute (42 U.S.C. § 1320a-7b) prohibits persons from knowingly and willfully soliciting, receiving, offering or providing remuneration, directly or indirectly, to induce either the referral of an individual, or the furnishing, recommending or arranging for a good or service, for which payment may be made under a federal healthcare program such as the Medicare and Medicaid programs. Remuneration is not defined in the federal Anti-Kickback Statute and has been broadly interpreted to include anything of value, including for example, gifts, discounts, coupons, the furnishing of supplies or equipment, credit arrangements, payments of cash, waivers of payments, ownership interests and providing anything at less than its fair market value. Under the federal Anti-Kickback Statute and the applicable criminal healthcare fraud statutes contained within 42 U.S.C. § 1320a-7b, a person or entity need not have actual knowledge of this statute or specific intent to violate it in order to have committed a violation. In addition, the government may assert that a claim, including items or services resulting from a violation of 42 U.S.C. § 1320a-7b, constitutes a false or fraudulent claim for purposes of the civil False Claims Act (discussed below) or the civil monetary penalties statute, which imposes fines against any person who is determined to have presented or caused to be presented claims to a federal Anti-Kickback Statute and implementing regulations provide for certain exceptions for "safe harbors" for certain discounting, rebating or personal services any government authorities, make compliance with the law difficult. Violations of the Anti-Kickback Statute can result in significant criminal fines, exclusion from participation in Medicare and Medicare and Medicare and Medicare and Medicare and Medicare and have actual knowledge of the advision of 42 U.S.C. § 1320a-7b, constitutes a false or fraudulent. The federal Anti-Kickback Statute and implementing regulations provide for certain exceptions

In October 2019, the OIG issued a proposed rule to, among other things, add new safe harbors for certain value-based arrangements. Although the valuebased proposals would not include pharmaceutical manufacturers among the entities that could permissibly enter into such contracting arrangements, the general trend toward outcomes and value-based contracts in the healthcare industry may continue. It is possible that payors, among other customers, could push manufacturers for novel contracting approaches, including those that would incorporate value-based principles, and these efforts could affect our business.

The civil False Claims Act and similar state laws impose liability on any person or entity who, among other things, knowingly presents, or causes to be presented, a false or fraudulent claim for payment by a federal healthcare program. The qui tam provisions of the False Claims Act and similar state laws allow a private individual to bring civil actions on behalf of the federal or state government and to share in any monetary recovery. The Federal Physician Payments Sunshine Act and similar state laws impose reporting requirements for various types of payments to physicians and teaching hospitals. Failure to comply with reporting requirements under these laws could subject manufacturers and others to substantial civil money penalties.

The Health Insurance Portability and Accountability Act of 1996 ("HIPAA") imposes criminal liability and amends provisions on the reporting, investigation, enforcement, and penalizing of civil liability for, among other things, knowingly and willfully executing, or attempting to execute, a scheme to defraud any healthcare benefit program, or knowingly and willfully falsifying, concealing or covering up a material fact or making any materially false statement in connection with the delivery of, or payment for healthcare benefits, items or services by a healthcare benefit program, which includes both government and privately funded benefits programs; similar to the federal Anti-Kickback Statute, a person or entity does not need to have actual knowledge of the statute or specific intent to violate it in order to have committed a violation.

Other Healthcare Laws and Compliance Requirements

In the United States, our activities are potentially subject to regulation by various federal, state and local authorities in addition to the FDA, including the Centers for Medicare and Medicaid Services (formerly the Health Care Financing Administration), other divisions of the United States Department of Health and Human Services, the United States Department of Justice and individual United States Attorney offices within the Department of Justice, and state and local governments.

Drug Quality and Security Act ("DQSA")

DQSA was enacted by Congress on November 27, 2013. Title II of DQSA, the Drug Supply Chain Security Act ("DSCSA"), outlines steps to build an electronic, interoperable system to identify and trace certain prescription drugs as they are distributed in the United States. This is intended to enhance the FDA's ability to help protect consumers from exposure to drugs that may be counterfeit, stolen, contaminated, or otherwise harmful. The system is also intended to improve detection and removal of potentially dangerous drugs from the drug supply chain to protect U.S. consumers.

Additionally, the DSCSA directs FDA to establish national licensure standards for wholesale distributors and third-party logistics providers, and requires these entities report licensure and other information to FDA annually. The implementation and enforcement of complete unit level traceability of verifiable return serialization, including aggregation throughout the whole supply chain, is not required as of November 27, 2023. Although the rule regarding wholesale distributor verification of saleable returned products does not directly apply to our Company, we are required to assist our wholesale distributor customers by setting in place mechanics that would allow for traceability of returns in the supply chain. If we are not able to come into compliance of this rule, our wholesale distributor customers may not accept our returns on our behalf.

We are subject to, and required to be in compliance with, the DQSA. Our Company remains in compliance with the requirements promogulated by the DSCSA and intends on remaining vigilant with regards to any potential modifications to the act. For purposes of our business, we are considered both manufacturers and re-packagers under the act. Currently, we are in compliance with the DSCSA as it relates to our business and operations.

Item 1A. Risk Factors

The following information sets forth risk factors that could cause our actual results to differ materially from those contained in forward-looking statements we have made in this report and those we may make from time to time. You should carefully consider the risks described below, in addition to the other information contained in this report and our other public filings, before making an investment decision. Our business, financial condition or results of operations could be harmed by any of these risks. The risks and uncertainties described below are not the only ones we face. Additional risks not presently known to us or other factors not perceived by us to present significant risks to our business at this time also may impair our business operations.

Risks Related to Our Business, Industry and Existing Operating Revenue Stream

Future revenue from sales of our dermatology products may be lower than expected or lower than in previous periods.

The vast majority of our operating income for the foreseeable future is expected to come from the sale of our dermatology products. Any setback that may occur with respect to such products could significantly impair our operating results and/or reduce our revenue and the value of our securities. Setbacks for such products could include, but are not limited to, issues related to: supply chain, shipping; distribution; demand; manufacturing; product safety; product quality; marketing; government regulation; pricing; reimbursement; licensing and approval; intellectual property rights; competition with existing or new products; product acceptance by physicians, other licensed medical professionals and patients; and higher than expected total rebates, returns or recalls.

Also, the majority of our sales derive from products that are without patent protection and/or are or may become subject to third-party generic competition, the introduction of new competitor products, or increased market share of existing competitor products, any of which could have a significant adverse effect on our operating income.

We face challenges as our products face generic competition and/or losses of exclusivity.

Our products do and may compete with well-established products, both branded and generic, with similar or the same indications. We face increased competition from manufacturers of generic pharmaceutical products, who may submit applications to the FDA seeking to market generic versions of our products. In connection with these applications, the generic drug companies may seek to challenge the validity and enforceability of our patents through litigation. When patents covering certain of our products expire or are successfully challenged through litigation or in USPTO proceedings, if a generic company launches a competing product "at risk," or when the regulatory or licensed exclusivity for our products expires or is otherwise lost, we may face generic competition as a result.



The majority of our sales derive from products that are without patent protection and/or are or may become subject to third-party generic competition, the introduction of new competitor products, or an increase in market share of existing competitor products, any of which could have a significant adverse impact on our operating income. Three of our marketed products, Qbrexza, Amzeeq, and Zilxi as well as DFD-29, currently have patent protection. Three of our marketed products, Accutane, Targadox, and Exelderm, do not have patent protection or otherwise are not eligible for patent protection. Accutane currently competes in the Isotretinoin market with five other therapeutic equivalent ("AB rated") products. Targadox faces AB rated generic competition. Exelderm may face AB rated generic competition in the future.

Generic versions are generally significantly less expensive than branded versions, and, where available, may be required to be utilized before or in preference to the branded version under third-party reimbursement programs, or substituted by pharmacies. Accordingly, when a branded product loses its market exclusivity, it normally faces intense price competition from generic forms of the product. To successfully compete for business with managed care and pharmacy benefits management organizations, we must often demonstrate that our products offer not only medical benefits, but also cost advantages as compared with other forms of care. If we fail to do so, our results of operations, financial condition or cash flows may be materially adversely affected.

Any disruptions to the capabilities, composition, size or existence of our field sales force may have a significant adverse impact on our existing revenue stream. Further, our ability to effectively market and sell any future products that we may develop will depend on our ability to establish and maintain sales and marketing capabilities or to enter into agreements with third parties to market, distribute and sell any such products.

Our field sales force has been and is expected to continue to be an important contributor to our commercial success. Any disruptions to our relationship with our field sales force could materially adversely affect our product sales. We may rely on professional employer organizations and staffing organizations for the employment of our field sales force in the future.

The establishment, development, and/or expansion of a field sales force, either by us or certain of our partners or vendors, or the establishment of a field sales force to market any products for which we may have or receive marketing approval, is expensive and time-consuming and could delay any such product launch or compromise the successful commercialization of such products. If we are unable to establish and maintain sales and marketing capabilities or any other non-technical capabilities necessary to commercialize any products that may be successfully developed, we will need to contract with third parties to market and sell such products. We may not be able to establish or maintain arrangements with third parties on commercially reasonable terms, or at all, which would have a material adverse effect on our business, prospects, results of operations, financial condition or cash flows.

Our current and potential future product candidates may not receive regulatory approval, or such approval may be delayed, which would have a material adverse effect on our business and financial condition. Further, even if a product receives regulatory approval, such product will remain subject to substantial regulatory scrutiny.

Our current and potential future product candidates and the activities associated with their development and commercialization, including their design, testing, manufacture, safety, efficacy, recordkeeping, labeling, storage, approval, advertising, promotion, sale and distribution, are subject to comprehensive regulation by the FDA and other regulatory agencies in the United States and abroad. Our failure to obtain marketing approval for any current or future product candidates will prevent us from commercializing the product candidates. Further, any products or future products candidates we license or acquire will be subject to ongoing requirements and review by such regulatory authorities.

We have limited experience in filing and supporting the applications necessary to gain marketing approvals and expect to rely on third parties to assist us in this process. To secure marketing approval, we will be required to establish a product candidate's safety and efficacy by submitting extensive preclinical and clinical data and supporting information for each therapeutic indication. We will further be required to submit information about the product manufacturing and to undergo regulatory inspection of our third-party manufacturing facilities to ensure ongoing compliance with cGMP requirements.

Any of our current or future product candidates may not be effective, may be only moderately effective or may prove to have undesirable or unintended side effects, toxicities or other characteristics that may preclude us from obtaining marketing approval or prevent or limit commercial use. If our current or future product candidates receive(s) marketing approval, the accompanying label may limit the approved use of our drug in this way, which could limit sales of the product.



The marketing approval process, both in the United States and abroad, is time consuming and expensive. Approval may take many years, and if it is granted can vary substantially based upon a variety of factors, including the type, complexity and novelty of the product candidates involved. Our product candidates could be delayed in receiving, or fail to receive, regulatory approval for many reasons, including: the FDA or comparable foreign regulatory authorities may disagree with the design or implementation of our clinical trials; the FDA or comparable foreign regulatory authorities that a drug candidate is safe and effective for its proposed indication or is suitable to identify appropriate patient populations; the results of clinical trials may not meet the level of statistical significance required by the FDA or comparable foreign regulatory authorities for approval; we may be unable to demonstrate that a product candidate's clinical and other benefits outweigh its safety risks.

Changes to marketing approval policies or the regulatory landscape during the development period may cause rejection of or delays in the approval of an application. Regulatory authorities have substantial discretion in the approval process and may refuse to accept any application or decide that our data is insufficient for approval and require costly additional preclinical studies or clinical trials. In addition, varying interpretations of the data obtained from preclinical and clinical testing could delay, limit or prevent marketing approval of a product candidate. Any marketing approval we ultimately obtain may be limited or subject to restrictions or post-approval commitments that render the approved product not commercially viable. If we experience delays in obtaining or fail to obtain or maintain any necessary approvals of any current or future product candidates, receive approval for fewer or more limited indications than we request or without including the labeling claims we desire, our future commercial prospects may be harmed and our ability to generate revenue may be materially impaired. Even if we do receive approval, it may be contingent on the performance of costly post-marketing clinical trials to verify whether or not the drug provides the anticipated clinical benefit, in order to maintain the approval.

Regulatory approval for any approved product is limited by the FDA to those specific indications and conditions for which clinical safety and efficacy have been demonstrated.

Any regulatory approval is limited to those specific diseases and indications for which a product is deemed to be safe and effective. If the FDA or any regulatory authority limits the scope of our indication, or if we are unable to obtain FDA approval for any desired future indications for our products, our ability to effectively market and sell our products may be reduced and our business may be adversely affected. Further, we are only permitted to promote our products for those indications specifically approved by the FDA and there are restrictions around making communications regarding uses not approved and described in the product's labeling. If our promotional activities fail to comply with these regulations or guidelines, we may be subject to advisory or enforcement action by these authorities. In addition, our failure to follow FDA requirements or guidelines relating to promotion and advertising may cause the FDA to suspend or withdraw an approved product from the market, require a recall or institute fines, or could result in disgorgement of money, operating restrictions, injunctions or criminal prosecution, any of which could harm our business.

If the FDA does not conclude that a product candidate satisfies the requirements for the Section 505(b)(2) regulatory approval pathway, or if the requirements for such product candidate under Section 505(b)(2) are not as we expect, the approval pathway for the product candidate will likely take significantly longer, cost significantly more and entail significantly greater complications and risks than anticipated, and in either case may not be successful.

The Drug Price Competition and Patent Term Restoration Act of 1984, also known as the Hatch-Waxman Act, added Section 505(b)(2) to the FDCA. Section 505(b)(2) permits the filing of an NDA where at least some of the information required for approval comes from studies that were not conducted by or for the applicant and for which the applicant has not obtained a right of reference. Section 505(b)(2), if applicable to us under the FDCA, would allow an NDA we submit to FDA to rely in part on data in the public domain or the FDA's prior conclusions regarding the safety and effectiveness of approved compounds, which could expedite the development program for our product candidates by potentially decreasing the amount of clinical data that we would need to generate in order to obtain FDA approval. If the FDA does not allow us to pursue the Section 505(b)(2) regulatory pathway as anticipated, we may need to conduct additional clinical trials, provide additional data and information, and meet additional standards for regulatory approval. If this were to occur, the time and financial resources required to obtain FDA approval for these product candidates, and complications and risks associated with these product candidates, would likely substantially increase. We could need to obtain more additional funding, which could result in significant dilution to the ownership interests of our then existing stockholders to the extent we issue equity securities or convertible debt. We cannot assure you that we would be obtain use hadditional financing on terms acceptable to us, if at all. Moreover, inability to pursue the Section 505(b)(2) regulatory pathway dversely impact our competitive products reaching the market more quickly than our product candidates, which would likely materially adversely impact our competitive position and prospects. Even if we are allowed to pursue the Section 505(b)(2) regulatory pathway, we cannot assure you that our product candidates will receive the requisite approvals for commercialization.



In addition, notwithstanding the approval of a number of products by the FDA under Section 505(b)(2) over the last few years, certain brand-name pharmaceutical companies and others have objected to the FDA's interpretation of Section 505(b)(2). If the FDA's interpretation of Section 505(b)(2) is successfully challenged, the FDA may change its Section 505(b)(2) policies and practices, which could delay or even prevent the FDA from approving any NDA that we submit under Section 505(b)(2). In addition, the pharmaceutical industry is highly competitive, and Section 505(b)(2) NDAs are subject to special requirements designed to protect the patent rights of sponsors of previously approved drugs that are referenced in a Section 505(b)(2) NDA. These requirements may give rise to patent litigation and mandatory delays in approval of our NDAs for up to 30 months or longer depending on the outcome of any litigation. It is not uncommon for a manufacturer of an approved product to file a citizen petition with the FDA seeking to delay approval of, or impose additional approval requirements for, pending competing products. If successful, such petitions can significantly delay, or even prevent, the approval of the petition. In addition, even if the FDA ultimately denies such a petition, the FDA may substantially delay approval while it considers and responds to the petition. In addition, even if we are able to utilize the Section 505(b)(2) regulatory pathway, there is no guarantee this would ultimately lead to faster product development or earlier approval.

Moreover, even if our product candidates are approved under Section 505(b)(2), the approval may be subject to limitations on the indicated uses for which the products may be marketed or to other conditions of approval, or may contain requirements for costly post-marketing testing and surveillance to monitor the safety or efficacy of the products.

If any of our contract manufacturers fails to produce our products in the volumes that we require on a timely basis, or to comply with stringent regulations applicable to pharmaceutical drug manufacturers, we may face delays in the commercialization of this product candidate or be unable to meet market demand, and may lose potential revenues.

The manufacture of pharmaceutical products requires significant expertise and capital investment, including the development of advanced manufacturing techniques and process controls, and the use of specialized processing equipment. Any termination or disruption of any current or future relationships relating to product development may materially harm our business and financial condition and frustrate any commercialization efforts for affected current or future product candidates.

Any current or future contract manufacturers we engage must comply with strictly enforced federal, state and foreign regulations, including cGMP requirements enforced by the FDA through its establishment inspection program. Despite the existence of contract manufacturing agreements and shared cGMP responsibilities, our contract manufacturers' may ignore these contractual provisions, or otherwise fail to meet the minimum standards set forth in the cGMP regulations, resulting in manufacturing non-compliance. This may go unnoticed or uncorrected despite our best efforts to regulatory audit or confirm the CMOs regulatory responsibilities. Any failure to comply with applicable regulations may result in fines and civil penalties, suspension of production, suspension or delay in product approval, product seizure or recall, or withdrawal of product approval, and would limit the availability of our product. Any manufacturing defect or error discovered after products have been produced and distributed could result in even more significant consequences, including costly recalls, re-stocking costs, damage to our reputation and potential for product liability claims.

If the CMOs upon which we rely to manufacture any current products, and any potential product candidates we may in-license or acquire, fail to deliver the required commercial quantities on a timely basis at commercially reasonable prices, we would likely be unable to meet demand for our products and we would lose potential revenues.

If serious adverse or unacceptable side effects are identified during the development of any current or future product candidates, we may need to abandon or limit our development of some of the other potential product candidates.

If any current or future product candidates are associated with undesirable side effects, toxicities, or other negative characteristics, we may need to abandon such products' commercialization, development or limit development to more narrow uses or subpopulations. Such side effects may affect patient recruitment or the ability of enrolled patients to complete the trial and could result in potential product liability claims. Many compounds that show initial promise in early-stage testing are later found to cause side effects that prevent further development. If our clinical trials reveal severe or prevalent side effects, our trials could be suspended or terminated, we may be unable to recruit patients and enrolled patients may be unable to complete the trials, and the FDA or comparable foreign regulatory authorities could order issue a clinical hold, or order us to cease further development or deny approval of the product candidate. The FDA may also request additional data, which it has done with increased prevalence in recent years, which has resulted in substantial delays in new drug approvals. Undesirable side effects caused by any current or future product candidates could also result in the inclusion of unfavorable information in our product labeling, denial of regulatory approval by the FDA or other regulatory authorities for any or all targeted indications, and in turn prevent us from commercializing and generating revenues from the sale of such product candidate.

If one or more of our current products or any future product candidate receives marketing approval and we or others later identify undesirable adverse events or side effects caused by this product, or we fail to comply with post-market regulatory requirements, a number of potentially significant negative consequences could result, including:

- regulatory authorities may require the addition of unfavorable labeling statements, specific warnings or a contraindication;
- regulatory authorities may suspend or withdraw their approval of the product, or require it to be removed from the market;
- we may be required to change the way the product is administered, conduct additional clinical trials or change the labeling of the product; or
- our reputation may suffer.

Any of these events could prevent us from achieving or maintaining market acceptance of any current or future product candidate or could substantially increase our development and commercialization costs and expenses, which could delay or prevent us from generating significant revenues.

All of our current and future products will remain subject to substantial regulatory scrutiny even after receiving regulatory approval.

Any products or current or future product candidates we may license or acquire will be subject to ongoing regulatory and compliance requirements and oversight by the FDA and other regulatory authorities. These requirements include labeling, packaging, storage, advertising, promotion, record-keeping and submission of safety and other post-market information and reports, registration and listing requirements, cGMP requirements relating to manufacturing, quality control, quality assurance and corresponding maintenance of records and documents, requirements regarding the distribution of samples to physicians and other licensed medical professionals and recordkeeping of the drug. The Food and Drug Administration Amendments Act of 2007 granted significant expanded authority to the FDA, much of which was aimed at improving the safety of drug products before and after approval. The FDA may also impose requirements for costly post-marketing studies or clinical trials and surveillance to monitor the safety or efficacy of the product. The FDA closely regulates the post-approval marketing and promotion of drugs to ensure drugs are marketed only for the approved indications and in accordance with the provisions of the approved labeling. The FDA imposes stringent restrictions on manufacturers' communications regarding off-label use and if we do not market our products for only their approved indications, we may be subject to enforcement action for off-label marketing. While physicians and other healthcare providers may choose to prescribe drugs for uses that are not described in the product's labeling and for uses that differ from those tested in clinical studies and approved by the regulatory authorities, our ability to promote the products is limited to those indications that are specifically approved by the FDA. These "off-label" uses are common across medical specialties and may constitute an appropriate treatment for some patients in varied circumstances. Regulatory authorities in the U.S. generally do not regulate the practice of medicine, including the clinical behavior of physicians and other healthcare providers in their choice of treatments. Regulatory authorities do; however, restrict communications by pharmaceutical companies on the subject of off-label use.

Violations of the FDCA relating to the promotion of prescription drugs may lead to investigations alleging violations of federal and state health care fraud and abuse laws, as well as state consumer protection laws.

In addition, later discovery of previously unknown adverse events or other problems with our products, manufacturers or manufacturing processes, or failure to comply with regulatory requirements, may yield various results, including:

- restrictions on such products or their manufacturers or manufacturing processes;
- restrictions on the labeling or marketing of a product;
- restrictions on product distribution or use;
- requirements to conduct post-marketing studies or clinical trials;
- warning letters, untitled letters, or Form 483s;



- withdrawal of the products from the market;
- refusal to approve pending applications or supplements to approved applications that we submit;
- recall of products;
- fines, restitution or disgorgement of profits;
- suspension or withdrawal of marketing or regulatory approvals;
- suspension of any ongoing clinical trials;
- denial of permits to import or export our products;
- product seizure; or
- injunctions or the imposition of civil or criminal penalties.

The FDA's policies may change, and additional government regulations may be enacted that could prevent, limit or delay regulatory approval of our current or future product candidates. If we are slow or unable to adapt to changes in existing requirements or the adoption of new requirements or policies, or if we are not able to maintain regulatory compliance, we may lose any marketing approval that we may have obtained.

Our current and future relationships with customers and third-party payors in the United States and elsewhere may be subject, directly or indirectly, to applicable anti-kickback, fraud and abuse, false claims, transparency, health information privacy and security and other healthcare laws and regulations, which could expose us to criminal sanctions, civil penalties, contractual damages, reputational harm, administrative burdens and diminished profits and future earnings.

Healthcare providers, physicians and third-party payors in the United States and elsewhere play a primary role in the recommendation and prescription of our current products or current or future product candidates for which we obtain marketing approval. Our current and future arrangements with third-party payors for the sales of our products and sales to customers may expose us to broadly applicable fraud and abuse and other healthcare laws and regulations, including, without limitation, the federal Anti-Kickback Statute ("AKS") and the federal False Claims Act, which may constrain the business or financial arrangements and relationships through which we sell, market and distribute any current products or current or future product candidates for which we obtain marketing approval. In addition, we may be subject to transparency laws and patient privacy regulation by U.S. federal and state governments and by affect our ability to operate include:

- AKS, which prohibits, among other things, persons from knowingly and willfully soliciting, offering, receiving or providing remuneration, directly
 or indirectly, in cash or in kind, to induce or reward, or in return for, either the referral of an individual for, or the purchase, order or
 recommendation of, any good or service, for which payment may be made under federal and state healthcare programs, such as Medicare and
 Medicaid. The OIG continues to make modifications to existing AKS safe harbors which may increase liability and risk as well as adversely impact
 sales relationships. The majority of states also have statutes or regulations similar to these federal laws, which apply to items and services
 reimbursed under Medicaid and other state programs, or, in several states, apply regardless of the payor;
- federal civil and criminal false claims laws and civil monetary penalty laws, including the federal False Claims Act, which impose criminal and civil
 penalties, including civil whistleblower or qui tam actions, against individuals or entities for knowingly presenting, or causing to be presented, to the
 federal government, including the Medicare and Medicaid programs, claims for payment that are false or fraudulent or making a false statement to
 avoid, decrease or conceal an obligation to pay money to the federal government;
- the federal Health Insurance Portability and Accountability Act of 1996 ("HIPAA"), which imposes criminal and civil liability for executing a scheme to defraud any healthcare benefit program or making false statements relating to healthcare matters;



- HIPAA, as amended by the Health Information Technology for Economic and Clinical Health Act of 2009, and their respective implementing regulations, which impose obligations on covered healthcare providers, health plans, and healthcare clearinghouses, as well as their business associates that create, receive, maintain or transmit individually identifiable health information for or on behalf of a covered entity, with respect to safeguarding the privacy, security and transmission of individually identifiable health information;
- the federal Open Payments program, which requires manufacturers of drugs, devices, biologics and medical supplies for which payment is available
 under Medicare, Medicaid or the Children's Health Insurance Program, with specific exceptions, to report annually to the Centers for Medicare &
 Medicaid Services ("CMS"), information related to "payments or other transfers of value" made to physicians, which is defined to include doctors,
 dentists, optometrists, podiatrists, chiropractors, physician assistants, nurse practitioners, clinical nurse specialists, certified nurse anesthetists,
 certified nurse-midwives and teaching hospitals and applicable manufacturers and applicable group purchasing organizations to report annually to
 CMS ownership and investment interests held by the physicians and their immediate family members;
- Increased OIG scrutiny on the sale of our products through specialty pharmacies by means of direct investigation or by issuance of unfavorable Opinion Letters which may curtail or hinder the sales of our products based on risk of enforcement upon our-selves or our buyers;
- analogous state and foreign laws and regulations, such as state anti-kickback and false claims laws, which may apply to sales or marketing arrangements and claims involving healthcare items or services reimbursed by non-governmental third-party payors, including private insurers; state and foreign laws that require pharmaceutical companies to comply with the pharmaceutical industry's voluntary compliance guidelines and the relevant compliance guidance promulgated by the federal government or otherwise restrict payments that may be made to healthcare providers;
- state and foreign laws that require drug manufacturers to report information related to payments and other transfers of value to physicians and other healthcare providers or marketing expenditures; and
- state and foreign laws governing the privacy and security of health information in certain circum-stances, many of which differ from each other in
 significant ways and often are not preempted by HIPAA, thus complicating compliance efforts.

Efforts to ensure that our business arrangements with third parties will comply with applicable healthcare laws and regulations may involve substantial costs. It is possible that governmental authorities will conclude that our business practices may not comply with current or future statutes, regulations or case law involving applicable fraud and abuse or other healthcare laws and regulations. If our operations are found to be in violation of any of these laws or any other governmental regulations that may apply to us, we may be subject to significant civil, criminal and administrative penalties, including, without limitation, damages, fines, imprisonment, exclusion from participation in government healthcare programs, such as Medicare and Medicaid, and the curtailment or restructuring of our operations, which could have a material adverse effect on our business. If any of the physicians or other healthcare providers or entities with whom we expect to do business, including our collaborators, is found not to be in compliance with applicable laws, it may be subject to criminal, civil or administrative sanctions, including exclusions from participation in government healthcare programs, which could also materially affect our business.

We have established and implemented a corporate compliance program designed to prevent, detect and correct violations of state and federal healthcare laws, including laws related to advertising and promotion of our products. Nonetheless, enforcement agencies or private plaintiffs may take the position that we are not in compliance with such requirements and, if such noncompliance is proven, the Company and, in some cases, individual employees, may be subject to significant liability, including the aforementioned administrative, civil and criminal sanctions.

We are subject to new legislation, regulatory proposals and managed care initiatives that may increase our costs of compliance and adversely affect our ability to market our products, obtain collaborators and raise capital.

In the United States and certain foreign jurisdictions, there have been, and we expect there will continue to be, a number of legislative and regulatory changes to the healthcare system that could impact our ability to sell our licensed products profitably. In March 2010, the Patient Protection and Affordable Care Act, as amended by the Health Care and Education Reconciliation Act of 2010 (the "PPACA" or collectively, the "ACA"), was signed into law, which substantially changed the way healthcare is financed by both governmental and private insurers in the United States. By way of example, the ACA: increased the minimum level of Medicaid rebates payable by manufacturers of brand name drugs from 15.1% to 23.1%; required collection of rebates for drugs paid by Medicaid managed care organizations; imposed a non-deductible annual fee on pharmaceutical manufacturers who sell certain "branded prescription drugs" to specified federal government programs; implemented a new methodology under which rebates owed by manufacturers of drugs that are inhaled, infused, instilled, implanted, or injected; expanded the eligibility criteria effectiveness research, along with funding for such research; and established a Center for Medicare and Medicaid Innovation ("CMMI") at the CMS, to test innovative payment and service delivery models to lower Medicare and Medicaid spending, potentially including prescription drug spending.

Since its enactment, there have been executive, judicial and Congressional challenges to certain aspects of the ACA, and we expect there will be additional challenges and amendments to the ACA in the future. On January 20, 2017, President Trump signed an executive order stating that his administration intended to seek prompt repeal of the Affordable Care Act, and, pending repeal, directed the U.S. Department of Health and Human Services and other executive departments and agencies to take all steps necessary to limit any fiscal or regulatory burdens of the Affordable Care Act. On January 28, 2021, President Biden signed an Executive Order on Strengthening Medicaid and the Affordable Care Act and stated his administration's intentions to reverse the actions of his predecessor and strengthen the ACA. As part of this Executive Order, the Department of Health and Human Services, United States Treasury, and the Department of Labor are directed to review all existing regulations, orders, guidance documents, policies, and agency actions to consider if they are consistent with ensuring both coverage under the ACA and making high-quality healthcare affordable and accessible to Americans. On March 11, 2021, President Biden signed into law the American Rescue Plan Act of 2021 to further strengthen Medicaid and the ACA and on April 5, 2022, President Biden signed the Executive Order on Continuing to Strengthen Americans' Access to Affordable, Quality Health Coverage in which he celebrated the significant progress his administration believes has been made across the U.S. in making healthcare more affordable and accessible. In this Executive Order, President Biden directed agencies "with responsibilities related to Americans' access to health coverage" to "review agency actions to identify ways to continue to expand the availability of affordable health coverage." The continued expansion of the government's role in the U.S. healthcare industry may further lower rates of reimbursement for pharmaceutical products. While we are

President Biden intends, as his predecessor did, to take action against drug prices which are considered "high." Drug pricing continues to be a subject of debate at the executive and legislative levels of U.S. government. The American Rescue Plan Act of 2021 signed into law by President Biden on March 14, 2021 includes a provision that will eliminate the statutory cap on rebates drug manufacturers pay to Medicaid beginning in January 2024. With the elimination of the rebate cap, manufacturers may be required to compensate states in an amount greater than what the state Medicaid programs pay for the drug. Additionally, the Inflation Reduction Act of 2022 contains substantial drug pricing reforms, including the establishment of a drug price negotiation program within the U.S. Department of Health and Human Services that would require manufacturers to charge a negotiated "maximum fair price" for certain selected drugs or pay an excise tax for noncompliance, the establishment of rebate payment requirements on manufacturers of certain drugs payable under Medicare Parts B and D to penalize price increases that outpace inflation, and requires manufacturers to provide discounts on Part D drugs. Substantial penalties can be assessed for noncompliance with the drug pricing provisions in the Inflation Reduction Act of 2022. The Inflation Reduction Act of 2022 could have the effect of reducing the prices we can charge and reimbursement we receive for our products, if approved, thereby reducing our profitability, and could have a material adverse effect on our financial condition, results of operations and growth prospects. The effect of Inflation Reduction Act of 2022 on our business and the pharmaceutical industry in general is not yet known.

At the state level, legislatures have increasingly passed legislation and implemented regulations designed to control pharmaceutical product pricing, including price or patient reimbursement constraints, discounts, restrictions on certain product access and marketing cost disclosure and transparency measures, and, in some cases, designed to encourage importation from other countries and bulk purchasing. We expect that additional federal, state and foreign healthcare reform measures will be adopted in the future, any of which could limit the amounts that federal and state governments will pay for healthcare products and services, which could result in limited coverage and reimbursement and reduced demand for our products, once approved, or additional pricing pressures.

These and other healthcare reform measures that may be adopted in the future may result in more rigorous coverage criteria and in additional downward pressure on the price that we receive for any current product or future product candidate. Any reduction in reimbursement from Medicare or other government healthcare programs may result in a similar reduction in payments from private payors. The implementation of cost containment measures or other healthcare reforms may prevent us from being able to generate revenue, attain profitability or commercialize our products. Legislative and regulatory proposals have been made to expand post-approval requirements and restrict sales and promotional activities for drugs. We cannot be sure whether additional legislative changes will be enacted, or whether the FDA regulations, guidance or interpretations will be changed, or what the impact of such changes on the marketing approvals of any current or future product candidates, if any, may be. In addition, increased Congressional scrutiny of the FDA's approval process may significantly delay or prevent marketing approval, as well as subject us to more stringent product labeling and post-marketing testing and other requirements.

Public concern regarding the safety of any of our current or future drug products could delay or limit our ability to obtain regulatory approval, result in the inclusion of unfavorable information in our labeling, or require us to incur additional costs.

In light of widely publicized events concerning the safety risk of certain drug products, the FDA, members of Congress, the Government Accountability Office, medical professionals and the general public have raised concerns about potential drug safety issues. These events have resulted in the withdrawal of drug products, revisions to drug labeling that further limit use of the drug products, and the establishment of risk management programs. The increased attention to drug safety issues may result in a more cautious approach by the FDA in its review of data from our clinical trials. Data from clinical trials may receive greater scrutiny, particularly with respect to safety, which may make the FDA or other regulatory authorities more likely to require additional preclinical studies or clinical trials. If the FDA requires us to conduct additional preclinical studies or clinical trials or proving any other potential future product candidate, our ability to obtain of such product candidate, the indications for which such product candidate is approved may be limited or there may be specific warnings or limitations on dosing, and our efforts to commercialize potential future product candidate may be otherwise adversely impacted.

If we experience delays or difficulties in the enrollment of patients in any future clinical trials, our receipt of necessary regulatory approvals could be delayed or prevented.

We may not be able to initiate any future clinical trials for any current or future product candidates if we are unable to locate and enroll a sufficient number of eligible patients to participate in these trials as required by the FDA or similar regulatory authorities outside the United States. Some of our competitors may have ongoing clinical trials for product candidates that treat the same indications as our current or potential future product candidates, and patients who would otherwise be eligible for any future clinical trials may instead enroll in clinical trials of our competitors' product candidates. Patient enrollment is affected by other factors, including:

- the severity of the disease under investigation;
- the eligibility criteria for a study;
- the perceived risks and benefits of the product candidate under study;
- the efforts to facilitate timely enrollment in clinical trials;
- the patient referral practices of physicians;
- · the ability to monitor patients adequately during and after treatment; and



the proximity and availability of clinical trial sites for prospective patients.

Our inability to enroll a sufficient number of patients for any future clinical trials would result in significant delays and could require us to abandon any future clinical trials altogether. Enrollment delays in any future clinical trials may result in increased development costs for any current or future product candidates, which would cause the value of our company to decline and limit our ability to obtain additional financing.

We expect intense competition for our products and current or future product candidates, and new products may emerge that provide different or better therapeutic alternatives for our targeted indications.

We face, and will continue to face, competition in the development and marketing of products from academic institutions, government agencies, research institutions and biotechnology and pharmaceutical companies, including specialty and other large pharmaceutical companies, and OTC companies and generic manufacturers. The dermatology competitive landscape is highly fragmented, with many mid-size and smaller companies competing in the prescription sector. Our competitors are pursuing the development and/or acquisition of pharmaceuticals, medical devices and OTC products targeting the same diseases, conditions, and indications as our products. There can be no assurance that our competitors' developments, including the development of other drug technologies and methods of preventing the incidence of disease, will not render our current products or current or future product candidates obsolete or noncompetitive.

If patents covering any of our currently marketed products expire or are successfully challenged, or when the regulatory or licensed exclusivity for our products expires or is otherwise lost, we will face increased competition from generic versions of our products. Generic versions are generally significantly less expensive than branded versions and third-party reimbursement programs may require or prefer that a generic version is used before the branded version. Accordingly, when a branded product loses market exclusivity, the product faces intense price competition from generic versions. To successfully compete for business with managed care and pharmacy benefits management organizations, we must demonstrate that our products offer medical and cost advantages when compared with other forms.

Competitive factors vary by product line and geographic area in which the products are sold. The principal methods of competition for our products include quality, efficacy, market acceptance, price, and marketing and promotional efforts. The commercial opportunity for our products and/or product future candidates could be significantly harmed if competitors are able to develop alternative formulations outside the scope of our in-licensed intellectual property. Many of our potential competitors have substantially greater capital resources, development resources, including personnel and technology, clinical trial and regulatory experience, expertise in the prosecution of intellectual property rights, and manufacturing, distribution, and sales and marketing than we do.

As a result of these factors, our competitors may obtain regulatory approval of their products more rapidly than we are able to or may obtain patent protection or other intellectual property rights that limit our ability to develop or commercialize any current or future product candidates. Our competitors may also develop drugs or products that are more effective, safe, useful and less costly than ours and may be more successful than us in manufacturing and marketing their drugs or products. If we are unable to compete effectively, our business, our business, prospects, results of operations, financial condition or cash flows may be materially adversely affected.

If our products do not achieve broad market acceptance, including by government and third-party payors, the revenues that we generate from sales will be limited.

The commercial success of our products or any current or future product candidates will depend upon their acceptance by the medical community and coverage and reimbursement for our products by third-party payors, including government payors. The degree of market acceptance of our products or any other potential product candidate we may develop, license or acquire will depend on a number of factors, including:

- the success of any potential clinical studies during the drug development process;
- limitations of use, contraindications, or warnings contained in the product's FDA-approved labeling;
- changes in the standard of care for the targeted indications for any current or future product candidates, which could reduce the marketing impact of
 any superiority claims that we could make following FDA approval;



- ability to be listed on formularies (lists of recommended or approved medicines and other products) and reimbursement lists by demonstrating the qualities and treatment benefits of our products within their approved indications; and
- potential advantages over, and availability of, alternative treatments.

Our ability to effectively promote and sell our products and any other current or future product candidates we may develop, license or acquire in the marketplace will also depend on pricing and cost effectiveness, including our ability to produce a product at a competitive price and achieve acceptance of the product onto formularies, as well as our ability to obtain sufficient third-party coverage or reimbursement. Since many insurance plans are members of group purchasing organizations, which leverage the purchasing power of a group of entities to obtain discounts based on the collective buying power of the group, our ability to attract customers in the marketplace will also depend on our ability to effectively promote any current or future product candidates to group purchasing organizations. We will also need to demonstrate acceptable evidence of safety and efficacy, as well as relative convenience and ease of administration. Market acceptance could be further limited depending on the prevalence and severity of any expected or unexpected adverse side effects associated with any current or future product candidates. If any current or future products, and we may not become or remain profitable. In addition, our efforts to educate the medical community and third-party payors on the benefits of any current or future product candidates may require significant resources and may never be successful.

Further, in both domestic and foreign markets, any future product sales will depend in part upon the availability of coverage and reimbursement from thirdparty payors. Such third-party payors include government health programs such as Medicare, managed care providers, private health insurers and other organizations. We may need to conduct post-marketing studies in order to demonstrate the cost-effectiveness of any future products to the satisfaction of target customers and their third-party payors. Such studies might require us to commit a significant amount of management time and financial and other resources. Our current or future products might not ultimately be considered cost-effective. Adequate third-party coverage and reimbursement might not be available to enable us to maintain price levels sufficient to realize an appropriate return on investment in product development.

Risks Related to Our Reliance on Third Parties

If we are unable to maintain sales, marketing, and distribution capabilities, or to enter into agreements with third parties to market and sell current or future product candidates, we may not be successful in generating revenues from selling and commercializing any such product candidates.

In order to commercialize any current or future product candidates that have not yet received marketing approval or for which we have not yet acquired rights, we may need to build additional marketing, sales, distribution, managerial and other non-technical capabilities or make arrangements with third parties to perform these services tailored to those products, and we may not be successful in doing so. In the event of successful development and regulatory approval of any potential new product candidate, or a new product acquisition, we expect to build a targeted specialist field sales force to market or co-promote that specific product. There are risks involved with establishing our own sales, marketing and distribution capabilities. For example, recruiting and training a field sales force is expensive and time consuming and could delay any product launch. If the commercial launch of a future product candidate or acquired product for which we recruit a field sales force and establish marketing capabilities is delayed or does not occur for any reason, we would have prematurely or unnecessarily incurred these commercialization expenses. This may be costly, and our investment would be lost if we cannot retain or reposition our sales and marketing personnel.

Factors that may inhibit our efforts to maintain our current products' marketing and sales organizations and/or commercialize any future products on our own include:

- our inability to recruit, train and retain adequate numbers of effective sales and marketing personnel;
- the inability of sales personnel to obtain access to physicians and other healthcare providers or persuade adequate numbers of physicians and other healthcare providers to prescribe any future products;
- the lack of complementary or other products to be offered by sales personnel, which may put us at a competitive disadvantage from the perspective
 of sales efficiency relative to companies with more extensive product lines; and
- unforeseen costs and expenses associated with creating an independent sales and marketing organization.

We are dependent on third parties to supply raw materials used in our products, to manufacture our products, and to provide services for certain core aspects of our business. Any interruption or failure by these suppliers, distributors, and collaboration partners to meet their contractual obligations to us or obligations pursuant to applicable laws and regulations may materially adversely affect our business, financial condition, results of operations and cash flows.

We rely on third parties to supply raw materials, to manufacture, warehouse, and distribute our products, as well as to provide customer service support, medical affairs services, clinical studies, sales, and other technical and financial services. All third-party suppliers and contractors are subject to FDA requirements, as well as those of comparable regulatory authorities. Our business and financial viability are dependent on the continued supply of goods and services by these third parties, the regulatory compliance of these third parties and on the strength, validity and terms of our various contracts with these third parties. Any interruption or failure by our suppliers, distributors and collaboration partners to meet their obligations pursuant to various agreements with us on schedule or in accordance with our expectations, misappropriation of our proprietary information, including trade secrets and knowhow, or any termination by these third parties of their arrangements with us, which, in each case, could be the result of one or many factors outside of our control, could delay or prevent the future development, future approval, manufacture or commercialization of our products, result in non-compliance with applicable laws and regulations, cause us to incur failure-to-supply penalties with our wholesale customers, disrupt our operations and cash flows. We may also be unsuccessful in resolving any underlying issues with such suppliers, distributors and partners or replacing them within a reasonable time and on commercially reasonable terms.

We do not expect to have the resources or capacity to commercially manufacture any future approved product candidates ourselves. We will likely continue to be heavily dependent upon third-party manufacturers, over whose manufacturing practices and processes we will have oversight, but not direct control, which may adversely affect our ability to develop and commercialize products in a timely or cost-effective manner, if at all. If any of our third-party manufacturers should become unavailable to us for any reason, including as a result of capacity constraints, differing priorities, financial difficulties or insolvency, we would likely incur added costs and delays in identifying or qualifying replacements. We may be unable to establish agreements with such replacement manufacturers or to do so on terms acceptable to us, and our reputation, business, financial condition and results of operations could be negatively impacted.

The pharmaceutical manufacturing process requires significant expertise and capital investment, including the development of advanced manufacturing techniques and process controls, and the use of specialized processing equipment. Further, the CMOs with which we contract must comply with strictly enforced federal, state, and foreign regulations, including the cGMP requirements enforced by the FDA. We will rely on our CMOs to comply with all such regulatory requirements, including cGMP requirements, and failure to do so may result in fines and civil penalties, suspension of production, suspension, delay, or withdrawal of product approval, product seizure or recall, and may limit the availability of our product. Any manufacturing defect or error discovered after products have been produced and distributed could result in costly recall procedures, re-stocking costs, damage to our reputation and potential for product liability claims. The FDA would likely hold us ultimately responsible for any product our CMO manufactures and regulatory enforcement for failure to meet FDA requirements would impact both the CMO and ourselves. The FDA considers the owners of drug products to be ultimately responsible for their products, even where a CMO or other third-party manufacturing activities, any failure by a CMO to meet the requirements of the regulations would have an adverse impact on both the CMO and ourselves.

We also may rely on third-party manufacturers to purchase from third-party suppliers the materials necessary to produce our current or future product candidates for anticipated clinical trials. There are a small number of suppliers for certain capital equipment and raw materials that are used to manufacture those products. We do not have any control over the process or timing of the acquisition of these raw materials by our third-party manufacturers. Moreover, we currently do not have any agreements for the commercial production of these raw materials. Any significant delay in the supply of raw material components related to an ongoing clinical trial could considerably delay completion of our clinical trials, product testing and potential regulatory approval.

We rely, and expect to continue to rely, on third parties to conduct any future preclinical studies and clinical trials, and those third parties may not perform satisfactorily, including by failing to meet deadlines for the completion of such trials or to comply with applicable regulatory requirements.

We expect to rely on third-party contract and clinical research organizations, clinical data management organizations, and medical institutions and clinical investigators to conduct future preclinical studies and clinical trials. Any future agreements with these third parties might terminate for a variety of reasons, including a failure to perform by the third parties. If we need to enter into alternative arrangements, that could delay any future product development activities.

Our reliance on any third parties for research and development activities will reduce our own control over these activities but will not relieve us of our responsibilities. We will remain responsible for ensuring that each of any future preclinical studies and clinical trials are conducted in accordance with the general investigational plan and protocols for the trial and for ensuring that any future preclinical studies are conducted in accordance with good laboratory practice ("GLP") as appropriate. Moreover, the FDA requires us to comply with standards, commonly referred to as good clinical practices ("GCPs") for conducting, recording and reporting the results of clinical trials to assure that data and reported results are credible and accurate and that the rights, integrity and confidentiality of trial participants are protected. Regulatory authorities enforce these requirements through periodic inspections of trial sponsors, clinical trials investigators and trial sites. If we or any of our future clinical research organizations fail to comply with applicable GCPs, the clinical trials before approving our marketing applications. We cannot assure you that any such regulatory authorities may require us to perform additional clinical trials before approving our marketing applications. In addition, any future clinical trials must be conducted with product produced under cGMP regulations and subject to an IND. Our failure to comply with these regulations may require us to repeat clinical trials, which would delay the regulatory approval process. We also are required to register ongoing clinical trials and post the results of completed clinical trials on a government-sponsored database, ClinicalTrials.gov, within specified timeframes. Failure to do so can result in fines, adverse publicity and civil and criminal sanctions.

The third parties with whom we may contract to help perform future preclinical studies or clinical trials may also have relationships with other entities, some of which may be our competitors. If these third parties do not successfully carry out their contractual duties, meet expected deadlines or conduct our preclinical studies or clinical trials in accordance with regulatory requirements or our stated protocols, we will not be able to obtain, or may be delayed in obtaining, marketing approvals for any current or future product candidates and will not be able to, or may be delayed in our efforts to, successfully commercialize such product candidates.

If any of our future relationships with these third-party contract research organizations or clinical research organizations terminate, we may not be able to enter into arrangements with alternative contract research organizations or clinical research organizations or to do so on commercially reasonable terms. Switching or additional contract research organizations or clinical research organizations involves additional cost and requires management time and focus. In addition, there is a natural transition period when a new contract research organization or clinical research organization commences work. As a result, delays could occur, which could compromise our ability to meet our desired development timelines. Though we will carefully manage any future relationships with contract research organizations or clinical research organizations, there can be no assurance that we will not encounter similar challenges or delays in the future.

We rely on clinical data and results obtained by third parties that could ultimately prove to be inaccurate or unreliable.

As part of our strategy to mitigate development risk, we intend on developing product candidates with validated mechanisms of action and assess potential clinical efficacy early in the development process or otherwise acquire the rights to products for which marketing approval has already been obtained. This strategy necessarily relies upon clinical data and other results obtained by third parties that may ultimately prove to be inaccurate or unreliable. If the third-party data and results we rely upon prove to be inaccurate, unreliable or not applicable to future product candidates or acquired products, we could make inaccurate assumptions and conclusions about current or future product candidates and our research and development efforts could be compromised.

If successful product liability claims are brought against us, we may incur substantial liability, and may have to limit the commercialization of certain current or future products or product candidates.

The use of our products and any current or future product candidate we may license, acquire or develop in clinical trials and the sale of any products for which we obtain marketing approval expose us to the risk of product liability claims. For example, we may be sued if any product or product candidate we develop or sell allegedly causes injury or is found to be otherwise unsuitable during clinical testing, manufacturing, marketing, or sale. Product liability claims might be brought against us by consumers, health care providers or others who use, administer, or sell our products. If we cannot successfully defend ourselves against these claims, we will incur substantial liabilities. Regardless of merit or eventual outcome, liability claims may result in:

- termination of clinical trial sites or entire trial programs or withdrawal of clinical trial participants;
- regulatory investigations by governmental authorities related to regulatory issues or alleged non-compliances;
- litigation costs and potential monetary awards to patients or other claimants;

- harm to our reputation and/or decreased demand for our products and corresponding revenue loss;
- reduced resources of our management to pursue our business strategy; and
- the inability to commercialize our current products or any current or future product candidates.

We have obtained or will obtain limited product liability insurance coverage for any and all current or future clinical trials. However, our insurance coverage may not reimburse us or may not be sufficient to reimburse us for any expenses or losses we may suffer. Moreover, insurance coverage is becoming increasingly expensive, and, in the future, we may not be able to maintain insurance coverage at a reasonable cost or in sufficient amounts to protect us against losses due to liability. Our current insurance for any products the sale of commercial products, but we may be unable to maintain or obtain commercially reasonable product liability insurance for any products approved for marketing. On occasion, large judgments have been awarded in class action lawsuits based on drugs that had unanticipated side effects. A successful product liability claim or series of claims brought against us could cause our stock price to fall and, if judgments exceed our insurance coverage, could decrease our cash and materially adversely affect our business, results of operations, financial condition or cash flows.

We began marketing and promoting Accutane®, an isotretinoin product in the second quarter of 2021. Isotretinoin has a black box warning for use in pregnant women. Isotretinoin also has warnings for side effects related to psychiatric disorders and inflammatory bowel disease, among others. Historically, isotretinoin has been the subject of significant product liability claims, mainly related to irritable bowel disease. Currently, there is no significant isotretinoin product liability litigation. In 2014, the federal multi-district litigation ("MDL") court ruled that the warning label for isotretinoin was adequate and dismissed all remaining federal isotretinoin cases. The MDL dissolved in 2015, effectively ending federal isotretinoin lawsuits. Isotretinoin cases continued in New Jersey state court until 2017, when the trial court judge dismissed the remaining isotretinoin product liability claim arise related to isotretinoin. However, we cannot predict the ultimate outcome of any litigation and the Company may be required to pay significant amounts as a result of settlement or judgments should any new product liability claim be brought.

We are subject to numerous environmental, health and safety laws and regulations, including those governing laboratory procedures and the handling, use, storage, treatment and disposal of hazardous materials and wastes. Our operations involve the use of hazardous and flammable materials, including chemicals and biological materials. Our operations also produce hazardous waste products. We generally contract with third parties for the disposal of these materials and wastes. We cannot eliminate the risk of contamination or injury from these materials. Although we believe that the safety procedures for handling and disposing of these materials comply with the standards prescribed by these laws and regulations, we cannot eliminate the risk of accidental contamination or injury from these materials. In the event of contamination or injury resulting from our use of hazardous materials, we could be held liable for any resulting damages, and any liability could exceed our resources. If we fail to comply with environmental, health and safety laws and regulations, we could become subject to fines or penalties or incur costs that could harm our business.

Although we maintain workers' compensation insurance to cover us for costs and expenses we may incur due to injuries to our employees resulting from the use of hazardous materials, this insurance may not provide adequate coverage against potential liabilities. We do not maintain insurance for environmental liability or toxic tort claims that may be asserted against us in connection with our storage or disposal of biological, hazardous or radioactive materials.

In addition, we may incur substantial costs in order to comply with current or future environmental, health and safety laws and regulations. These current or future laws and regulations may impair our research, development or production efforts. Our failure to comply with these laws and regulations also may result in substantial fines, penalties or other sanctions.

Risks Related to our Growth

A significant part of our future growth may depend on our ability to identify and acquire or in-license products, and if we do not successfully identify and acquire or in-license related product candidates or integrate them into our operations, we may have limited growth opportunities.

An important part of our business strategy is to continue to develop a pipeline of product candidates by acquiring or in-licensing products, product candidates, businesses or technologies that we believe are a strategic fit with our focus on the dermatological marketplace. Future in-licenses or acquisitions may entail numerous operational and financial risks, including:

- exposure to unknown liabilities;
- disruption of our business and diversion of our management's time and attention to develop acquired products or technologies;
- difficulty or inability to secure financing to fund development activities for such acquired or in-licensed technologies in the current economic environment;
- incurrence of substantial debt or dilutive issuances of securities to pay for acquisitions;
- higher than expected acquisition and integration costs;
- increased amortization expenses;
- difficulty and cost in combining the operations and personnel of any acquired businesses with our operations and personnel;
- impairment of relationships with key suppliers or customers of any acquired businesses due to changes in management and ownership; and
- inability to retain key employees of any acquired businesses.

We have limited resources to identify and execute the acquisition or in-licensing of third-party products, current or future product candidates, businesses, and technologies and to integrate them into our current infrastructure. As a result, we focus on research programs and product candidates that we identify for specific indications, which may cause us to forego or delay pursuit of opportunities with other product candidates or for other indications that may have greater commercial potential. Further, we may devote resources to potential acquisitions or in-licensing opportunities that are ultimately not completed or of which we do not realize the anticipated benefits. Our resource allocation decisions may cause us to fail to capitalize on viable commercial products or profitable market opportunities. Our spending on future research and development programs and product candidates for specific indications may not yield any commercially viable products. If we do not accurately evaluate the commercial potential or target market for a particular product candidate, we may relinquish valuable rights to that product candidate through collaboration, licensing or other royalty arrangements in cases in which it would have been more advantageous for us to retain sole development and commercialization rights to such product candidate.

Additionally, we may compete with larger pharmaceutical companies and other competitors for new collaborations and in-licensing opportunities. These competitors likely will have greater financial resources than we do and may have greater expertise in identifying and evaluating new opportunities. The realization of any of the foregoing risks related to our acquisition and in-license strategy could materially adversely affect our business, results of operations, financial condition or cash flows.

Our growth is subject to economic and political conditions.

Our business is affected by global and local economic and political conditions as well as the state of the financial markets, inflation, recession, financial liquidity, currency volatility, growth, and policy initiatives. There can be no assurance that global economic conditions and financial markets will not worsen and that we will not experience any adverse effects that may be material to our consolidated cash flows, results of operations, financial position or our ability to access capital, such as the adverse effects resulting from a prolonged shutdown in government operations both in the United States and internationally. Political changes, including war or other conflicts, some of which may be disruptive, could interfere with our supply chain, our customers and all of our activities in a particular location.

Our operating history may make it difficult to evaluate our business and prospects as it relates to clinical trials or regulatory approvals.

We were incorporated in October 2014 and have only been conducting commercial operations with respect to our products since 2015. We have not yet demonstrated an ability to successfully complete clinical trials or obtain regulatory approvals. Consequently, any predictions about our future performance may not be as accurate as they could be if we had a history of successfully developing and commercializing future pharmaceutical products.

In addition, we may encounter unforeseen expenses, difficulties, complications, delays, and other known and unknown factors. We will need to expand our capabilities to support any future commercial activities. We may not be successful in adding such capabilities.

We expect our financial condition and operating results to continue to fluctuate from quarter to quarter and year to year due to a variety of factors, many of which are beyond our control. Accordingly, you should not rely upon the results of any past quarterly period as an indication of future operating performance.

We may not be able to manage our business effectively if we are unable to attract and retain key personnel.

We may not be able to attract or retain qualified management and commercial, sales, scientific and clinical personnel in the future due to the intense competition for qualified personnel among biotechnology, pharmaceutical and other businesses. If we are not able to attract and retain necessary personnel to accomplish our business objectives, we may experience constraints that will significantly impede sales growth of our branded and generic products, the achievement of our development objectives, our ability to raise additional capital and our ability to implement our business strategy.

We may decide to sell assets, which could adversely affect our prospects and opportunities for growth.

We may from time to time consider selling certain assets if we determine that such assets are not critical to our strategy or we believe the opportunity to monetize the asset is attractive or for various other reasons, including for the reduction of indebtedness. Although our expectation is to engage in asset sales only if they advance or otherwise support our overall strategy, we may be forced to sell assets in response to liquidation or other claims described herein, and any such sale could reduce the size or scope of our business, our market share in particular markets or our opportunities with respect to certain markets, products or therapeutic categories. As a result, any such sale could have a material adverse effect on our business, financial condition, results of operations and cash flows.

There is substantial doubt regarding our ability to continue as a going concern. We may need to raise additional funding (which may not be available on acceptable terms to the Company, or at all) and/or to delay, limit or terminate certain of our product development and commercialization efforts or other operations.

Our current assumptions, projected commercial sales of our products, clinical development plans and regulatory submission timelines are uncertain and may not emerge as expected. Additionally, as a result of recurring losses from operations, we have concluded that there is substantial doubt regarding our ability to continue as a going concern for a period of at least 12 months from the date of the issuance of the financial statements included in this Annual Report on Form 10-K for the year ended December 31, 2023. In addition to reductions in sales force and marketing expenses, we may also seek to raise capital through additional debt or equity financing, which may include sales of securities under our existing shelf registration statement on Form S-3, including under the Sales Agreement with B. Riley, or under a new registration statement.



Our efforts to raise additional funding may divert our management from its day-to-day activities, which may adversely affect our ability to develop and commercialize our products. In addition, we cannot guarantee that financing will be available in sufficient amounts or on terms acceptable to us, if at all. Moreover, the terms of any financing may adversely affect the holdings or the rights of our stockholders and the issuance of additional securities, whether equity or debt, by us, or the possibility of such issuance, may cause the market price of our common stock to decline. The sale of additional equity or convertible securities would dilute all of our stockholders. Potential indebtedness in addition to our current facility with SWK, if incurred, would result in increased fixed payment obligations, and we may be required to agree to further certain restrictive covenants, such as limitations on our ability to incur additional debt, limitations on our ability to acquire, sell or license intellectual property rights and other operating restrictions that could adversely impact our ability to conduct our business. We may be required to relinquish rights to some of our technologies or products or otherwise agree to terms unfavorable to us, any of which may have a material adverse effect on our business, operating results and prospects.

If funding for our operations is not available or not available on terms acceptable to us, our strategic plans may be limited. In addition, in order to address our current funding constraints, we may be required to further revise our business plan and strategy, which may result in us (i) significantly curtailing, delaying or discontinuing our DFD-29 research or development programs or the commercialization of any other products, (ii) selling certain of our assets and/or (iii) being unable to expand our operations or otherwise capitalize on our business opportunities. Such measures may become necessary whether or not we are able to raise additional capital. As a result, our business, financial condition, and results of operations could be materially affected.

Risks Related to Development and Regulatory Approval of Our Product Candidates

Our business is dependent on the successful development and regulatory approval of our current and any future product candidates.

As of December 31, 2023, our major marketed products that have been approved by the FDA for sale in the United States include Qbrexza®, Accutane®, Amzeeq®, Zilxi®, Exelderm®, Targadox® and Luxamend®. However, our business remains dependent on the successful development and regulatory approval of additional product candidates.

On June 29, 2021, we entered into a license, collaboration, and assignment agreement with DRL to initiate a Phase III clinical development program for a collaborative product candidate, DFD-29, that is being evaluated for the treatment of inflammatory lesions of rosacea. The success of our business, including our ability to finance our company and generate additional revenue in the future, may depend on the successful development and regulatory approval of the DFD-29 product candidate and any future product candidates that we may develop, in-license or acquire.

The clinical success of our current and any future product candidates will depend on a number of factors, including the following:

- our ability to raise additional capital on acceptable terms, or at all;
- timely completion of our clinical trials, which may be significantly slower or cost more than we currently anticipate and will depend substantially
 upon the performance of third-party contractors as well as our ability to timely recruit and enroll patients in our clinical trials, which may be delayed
 due to numerous factors, including the prevalence of other companies' clinical trials for their product candidates for the same or similar indications;
- whether we are required by the FDA or similar foreign regulatory agencies to conduct additional clinical trials or other studies beyond those planned to support the approval and commercialization of our current or any future product candidates;
- acceptance of our proposed indications and primary endpoint assessments relating to the proposed indications of our current or any future product candidates by the FDA and similar foreign regulatory authorities;
- our ability to demonstrate to the satisfaction of the FDA and similar foreign regulatory authorities the safety, efficacy and acceptable risk to benefit
 profile of our current or any future product candidates;
- the prevalence, duration and severity of potential side effects experienced with our current or any future product candidates;
- the timely receipt of necessary marketing approvals from the FDA and similar foreign regulatory authorities;

- achieving and maintaining, and, where applicable, ensuring that our third-party contractors achieve and maintain, compliance with our contractual
 obligations and with all regulatory requirements applicable to our current or any future product candidates;
- our ability to successfully obtain the substances and materials used in our current or any future product candidates from third parties and to have finished products manufactured by third parties in accordance with regulatory requirements and in sufficient quantities for preclinical and clinical testing;
- the ability of third parties with whom we contract to manufacture clinical trial supplies of our current or any future product candidates, remain in good standing with regulatory agencies and develop, validate and maintain commercially viable manufacturing processes that are compliant with cGMP; and
- a continued acceptable safety profile during clinical development of our current or any future product candidates.

If we do not achieve one or more of these factors, many of which are beyond our control, in a timely manner or at all, we could experience significant delays or an inability to successfully complete and obtain regulatory approvals of our current or any future product candidates, which could materially adversely affect our business, results of operations, financial condition or cash flows.

Clinical drug development is very expensive, time-consuming and uncertain. Our clinical trials may fail to adequately demonstrate the safety and efficacy of our current or any future product candidates, which could prevent or delay regulatory approval and commercialization.

Clinical drug development is very expensive, time-consuming and difficult to design and implement, and its outcome is inherently uncertain. Before obtaining regulatory approval for the commercial sale of a product candidate, we must demonstrate through clinical trials that a product candidate is both safe and effective for use in the target indication. Most product candidates that commence clinical trials are never approved by regulatory authorities for commercialization. The clinical trials for these product candidates may take significantly longer than expected to complete. In addition, we, any partner with which we currently or may in the future collaborate, the FDA, an institutional review board ("IRB") or other regulatory authorities, including state and local agencies and counterpart agencies in foreign countries, may suspend, delay, require modifications to or terminate our clinical trials at any time, for various reasons, including:

- discovery of serious or unexpected adverse events, toxicities, or side effects experienced by study participants or other safety issues;
- lack of effectiveness of any product candidate during clinical trials or the failure of a product candidate to meet specified endpoints;
- slower than expected rates of subject recruitment and patient enrollment in clinical trials resulting from numerous factors, including the prevalence
 of other companies' clinical trials for their product candidates for the same indication, such as atopic dermatitis;
- difficulty in retaining subjects who have initiated participation in a clinical trial but may withdraw at any time due to adverse side effects from the therapy, insufficient efficacy, fatigue with the clinical trial process or for any other reason;
- difficulty in obtaining IRB approval for studies to be conducted at each site;
- delays in manufacturing or obtaining, or inability to manufacture or obtain, sufficient quantities of materials for use in clinical trials;
- inadequacy of or changes in our manufacturing process or the product formulation or method of delivery;
- changes in applicable laws, regulations and regulatory policies;
- delays or failure in reaching agreement on acceptable terms in clinical trial contracts or protocols with prospective CROs, clinical trial sites and other third-party contractors;

- inability to add a sufficient number of clinical trial sites;
- uncertainty regarding proper dosing;
- failure of our contract research organizations ("CROs") or other third-party contractors to comply with contractual and regulatory requirements or to
 perform their services in a timely or acceptable manner;
- failure by us, our employees, our CROs or their employees or any partner with which we may collaborate or their employees to comply with
 applicable FDA or other regulatory requirements relating to the con-duct of clinical trials or the handling, storage, security and recordkeeping for
 drug and biologic products;
- scheduling conflicts with participating clinicians and clinical institutions;
- failure to design appropriate clinical trial protocols;
- inability or unwillingness of medical investigators to follow our clinical protocols;
- · difficulty in maintaining contact with subjects during or after treatment, which may result in incomplete data; or
- insufficient data to support regulatory approval.

We or any partner with which we may collaborate may suffer significant setbacks in our clinical trials similar to the experience of a number of other companies in the pharmaceutical and biotechnology industries, even after receiving promising results in earlier trials. In the event that we or our potential partners abandon or are delayed in the clinical development efforts related to our current or any future product candidates, we may not be able to execute on our business plan effectively and our business, financial condition, operating results and prospects would be harmed.

We expect to rely on third-party CROs and other third parties to conduct and oversee our clinical trials, other aspects of our product development and our regulatory submission process for our product candidates. If these third parties do not meet our requirements, conduct the trials as required or otherwise provide services as anticipated, we may not be able to satisfy our contractual obligations or obtain regulatory approval for, or successfully commercialize, our current or any future product candidates when expected or at all.

We expect to rely on third-party CROs and other third parties to conduct and oversee our clinical trials, other aspects of our product development and our regulatory submission process. We will also rely upon various medical institutions, clinical investigators and contract laboratories to conduct our trials in accordance with our clinical protocols and all applicable regulatory requirements, including the FDA's regulations and GCPs, which are meant to protect the rights, integrity, and confidentiality of study subjects and to define the roles of clinical trial sponsors, administrators and monitors, and state regulations governing the handling, storage, security and recordkeeping for drug and biologic products. These CROs and other third parties play a significant role in the conduct of our clinical trials, the subsequent collection and analysis of data from the clinical trials, the preparation for and submission of our filings with the FDA and comparable foreign regulatory authorities and the successful commercialization of our product.

We rely heavily on third parties for the execution of our clinical trials and preclinical studies, and control only certain aspects of their activities. For example, our agreement with DRL for the regulatory submission and approval for DFD-29 is heavily reliant on DRL's ability to conduct clinical manufacturing for clinical supply of product, attending FDA meetings, advising on the Phase III study design, assisting in identifying third-party CROs, and drafting and advising on the NDA and other regulatory submissions. We and our CROs and other third-party contractors are required to comply with GCP and GLP requirements, which are regulations and guidelines enforced by the FDA and comparable foreign regulatory authorities for products in clinical development. Regulatory authorities enforce these GCP and GLP requirements through periodic inspections of trial sponsors, principal investigators and trial sites. If we or any of these third parties fail to comply with applicable GCP and GLP requirements, the clinical data generated in our clinical trials may be deemed unreliable and the FDA or other regulatory authorities may not accept or data, or may require us to perform additional clinical trials before approving our or our partners' marketing applications. We cannot provide assurances that upon inspection by a given regulatory authority, such regulatory authority will determine that any of our clinical trials or preclinical studies complies with applicable GCP and GLP requirements. In addition, our clinical trials must generally be conducted with products manufactured and produced under cGMP



regulations. Our failure to comply with these regulations and policies may require us to repeat clinical trials, which would delay the regulatory approval process.

If any of our CROs or clinical trial sites terminate their involvement in our clinical trials for any reason, we may not be able to enter into arrangements with alternative CROs or clinical trial sites in a timely manner, or do so on commercially reasonable terms or at all. In addition, if our relationship with clinical trial sites is terminated, we may experience the loss of follow-up information on patients enrolled in our ongoing clinical trial unless we are able to transfer the care of those patients to another qualified clinical trial site. In addition, principal investigators for our clinical trials may serve as scientific advisors or consultants to us from time to time and could receive cash compensation in connection with such services. If these relationships and any related compensation result in perceived or actual conflicts of interest, the integrity of the data generated at the applicable clinical trial site may be questioned by the FDA and comparable foreign regulatory authorities.

Additionally, the regulatory submission process for a product candidate is complex. We expect to rely on a third-party service provider for the preparation and submission of filings with the FDA and comparable foreign regulatory authorities for approval of our current and any future product candidates. Our reliance on third-party CROs may adversely affect our development timelines if the third-party CROs do not meet the requirements or satisfy the obligations required to obtain regulatory approval. Any significant delays caused by our collaboration partner or third-party CROs may have an adverse effect on our development timelines or otherwise may delay approval and commercialization of DFD-29. If our relationship with such service provider is terminated prior to completion of our regulatory submission process, we may not be able to enter into an arrangement with an alternative service provider in a timely manner, or do so on commercially reasonable terms, and our submission may be substantially delayed.

We are currently dependent on DRL for the manufacture and clinical supply of DFD-29 drug product. Any interruption in our supply may cause serious delays in the timing of our clinical trials, increase our costs and adversely impact our financial results.

Pursuant to the terms of our agreement with DRL for the exclusive, worldwide rights to develop and commercialize DFD-29 for the evaluation of treatment, among other potential indications, inflammatory lesions of rosacea (the "DFD-29 Agreement"), DRL is responsible for the manufacture and supply to us of DFD-29 drug product and we are completely reliant upon DRL to provide us with adequate supply for our use. We may experience an interruption in supply if, among other reasons, we incorrectly forecast our supply requirements, DRL allocates supply to its own development programs, DRL incorrectly plans its manufacturing production or DRL is unable to manufacture DFD-29 drug product in a timely manner to match our development or commercial needs. Transferring technology to a new manufacturer will require additional processes, technologies and validation studies, which are costly, may take considerable amounts of time, may not be successful and require review and approval by the FDA and applicable foreign regulatory bodies. Such manufacturer must comply with cGMP requirements enforced by the FDA and applicable foreign regulatory bodies through facilities inspection programs and review of submitted technical information.

We may be unable to obtain regulatory approval for our current or any of our future product candidates under applicable regulatory requirements. The FDA and foreign regulatory bodies have substantial discretion in the approval process, including the ability to delay, limit or deny approval of product candidates. The delay, limitation or denial of any regulatory approval would adversely impact our business and our operating results.

We may never obtain regulatory approval to commercialize our current or any future product candidates. The research, testing, manufacturing, safety surveillance, efficacy, quality control, recordkeeping, labeling, packaging, storage, approval, sale, marketing, distribution, import, export and reporting of safety and other post-market information related to our current and any future product candidates are subject to extensive regulation by the FDA and other regulatory authorities in the United States and in foreign countries, and such regulations differ from country. We are not permitted to market any of our current or any future product candidates in NDA, BLA or other applicable regulatory filing from the FDA. We are also not permitted to market our product or our current or any future product candidates in any foreign countries until we receive the requisite approval from the applicable regulatory authorities of such countries.

To gain approval to market a new drug, the FDA and foreign regulatory authorities must receive preclinical, clinical and chemistry, manufacturing and controls data that adequately demonstrate the safety, purity, potency, efficacy and compliant manufacturing of the product for the intended indication applied for in an NDA, BLA or other applicable regulatory filing. The development and approval of new drug products and biologic products involves a long, expensive and uncertain process. A delay or failure can occur at any stage in the process. A number of companies in the pharmaceutical and biopharmaceutical industry have suffered significant setbacks in clinical trials, including in Phase 3 clinical development, even after promising results in earlier preclinical studies or clinical trials. These



setbacks have been caused by, among other things, findings made while clinical trials were underway and safety or efficacy observations made in clinical trials, including previously unreported adverse events. Success in preclinical testing and early clinical trials does not ensure that later clinical trials will be successful, and the results of clinical trials by other parties may not be indicative of the results in trials we or our partners may conduct.

The FDA and foreign regulatory bodies have substantial discretion in the drug approval process, including the ability to delay, limit or deny approval of product candidates for many reasons, including:

- the FDA or the applicable foreign regulatory body may disagree with the design, implementation, choice of dose, analysis plans or interpretation of the outcome of one or more clinical trials;
- the FDA or the applicable foreign regulatory body may not deem a product candidate safe and effective for its proposed indication, or may deem a product candidate's safety or other perceived risks to out-weigh its clinical or other benefits;
- the FDA or the applicable foreign regulatory body may not find the data from preclinical studies and clinical trials, including the number of subjects in the safety database, sufficient to support approval, or the results of clinical trials may not meet the level of statistical or clinical significance required by the FDA or the applicable foreign regulatory body for approval;
- the FDA or the applicable foreign regulatory body may disagree with our interpretation of data from pre-clinical studies or clinical trials performed by us or third parties, or with the interpretation of any partner with which we may collaborate;
- the data collected from clinical trials may not be sufficient to support the submission and approval of an NDA, BLA or other applicable regulatory filing;
- the FDA or the applicable foreign regulatory body may require additional preclinical studies or clinical trials;
- the FDA or the applicable foreign regulatory agency may identify deficiencies in the formulation, manufacturing, quality control, labeling or specifications of our current or any future product candidates;
- the FDA or the applicable foreign regulatory agency may require clinical trials in pediatric patients in order to establish pharmacokinetics or safety for this more drug-sensitive population;
- the FDA or the applicable foreign regulatory agency may grant approval contingent on the performance of costly additional post-approval clinical trials;
- the FDA or the applicable foreign regulatory agency may grant approval but impose substantial and costly post-approval requirements;
- the FDA or the applicable foreign regulatory agency may approve our current or any future product candidates for a more limited indication or a narrower patient population than we originally requested;
- the FDA or applicable foreign regulatory agency may not approve the labeling that we believe is necessary or desirable for the successful commercialization of our current or any future product candidates;
- the FDA or the applicable foreign regulatory body may not approve of the manufacturing processes, controls or facilities of third-party manufacturers or testing labs with which we contract; or
- the FDA or the applicable foreign regulatory body may change its approval policies or adopt new regulations in a manner rendering our clinical data or regulatory filings insufficient for approval.

Of the large number of drugs and biologics in development, only a small percentage successfully complete the FDA or other regulatory approval processes and are commercialized. Our current and any future product candidates may not be approved by the FDA or applicable foreign regulatory agencies even though they meet specified endpoints in our clinical trials. The FDA or applicable foreign regulatory agencies may ask us to conduct additional costly and time-consuming clinical trials in order to obtain marketing approval or



approval to enter into an advanced phase of development, or may change the requirements for approval even after such agency has reviewed and commented on the design for the clinical trials. Any delay in obtaining, or inability to obtain, applicable regulatory approval for any of our product candidates would delay or prevent commercialization of our current and any future product candidates and would harm our business, financial condition, operating results and prospects.

We may conduct clinical trials for our current and any future product candidates, in whole or in part, outside of the United States and the FDA and applicable foreign regulatory authorities may not accept data from such trials, which would likely result in additional costs to us and delay our business plan.

We may in the future choose to conduct, one or more of our clinical trials outside the United States. Although the FDA or applicable foreign regulatory authority may accept data from clinical trials conducted outside the United States or the applicable jurisdiction, acceptance of such study data by the FDA or applicable foreign regulatory authority may be subject to certain conditions. Where data from foreign clinical trials are intended to serve as the basis for marketing approval in the United States, the FDA will not approve the application on the basis of foreign data alone unless those data are applicable to the U.S. population and U.S. medical practice; the studies were performed by clinical investigators of recognized competence; and the data are considered valid without the need for an on-site inspection by the FDA or, if the FDA considers such an inspection to be necessary, the FDA is able to validate the data through an on-site inspection or other appropriate means. Many foreign regulatory bodies have similar requirements. In addition, such foreign studies would be subject to the applicable local laws of the foreign jurisdictions where the studies are conducted. There can be no assurance the FDA or applicable foreign regulatory authority will accept data from trials conducted outside of the United States or the applicable jurisdiction. If the FDA or applicable foreign regulatory authority does not accept such data, it would likely result in the need for additional clinical trials, which would be costly and time consuming and delay aspects of our business plan.

Risks Related to Intellectual Property, Generic Competition and Paragraph IV Litigation

If we are unable to obtain and maintain patent protection for our technology and products or if the scope of the patent protection obtained is not sufficiently broad, our competitors could develop and commercialize technology and products similar or identical to ours, and our ability to successfully commercialize our technology and products may be impaired.

Our commercial success will depend in part on obtaining and maintaining patent protection and trade secret protection in the United States and other countries with respect to our products or any current or future product candidates that we may license or acquire and our manufacturing methods, as well as successfully defending these patents and trade secrets against third-party challenges, which is expensive and time-consuming. A patent is the grant of a property right which allows its holder to exclude others from, among other things, selling the subject invention in, or importing such invention into, the jurisdiction that granted the patent. We have obtained, acquired or in-licensed a number of patents and patent applications covering key aspects of certain of our principal products. In the aggregate, our patents are of material importance to our business taken as a whole. We seek to protect our proprietary position by filing or obtaining licenses under patent applications in the United States and abroad related to our products and any other current or future product candidates. We will only be able to protect our technologies from unauthorized use by third parties to the extent that valid and enforceable patents cover them. Our success is predicated, in part, by our ability to maintain the integrity of our trade secrets.

It is possible that we or our licensors will fail to timely identify patentable aspects of our research and development output before it is too late to obtain patent protection, which may result in third parties using our proprietary information, impairing our abilities to compete in the market, to generate revenues, and to achieve profitability. Moreover, should we enter into other collaborations, we may be required to consult with or cede control to collaborators regarding the prosecution, maintenance and enforcement of our patents. Therefore, such patents and applications may not be prosecuted and enforced in a manner consistent with the best interests of our business. The patent prosecution process is expensive and time-consuming, and we may not be able to file and prosecute all necessary or desirable patent applications at a reasonable cost or in a timely manner.

The patent position of biotechnology and pharmaceutical companies generally is highly uncertain, involves complex legal and factual questions and has in recent years been the subject of much litigation. In addition, no consistent policy regarding the breadth of claims allowed in pharmaceutical or biotechnology patents has emerged to date in the U.S. The patent situation outside the U.S. is even more uncertain. The laws of foreign countries may not protect our rights to the same extent as the laws of the United States. For example, European patent law restricts the patentability of methods of treatment of the human body more than United States law does. Publications of discoveries in the scientific literature often lag behind the actual discoveries, and patent applications in the United States and other jurisdictions are typically not published until eighteen (18) months after a first filing, if at all. Therefore, we cannot know with certainty whether we or our licensors were the first to make the inventions claimed in our owned or licensed patents or pending patent applications,



or that we were the first to file for patent protection of such inventions. In the event that a third party has also filed a U.S. patent application relating to any current or future product candidates or a similar invention, we may have to participate in derivation proceedings declared by the USPTO to determine proper inventorship of a claimed invention. The costs of these proceedings could be substantial, and it is possible that our efforts would be unsuccessful, resulting in a material adverse effect on our U.S. patent position. As a result, the issuance, scope, validity, enforceability and commercial value of our patent rights are highly uncertain. Our pending and future patent applications may not result in patents being issued which protect our technology or products, in whole or in part, or which effectively prevent others from commercializing competitive technologies and products. Changes in either the patent laws or interpretation of the patent laws in the United States and other countries may diminish the value of our patents or narrow the scope of our patent protection. Accordingly, we cannot predict the breadth of claims that may be allowed or enforced in our patents or in third-party patents.

Recent patent reform legislation could increase the uncertainties and costs surrounding the prosecution of our patent applications and the enforcement or defense of our issued patents. On September 16, 2011, the Leahy-Smith America Invents Act (the "Leahy-Smith Act"), was signed into law. The Leahy-Smith Act includes a number of significant changes to United States patent law. These include provisions that affect the way patent applications are prosecuted and may also affect patent litigation. The United States Patent Office recently developed new regulations and procedures to govern administration of the Leahy-Smith Act, and many of the substantive changes to patent law associated with the Leahy-Smith Act, and in particular, the first-inventor-to-file provisions, which became effective on March 16, 2013. Courts continue to consider the constitutionality of certain provisions of the Leahy-Smith Act, including the Supreme Court in a recent decision *affecting inter partes* review procedures. Accordingly, it is not clear what, if any, impact the Leahy-Smith Act will have on the operation of our business. However, the Leahy-Smith Act and its implementation could increase the uncertainties and costs surrounding the prosecution of our patent applications and the enforcement or defense of our issued patents, all of which could have a material adverse effect on our business and financial condition.

Moreover, we may be subject to a third-party pre-issuance submission of prior art to the USPTO, or become involved in opposition, derivation, reexamination, *inter partes* review, post-grant review or other administrative proceedings challenging our patent rights or the patent rights of others. An adverse determination in any such submission, proceeding or litigation could reduce the scope of, render unenforceable, or invalidate, our patent rights, allow third parties to commercialize our technology or products and compete directly with us, without payment to us. We may also be unable to manufacture or commercialize products without infringing third-party patent rights, under which a license might not be available. In addition, if the breadth or strength of protection provided by our patents and patent applications is threatened, it could dissuade companies from collaborating with us to license, develop, or commercialize our current or future product candidates.

Even if our patent applications issue as patents, they may not issue in a form that will provide us with any meaningful protection, prevent competitors from competing with us, or otherwise provide us with any competitive advantage. Our competitors may be able to circumvent our owned or licensed patents by developing similar or alternative technologies or products in a non-infringing manner.

The issuance of a patent does not foreclose challenges to its inventorship, scope, validity or enforceability. Therefore, our owned and licensed patents may be challenged in the courts or patent offices in the United States and abroad. Such challenges may result in loss of exclusivity or in patent claims being narrowed, invalidated or held unenforceable, in whole or in part, which could limit our ability to stop others from using or commercializing similar or identical technology and products, or limit the duration of the patent protection of our technology and products. Given the amount of time required for the development, testing and regulatory review of new product candidates, patents protecting such product candidates might expire before or shortly after such product candidates are commercialized. As a result, our owned and licensed patent portfolio may not provide us with sufficient rights to exclude others from commercializing products similar or identical to ours.

Generic drug approvals and successful challenges against the validity of our patents may cause us to lose exclusivity of some of our products.

In the U.S., the Hatch-Waxman Act provides non-patent regulatory exclusivity for five years from the date of the first FDA approval of a new drug compound in an NDA. The FDA, with one exception, is prohibited during those five years from accepting for filing a generic, or ANDA that references the NDA. In reference to the foregoing exception, if a patent is indexed in the FDA Orange Book for the new drug compound, a generic may file an ANDA four years from the NDA approval date if it also files a Paragraph IV Certification with the FDA challenging the patent. Protection under the Hatch-Waxman Act will not prevent the filing or approval of another full NDA. However, the NDA applicant would be required to conduct its own pre-clinical and adequate and well-controlled clinical trials to independently demonstrate safety and effectiveness.

Generic drug companies may submit applications seeking approval to market generic versions of our products. In connection with these applications, generic drug companies may seek to challenge the validity and enforceability of our patents through litigation and/or with the USPTO. Such challenges may subject us to costly and time-consuming litigation and/or USPTO proceedings). Such challenges may subject us to costly and time-consuming litigation and/or USPTO proceedings. As a result of the loss of any patent protection from such litigation or USPTO proceedings, or the "at-risk" launch by a generic competitor of our products, our products could be sold at significantly lower prices, and we could lose a significant portion of sales of that product in a short period of time, which could adversely affect our business, financial condition, operating results and prospects.

Enforcing our proprietary rights is difficult and costly and we may be unable to ensure their protection.

The degree of future protection for our proprietary rights is uncertain, as legal means afford only limited protection and may not adequately protect our rights or permit us to gain or keep our competitive advantage. For example:

- our licensors might not have been the first to make the inventions covered by each of our pending patent applications and issued patents;
- our licensors might not have been the first to file patent applications for these inventions;
- others may independently develop similar or alternative technologies or duplicate our products or our current or future product candidates' technologies;
- it is possible that none of the pending patent applications licensed to us will result in issued patents;
- the issued patents covering our products or any current or future product candidates may not provide a basis for commercially viable active products, may not provide us with any competitive advantages, or may be challenged and defeated by third parties;
- we may not develop additional proprietary technologies that are patentable; or
- patent rights of others may have an adverse effect on our business.

Furthermore, competitors may infringe our issued patents or other intellectual property (collectively, our "IP"), which may require us to file infringement claims, which is expensive and time consuming, and the outcome uncertain. Any claims we assert against perceived infringers could provoke counterclaims alleging that our IP rights are invalid, unenforceable, or not infringed or that we have infringed upon misappropriated others' intellectual property. In response, a court may decide that a patent of ours is wholly or partially invalid or unenforceable, construe the patent's claims narrowly, or refuse to stop the accused party from using the technology at issue.

Additionally, some of our products, including Accutane, Targadox and Exelderm, do not have patent protection because they are not eligible or qualify for such protection. This creates greater risk of competition with generic drug manufacturers and may otherwise adversely affect our business or result of operations.

Further, we rely on trade secrets, including unpatented know-how, to maintain our competitive position. We enter into non-disclosure and confidentiality agreements to protect these trade secrets but cannot guarantee that counterparties will not breach the agreements and disclose our proprietary information, including trade secrets. Enforcing a claim that a party illegally disclosed or misappropriated trade secrets is costly, difficult, and time consuming, and we may be unable to obtain adequate remedy. If any of our trade secrets were to be disclosed to or independently developed by a competitor, our competitive position would be harmed.

If we are sued for infringing intellectual property rights of third parties, it will be costly and time consuming, and an unfavorable outcome in any litigation would harm our business.

Our ability to develop, manufacture, market and sell our products or any current or future product candidates depends upon our ability to avoid infringing the proprietary rights of third parties. There are many U.S. and foreign issued patents and pending patent applications owned by third parties in the dermatology field that cover numerous compounds and formulations in our targeted markets. Because of the uncertainty inherent in any patent or other litigation involving proprietary rights, we and our licensors may not be successful in defending against intellectual property claims raised by third parties, which could have a material adverse effect on our results of



operations. Regardless of the outcome of any litigation, defending the litigation may be expensive, time-consuming and distracting to management. In addition, because patent applications can take many years to issue, there may be currently pending applications, unknown to us, which may later result in issued patents that our commercial activities relating to our products or current or future product candidates may infringe. There could also be existing patents of which we are not aware that our products or current or future product candidates may infringe.

There is a substantial amount of litigation involving patent and other intellectual property rights in the biotechnology and biopharmaceutical industries generally. If a third party claims that we infringe on their products or technology, in addition to costly and time-consuming litigation, we could face a number of issues, including:

- diversion of management's attention from our core business;
- substantial damages for past infringement;
- injunctions prohibiting us from selling or licensing our product unless the patent holder licenses the patent to us, which it would not be required to do;
- requirements that we pay substantial royalties or grant cross licenses under our patents;
- redesigning our processes so they do not infringe, which may not be possible or could require substantial funds and time; and
- harm to our reputation and subsequent adverse effect on the valuation of our securities and revenue.

Intellectual property litigation could cause us to spend substantial resources and distract our personnel from their normal responsibilities.

Even if resolved in our favor, litigation or other legal proceedings relating to intellectual property claims may cause us to incur significant expenses and could distract our technical and management personnel from their normal responsibilities. In addition, there could be public announcements of the results of hearings, motions or other interim proceedings or developments and if securities analysts or investors perceive these results to be negative, it could have a substantial adverse effect on the valuation of our securities. Such litigation or proceedings could substantially increase our operating losses and reduce the resources available for development activities or any future sales, marketing or distribution activities. We may not have sufficient financial or other resources to conduct such litigation or proceedings adequately. Some of our competitors may be able to sustain the costs of such litigation or proceedings more effectively than we can because of their greater financial resources. Uncertainties resulting from the initiation and continuation of patent litigation or other proceedings could compromise our ability to compete in the marketplace. The occurrence of any of the above-described risks could materially adversely affect our business, results of operations, financial condition or cash flows.

We may be subject to claims that our employees have wrongfully used or disclosed alleged trade secrets of their former employers.

As is common in the biotechnology and pharmaceutical industry, we employ individuals who were previously employed at other biotechnology or pharmaceutical companies, including our competitors or potential competitors. Although no claims against us are currently pending, we may be subject to claims that these employees or we have inadvertently or otherwise used or disclosed trade secrets or other proprietary information of their former employers. Litigation may be necessary to defend against these claims. Even if we are successful in defending against these claims, litigation could result in substantial costs and be a distraction to management.

We may need to license certain intellectual property from third parties, and such licenses may not be available or may not be available on commercially reasonable terms.

A third party may hold intellectual property rights, including patent rights, that are important or necessary to the development of our products or current or future product candidates. It may be necessary for us to use the patented or proprietary technology of third parties to commercialize our products or current or future product candidates, in which case we would be required to obtain a license from these third parties, if available, on commercially reasonable terms, or our business could be harmed, possibly materially. Our inability to obtain such rights on acceptable terms, or at all, could materially adversely affect our business, results of operations, financial condition or cash flows.



If we fail to comply with our obligations in our intellectual property licenses and funding arrangements with third parties, or if we breach an agreement under which we license rights to any product or future product candidate, we could lose rights that are important to our business.

If we fail to comply with our obligations under current or future license and funding agreements, our counterparties may have the right to terminate these agreements, in which event we might not be able to develop, manufacture, or market any product or utilize any technology that is covered by these agreements or may face other penalties under the agreements. Such an occurrence could materially and adversely affect the value of a product candidate being developed under any such agreement or could restrict our drug discovery activities. Termination of these agreements or reduction or elimination of our rights under these agreements may result in our having to negotiate new or reinstated agreements with less favorable terms or cause us to lose our rights under these agreements, including our rights to important intellectual property or technology. Further, any uncured, material breach under our license agreement with any current or future licensor could result in our loss of rights to our products or current or future product candidates and may lead to a complete termination of any future product development efforts.

Risks Related to our Platform and Data

Our business and operations would suffer in the event of computer system failures, cyber-attacks, or deficiencies in our or third parties' cybersecurity.

We are increasingly dependent upon information technology systems, infrastructure, and data to operate our business. In the ordinary course of business, we collect, store, and transmit confidential information, including, but not limited to, information related to our intellectual property and proprietary business information, personal information, and other confidential information. It is critical that we maintain such confidential information in a manner that preserves its confidentiality and integrity. Furthermore, we have outsourced elements of our operations to third party vendors, who each have access to our confidential information, which increases our disclosure risk.

While we recently implemented internal security and business continuity measures, our internal computer systems and those of current and future third parties on which we rely may fail and are vulnerable to damage from computer viruses and unauthorized access. Our information technology and other internal infrastructure systems, including corporate firewalls, servers, data center facilities, lab equipment, and connection to the internet, face the risk of breakdown or other damage or interruption from service interruptions, system malfunctions, natural disasters, terrorism, war, and telecommunication and electrical failures, as well as security breaches from inadvertent or intentional actions by our employees, contractors, consultants, business partners, and/or other third parties, or from cyber-attacks by malicious third parties (including the deployment of harmful malware, ransomware, denial-of-service attacks, social engineering and other means to affect service reliability and threaten the confidentiality, integrity and availability of information), each of which could compromise our system infrastructure or lead to the loss, destruction, alteration, disclosure, or dissemination of, or damage or unauthorized access to, our data or data that is processed or maintained on our behalf, or other assets.

If such an event were to occur and cause interruptions in our operations, it could result in a material disruption of our development programs and our business operations, and could result in financial, legal, business, and reputational harm to us. For example, in 2021, we were the victim of a cybersecurity incident that affected our accounts payable function and led to approximately \$9.5 million in wire transfers being misdirected to fraudulent accounts. The matter was reported to the FBI and remains under their investigation. The cybersecurity incident does not appear to have compromised any personally identifiable information or protected health information. Fortress, as our controlling stockholder and supporting partner in our back-office functions, provided us with \$9.5 million to ensure our accounts payable operations continued to function smoothly. The \$9.5 million of support was in the form of a related party note which the boards of both companies have agreed and converted into 1,476,044 shares of our common stock upon the consummation of our IPO in November 2021 at the IPO price. The federal government has been able to trace and seize the fraudulently transferred cryptocurrency associated with the breach. Once the cryptocurrency has been converted back into U.S. dollars, we expect to receive a notification letter to initiate the return of the cash to the Company. This process could take as long as six months or more to complete. Given the recent market declines, volatility, and liquidity issues with cryptocurrency, there is no certainty as to the amount we will ultimately recover. We may incur additional expenses and losses as a result of this cybersecurity incident, including related to investigation fees and remediation costs.

In addition, the loss or corruption of, or other damage to, clinical trial data from completed or future clinical trials could result in delays in our regulatory approval efforts and significantly increase our costs to recover or reproduce the data. Likewise, we rely on third parties for the manufacture of our drug candidates or any future drug candidates and to conduct clinical trials, and similar events relating to

their systems and operations could also have a material adverse effect on our business and lead to regulatory agency actions. The risk of a security breach or disruption, particularly through cyber-attacks or cyber intrusion, including by computer hackers, foreign governments, and cyber terrorists, has generally increased as the number, intensity, and sophistication of attempted attacks and intrusions from around the world have increased. Sophisticated cyber attackers (including foreign adversaries engaged in industrial espionage) are skilled at adapting to existing security technology and developing new methods of gaining access to organizations' sensitive business data, which could result in the loss of proprietary information, including trade secrets. We may not be able to anticipate all types of security threats, and we may not be able to implement preventive measures effective against all such security threats. The techniques used by cyber criminals change frequently, may not be recognized until launched, and can originate from a wide variety of sources, including outside groups such as external service providers, organized crime affiliates, terrorist organizations, or hostile foreign governments or agencies.

Any security breach or other event leading to the loss or damage to, or unauthorized access, use, alteration, disclosure, or dissemination of, personal information, including personal information regarding clinical trial subjects, contractors, directors, or employees, our intellectual property, proprietary business information, or other confidential or proprietary information, could directly harm our reputation, enable competitors to compete with us more effectively, compel us to comply with federal and/or state breach notification laws and foreign law equivalents, subject us to mandatory corrective action, or otherwise subject us to liability under laws and regulations that protect the privacy and security of personal information. Each of the foregoing could result in significant legal and financial exposure and reputational damage that could adversely affect our business. Notifications and follow-up actions related to a security incident could impact our reputation or cause us to incur substantial costs, including legal and remediation costs, in connection with these measures and otherwise implement our internal security and business continuity measures, and actual, potential, or anticipated attacks may cause us to incur increasing costs, including costs to deploy additional personnel and protection technologies, train employees, and engage third-party experts and consultants. We may face increased costs and find it necessary or appropriate to expend substantial resources in the event of an actual or perceived security breach.

The costs related to significant security breaches or disruptions could be material and our insurance policies may not be adequate to compensate us for the potential losses arising from any such disruption in, or failure or security breach of, our systems or third-party systems where information important to our business operations or commercial development is stored or processed. In addition, such insurance may not be available to us in the future on economically reasonable terms, or at all. Further, our insurance may not cover all claims made against us and could have high deductibles in any event, and defending a suit, regardless of its merit, could be costly and divert management attention. Furthermore, if the information technology systems of our third-party vendors and other contractors and consultants become subject to disruptions or security breaches, we may have insufficient recourse against such third parties and we may have to expend significant resources to mitigate the impact of such an event, and to develop and implement protections to prevent future events of this nature from occurring.

Risks Related to Our Finances and Capital Requirements

We have incurred net losses in recent fiscal years, and we may incur losses for the foreseeable future and may not be able to achieve or maintain profitability.

Even though we are a cash generating, commercial organization, we have a limited operating history. We have focused primarily on in-licensing, developing, commercializing and/or manufacturing and selling our products. Potential future losses, among other things, will have an adverse effect on our stockholders' equity and working capital. Because of the numerous risks and uncertainties associated with commercialization and/or developing pharmaceutical products, we are unable to predict the timing or amount of increased expenses or if we will be able to maintain profitability. Any future net losses may fluctuate significantly from quarter to quarter and year to year. We anticipate that our expenses will increase substantially if:

- our current or any future product candidates are approved for commercial sale, due to our ability to establish the necessary commercial infrastructure to launch this product candidate without substantial delays, including hiring sales and marketing personnel and contracting with third parties for warehousing, distribution, cash collection and related commercial activities;
- we acquire or in-license new products for development and/or sale;
- we are required by the FDA, or foreign regulatory authorities, to perform studies in addition to those currently expected;

- there are any delays in completing our clinical trials or the development of our current or any future product candidates;
- we execute other collaborative, licensing or similar arrangements that require us to make payments and/ or expend funds;
- there are variations in the level of expenses related to our future development programs;
- there are any product liability or intellectual property infringement lawsuits in which we may become involved;
- · there are any regulatory developments affecting our products, current or future product candidates, or the product candidates of our competitors; or
- the level of underlying demand for our products and wholesalers' buying patterns.

Our ability to maintain profitability depends upon our ability to generate and sustain revenue. Our ability to generate and sustain revenue depends on a number of factors, including, but not limited to, our ability to:

- · obtain and maintain regulatory approval for our products, or any other current or future product candidates that we may license or acquire;
- manufacture commercial quantities of our current products or current or future product candidates, if approved, at acceptable cost levels; and
- maintain and/or expand our commercial organization and the supporting infrastructure required to successfully market and sell our products or current or future product candidates, if approved.

Even if we do achieve sustainable profitability, we may not be able to maintain or increase profitability on a quarterly or annual basis. Our failure to remain profitable would depress our value and could impair our ability to raise capital, expand our business, maintain or initiate any research and development efforts, diversify our product offerings or even continue our operations. A decline in our value could also cause you to lose all or part of your investment.

We may need additional funding and may be unable to raise capital when needed, which would force us to delay, reduce or eliminate any future product development programs or commercialization, manufacturing and/or sales efforts.

Selling and developing products for dermatological use, conducting clinical trials, establishing outsourced manufacturing relationships and successfully manufacturing and marketing drugs that we may develop is expensive. We may need to raise additional capital to:

- fund our operations and continue our efforts to hire additional personnel;
- qualify and outsource the commercial-scale manufacturing of our products under cGMP; and
- acquire, in-license and/ or develop additional product candidates.

Our future funding requirements will depend on many factors, including, but not limited to:

- the potential for delays in our efforts to seek regulatory approval for any current or future product candidates, and any costs associated with such delays;
- the costs of maintaining and/or establishing a commercial organization to sell, market and distribute our products and/or current or future product candidates;
- the rate of progress and costs of our efforts to prepare for the submission of NDA or BLA for any product candidates that we may in-license or acquire in the future, and the potential that we may need to conduct additional clinical trials to support applications for regulatory approval;

- the costs of filing, prosecuting, defending and enforcing any patent claims and other intellectual property rights associated with any current or future product candidates, including any such costs we may be required to expend if licensors are unwilling or unable to do so;
- the cost and timing of securing sufficient supplies of our products and current or future product candidates from our contract manufacturers in preparation for commercialization, manufacture, and/or sale;
- the effect of competing technological and market developments;
- the terms and timing of any collaborative, licensing, co-promotion or other arrangements that we may establish;
- the potential that we may be required to file a lawsuit to defend our patent rights or regulatory exclusivities from challenges by companies seeking to market generic versions of our branded products; and
- the success of sales efforts of our current products and/or the commercialization of any current or future product candidates.

Future capital requirements will also depend on the extent to which we acquire or invest in additional complementary businesses, products and technologies, but we currently have no commitments or agreements relating to any of these types of transactions.

We may need to finance future cash needs through public or private equity offerings, debt financings or corporate collaboration and licensing arrangements, as well as through interest income earned on cash and investment balances. We cannot be certain that additional funding will be available on acceptable terms, or at all. If adequate funds are not available, we may be required to delay, reduce the scope of or eliminate one or more of future development programs, acquisition plans or our future commercialization efforts, which could materially adversely affect our business, prospects and the trading price of our common stock.

Raising additional capital may cause dilution to our existing stockholders, restrict our operations or require us to relinquish proprietary rights.

To the extent that we raise additional capital through the sale of equity or convertible debt securities, your ownership interest will be diluted, and the terms of these securities may include liquidation or other preferences that adversely affect your rights. Debt financing and preferred equity financing, if available, may involve agreements that include covenants limiting or restricting our ability to take specific actions, such as incurring additional debt, making capital expenditures or declaring dividends.

If we raise additional funds through collaborations, strategic alliances or marketing, distribution or licensing arrangements with third parties, we may have to relinquish valuable rights to our technologies, future revenue streams, research programs or product candidates or grant licenses on terms that may not be favorable to us. If we are unable to raise additional funds through equity or debt financings when needed, we may be required to delay, limit, reduce, or terminate future product development or current or future commercialization efforts or grant rights to develop and market product candidates that we would otherwise prefer to develop and market ourselves.

If we fail to raise the additional funds needed to complete the development of our current products or current or future product candidates, or the funds needed to complete the development of our current or future product candidates, we will be unable to execute our current business plan.

Risks Related to Owning our Common Stock

If we fail to maintain or implement effective internal controls, we may not be able to report financial results accurately or on a timely basis, or to detect fraud, which could have a material adverse effect on our business and the per share price of our common stock.

The Sarbanes-Oxley Act requires, among other things, that we maintain effective disclosure controls and procedures, and internal control over financial reporting. We are continuing to develop and refine our disclosure controls and other procedures that are designed to ensure that information required to be disclosed by us in the reports that we will file with the SEC is recorded, processed, summarized and reported within the time periods specified in SEC rules and forms. We are also continuing to improve our internal control over financial reporting. We have expended, and anticipate that we will continue to expend, significant resources in order to maintain and improve the effectiveness of our disclosure controls and procedures and internal control over financial reporting.



Our current controls and any new controls that we develop may become inadequate because of changes in conditions in our business. Further, weaknesses in our disclosure controls or our internal control over financial reporting may be discovered in the future. Any failure to develop or maintain effective controls, or any difficulties encountered in their implementation or improvement, could harm our operating results or cause us to fail to meet our reporting obligations and may result in a restatement of our financial statements for prior periods. Any failure to implement and maintain effective internal control over financial reporting could also adversely affect the results of management reports and independent registered public accounting firm audits of our internal control over financial reporting could also cause investors to lose confidence in our reported disclosure controls and procedures, and ineffective internal control over financial and other information, which would likely have a negative effect on the market price of our common stock. In addition, if we are unable to continue to meet these requirements, we may not be able to remain listed on The Nasdaq Capital Market ("Nasdaq").

Our independent registered public accounting firm is not required to audit the effectiveness of our internal control over financial reporting until after we are no longer an "emerging growth company" ("EGC"), as defined in the JOBS Act. At such time, our independent registered public accounting firm may issue a report that is adverse in the event it is not satisfied with the level at which our internal control over financial reporting is documented, designed or operating.

Any failure to maintain effective disclosure controls and internal control over financial reporting could have a material and adverse effect on our business and operating results, and cause a decline in the market price of our common stock.

Our charter documents and Delaware law could discourage takeover attempts and other corporate governance changes.

Our Third Amended and Restated Certificate of Incorporation and bylaws contain provisions that could delay or prevent a change in control of our Company. These provisions could also make it difficult for stockholders to elect directors that are not nominated by the current members of our board of directors or take other corporate actions, including effecting changes in our management. These provisions include certain provisions that:

- permit the board of directors to establish the number of directors and fill any vacancies and newly created directorships;
- provide that, after a removal for cause, vacancies on our board of directors may be filled only by a majority of directors then in office, even though less than a quorum;
- prohibit cumulative voting in the election of directors;
- require majority voting to amend our certificate of incorporation and bylaws;
- authorize the issuance of "blank check" preferred stock that our board of directors could use to implement a stockholder rights plan;
- restrict the forum for certain litigation against us to Delaware or federal courts;
- establish advance notice requirements for nominations for election to our board of directors or for proposing matters that can be acted upon by stockholders at annual stockholder meetings; and
- bestow majority control of the stockholder vote to Fortress by virtue of their exclusive ownership of our Class A Common Stock.

In addition, as a Delaware corporation, we are subject to Section 203 of the Delaware General Corporation Law. These provisions may prohibit large stockholders, in particular those owning 15% or more of our outstanding voting stock, from merging or combining with us for a period of time without the approval of our board of directors. In addition, our credit facility includes, and other debt instruments we may enter into in the future may include, provisions entitling the lenders to demand immediate repayment of all borrowings upon the occurrence of certain change of control events relating to our company, which also could discourage, delay or prevent a business combination transaction.

Our Third Amended and Restated Certificate of Incorporation provides, subject to limited exceptions, that the Court of Chancery of the State of Delaware is the sole and exclusive forum for certain stockholder litigation matters, which could limit our stockholders' ability to obtain a chosen judicial forum for disputes with us or our directors, officers, employees or stockholders.

Our Third Amended and Restated Certificate of Incorporation requires to the fullest extent permitted by law, that derivative actions brought in our name, actions against directors, officers and employees for breach of fiduciary duty and other similar actions must be brought in the Court of Chancery in the State of Delaware or, if that court lacks subject matter jurisdiction, another federal or state court situated in the State of Delaware. Any person or entity purchasing or otherwise acquiring any interest in shares of our capital stock shall be deemed to have notice of and consented to the forum provisions in our certificate of incorporation. In addition, our Third Amended and Restated Certificate of Incorporation provides that the federal district courts of the United States are the exclusive forum for the resolution of any complaint asserting a cause of action under the Securities Act and the Exchange Act.

This choice of forum provision may limit a stockholder's ability to bring a claim in a judicial forum of its choosing for disputes with us or any of our directors, officers, other employees or stockholders, which may discourage lawsuits with respect to such claims. Alternatively, if a court were to find the choice of forum provision contained in our Third Amended and Restated Certificate of Incorporation to be inapplicable or unenforceable in an action, we may incur additional costs associated with resolving such action in other jurisdictions, which could harm our business, operating results and financial condition.

The requirements of being a public company may strain our resources, divert our management's attention and affect our ability to attract and retain qualified board members.

As a public company, we are subject to the reporting requirements of the Securities Exchange Act of 1934, as amended, and are required to comply with the applicable requirements of the Sarbanes-Oxley Act and the Dodd-Frank Wall Street Reform and Consumer Protection Act, the listing requirements of Nasdaq, and other applicable securities rules and regulations. Compliance with these rules and regulations increases our legal and financial compliance costs, makes some activities more difficult, time-consuming or costly and increases demand on our systems and resources. Among other things, the Exchange Act requires that we file annual, quarterly and current reports with respect to our business and operating results and maintain effective disclosure controls and procedures and internal controls over financial reporting. Significant resources and management oversight is required to maintain and, if required, improve our disclosure controls and procedures and internal controls over financial reporting to meet this standard. As a result, management's attention may be diverted from other business concerns, which could harm our business and operating results. Although we hired additional employees to comply with these requirements, we may need to hire even more employees in the future, which will increase our costs and expenses.

Reduced reporting and disclosure requirements applicable to us as an EGC could make our common stock less attractive to investors.

We are an EGC and, as long as we continue to be an EGC, we may continue to avail ourselves of exemptions from various reporting requirements applicable to other public companies. Consequently, we are not required to have our independent registered public accounting firm audit our internal control over financial reporting under Section 404 of the Sarbanes-Oxley Act, and we are subject to reduced disclosure obligations regarding executive compensation in our periodic reports and proxy statements and exemptions from the requirements of holding a nonbinding advisory vote on executive compensation and stockholder approval of any golden parachute payments not previously approved. In addition, the JOBS Act provides that an EGC can take advantage of an extended transition period for complying with new or revised accounting standards. As a result, our financial statements may not be companies that comply with new or revised accounting pronouncements as of the dates such pronouncements are effective for public companies. We could be an EGC for up to five years following the completion of our November 2021 public offering. We will cease to be an EGC upon the earliest of: (i) the end of the fiscal year following the fifth anniversary of the aforementioned offering, (ii) the first fiscal year after our annual gross revenue is \$1.235 billion or more, (iii) the date on which we have, during the previous three-year period, issued more than \$1.0 billion in nonconvertible debt second quarter of that fiscal year in which the market value of our common stock held by non-affiliates exceeded \$700 million as of the end of the end of the end of the rely of the editor will find our common stock held by non-affiliates exceeded \$700 million as of the end of the rely of the editor whether investors will find our common stock held by non-affiliates exceeded \$700 million as of the editor of any charge our common stock held by non-affiliates exceeded \$700 million as of the exemptions. If some investors find our common stock held by non-affil

Our shares of common stock are subject to potential delisting if we do not continue to maintain the listing requirements of Nasdaq.

We list our shares of common stock on Nasdaq under the symbol "DERM." Nasdaq has rules for continued listing, including, without limitation, minimum market capitalization and other requirements. Failure to maintain our listing, or de-listing from Nasdaq, would make it more difficult for shareholders to sell our securities and more difficult to obtain accurate price quotations on our securities. This could have an adverse effect on the price of our common stock. Our ability to issue additional securities for financing or other purposes, or otherwise to arrange for any financing we may need in the future, may also be materially and adversely affected if our common stock is not traded on a national securities exchange.

Because we do not anticipate paying any cash dividends on our common stock in the foreseeable future, capital appreciation, if any, will be your sole source of gains.

We currently intend to retain any earnings to finance the operation and expansion of our business, and we do not anticipate paying any cash dividends in the foreseeable future. Any determination to pay dividends in the future will be at the discretion of our board of directors. In addition, the terms of our existing debt arrangements preclude us from paying dividends and our future debt agreements, if any, may contain similar restrictions. As a result, you may only receive a return on your investment in our common stock if the market price of our common stock increases.

The trading price of the shares of our common stock is likely to be volatile, and purchasers of our common stock could incur substantial losses.

The trading price of our common stock fluctuates substantially. These fluctuations could cause you to incur substantial losses, including all of your investment in our common stock. Factors that could cause fluctuations in the trading price of our common stock include the following:

- significant volatility in the market price and trading volume of companies in our industry;
- announcements of new solutions or technologies, commercial relationships, acquisitions, or other events by us or our competitors;
- price and volume fluctuations in the overall stock market from time to time;
- changes in how customers perceive the benefits of our products and future offerings;
- the public's reaction to our press releases, other public announcements, and filings with the SEC;
- fluctuations in the trading volume of our shares or the size of our public float;
- actual or anticipated changes or fluctuations in our results of operations or financial projections;
- changes in actual or future expectations of investors or securities analysts;
- litigation involving us, our industry, or both;
- governmental or regulatory actions or audits;
- regulatory developments applicable to our business, including those related to privacy in the United States or globally;
- general economic conditions and trends;
- major catastrophic events in our domestic and foreign markets; and
- departures of key employees.



Risks Related to our Relationship with Fortress Biotech, Inc.

Fortress controls a voting majority of our common stock, which could be detrimental to our other shareholders.

Pursuant to the terms of the Class A Common Stock held by Fortress, Fortress will be entitled to cast, for each share of Class A Common Stock held by Fortress, the number of votes that is equal to 1.1 times a fraction, the numerator of which is the number of shares of our outstanding common stock and the denominator of which is the number of shares of outstanding Class A Common Stock (the "Class A Common Stock Ratio"). Thus, Fortress will at all times have voting control of Journey. Further, for a period of ten (10) years from the date of the first issuance of shares of Class A Common Stock, the holders of record of the shares of Class A Common Stock (or other capital stock or securities issued upon conversion of or in exchange for the Class A Common Stock), exclusively and as a separate class, shall be entitled to appoint or elect the majority of the directors of Journey. This concentration of voting power may delay, prevent or deter a change in control of us even when such a change may be in the best interests of all stockholders, could deprive our stockholders of an opportunity to receive a premium for their shares of common stock as part of a sale of Journey or our assets, and might affect the prevailing market price of our common stock.

We are a "controlled company" within the meaning of Nasdaq listing standards and, as a result, qualify for exemptions from certain corporate governance requirements. Although we do not presently intend to take advantage of these exemptions, we may do so in the future.

We are a "controlled company" within the meaning of Nasdaq listing standards. Under these rules, a company of which more than 50% of the voting power is held by an individual, a group or another company is a "controlled company" and may elect not to comply with certain corporate governance requirements of Nasdaq, including (i) the requirement that a majority of the Board of Directors consist of independent directors, (ii) the requirement that we have a nominating and corporate governance committee that is composed entirely of independent directors with a written charter addressing the committee's purpose and responsibilities and (iii) the requirement that we have a compensation committee that is composed entirely of independent directors with a written charter addressing the committee's purpose and responsibilities. Although we presently are not taking advantage of these exemptions, we may do so in the future. Accordingly, our stockholders may not have the same protections afforded to stockholders of companies that are subject to all of the corporate governance requirements of Nasdaq. Investors may find our common stock less attractive as a result of our reliance on these exemptions. If some investors find our common stock less attractive as a result of our stock and our stock price may be more volatile.

We may have received better terms from unaffiliated third parties than the terms we receive in our arrangements with Fortress.

We have arrangements with Fortress in connection with management and administration services for the Company. While we believe the terms of these arrangements are reasonable, they might not reflect terms that would have resulted from arm's-length negotiations between unaffiliated third parties. The terms of the arrangement relate to, among other things, systems, insurance, accounting, legal, finance, tax and human resources. We might have received better terms from third parties because, among other things, third parties might have competed with each other to win our business.

The ownership by our executive officers and some of our directors of shares of equity securities of Fortress and/or rights to acquire equity securities of Fortress might create, or appear to create, conflicts of interest.

Because of their current or former positions with Fortress, some of our executive officers and directors own shares of Fortress common stock and/or options to purchase shares of Fortress common stock. Their individual holdings of common stock and/or options to purchase common stock of Fortress may be significant compared to their total assets. Ownership by our directors and officers, after our separation, of common stock and/or options to purchase common stock of Fortress might appear to create conflicts of interest when these directors and officers are faced with decisions that could have different implications for Fortress than for us.

Fortress' current or future financial obligations and arrangements, or an event of default thereon, may change the ownership dynamic of us by Fortress.

Any default or breach by Fortress under any current or future credit agreement or arrangements may have an adverse effect on our business. Fortress has pledged as collateral to certain of its creditors equity in the Company. If Fortress were to default on its obligations to any such creditor, that creditor, whose interests may not align with those of our other stakeholders, could acquire a controlling interest in the Company. In addition, Fortress' current credit agreement with Oaktree Capital (the "Oaktree Credit Agreement") contains certain affirmative and negative covenants and events of default that apply in different instances to Fortress itself, its private subsidiaries, its public subsidiaries, or combinations of the foregoing. Although we are not a party to the Oaktree Credit Agreement, because Fortress cannot allow so as to remain in compliance with the Oaktree Credit Agreement.

General Risks

Failure to manage our growth effectively could cause our business to suffer and have an adverse effect on our business, operating results and financial condition.

We have experienced significant growth in a short period of time. To manage our growth effectively, we must continually evaluate and evolve our organization. We must also manage our employees, operations, finances and capital investments efficiently. Our efficiency, productivity and the quality of our products may be adversely impacted if we do not train our new personnel, particularly our sales and support personnel, quickly and effectively, or if we fail to appropriately coordinate across our organization. Additionally, our rapid growth may place a strain on our resources, infrastructure and ability to maintain the quality of our products. You should not consider our revenue growth and levels of profitability in recent periods as indicative of future periods, our revenue or profitability could decline or grow more slowly than we expect. Failure to manage our growth effectively could cause our business to suffer and have an adverse effect on our operating results and financial condition.

If securities or industry analysts do not publish research or reports about our business, or publish inaccurate or unfavorable research reports about our business, our share price and trading volume could decline.

The trading market for our common stock partially depends on the research and reports that securities or industry analysts publish about us or our business. We do not have any control over these analysts. If one or more of the analysts who cover us should downgrade our shares or change their opinion of our business prospects, our share price would likely decline. If one or more of these analysts ceases coverage of our company or fails to regularly publish reports on us, we could lose visibility in the financial markets, which could cause our share price or trading volume to decline.

Our reported financial results may be adversely affected by changes in accounting principles generally accepted in the United States. If our estimates or judgments relating to our critical accounting policies prove to be incorrect, our results of operations could be adversely affected.

U.S. generally accepted accounting principles ("GAAP") are subject to interpretation by the Financial Accounting Standards Board ("FASB"), the SEC and various bodies formed to promulgate and interpret appropriate accounting principles. A change in these principles or interpretations could have a significant effect on our reported results of operations and could affect the reporting of transactions already completed before the announcement of a change.

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in our consolidated financial statements and accompanying notes appearing elsewhere in this report on Form 10-K. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances, as discussed in the section titled "*Management's Discussion and Analysis of Financial Condition and Results of Operations — Critical Accounting Policies and Estimates.*" The results of these estimates form the basis for making judgments about the carrying values of assets, liabilities and equity, and the amount of revenue and expenses that are not readily apparent from other sources. Significant estimates, judgments, and assumptions used in our financial statements include, but are not limited to, those related to revenue recognition, accounts receivable and related reserves, useful lives and realizability of long-lived assets, research and development costs, assumptions used in the valuation of warrants, accounting for share-based compensation, and valuation allowances against deferred tax assets. These estimates are periodically reviewed for any changes in circumstances, facts, and experience. Our results of operations may be adversely affected if our assumptions change or if actual circumstances differ from those in our assumptions, which could cause our results of operations to fall below the expectations of securities analysts and investors, resulting in a decline in the market price of our common stock.

Global and national financial events may have an impact on our business and financial condition in ways that we currently cannot predict.

A credit crisis, turmoil in the global or U.S. financial system, recession or similar possible events in the future could negatively impact us. A financial crisis or recession may limit our ability to raise capital through credit and equity markets. The prices for the products and services that we intend to provide may be affected by a number of factors, and it is unknown how these factors may be impacted by a global or national financial event.

If our estimates or judgments relating to our critical accounting policies are erroneous or based on assumptions that change or prove to be incorrect, our operating results could fall below the expectations of securities analysts and investors, resulting in a decline in our stock price.

The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the amounts reported in the consolidated financial statements and accompanying notes. We base our estimates on our best judgment, historical experience, information derived from third parties and on various other assumptions that we believe to be reasonable under the circumstances, as discussed in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations," the results of which form the basis for making judgments about the carrying values of assets, liabilities, equity, revenue and expenses that are not readily apparent from other sources. Our operating results may be adversely affected if our judgments prove to be wrong, assumptions change or actual circumstances differ from those in our assumptions, which could cause our operating results to fall below the expectations of securities analysts and investors, resulting in a decline in our stock price. Significant assumptions and estimates used in preparing our consolidated financial statements include those related to revenue recognition, share-based compensation and income taxes.

Changes in tax laws or regulations that are applied adversely to us may have a material adverse effect on our business, cash flow, financial condition or results of operations.

New income, sales, use or other tax laws, statutes, rules, regulations or ordinances could be enacted at any time, which could adversely affect our business operations and financial performance. For example, the United States recently passed the Inflation Reduction Act, which provides for a minimum tax equal to 15% of the adjusted financial statement income of certain large corporations, as well as a 1% excise tax on certain share buybacks by public corporations that would be imposed on such corporations. In addition, it is uncertain if and to what extent various states will conform to newly enacted federal tax legislation. Changes in corporate tax rates, the realization of net deferred tax assets relating to our operations, the taxation of foreign earnings, and the deductibility of expenses could have a material impact on the value of our deferred tax assets, could result in significant one-time charges, and could increase our future U.S. tax expense.

Item 1B. Unresolved Staff Comments

None.

Item 1C. Cybersecurity

Cybersecurity Risk Management and Strategy

We recognize the importance of assessing, identifying, and managing material risks associated with cybersecurity threats, as such term is defined in Item 106(a) of Regulation S-K. These risks include, among other things: operational risks, intellectual property theft, fraud, extortion, harm to employees or customers and violation of data privacy or security laws.

Identifying and assessing cybersecurity risk is integrated into our overall risk management systems and processes. Cybersecurity risks related to our business, operations, privacy and compliance issues are identified and addressed through a multi-faceted approach. To defend, detect and respond to cybersecurity incidents, we, among other things: conduct proactive privacy and cybersecurity reviews of systems and applications, conduct employee training, monitor emerging laws and regulations related to data protection and information security and implement appropriate changes.

We have implemented a cybersecurity risk management program that leverages the National Institute of Standards and Technology ("NIST") framework, which organizes cybersecurity risks into five categories: identify, protect, detect, respond and recover. We regularly assess the threat landscape and take a holistic view of cybersecurity risks, with a layered cybersecurity strategy based on prevention, detection and mitigation.

Security events and data incidents are evaluated, ranked by severity and prioritized for response and remediation. Our cybersecurity team collaborates with stakeholders across our business units to further analyze the risk to the company, and form detection, mitigation and remediation strategies. Our risk management program also assesses third-party cybersecurity risks and we perform third-party risk management to identify and mitigate risks from third parties such as vendors, suppliers, and other business partners associated with our use of third-party service providers.

We describe whether and how risks from identified cybersecurity threats, including as a result of any previous cybersecurity incidents, have materially affected or are reasonably likely to materially affect us, including our business strategy, results of operations, or financial condition, under the heading "Our business and operations would suffer in the event of computer system failures, cyber-attacks, or deficiencies in our or third parties' cybersecurity" in our risk factor disclosures in Item IA of this Annual Report on Form 10-K.

Cybersecurity Governance

Cybersecurity is an important part of our risk management processes and an area of focus for our management. Our executive management is responsible for the oversight of risks from cybersecurity threats. Members of our board of directors receive updates from our executive management team regarding matters of cybersecurity. This includes existing and new cybersecurity risks, status on how management is addressing and/or mitigating those risks, cybersecurity and data privacy incidents (if any) and status on key information security initiatives.

Our cybersecurity risk management and strategy processes are overseen by leaders from our information security, compliance and legal teams. These individuals are informed about, and monitor the prevention, mitigation, detection and remediation of cybersecurity incidents through their management of, and participation in, the cybersecurity risk management and strategy processes described above, including the operation of our incident response plan, and report to our board of directors on any appropriate items.

Item 2. Properties

Our executive offices are located at 9237 E Via de Ventura Blvd. Suite 105, Scottsdale, AZ 85258. We believe that our existing facilities are adequate to meet our current requirements. We do not own any real property.

In September 2022, we amended our lease and entered into a new 25-month extension of the office space in Scottsdale, AZ at an average annual rent of \$0.1 million. The term of this amended lease commenced on January 1, 2023 and will expire on January 31, 2025.



Item 3. Legal Proceedings

To our knowledge, there are no legal proceedings pending against us, other than routine actions, administrative proceedings, and other actions not deemed material, that are expected to have a material adverse effect on our financial condition, results of operations, or cash flows. In the ordinary course of business, however, the Company may be subject to both insured and uninsured litigation. Suits and claims may be brought against the Company by customers, suppliers, partners and/or third parties (including tort claims for personal injury arising from clinical trials of the Company's product candidates and property damage) alleging deficiencies in performance, breach of contract, etc., and seeking resulting alleged damages.

Item 4. Mine Safety Disclosures

Not applicable.

PART II

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

Market information

Our common stock is listed on the Nasdaq Capital Market and trades under the symbol "DERM." We commenced trading on the Nasdaq Capital Market on November 12, 2021. Prior to November 12, 2021 there was no public market for our common stock.

Equity Compensation Plans

We do not maintain any deferred compensation, retirement, pension or profit-sharing plans. Our board of directors has adopted an incentive plan, allowing for the grant of equity and cash-based awards to our employees and directors.

Sales of Unregistered Securities

None.

Use of Proceeds from Sales of Registered Securities

On December 30, 2022, we filed a shelf registration statement on Form S-3 (File No. 333 - 269079), which was declared effective by the SEC on January 26, 2023. This shelf registration statement covers the offering, issuance and sale by us of up to an aggregate of \$150.0 million of the Company's common stock, preferred stock, debt securities, warrants, and units (the "2022 Shelf"). In connection with the 2022 Shelf, we entered into an At Market Issuance Sales Agreement (the "Sales Agreement") with B. Riley Securities, Inc. ("B. Riley"), relating to shares of our common stock. In accordance with the terms of the Sales Agreement, we may offer and sell up to 4,900,000 shares of our common stock, and wile \$0.0001 per share, from time to time through or to B. Riley acting as our agent or principal. During 2023, we issued 748,703 shares of common stock under the 2022 Shelf, generating net proceeds of \$4.5 million. At December 31, 2023, 4,151,297 shares remain available for issuance under the Sales Agreement.

We currently intend to use the net proceeds from this offering for general corporate purposes, including working capital, research and development, payments for research and development — licenses acquired, sales and marketing activities, general administrative matters, operating expenses and capital expenditures.

Holders

As of March 28, 2024, there were approximately 44 holders of record for our common stock and 1 holder of record for our Class A common stock. The actual number of stockholders is greater than this number of record holders and includes stockholders who are beneficial owners and whose shares are held in street name by brokers and other nominees. This number of holders of record also does not include stockholders whose shares may be held in trust by other entities.

Dividends

We have never paid cash dividends on any of our capital stock and currently intend to retain our future earnings, if any, to fund the development and growth of our business.

Item 6. [RESERVED.]



Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

Forward-Looking Statements

You should read the following discussion and analysis of our financial condition and results of operations in conjunction with our consolidated financial statements and the related notes included elsewhere in this Form 10-K. Our consolidated financial statements have been prepared in accordance with U.S. GAAP. The following discussion and analysis contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934 (the "Exchange Act"). Please see the section titled "Special Cautionary Notice Regarding Forward-Looking Statements" elsewhere in this Annual Report on Form 10-K for more information. In evaluating our business, you should carefully consider the information set forth under the heading "Risk Factors" herein. As used below, the words "we," "us" and "our" refer to Journey Medical Corporation and its consolidated subsidiaries.

Overview

We are a commercial-stage pharmaceutical company founded in October 2014 that focuses on the development and commercialization of pharmaceutical products for the treatment of dermatological conditions. Our current portfolio includes seven branded and two authorized generic prescription drugs for dermatological conditions that are marketed in the U.S. We are managed by experienced life science executives with a track record of creating value for their stakeholders and bringing novel medicines to the market, enabling patients to experience increased quality of life, and enabling physicians and other licensed medical professionals to provide better care for their patients. We aim to acquire rights to future products by licensing or otherwise acquiring an ownership interest in, funding the research and development of, and eventually commercializing, the products through our field sales organization. Since inception, we have made significant investments to build out our commercial product portfolios, which we believe, coupled with our experienced dermatology sales leadership team and field sales force, will position our business for growth. We are a majority-owned subsidiary of Fortress.

Critical Accounting Policies and Uses of Estimates

Our consolidated financial statements have been prepared in accordance with U.S. GAAP. The preparation of these consolidated financial statements requires us to make difficult, subjective or complex judgments, often as a result of the need to make estimates and assumptions about the effect of matters that are inherently uncertain in the reported amounts of assets and liabilities, and the disclosure of contingent assets and liabilities at the date of the financial statements, as well as the reported expenses incurred during the reporting periods. Our estimates are based on our historical experience and on various other factors that we believe are reasonable under the circumstances, the results of which form the basis for making judgments about the carrying value of assets and liabilities that are not readily apparent from other sources. Actual results may differ from these estimates under different assumptions or conditions.

While our significant accounting policies are described in greater detail in Note 2, "Basis of Presentation and Summary of Significant Accounting Policies" in our consolidated financial statements, appearing under Part II, Item 8 and beginning at page F-1 of this Annual Report on Form 10-K, we believe that the following accounting policies and estimates are those most critical to the judgments and estimates used in the preparation of our consolidated financial statements in understanding our historical and future performance. These policies relate to the more significant areas involving management's judgments and estimates.

Revenue Recognition

Our gross product revenues are subject to a variety of deductions, which generally are estimated and recorded in the same period that the revenues are recognized. Such variable consideration represents chargebacks, coupons, discounts, other sales allowances and sales returns. These deductions represent estimates of the related obligations and, as such, knowledge and judgment are required when estimating the impact of these revenue deductions or gross sales for a reporting period. Historically, adjustments to these estimates to reflect actual results or updated expectations have not been material to our overall business. Coupons, however, can have a significant impact on year-over-year individual product revenue growth trends. If any of our ratios, factors, assessments, experiences, or judgments are not indicative or accurate estimates of our future experience, our results could be materially affected. The potential of our estimates to vary differs by program, product, type of customer and geographic location.

Recent Accounting Pronouncements

See Note 2, "Basis of Presentation and Summary of Significant Accounting Policies" in our consolidated financial statements, appearing under Part II, Item 8 and beginning at page F-1 of this Annual Report on Form 10-K for information about recent accounting pronouncements, the timing of their adoption, if applicable, and our assessment, if any, of their potential impact on our financial condition and results of operations.

Emerging Growth Company and Smaller Reporting Company Status

We are an emerging growth company, as defined in the Jumpstart Our Business Startups Act of 2012 ("JOBS Act"). Under the JOBS Act, emerging growth companies can delay the adoption of new or revised accounting standards issued subsequent to the enactment of the JOBS Act until such time as those standards apply to private companies. Other exemptions and reduced reporting requirements under the JOBS Act for emerging growth companies include presentation of only two years of audited financial statements in our annual reports on Form 10-K, an exemption from the requirement to provide an auditor's report on internal controls over financial reporting pursuant to Section 404 of the Sarbanes-Oxley Act of 2002, as amended, an exemption from any requirement that may be adopted by the Public Company Accounting Oversight Board regarding mandatory audit firm rotation and less extensive disclosure about our executive compensation arrangements. We have elected to use the extended transition period for complying with new or revised accounting growth company or (ii) we affirmatively and irrevocably opt out of the extended transition period provided in the JOBS Act.

We are also a "smaller reporting company," meaning that either (i) the market value of our shares held by non-affiliates is less than \$250 million or (ii) the market value of our shares held by non-affiliates is less than \$700 million and our annual revenue was less than \$100 million during the most recently completed fiscal year. We may continue to be a smaller reporting company if either (i) the market value of our shares held by non-affiliates is less than \$250 million or (ii) our annual revenue was less than \$100 million during the most recently completed fiscal year. We may continue to be a smaller reporting company if either (i) the market value of our shares held by non-affiliates is less than \$100 million during the most recently completed fiscal year and the market value of our shares held by non-affiliates is less than \$100 million during the most recently completed fiscal year and the market value of our shares held by non-affiliates is less than \$100 million. As a smaller reporting company, we may choose to present only the two most recent fiscal years of audited financial statements in our Annual Report on Form 10-K, have reduced disclosure obligations regarding executive compensation, and smaller reporting companies are permitted to delay adoption of certain recent accounting pronouncements discussed in Note 2 See Note 2, "Basis of Presentation and Summary of Significant Accounting Policies" in our consolidated financial statements, appearing under Part II, Item 8 and beginning at page F-1 of this Annual Report on Form 10-K.

Results of Operations

Comparison of the Years Ended December 31, 2023 and 2022

The following table summarizes our results of operations for the years ended December 31, 2023 and 2022:

	Fo	For the Years Ended December 31,						
(\$ in thousands, except per share data)		2023		2022	\$	%		
Revenue:								
Product revenue, net	\$	59,662	\$	70,995	\$ (11,333)	(16)%		
Other revenue		19,519		2,674	16,845	630 %		
Total revenue		79,181		73,669	5,512	7 %		
Operating expenses								
Cost of goods sold - product revenue		26,660		30,775	(4,115)	(13)%		
Research and development		7,541		10,943	(3,402)	(31)%		
Selling, general and administrative		43,910		59,468	(15,558)	(26)%		
Loss on impairment of intangible assets		3,143			3,143	100 %		
Total operating expenses		81,254		101,186	(19,932)	(20)%		
Loss from operations		(2,073)		(27,517)	25,444	(92)%		
Other expense								
Interest income		(322)		(60)	(262)	437 %		
Interest expense		1,698		2,019	(321)	(16)%		
Foreign exchange transaction losses		183		89	94	106 %		
Total other expense		1,559		2,048	(489)	(24)%		
Loss before income taxes		(3,632)		(29,565)	25,933	(88)%		
Income tax expense		221		63	158	251 %		
Net Loss		(3,853)		(29,628)	25,775	(87)%		

Revenues

The following table reflects our revenue by product for the years ended December 31, 2023 and 2022:

	For the Years Ended December 31,					Change		
(\$in thousands)	2023		2022		_	\$	%	
Qbrexza®	\$	25,410	\$	26,715	\$	(1,305)	(5)%	
Accutane®		20,168		18,373		1,795	10 %	
Amzeeq		6,201		7,242		(1,041)	(14)%	
Targadox®		3,204		7,972		(4,768)	(60)%	
Exelderm®		2,395		3,463		(1,068)	(31)%	
Zilxi		1,962		2,273		(311)	(14)%	
Ximino®		287		4,957		(4,670)	(94)%	
Luxamend®		35		_		35	100 %	
Total net product revenue	\$	59,662	\$	70,995	\$	(11,333)	(16)%	

Total net product revenues decreased \$11.3 million, or 16%, to \$59.7 million for the year ended December 31, 2023, from \$71.0 million for the year ended December 31, 2022. The decrease is primarily due to lower unit volumes from our legacy products, Targadox, Ximino and Exelderm driven specifically by continued generic competition for Targadox and the winding down, and ultimate discontinuation of Ximino, during the third quarter of 2023. Despite unit volume increases from period-to-period for Qbrexza, Amzeeq and Zilxi, net revenues for these products were negatively impacted by higher managed care rebates due to higher managed care program costs. In addition, Qbrexza net revenue was negatively impacted by coupon deductible rate resets in the beginning of 2023, and isolated charges in the first quarter of 2023 for higher-than-anticipated returns from the Dermira product lots purchased in 2021, as well as higher government rebates from increases in certain state rebate programs. As of July 1, 2023, we no longer participate in these programs. Accutane net product revenue increased \$1.8 million from 2022 due to increased unit volume resulting from our focused sales and marketing efforts.

Other revenue

	For the Yo Decem		Chang	e
(Sin thousands)	2023	2022	\$	%
Non-refundable upfront payment from Maruho	\$ 19,000	\$ _	\$ 19,000	100 %
Net milestone payment from Maruho	_	2,500	(2,500)	(100)%
Royalties on sales of Rapifort® Wipes 2.5%	519	174	345	198 %
Total other revenue	\$ 19,519	\$ 2,674	\$ 16,845	630 %

Other revenues increased approximately \$16.8 million, to \$19.5 million for the year ended December 31, 2023, from \$2.7 million for the year ended December 31, 2022. Other revenue for the year ended December 31, 2023 includes a \$19.0 million non-refundable upfront payment from Maruho under the New License Agreement. Royalties on sales of Rapifort Wipes 2.5% in Japan were \$0.5 million for the year ended December 31, 2023 as compared to \$0.2 million for the year ended December 31, 2022. Other revenue for the year ended December 31, 2022 includes a net \$2.5 million milestone payment from Maruho. In January 2022, Maruho received manufacturing and marketing approval in Japan for Rapifort Wipes 2.5% (Japanese equivalent to U.S. FDA approved QBREXZA®), for the treatment of primary axillary hyperhidrosis, triggering the one-time net payment. Sales of Rapifort in Japan will no longer be subject to a royalty after October 1, 2023 in accordance with the Second A&R License Agreement.

Gross - to - Net Sales Accruals

We record gross-to-net sales accruals for chargebacks, distributor service fees, prompt pay discounts, sales returns, coupons, managed care rebates, government rebates, and other allowances customary to the pharmaceutical industry.

Gross-to-net sales accruals and the balance in the related allowance accounts for the years ended December 31, 2023, 2022 and 2021 were as follows:

(\$'s in thousands)	a	argebacks nd other lowances	stributor Service Fees	Prompt Pay Discounts	1	Returns	Coupons	Managed Care Rebates	Gov't Rebates	Total
Balance as of December 31, 2021	\$	622	\$ 791	\$ 197	\$	3,240	\$ 4,992	\$ 3,492	\$ 690	\$ 14,024
Current provision related to sales in the										
current period		2,663	5,868	1,104		5,387	117,883	22,654	3,651	159,210
Checks/credits issued to third parties		(3,032)	 (5,730)	 (1,094)	_	(4,938)	 (121, 179)	 (22,552)	 (3,331)	 (161,856)
Balance as of December 31, 2022	\$	253	\$ 929	\$ 207	\$	3,689	\$ 1,696	\$ 3,594	\$ 1,010	\$ 11,378
Current provision related to sales in the										
current period		1,856	5,439	976		5,483	94,822	22,934	5,191	136,701
Checks/credits issued to third parties		(2,016)	(5,470)	(1,041)		(5,095)	(93,074)	(21,318)	(5,948)	(133,962)
Balance as of December 31, 2023	\$	93	\$ 898	\$ 142	\$	4,077	\$ 3,444	\$ 5,210	\$ 253	\$ 14,117

The increase in our reserves for gross-to-net sales accruals from period-to-period is driven by increases in our reserves for coupons and managed care rebates of \$1.7 million and \$1.6 million, respectively. Our provision for coupons was \$3.4 million at December 31, 2023 compared to \$1.7 million at December 31, 2022. The increase in the coupon reserve is primarily due to an increase in our channel reserve at December 31, 2023 for rebates not credited at the end of the year as a result of the timing of receipt. Our provision for managed care rebates was \$5.2 million at December 31, 2023 compared to \$3.6 million at December 31, 2022. The increase in the managed care rebate reserve is primarily due to the timing of invoices received.

Cost of Goods Sold

Cost of goods sold decreased by \$4.1 million, or 13%, to \$26.7 million for the year ended December 31, 2023, from \$30.8 million for the year ended December 31, 2022. The decrease is mainly due to lower-than-prior-year product royalties driven by lower sales of products from period-to-period, and a permanent contractual decrease in the Qbrexza royalty percentage from the prior-year period.

Research and Development

Research and development expense decreased by \$3.4 million, or 31%, to \$7.5 million for the year ended December 31, 2023 from \$10.9 million for the year ended December 31, 2022. The decrease is related to lower clinical trial expenses to develop our DFD-29 product as the project winds down and eventually concludes.

Selling, General and Administrative Expenses ("SG&A")

Selling, general and administrative expenses decreased by \$15.6 million, or 26%, to \$43.9 million for the year ended December 31, 2023, from \$59.5 million for the year ended December 31, 2022. The decrease is mainly due to our expense reduction efforts primarily in sales and marketing and other SG&A areas. During the last quarter of 2022, we began implementing a cost reduction initiative designed to improve operational efficiencies, optimize expenses, and reduce overall costs. The initiative is intended to reduce selling, general, and administrative expenses to better align costs with their revenue-generating capabilities. In connection with the cost reduction initiative, during the last quarter of 2022 and the first two quarters of 2023, we executed a headcount reduction to our sales force and implemented marketing and other cost cuts.

Loss on impairment of intangible assets

We recorded a loss on the impairment of intangible assets of \$3.1 million during 2023 related to the impairment of the Ximino intangible asset as a result of lower net product revenues and gross profit levels for the Ximino products. We discontinued selling Ximino on September 29, 2023.

Interest Expense

Interest expense decreased \$0.3 million to \$1.7 million for the year ended December 31, 2023, from \$2.0 million for the year ended December 31, 2022. This decrease was driven in part by the total repayment of our prior credit facility with East West Bank during the third quarter of 2023 and no additional borrowing of funds until entering into the Credit Facility with SWK in December 2023. As we utilize this Credit Facility during 2024 to help us fund our operations, we expect interest expenses may increase year-over-year from the fiscal year ended December 31, 2023.

Income tax expense

Income tax expense increased by \$0.2 million during 2023 due to an increase in certain state taxes driven by the Maruho New License Agreement.

Liquidity and Capital Resources

On December 27, 2023, we entered into the Credit Agreement with SWK. The Credit Agreement provides for a term loan Credit Facility in the original principal amount of up to \$20.0 million. On the closing date, we drew \$15.0 million. The remaining \$5.0 million may be drawn upon our request within 12 months after the closing date. The Term Loans mature on December 27, 2027, and bear interest at a rate per annum equal to the three-month term SOFR (subject to a SOFR floor of 5%) plus 7.75%. The interest rate resets quarterly. Interest payments begin in February 2024 and are paid quarterly. Beginning in February 2026, we are required to repay the outstanding principal of the Term Loans quarterly in an amount equal to 7.5% of the principal amount of funded Term Loans. If our total revenue, measured on a trailing twelve-month basis, is greater than \$70.0 million as of December 31, 2025, principal repayment is not required until February 2027, at which point we are required to repay a portion of the outstanding principal amount of funded Term Loans. The SWK Credit Facility also includes both revenue and liquidity covenants, restrictions as to payment of dividends, and is secured by substantially all assets of the Company. As of December 31, 2023, and as of the date of this Annual Report on Form 10-K, the Company was in compliance with the financial covenants under the SWK Credit Facility

On August 31, 2023, we entered into the New License Agreement with Maruho, whereby we granted an exclusive license to Maruho to develop and commercialize Qbrexza® for the treatment of primary axillary hyperhidrosis in South Korea, Taiwan, Hong Kong, Macau, Thailand, Indonesia, Malaysia, Philippines, Singapore, Vietnam, Brunei, Cambodia, Myanmar and Laos (the "Territory"). Under the terms of the New License Agreement, in exchange for the exclusive rights to Qbrexza® in the Territory, Maruho paid us \$19.0 million as a non-refundable upfront payment.

In July 2023, we satisfied all of the outstanding debt obligations we had with East West Bank ("EWB") by voluntarily repaying the outstanding balance on the term loan under the Loan and Security Agreement, dated March 31, 2021 (as Amended, the "EWB facility").

On December 30, 2022, we filed a shelf registration statement on Form S-3 (File No. 333 - 269079), which was declared effective by the SEC on January 23, 2023. This shelf registration statement covers the offering, issuance and sale by us of up to an aggregate of \$150.0 million of our common stock, preferred stock, debt securities, warrants, and units (the "2022 Shelf"). In connection with the 2022 shelf, we have entered into the Sales Agreement with B. Riley relating to shares of our common stock in an at-the-market sales program. In accordance with the terms of the Sales Agreement, we may offer and sell up to 4,900,000 shares of our common stock, from time-to-time through B. Riley acting as our agent or principal. During 2023, we issued 748,703 shares of common stock under the 2022 Shelf, generating net proceeds of \$4.5 million. At December 31, 2023, 4,151,297 shares remain available for issuance under the 2022 Shelf.

We regularly evaluate market conditions, our liquidity profile, and financing alternatives, including out-licensing arrangements for our products, to enhance our capital structure. We may seek to raise capital through debt or equity financings, to expand our product portfolio, and for other strategic initiatives, which may include sales of securities under either our 2022 Shelf or a new registration statement or drawing on the SWK Credit Facility. We cannot make any assurances that such additional financing will be available to us and, if available, the terms may negatively impact our business and operations. At December 31, 2023, we had cash and cash equivalents of approximately \$27.4 million. Our current assumptions, projected commercial sales of our products, clinical development plans and regulatory submission timelines are uncertain and may not emerge as expected. Additionally, as a result of these financial statements.

Cash Flows for the Years Ended December 31, 2023 and 2022

	For	the Years end	Increase		
(\$'s in thousands)		2023	2022	(Decrease)	
Net cash provided by (used in) operating activities	\$	5,240	\$ (13,534)	\$ 18,774	
Net cash used in investing activities		(5,000)	(20,000)	15,000	
Net cash provided by (used in) financing activities		(4,804)	 16,456	(21,260)	
Net change in cash and cash equivalents		(4,564)	 (17,078)	(12,514)	

Operating Activities

Net cash from operating activities changed by \$18.8 million from period-to-period, from \$13.5 million cash used in operating activities for the year ended December 31, 2022 to \$5.2 million net cash provided by operating activities for the year ended December 31, 2023. The change was driven primarily by the lower net loss from period-to-period, driven by our lower expense base and the \$19.0 million payment from Maruho. This was offset by vendor payments as we utilized operating cash and the proceeds of the SWK facility to aggressively pay down our current liabilities.

Investing Activities

Net cash used in investing activities decreased by \$15.0 million, to \$5.0 million for the year ended December 31, 2023, from \$20.0 million for the year ended December 31, 2022. The year ended December 31, 2023 reflects the \$5.0 million deferred cash payment paid in January 2023 related to the VYNE Product Acquisition. The year ended December 31, 2022 reflects the upfront \$20.0 million payment for the VYNE Product Acquisition.

Financing Activities

Net cash used in financing activities increased by \$21.3 million, to \$4.8 million for the year ended December 31, 2023, from \$16.5 million of cash flows provided by financing activities for the year ended December 31, 2022. The increase reflects a cash outflow of \$20.0 million for the repayment of principal on the EWB term loan and net cash outflows of \$2.9 million from the repayment of the EWB revolving line of credit, offset by net proceeds of \$14.6 million from the SWK Term Loan and \$4.5 million from the issuance of common stock under the 2022 Shelf. Net cash provided by financing activities for the year ended December 31, 2022 reflects net proceeds of \$19.8 million from the EWB term loan and net proceeds of \$2.1 million from the EWB revolving line of credit, offset by \$5.0 million in payments of the installment notes related to our previously acquired products.

Material Cash Requirements

In the normal course of business, we enter into contractual obligations that contain cash requirements of which the most significant currently include the following:

• We are required to make regular payments under the SWK Credit Facility. Based on the amount currently outstanding under the SWK facility and current interest rates, and assuming we do not make further draws under the SWK facility, we expect to make the following payments:

		Payments by Period									
	Total	2024	2025	2026	2027						
		(\$'s in thousands)									
Interest	\$ 6,770	\$ 1,998	\$ 1,993	\$ 1,693	\$ 1,086						
Principal	15,000			4,500	10,500						
Exit fee	750		_		750						
Total	\$ 22,520	\$ 1,998	\$ 1,993	\$ 6,193	\$ 12,336						

Should we elect to borrow the remaining \$5.0 undrawn balance under the SWB facility, we would expect to repay additional amounts each year until maturity.

- Pursuant to the Vyne Product Acquisition Agreement, upon the achievement of net sales milestones with respect to the products purchased in the Vyne Product Acquisition, we are also required to pay contingent consideration consisting of a one-time payment, per product, of \$10.0 million and \$20.0 million upon each product reaching annual net sales of \$100 million and \$200 million, respectively. Each required payment must only be paid one time following the first achievement of the applicable annual net sales milestone amount.
- On June 29, 2021, we entered into the DFD-29 Agreement to obtain the global rights for the development and commercialization of DFD-29 with DRL. Based on the development and commercialization of DFD-29, additional contingent regulatory and commercial milestone payments totaling up to \$158.0 million may also become payable. Royalties ranging from ten percent to twenty percent are payable on net sales of the product. Additionally, the Company was required to fund and



oversee the Phase 3 clinical trials beginning upon the license of DFD-29 in 2021. The two Phase 3 clinical trials substantially concluded in July 2023 upon the receipt of positive topline results from the trials. In January 2024, the Company paid a \$4.0 million filing fee to the FDA upon filing of an NDA for DFD-29. The Company is obligated to make a \$3.0 million milestone payment to DRL in April 2024 based on the FDA's acceptance of the NDA filed in January 2024.

- We are contractually obligated to make installment milestone payments of \$3.0 million on Ximino, all of which is classified as current as it is due within a year of December 31, 2023.
- We are contractually obligated to make sales-based royalty payments to Dermira (for Qbrexza), Sun Pharmaceutical Industries (for Exelderm) and PuraCap Caribe (for Targadox). Due to the contingent nature of these obligations, the amounts of these payments cannot be reasonably predicted as of the date of this Annual Report on Form 10-K.

Item 7A. Quantitative and Qualitative Disclosures About Market Risks

We are a smaller reporting company as defined by Rule 12b-2 of the Exchange Act and are not required to provide the information otherwise required under this item.

Item 8. Financial Statements and Supplementary Data.

The information required by this Item is set forth in the financial statements and notes thereto beginning at page F-1 of this Annual Report on Form 10-K.

Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure.

None.

Item 9A. Controls and Procedures.

Disclosure Controls and Procedures

We maintain "disclosure controls and procedures," as such term is defined under Rule 13a-15(e) promulgated under the Exchange Act, designed to ensure that information required to be disclosed in our reports filed pursuant to the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the SEC's rules and forms, and that such information is accumulated and communicated to our management, including our principal executive officer and our principal financial officer, as appropriate to allow timely decisions regarding required disclosures.

In designing and evaluating the disclosure controls and procedures, we recognize that any controls and procedures, no matter how well designed and operated, can provide only reasonable assurance of achieving the desired control objectives, and we are required to apply our judgment in evaluating the cost-benefit relationship of possible controls and procedures. We have carried out an evaluation as of the end of the period covered by this Annual Report on Form 10-K under the supervision, and with the participation, of our management, including our Chief Executive Officer (who serves as our principal executive officer) and our Interim Chief Financial Officer (who serves as our principal financial officer), of the effectiveness of the design and operation of our disclosure controls and procedures.

Based on that evaluation, our Chief Executive Officer and Interim Chief Financial Officer concluded that our disclosure controls and procedures were effective as of the end of the period covered by this Annual Report on Form 10-K in providing reasonable assurance of achieving the desired control objectives.

Management's Report on Internal Control over Financial Reporting

Our management is responsible for establishing and maintaining adequate internal control over financial reporting, as such term is defined in Exchange Act Rules 13a-15(f) and 15d-15(f). Internal control over financial reporting refers to the process designed by, or under the supervision of, our principal executive officer and principal financial officer, and effected by our Board of Directors, management and other personnel, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles, and includes those policies and procedures that:

- (1) pertain to the maintenance of records that in reasonable detail accurately and fairly reflect the transactions and dispositions of our assets;
- (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with GAAP, and that our receipts and expenditures are being made only in accordance with authorization of our management and directors; and
- (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisitions, use or disposition of our assets that could have a material effect on the financial statements.

Internal control over financial reporting has inherent limitations. Internal control over financial reporting is a process that involves human diligence and compliance and is subject to lapses in judgment and breakdowns resulting from human failures. Internal control over financial reporting also can be circumvented by collusion or improper management override. Because of such limitations, there is a risk that material misstatements may not be prevented or detected on a timely basis by internal control over financial reporting. However, these inherent limitations are known features of the financial reporting process. Therefore, it is possible to design into the process safeguards to reduce, though not eliminate, this risk.

Our management assessed the effectiveness of our internal control over financial reporting as of December 31, 2023. In making the assessment, management used the criteria set forth by the Committee of Sponsoring Organizations of the Treadway Commission (COSO) in *Internal Control-Integrated Framework (2013)*. Based on the results of this assessment, management (including our Chief Executive Officer and our Interim Chief Financial Officer) has concluded that, as of December 31, 2023, our internal control over financial reporting was effective.

This Annual Report does not include an attestation report on internal control over financial reporting from our independent registered public accounting firm due to our status as an emerging growth company under the JOBS Act.



Changes in Internal Control over Financial Reporting

There have not been any changes in our internal control over financial reporting (as such term is defined in Rules 13a-15(f) and 15d-15(f) under the Exchange Act) during our most recent fiscal quarter ended December 31, 2023 to which this report relates that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

Item 9B. Other Information

None.

Item 9C. Disclosure Regarding Foreign Jurisdictions that Prevent Inspections.

None.

PART III

Item 10. Directors, Executive Officers and Corporate Governance

The information required by this Item is incorporated herein by reference from our Proxy Statement for our 2024 Annual Meeting of Stockholders.

Item 11. Executive Compensation

The information required by this Item is incorporated herein by reference from our Proxy Statement for our 2024 Annual Meeting of Stockholders.

Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

The information required by this Item is incorporated herein by reference from our Proxy Statement for our 2024 Annual Meeting of Stockholders.

Item 13. Certain Relationships and Related Transactions, and Director Independence

The information required by this Item is incorporated herein by reference from our Proxy Statement for our 2024 Annual Meeting of Stockholders.

Item 14. Principal Accounting Fees and Services

The information required by this Item is incorporated by reference from our Proxy Statement for our 2024 Annual Meeting of Stockholders.

PART IV

Item 15. Exhibits and Financial Statement Schedules

(a) Financial Statements.

The following financial statements are filed as part of this report:

Report of Independent Registered Public Accounting Firm (KPMG LLP, Short Hills, New Jersey; PCAOB# 185)	F-2
Consolidated Balance Sheets as of December 31, 2023 and 2022	F-3
Consolidated Statements of Operations for the years ended December 31, 2023 and 2022	F-4
Consolidated Statement of Changes in Stockholders' Equity for the years ended December 31, 2023 and 2022	F-5
Consolidated Statements of Cash Flows for the years ended December 31, 2023 and 2022	F-6
Notes to Consolidated Financial Statements	F-7

75

(b) Exhibits.

(b) Exhibits.	
Exhibit	
Number	Description
3.1	Third Amended and Restated Certificate of Incorporation of Journey Medical Corporation, filed on March 28, 2022 and incorporated herein
	by reference.
3.2	Amended and Restated Bylaws of Journey Medical Corporation, filed as Exhibit 3.2 to Form 10-K, filed on March 28, 2022 and
	incorporated herein by reference.
4.1	Form of Common Stock Certificate, filed as Exhibit 4.1 to Form S-1, filed on October 22, 2021 and incorporated herein by reference.
4.2	Description of Securities of Journey Medical Corporation, filed as Exhibit 4.2 to Form 10-K, filed on March 28, 2022 and incorporated
	herein by reference.
10.1	Journey Medical Corporation 2015 Stock Plan, filed as Exhibit 10.1 to Form S-1, filed on October 22, 2021 and incorporated herein by
	reference.#
10.2	Amendment to Journey Medical Corporation 2015 Stock Plan, filed as Exhibit 10.1 to Form 8-K filed on June 21, 2022 and incorporated
	herein by reference.#
10.3	Executive Employment Agreement with Claude Maraoui, dated September 22, 2014, filed as Exhibit 10.2 to Form S-1, filed on October 22,
10.4	2021 and incorporated herein by reference.#
10.4	Non-Employee Director Compensation Plan, filed as Exhibit 10.4 to Form S-1, filed on October 22, 2021 and incorporated herein by
10.5	reference.#
10.5	Journey Medical Corporation 2023 Employee Stock Purchase Plan, filed as Exhibit 10.1 to Form 8-K filed on June 23, 2023 and incorporated herein by reference.#
10.6	Asset Purchase Agreement for Obrexza, entered into by and between Journey Medical Corporation and Dermira, Inc., a subsidiary of Eli
10.0	Lilly and Company, dated as of March 31, 2021, filed as Exhibit 10.6 to Form S-1, filed on October 22, 2021 and incorporated herein by
	reference.**
10.7	Asset Purchase Agreement between VYNE Therapeutics Inc. and Journey Medical Corporation, dated as of January 12, 2022, filed as
10.7	Exhibit 10.1 to Form 8-K filed on January 13, 2022 and incorporated herein by reference.**
10.8	License and Supply Agreement for Accutane, entered into by and between Journey Medical Corporation and Dr. Reddy's Laboratories Ltd.,
10.0	dated as of July 29, 2020, filed as Exhibit 10.7 to Form S-1, filed on October 22, 2021 and incorporated herein by reference.**
10.9	License and Supply Agreement for Targadox, entered into by and between Journey Medical Corporation and Blu Caribe Inc., dated as of
	March 10, 2015, filed as Exhibit 10.8 to Form S-1, filed on October 22, 2021 and incorporated herein by reference.**
10.10	First Amendment to the License and Supply Agreement for Targadox, entered into by and between Journey Medical Corporation and Blu
	Caribe Inc., dated as of August 26, 2015, filed as Exhibit 10.9 to Form S-1, filed on October 22, 2021 and incorporated herein by
	reference.**
10.11	Asset Purchase Agreement for Exelderm, entered into by and between Journey Medical Corporation and Sun Pharmaceutical Industries,
	Inc., dated as of August 31, 2018, filed as Exhibit 10.10 to Form S-1, filed on October 22, 2021 and incorporated herein by reference.**
10.12	Amendment 1 to the Asset Purchase Agreement for Exelderm, entered into by and between Journey Medical Corporation and Sun
	Pharmaceutical Industries, Inc., dated as of September 5, 2018, filed as Exhibit 10.11 to Form S-1, filed on October 22, 2021 and
	incorporated herein by reference. **
10.13	Asset Purchase Agreement for Ximino, entered into by and between Journey Medical Corporation and Sun Pharmaceutical Industries, Inc.,
	dated as of July 22, 2019, filed as Exhibit 10.12 to Form S-1, filed on October 22, 2021 and incorporated herein by reference.**
10.14	Asset Purchase Agreement for the Anti-itch Product, entered into by and between Journey Medical Corporation and Sun Pharmaceutical
	Industries, Inc., dated as of December 18, 2020, filed as Exhibit 10.13 to Form S-1, filed on October 22, 2021 and incorporated herein by
	reference.**
10.15	License, Collaboration, and Assignment Agreement for DFD-29, entered into by and between Journey Medical Corporation and Dr.
	Reddy's Laboratories Ltd., dated as of June 29, 2021, filed as Exhibit 10.14 to Form S-1, filed on October 22, 2021 and incorporated herein
10.16	by reference.**
10.16	Asset Purchase Agreement between Journey Medical Corporation and VYNE Therapeutics Inc., dated as of January 12, 2022, filed as
	Exhibit 10.1 to the Form 8-K filed on January 13, 2022 and incorporated herein by reference. **

76

10.17	Fortress Promissory Note, dated as of June 6, 2015, filed as Exhibit 10.16 to Form S-1, filed on October 22, 2021 and incorporated herein
	by reference.
10.18	At Market Issuance Sales Agreement, dated as of December 30, 2022, by and between Journey Medical Corporation and B. Riley
	Securities, Inc., filed as Exhibit 1.2 to Form S-3, filed on December 30, 2022 and incorporated herein by reference.
10.19	License Agreement, dated as of August 31, 2023, between Journey Medical Corporation and Maruho Co., Ltd. filed as Exhibit 10.1 to Form
	<u>10 - Q filed on November 13, 2023.**</u>
10.20	Second Amended and Restated License Agreement, dated as of August 31, 2023, between Journey Medical Corporation and Maruho Co.,
	Ltd. filed as Exhibit 10.2 to Form 10 - Q filed on November 13, 2023.**
10.21	Credit Agreement, dated as of December 27, 2023, between Journey Medical Corporation with SWK Funding LLC. ***
21.1	List of Subsidiaries of Journey Medical Corporation.*
23.1	Consent of KPMG LLP.*
31.1	Certification of Principal Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002. *
31.2	Certification of Principal Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.*
32.1	Certification of Principal Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. ***
32.2	Certification of Principal Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. ***
97.1	Clawback Policy of Journey Medical Corporation*
101	The following financial information from the Company's Quarterly Report on Form 10-K for the period ended December 31, 2023,
	formatted in Extensible Business Reporting Language (XBRL): (i) the Consolidated Balance Sheets, (ii) the Consolidated Statements of
	Operations, (iii) the Consolidated Statement of Stockholders' Equity, (iv) the Consolidated Statements of Cash Flows, and (v) Notes to the
	Consolidated Financial Statements.
104	Cover page from the Company's Annual Report on Form 10-K for the year ended December 31, 2023, formatted in Inline XBRL.

Item 16. Form 10-K Summary

The Company has elected not to provide summary information.

^{*} Filed herewith.

^{**} Certain portions of this exhibit have been omitted pursuant to Item 601(b)(10) of Regulation S-K.

^{***} Furnished herewith.# Management Compensation Arrangement.

INDEX TO FINANCIAL STATEMENTS

Report of Independent Registered Public Accounting Firm(KPMG LLP, Short Hills, New Jersey; PCAOB# 185) F-2
Consolidated Balance Sheets as of December 31, 2023 and 2022 F-3
Consolidated Statements of Operations for the years ended December 31, 2023 and 2022 F-4
Consolidated Statement of Changes in Stockholders' Equity for the years ended December 31, 2023 and 2022 F-5
Consolidated Statements of Cash Flows for the years ended December 31, 2023 and 2022 F-6
Notes to Consolidated Financial Statements F-7

Report of Independent Registered Public Accounting Firm

To the Stockholders and Board of Directors

Journey Medical Corporation:

Opinion on the Consolidated Financial Statements

We have audited the accompanying consolidated balance sheets of Journey Medical Corporation and subsidiary (the Company) as of December 31, 2023 and 2022, the related consolidated statements of operations, changes in stockholders' equity and cash flows for each of the years then ended December 31, 2023, and the related notes (collectively, the consolidated financial statements). In our opinion, the consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2023 and 2022, and the results of its operations and its cash flows for each of the years then ended December 31, 2023, in conformity with U.S. generally accepted accounting principles.

Going Concern

The accompanying consolidated financial statements have been prepared assuming that the Company will continue as a going concern. As discussed in Note 1 to the consolidated financial statements, the Company has suffered recurring losses from operations, and as a result has concluded that this raises substantial doubt about its ability to continue as a going concern. Management's plans in regard to these matters are also described in Note 1. The consolidated financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Basis for Opinion

These consolidated financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these consolidated financial statements based on our audits. We are a public accounting firm registered with the Public Company Accounting Oversight Board (United States) (PCAOB) and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the consolidated financial statements are free of material misstatement, whether due to error or fraud. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. As part of our audits, we are required to obtain an understanding of internal control over financial reporting an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion.

Our audits included performing procedures to assess the risks of material misstatement of the consolidated financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the consolidated financial statements. We believe that our audits provide a reasonable basis for our opinion.

/s/ KPMG LLP

We have served as the Company's auditor since 2021.

Short Hills, New Jersey

March 28, 2024

JOURNEY MEDICAL CORPORATION CONSOLIDATED BALANCE SHEETS (in thousands, except share and per share data)

		Decem	ber 31,	1,			
		2023		2022			
ASSETS							
Current assets							
Cash and cash equivalents	\$	27,439	\$	32,003			
Accounts receivable, net of reserves		15,222		28,208			
Inventory		10,206		14,159			
Prepaid expenses and other current assets		3,588		3,309			
Total current assets		56,455		77,679			
Intangible assets, net		20,287		27,197			
Operating lease right-of-use asset, net		101		189			
Other assets		6		95			
Total assets	\$	76,849	\$	105,160			
LIABILITIES AND STOCKHOLDERS' EQUITY							
Current liabilities							
Accounts payable	\$	18,149	\$	36,570			
Due to related party	Φ	10,149	φ	413			
Accrued expenses		20.350		19,388			
Accrued interest		20,550		19,500			
Income taxes payable		53		35			
Line of credit		55		2,948			
Deferred cash payment, net of discount				4,991			
Installment payments – licenses, short-term		3,000		2,244			
Operating lease liability, short-term		99		83			
Total current liabilities		41.868		66,832			
Total current haonities		41,000		00,852			
Term loan, net of discount		14,622		19,826			
Installment payments – licenses, long-term				1,412			
Operating lease liability, long-term		9		108			
Total liabilities		56,499		88,178			
Commitments and continuous (Net 14)							
Commitments and contingencies (Note 14)							
Stockholders' equity							
Common stock, \$.0001 par value, 50,000,000 shares authorized, 13,323,952 and 11,765,700 shares issued and outstanding as of December 31, 2023 and December 31, 2022, respectively		1		1			
Common stock - Class A, \$.0001 par value, 50,000,000 shares authorized, 6,000,000 shares issued and outstanding as of		1		1			
December 31, 2023 and December 31, 2022		1		1			
Additional paid-in capital		92,703		85,482			
Accumulated deficit		(72,355)		(68,502)			
Total stockholders' equity		20,350		16,982			
	\$	76,849	\$	10,982			
Total liabilities and stockholders' equity	3	/0,849	3	105,100			

The accompanying notes are an integral part of these consolidated financial statements.

JOURNEY MEDICAL CORPORATION CONSOLIDATED STATEMENTS OF OPERATIONS (in thousands, except share and per share data)

	Year Ended	December 31,
	2023	2022
Revenue:		
Product revenue, net	\$ 59,662	\$ 70,995
Other revenue	19,519	2,674
Total revenue	79,181	73,669
Operating expenses		
Cost of goods sold – product revenue	26,660	30,775
Research and development	7,541	10,943
Selling, general and administrative	43,910	59,468
Loss on impairment of intangible assets	3,143	
Total operating expenses	81,254	101,186
Loss from operations	(2,073)	(27,517)
Other expense (income)		
Interest income	(322)	(60)
Interest expense	1,698	2,019
Foreign exchange transaction losses	183	89
Total other expense (income)	1,559	2,048
Loss before income taxes	(3,632)	(29,565)
Income tax expense	221	63
Net Loss	<u>\$ (3,853)</u>	\$ (29,628)
Net loss per common share:		
Basic and diluted	\$ (0.21)	\$ (1.69)
Weighted average number of common shares:		
Basic and diluted	18,232,422	17,531,274

The accompanying notes are an integral part of these consolidated financial statements.

JOURNEY MEDICAL CORPORATION CONSOLIDATED STATEMENT OF CHANGES IN STOCKHOLDERS' EQUITY (in thousands, except share data)

	Comme	on St	ock	Commo	n Sto	ck A	 dditional Paid-in	(A	Accumulated	Sh	Total areholders'
	Shares		Amount	Shares	A	mount	Capital		Deficit)		Equity
Balance as of December 31, 2021	11,316,344	\$	1	6,000,000	\$	1	\$ 80,915	\$	(38,874)	\$	42,043
Share-based compensation	_			_			4,425		_		4,425
Exercise of stock options for cash	155,649						142		_		142
Issuance of common stock for vested restricted stock units	293,707			_					_		
Net loss							—		(29,628)		(29,628)
Balance as of December 31, 2022	11,765,700	\$	1	6,000,000	\$	1	\$ 85,482	\$	(68,502)	\$	16,982
Share-based compensation							2,606		_		2,606
Exercise of stock options for cash	82,300			_			121		_		121
Issuance of common stock for vested restricted stock units	727,249						—		_		
Issuance of common stock, ATM offering, net of issuance costs											
of \$140	748,703						4,494		_		4,494
Net loss			—	_		_	_		(3,853)		(3,853)
Balance as of December 31, 2023	13,323,952	\$	1	6,000,000	\$	1	\$ 92,703	\$	(72,355)	\$	20,350

The accompanying notes are an integral part of these consolidated financial statements.

JOURNEY MEDICAL CORPORATION CONSOLIDATED STATEMENTS OF CASH FLOWS (in thousands)

S	2023 (3,853) 435 353 356 3,767 88 2,606 3,143 12,551 3,953 (279) 	\$	2022 (29,628 284 770 63 4,277 88 8 4,425
S	435 353 356 3,767 88 2,606 3,143 12,551 3,953	\$	284 77(6: 4,27' 81 4,42:
2	435 353 356 3,767 88 2,606 3,143 12,551 3,953	\$	284 77(62 4,27 88 4,42
	353 356 3,767 88 2,606 3,143 12,551 3,953		770 63 4,277 88 4,425
	353 356 3,767 88 2,606 3,143 12,551 3,953		770 63 4,277 88 4,425
	356 3,767 88 2,606 3,143 12,551 3,953		63 4,277 88 4,425
	3,767 88 2,606 3,143 12,551 3,953		4,27 88 4,42
	88 2,606 3,143 12,551 3,953		88 4,425
	2,606 3,143 12,551 3,953		4,425
	3,143 12,551 3,953		
	12,551 3,953		
	3,953		/#
	3,953		
			(5,380
	(279)		1,744
			(871
			55
	(18,421)		14,343
	(218)		(228
	962		(3,568
	(138)		160
	18		27
	(83)		(95
	5,240		(13,534
	(5,000)		(20,000
	(5,000)		(20,000
	121		142
	(1,000)		(5,000
	_		(214
	28,000		5,000
	(30,948)		(2,864
	15,000		19,763
	(20,000)		_
	(91)		
	(380)		
	4,494		_
			(371
	(4,804)		16,456
	(4,564)		(17,078
	32,003		49,081
\$	27,439	\$	32,003
s	1 127	8	993
			168
ې	181	Ф	108
0		\$	
\$			4,740
		$ \begin{array}{r} 121\\(1,000)\\ & & \\28,000\\(30,948)\\15,000\\(20,000)\\(91)\\(380)\\ & \\4,494\\ & \\\hline\\(4,804)\\(4,804)\\(4,564)\\(4,564)\\(4,564)\\(32,003\\\underline{5}\\27,439\\\hline\\\hline\\(5)\\1,127\\[5]\\181\end{array} $	(5,000) 121 (1,000) 28,000 (30,948) 15,000 (20,000) (91) (380) 4,494 (4,564) (4,564) 32,003 <u>\$ 27,439</u> <u>\$</u> <u>\$ 1,127</u> <u>\$ 1,127</u> <u>5 1,127</u>

The accompanying notes are an integral part of these consolidated financial statements.

Notes to Financial Statements

NOTE 1. ORGANIZATION AND PLAN OF BUSINESS OPERATIONS

Journey Medical Corporation (collectively "Journey" or the "Company") is a commercial-stage pharmaceutical company that focuses on the development and commercialization of pharmaceutical products for the treatment of dermatological conditions. The Company's current product portfolio includes seven branded and two authorized generic prescription drugs for dermatological conditions that are marketed in the U.S. The Company acquires rights to products and product candidates by licensing or otherwise acquiring an ownership interest in, funding the research and development of, and eventually commercializing, the products through its exclusive field sales organization.

As of December 31, 2023 and 2022, the Company is a majority-owned subsidiary of Fortress Biotech, Inc. ("Fortress" or "Parent").

Liquidity and Capital Resources

At December 31, 2023, the Company had \$27.4 million in cash and cash equivalents as compared to \$32.0 million at December 31, 2022.

On December 27, 2023, the Company entered into a Credit Agreement (the "Credit Agreement") with SWK Funding LLC ("SWK"). The Credit Agreement provides for a term loan facility (the "Credit Facility") in the original principal amount of up to \$20.0 million. On the closing date, the Company drew \$15.0 million. The remaining \$5.0 million may be drawn upon the Company's request within 12 months after the closing date. Loans under the Credit Facility (the "Term Loans") mature on December 27, 2027, and bear interest at a rate per annum equal to the three-month term Secured Overnight Financing Rate ("SOFR") (subject to a SOFR floor of 5%) plus 7.75%. The interest rate resets quarterly. Interest payments begin in February 2024 and are paid quarterly. Beginning in February 2026, the Company is required to repay a portion of the outstanding principal of the Term Loans quarterly in an amount equal to 7.5% of the principal amount of funded Term Loans.

On August 31, 2023, the Company entered into a license agreement (the "New License Agreement") with Maruho Co., Ltd., a Japanese company specializing in dermatology ("Maruho"), whereby the Company granted an exclusive license to Maruho to develop and commercialize Qbrexza® for the treatment of primary axillary hyperhidrosis in South Korea, Taiwan, Hong Kong, Macau, Thailand, Indonesia, Malaysia, Philippines, Singapore, Vietnam, Brunei, Cambodia, Myanmar and Laos (the "Territory"). Under the terms of the New License Agreement, in exchange for the exclusive rights to Qbrexza ® in the Territory, Maruho paid \$19.0 million to the Company as a non-refundable upfront payment.

On December 30, 2022, the Company filed a shelf registration statement on Form S-3 (File No. 333-269079), which was declared effective by the Securities and Exchange Commission ("SEC") on January 26, 2023. This shelf registration statement covers the offering, issuance and sale by the Company of up to an aggregate of \$150.0 million of the Company's common stock, preferred stock, debt securities, warrants, and units (the "2022 Shelf"). In connection with the 2022 Shelf, the Company entered into an At Market Issuance Sales Agreement (the "Sales Agreement") relating to shares of the Company's common stock. The Company may offer and sell up to 4,900,000 shares of its common stock, from time to time. During 2023, the Company issued 748,703 shares of common stock under the 2022 Shelf, generating net proceeds of \$4.5 million. At December 31, 2023, 4,151,297 shares remain available for issuance under the 2022 Shelf.

The Company regularly evaluates market conditions, its liquidity profile, and financing alternatives, including out-licensing arrangements for its products to enhance its capital structure. The Company may seek to raise capital through debt or equity financings to expand its product portfolio and for other strategic initiatives, which may include sales of securities under either the 2022 Shelf or a new registration statement or drawing on the SWK Credit Facility. The Company cannot make any assurances that such additional financing will be available and, if available, the terms may negatively impact the Company's business and operations. The Company's current assumptions, projected commercial sales of our products, clinical development plans and regulatory submission timelines are uncertain and may not emerge as expected. Additionally, as a result of recurring losses, substantial doubt exists about the Company's ability to continue as a going concern for a period of at least twelve months from the date of issuance of these financial statements.

The accompanying financial statements have been prepared on a going concern basis, which contemplates the realization of assets and satisfaction of liabilities in the ordinary course of business. The financial statements do not include any adjustments relating to the



Notes to Financial Statements

recoverability and classification of recorded asset amounts or the amounts and classification of liabilities that may be necessary if the Company is unable to continue as a going concern.

NOTE 2. BASIS OF PRESENTATION AND SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Presentation and Principles of Consolidation

The Company's consolidated financial statements have been prepared in conformity with accounting principles generally accepted in the United States of America ("GAAP"). The Company's consolidated financial statements include the accounts of the Company and the accounts of the Company's wholly-owned subsidiary, JG Pharma, Inc. ("JG" or "JG Pharma"). All intercompany balances and transactions have been eliminated.

Emerging Growth Company

From time to time, new accounting pronouncements are issued by the Financial Accounting Standards Board ("FASB") or other standard-setting bodies and adopted by the Company as of the specified effective date. Unless otherwise discussed, the impact of recently issued standards that are not yet effective will not have a material impact on the Company's audited consolidated financial statements upon adoption. Under the Jumpstart Our Business Startups Act of 2012, as amended, the Company meets the definition of an emerging growth company and elected the extended transition period for complying with new or revised accounting standards, which delays the adoption of these accounting standards until they would apply to private companies.

Use of Estimates

The preparation of consolidated financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and the reported amounts of expenses during the reporting period. Significant estimates made by management include provisions for coupons, chargebacks, wholesaler fees, specialty pharmacy discounts, managed care rebates, product returns, and other allowances customary to the pharmaceutical industry. Significant estimates made by management also include inventory realization, valuation of intangible assets, useful lives of amortizable intangible assets and share-based compensation. Actual results may differ materially and adversely from these estimates. To the extent there are material differences between the estimates and actual results, the Company's future results of operations will be affected.

Segment Information

Operating segments are defined as components of an enterprise about which separate discrete information is available for evaluation by the chief operating decision maker, or decision-making group, in deciding how to allocate resources and in assessing performance. The Company views its operations and manages its business in one segment, which reflects products for the treatment of dermatological conditions.

Concentrations of Credit Risk

Financial instruments that potentially subject the Company to significant concentration of credit risk consist primarily of cash. Periodically, the Company may maintain deposits in financial institutions in excess of government insured limits. Management believes that the Company is not exposed to significant credit risk as the Company's deposits are held at financial institutions that management believes to be of high credit quality. The Company has not experienced any losses on these deposits.



Notes to Financial Statements

The Company's accounts receivable primarily represent amounts due from drug wholesalers and specialty pharmacies in the United States. The Company performs periodic credit evaluations of customers and does not require collateral. An allowance for doubtful accounts is maintained for potential credit losses based on the aging of accounts receivable, historical bad debts experience, and the customer's current ability to pay its obligations to the Company. Accounts receivables balances are written off against the allowance when it is probable that the receivable will not be collected. See Note 17 for significant customers.

Cash and Cash Equivalents

The Company considers highly liquid investments with a maturity of three months or less when purchased to be cash equivalents. Cash and cash equivalents at December 31, 2023 and 2022 consisted entirely of cash and cash equivalents in institutions within the United States. Balances at certain institutions have exceeded Federal Deposit Insurance Corporation insured limits.

Accounts Receivable, Net

The Company's accounts receivable consists of amounts due from customers related to product sales and have standard payment terms. For certain customers, the accounts receivable for the customer are net of prompt payment or specialty pharmacy discounts. The Company monitors the financial performance and creditworthiness of its customers so that it can properly assess and respond to changes in their credit profile. The Company reserves against accounts receivable for estimated losses that may arise from a customer's inability to pay, and any amounts determined to be uncollectible are written off against the reserve when it is probable that the receivable will not be collected. The Company has historically not experienced significant credit losses. The allowance for doubtful accounts was \$0.5 million and \$0.4 million at December 31, 2023 and 2022, respectively.

Inventories

Inventories are recorded at the lower of cost or net realizable value, with cost determined on a first-in, first-out basis. The Company periodically reviews the composition of inventory in order to identify excess, obsolete, slow-moving or otherwise non-saleable items taking into account anticipated future sales compared with quantities on hand, and the remaining shelf life of goods on hand. If non-saleable items are observed and there are no alternate uses for the inventory, the Company records a write-down to net realizable value in the period that the decline in value is first recognized. The Company's inventory reserves were \$0.3 million and \$0.4 million at December 31, 2023 and 2022, respectively.

Leases

Arrangements meeting the definition of a lease are classified as operating or financing leases and are recorded on the consolidated balance sheet as both a right-of-use asset and lease liability, calculated by discounting fixed lease payments over the lease term at the rate implicit in the lease or the Company's incremental borrowing rate. Lease liabilities are increased by interest and reduced by payments each period, and the right-of-use asset is amortized over the lease term. For operating leases, interest on the lease liability and the amortization of the right-of-use asset result in straight-line rent expense over the lease term. Variable lease expenses are recorded when incurred.

In calculating the right-of-use asset and lease liability, the Company elects to combine lease and non-lease components.

Research and Development Costs

Research and development costs are expensed as incurred. Research and development costs primarily consist of personnel related expenses, payments made to third parties for license and milestone costs related to in-licensed products and technology, and payments made to third party contract research organizations.

Contingencies

The Company records accruals for contingencies and legal proceedings expected to be incurred in connection with a loss contingency when it is probable that a liability has been incurred and the amount can be reasonably estimated.



Notes to Financial Statements

If a loss contingency is not probable but is reasonably possible, or is probable but cannot be estimated, the nature of the contingent liability, together with an estimate of the range of possible loss if determinable and material, would be disclosed.

Fair Value Measurement

The Company follows accounting guidance on fair value measurements for financial assets and liabilities measured at fair value on a recurring basis. Under the accounting guidance, fair value is defined as an exit price, representing the amount that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. As such, fair value is a market-based measurement that should be determined based on assumptions that market participants would use in pricing an asset or a liability.

The accounting guidance requires fair value measurements be classified and disclosed in one of the following three categories:

- Level 1: Quoted prices in active markets for identical assets or liabilities.
- Level 2: Observable inputs other than Level 1 prices for similar assets or liabilities that are directly or indirectly observable in the marketplace.
- Level 3: Unobservable inputs which are supported by little or no market activity and that are financial instruments whose values are determined using pricing models, discounted cash flow methodologies, or similar techniques, as well as instruments for which the determination of fair value requires significant judgment or estimation.

The fair value hierarchy also requires an entity to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. Assets and liabilities measured at fair value are classified in their entirety based on the lowest level of input that is significant to the fair value measurement. The Company's assessment of the significance of a particular input to the fair value measurement in its entirety requires management to make judgments and consider factors specific to the asset or liability.

Certain of the Company's financial instruments are not measured at fair value on a recurring basis but are recorded at amounts that approximate their fair value due to their liquid or short-term nature, such as accounts payable, accrued expenses and other current liabilities.

Intangible Assets

Intangible assets are reported at cost, less accumulated amortization and impairments. Intangible assets with finite lives are amortized over their estimated useful lives, which represents the estimated life of the product. Amortization is calculated using the straight-line method.

During the ordinary course of business, the Company has entered into certain licenses and asset purchase agreements. Potential milestone payments for achieving sales targets or regulatory development milestones are recorded when it is probable of achievement. Upon a milestone payment being achieved, the milestone payment will be capitalized and amortized over the remaining useful life for approved products and expensed for milestones prior to FDA approval. Royalty payments are recorded as cost of goods sold as sales are recognized.

Impairment of Long-Lived Assets

The Company reviews long-lived assets, including intangible assets with finite useful lives, for impairment whenever events or changes in business circumstances indicate that the carrying amount of the assets may not be fully recoverable (a "triggering event"). Factors that the Company considers in deciding when to perform an impairment review include significant underperformance of the long-lived asset in relation to expectations, significant negative industry or economic trends, and significant changes or planned changes in the use of the assets. If an impairment review is performed to evaluate a long-lived asset for recoverability, the Company compares forecasts of undiscounted cash flows expected to result from the use and eventual disposition of the long-lived asset to its carrying value. An impairment loss would be recognized when estimated undiscounted future cash flows expected to result from the use of an asset are less than its carrying amount. The impairment loss would be based on the excess of the carrying value of the impaired asset over its fair value, determined based on discounted cash flows. During the year ended December 31, 2023, the Company recorded an impairment



Notes to Financial Statements

loss associated with its intangible asset balance. See Note 5 for further details. The Company did not record any impairment losses on long-lived assets for the year ended December 31, 2022.

Share-based Compensation

The Company has a share-based compensation plan in place and records the associated share-based compensation expense over the requisite service period. The share-based compensation plan and related compensation expense are discussed more fully in Note 16 to the Company's consolidated financial statements.

Compensation expense for service-based stock options is charged against operations on a straight-line basis over the vesting period, which is generally four years. Forfeitures are recorded as they occur. Share-based compensation costs are recorded in both research and development and selling, general and administrative expense in the Company's consolidated statements of operations. Options granted have a term of 10 years from the grant date.

The Company estimates the fair value of all service-based stock option awards as of the grant date by applying the Black-Scholes option pricing valuation model. The application of this valuation model involves assumptions, including the fair value of the common stock, expected volatility, risk-free interest rate, expected dividends and the expected term of the option. The assumptions used in calculating the fair value of share-based awards represent management's best estimates and involve inherent uncertainties and the application of management's judgment. The following inputs are used in the Black-Scholes calculation.

Expected term—The Company has elected to use the "simplified method" for estimating the expected term of options, whereby the expected term equals the arithmetic average of the vesting term and the original contractual term of the option (generally 10 years).

Expected volatility— Historical information is the primary basis for the selection of the expected volatility of options granted. However, as the Company has limited trading history for its common shares, the expected volatility was estimated based on the average volatility for comparable guideline publicly traded biotechnology companies over a period equal to the expected term of the stock option grants. The comparable companies were chosen based on their similar size, stage in the life cycle or area of specialty.

Risk-free interest rate— The risk-free interest rate is selected based upon yields of United States Treasury issues with a term equal to the expected life of the option being valued.

Expected dividend yield—The Company has not issued any dividends in our history and do not expect to issue dividends over the life of the options; therefore, the Company has estimated the dividend yield to be zero.

Restricted stock units ("RSU's") that are service based are recorded as deferred compensation and amortized into compensation expense on a straight-line basis over the vesting period, which ranges from three to four years in duration. Compensation cost for service based RSU's is based on the grant date fair value of the award, which is the closing market price of the Company's common stock on the grant date multiplied by the number of shares awarded.

Net (Loss) Income Per Share

Basic net (loss) income per share of common stock is calculated by dividing net (loss) income by the weighted-average number of shares of common stock outstanding during the reporting period. Diluted earnings per share is calculated by dividing net income by the weighted-average number of shares of common stock outstanding during the reporting period after giving effect to dilutive potential common shares for stock options and restricted stock units, determined using the treasury stock method. See Note 19 below.

Revenue Recognition

The Company records and recognizes revenue in a manner that depicts the transfer of promised goods or services to customers in an amount that reflects the consideration to which the company expects to be entitled in exchange for those goods or services. The Company's revenues primarily result from contracts with customers, which are generally short-term and have a single performance obligation – the delivery of product. The Company's performance obligation to deliver products is satisfied at the point in time that the goods are received by the customer, which is when the customer obtains title to and has the risks and rewards of ownership of the



Notes to Financial Statements

products. The transaction price is the amount of consideration to which the Company expects to be entitled in exchange for transferring promised goods to a customer. The consideration promised in a contract with a customer may include fixed amounts, variable amounts, or both.

Many of the Company's products sold are subject to a variety of deductions. Revenues are recorded net of provisions for variable consideration, including coupons, chargebacks, wholesaler fees, specialty pharmacy discounts, managed care rebates, product returns, and other deductions customary to the pharmaceutical industry. Accruals for these provisions are presented in the consolidated financial statements as reductions to gross sales in determining net sales and as a contra asset within accounts receivable, net (if settled via credit) and other current liabilities (if paid in cash). Amounts recorded for revenue deductions can result from a series of judgements about future events and uncertainties and can rely on estimates and assumptions. The following section briefly describes the nature of the Company's provisions for variable consideration and how such provisions are estimated:

Coupons — The Company offers coupons on products for qualified commercially-insured parties with prescription drug co-payments. Such product sales flow through both traditional wholesaler and specialty pharmacy channels. Coupons are processed and redeemed at the time of prescription fulfilment by the pharmacy. The majority of the coupon reserve accrual at the end of the period reflects expected redemptions for product in the distribution channel. The expected accrual reserve requires us to estimate the distribution channel inventory at period end, the expected redemption rates, and the cost per coupon claim that the Company expects to receive. The estimate of product remaining in the distribution channel is comprised of estimated inventory at the wholesaler as well as an estimate at the specialty pharmacies, which the Company estimates based upon historical ordering patterns. The estimated redemption rate is based on the tocupon rate.

Chargebacks and Government Chargebacks — The Company sells a portion of its products indirectly through wholesaler distributors to contracted indirect customers and qualified government healthcare providers. The Company enters into specific agreements with or provides discounts to these indirect customers and entities to establish pricing for the Company's products, and in-turn, the indirect customers and entities independently purchase these products. The Company's provision for chargebacks is based on expected sell-through levels by the Company's wholesale customers to the indirect customers and estimated wholesaler inventory levels as well as historical chargeback rates. The Company continually monitors its reserve for chargebacks and adjusts the reserve accordingly when expected chargebacks differ from actual experience.

Wholesaler fees — The Company provides allowances to its wholesale customers for sales order management, data, and distribution services. The Company also pays administrative and other fees to certain wholesale customers consistent with pharmaceutical industry practices. The Company records a provision for these fees based on contracted rates. Assumptions used to establish the provision include contract sales volumes and average contract pricing. The Company regularly reviews the information related to these estimates and adjusts the provision accordingly.

Specialty Pharmacy Discounts — The Company has in place contractual arrangements with specialty pharmacies and provides for contractually agreed upon discounts. These discounts are recorded at the time of sale based on the customer's contracted rate and recorded as a reduction of revenue.

Notes to Financial Statements

Managed Care Rebates — The Company is subject to rebates in connection with its agreements with certain contracted commercial payers. The Company estimates its managed care rebates based on the Company's estimated payer mix and the applicable contractual rebate rate. The Company's accrual for managed care rebates is based on an estimate of future claims that the Company expects to receive, which considers an estimate for inventory in the distribution channel. The accrual is recognized at the time of sale, resulting in a reduction of gross product revenue.

Product Returns — Consistent with industry practice, the Company offers customers a right to return any unused product. The customer's right of return commences six months prior to product expiration date and ends one year after product expiration date. Products returned for expiration are reimbursed at current wholesale acquisition cost or indirect contract price. The Company estimates the amount of its product sales that may be returned by the Company's customers and accrues this estimate as a reduction of revenue in the period the related product revenue is recognized. The Company estimates products returns as a percentage of sales to its customers.

Income Taxes

As of December 31, 2023, the Company was52.01% owned by Fortress Biotech, Inc. ("Fortress") and was filing consolidated federal tax returns and consolidated or combined state tax returns in multiple jurisdictions with Fortress for tax years prior to 2021. As the Company completed its initial public offering on November 12, 2021, it deconsolidated from the Fortress consolidated group for federal income tax purpose. The financial statements recognize the current and deferred income tax consequences that result from the activities during the current and preceding periods, as if the Company were a separate taxpayer rather than a member of the Fortress consolidated income tax return group. Fortress has agreed that the Company does not have to make payments to Fortress for the use of net operating losses ("NOLs") of Fortress (including other Fortress group members). Since Fortress does not require the Company to pay in any form for the utilization of the consolidated group's NOLs, the tax benefit realized have been recorded as a capital contribution.

The Company records income taxes using the asset and liability method. Deferred income tax assets and liabilities are recognized for the future tax effects attributable to temporary differences between the financial statement carrying amounts of existing assets and liabilities and their respective income tax bases, and operating loss and tax credit carryforwards. The Company establishes a valuation allowance if it believes it is more likely than not that the deferred tax assets will not be recovered based on an evaluation of objective verifiable evidence. The Company has considered its history of cumulative tax and book income/loss incurred since inception, and the other positive and negative evidence, and has concluded that it is not more likely than not that it will realize the benefits of the net deferred tax assets as of December 31, 2023 and 2022 and therefore a full valuation allowance on all of the deferred tax assets is required.

For tax positions that are more likely than not of being sustained upon audit, the Company recognizes the largest amount of the benefit that is greater than 50% likely of being realized. For tax positions that are not more likely than not of being sustained upon audit, the Company does not recognize any portion of the benefit. For the years ended December 31, 2023 and 2022, the Company had no unrecognized tax benefits and does not anticipate any significant change to the unrecognized tax benefit balance. The Company classifies interest and penalties related to uncertain tax positions as income tax expense, if applicable. There was no interest expense or penalties related to unrecognized tax benefits recorded through December 31, 2023 and 2022.

Comprehensive Income

The Company has no components of other comprehensive income, and therefore, comprehensive income equals net income.

Recently Issued Accounting Pronouncements

In November 2023, the FASB issued ASU No. 2023-07, Segment Reporting (Topic 280): Improvements to Reportable Segment Disclosures, which requires that an entity report segment information in accordance with Topic 280, Segment Reporting. The amendment in the ASU is intended to improve reportable segment disclosure requirements primarily through enhanced disclosures about significant segment expenses. The amendments in this update are effective for fiscal years beginning after December 15, 2023, and interim periods within fiscal years beginning after December 15, 2024. The Company is currently evaluating the impact of the new standard on its consolidated financial statements.

In December 2023, the FASB issued ASU No. 2023-09, Income Taxes (Topic 740): Improvements to Income Tax Disclosures, which expands disclosures in an entity's income tax rate reconciliation table and disclosures regarding cash taxes paid both in the U.S. and



Notes to Financial Statements

foreign jurisdictions. The update will be effective for annual periods beginning after December 15, 2024. The Company is currently evaluating the impact that this guidance will have on its consolidated financial statements and disclosures.

NOTE 3. INVENTORY

The Company's inventory consisted of the following at December 31, 2023 and 2022:

	December 31,			
(\$'s in thousands)		2023		2022
Raw materials	\$	4,640	\$	6,454
Work-in-process		884		395
Finished goods		4,987		7,739
Inventory at cost		10,511		14,588
Inventory reserves		(305)		(429)
Total Inventories	\$	10,206	\$	14,159

NOTE 4. ASSET ACQUISITION

In January 2022, the Company entered into an agreement with Vyne Therapeutics Inc. ("Vyne") to acquire two United States Food and Drug Administration ("FDA") approved topical minocycline products, Amzeeq (minocycline) topical foam, 4%, and Zilxi (minocycline) topical foam, 1.5%, and a Molecule Stabilizing TechnologyTM proprietary platform from Vyne for an upfront payment of \$ 20.0 million and an additional \$5.0 million payment on the one year anniversary of the closing (the "Vyne APA"). The Company also acquired the associated inventory related to the products.

The Vyne APA also provides for contingent net sales milestone payments, on a product-by-product basis. In the first calendar year in which annual net sales reach each of \$100 million and \$200 million, the Company is required to make a one-time payment of \$10.0 million and \$20.0 million, respectively, in that year only, per product. In addition, the Company will pay Vyne 10% of any upfront payment received by the Company from a licensee or sublicensee of the products in any territory outside of the United States, subject to exceptions for certain jurisdictions as detailed in the Vyne APA.

The following table summarizes the aggregate consideration transferred for the assets acquired by the Company in connection with the Vyne APA:

(\$'s in thousands)	Con	gregate sideration ansferred
Consideration transferred to Vyne at closing	\$	20,000
Fair Value of deferred cash payment due January 2023		4,740
Transaction costs		223
Total consideration transferred at closing	\$	24,963

The fair value of the deferred cash payment was accreted to the \$.0 million January 2023 cash payment over a one-year period through interest expense. The Company made the \$5.0 million deferred cash payment in January 2023.

Notes to Financial Statements

The following table summarizes the assets acquired in the Vyne APA:

(\$'s in thousands)	Assets Recognized
Inventory	\$ 6,041
Identifiable intangibles:	
Amzeeq intangible	15,162
Zilxi intangible	3,760
Fair value of net identifiable assets acquired	\$ 24,963

The intangible assets were valued using an income approach, while the inventory was valued using a final sales value less cost to dispose approach.

NOTE 5. INTANGIBLES

The Company's finite-lived intangible assets consist of acquired intangible assets. During the year ended December 31, 2023, the Company experienced lower net product revenues and gross profit levels for its Ximino products. Based on these results, the Company revised the financial outlook and plans for its Ximino products. The Company assessed the revised forecast for Ximino and determined that this constituted a triggering event, and the results of the analysis indicated the carrying amount was not expected to be recovered. The Company recorded an intangible asset impairment charge of \$3.1 million during the year ended December 31, 2023. This non-cash charge was recorded to loss on impairment of intangible assets in the consolidated statements of operations.

The table below provides a summary of the Company's intangible assets at December 31, 2023 and 2022, respectively:

	Estimated	nated Decem			Ι,
(\$'s in thousands)	Useful Lives (Years)	2023		2023	
Intangible assets - product licenses	3-9	\$	37,925	\$	37,925
Accumulated amortization			(14,495)		(10,728)
Accumulated impairment loss			(3,143)		
Total intangible assets		\$	20,287	\$	27,197

The Company's amortization expense for the years ended December 31, 2023 and 2022 was approximately \$3.8 million and \$4.3 million, respectively. Amortization expense is recorded as a component of cost of goods sold in the Company's consolidated statements of operations.

Future amortization of the Company's intangible assets is as follows:

		Total
For the years ended	Am	ortization
December 31, 2024	\$	3,257
December 31, 2025		3,257
December 31, 2026		2,471
December 31, 2027		1,775
Thereafter		5,585
Subtotal		16,345
Asset not yet placed in service		3,942
Total	\$	20,287

Notes to Financial Statements

NOTE 6. LICENSES ACQUIRED

DFD-29

In June 2021, the Company entered a license, collaboration, and assignment agreement (the "DFD-29 Agreement") to obtain global rights for the development and commercialization of a late-stage development modified release oral minocycline for the treatment of rosacea ("DFD-29") with Dr. Reddy's Laboratories, Ltd ("DRL"); provided, that DRL retained certain rights to the program in select markets including Brazil, Russia, India and China. Pursuant to the terms and conditions of the DFD-29 Agreement, the Company paid \$10.0 million. Based on the development and commercialization of DFD-29, additional contingent regulatory and commercial milestone payments totaling up to \$158.0 million may also become payable by the Company. The Company is required to pay royalties ranging from approximately ten percent to fifteen percent on net sales of the DFD-29 product, subject to certain reductions. Additionally, the Company was required to fund and oversee the Phase 3 clinical trials beginning upon the license of DFD-29 in 2021. The Phase 3 clinical trials substantially concluded in July 2023 upon the Company's receipt of positive topline results from the trials. From inception to date the Company has incurred approximately \$23.8 million in costs associated with the development of DFD-29.

Qbrexza

In March 2021, the Company executed an Asset Purchase Agreement (the "Qbrexza APA") with Dermira, Inc., a subsidiary of Eli Lilly and Company ("Dermira"). Pursuant to the terms of the Qbrexza APA, the Company acquired the rights to Qbrexza® (glycopyrronium), a prescription cloth towelette to treat primary axillary hyperhidrosis in patients nine years of age or older. The Company paid the upfront fee of \$12.5 million to Dermira. In addition, the Company is obligated to pay Dermira up to \$144.0 million in the aggregate upon the achievement of certain sales milestones. The royalty structure for the agreement is tiered with royalties for the first two years ranging from approximately 40% to 30%. Thereafter for a period of eight years royalties are approximately 12.0% to 19.0%. Royalty amounts are subject to certain reductions in the event there is a loss of exclusivity.

Accutane

In July 2020, the Company entered into an exclusive license and supply agreement for Accutane (the "Accutane Agreement") with DRL. Pursuant to the Accutane Agreement, the Company paid \$5.0 million. Three additional milestone payments totaling \$17.0 million are contingent upon the achievement of certain net sales milestones. The Company is required to pay royalties in an amount equal to a low-double digit percentage of net sales. The term of the Accutane Agreement is ten years and renewable upon mutual agreement. Each party may terminate the Accutane Agreement for an uncured material breach by the other party or for certain bankruptcy or insolvency related events. The Company may also terminate the Accutane Agreement without cause upon 180 days written notice to DRL.

NOTE 7: FAIR VALUE MEASUREMENTS

Financial assets and liabilities measured at fair value on a recurring basis are summarized below:

		December 31, 2023				
(\$'s in thousands)	Level 1	Level 2	Lev	el 3		Total
Assets		_	_			
Cash and cash equivalents	\$ 27,439	s —	\$		\$	27,439
Total	\$ 27,439	<u>s</u> —	\$	_	\$	27,439
1000	<u> </u>					
			er 31, 2022			Total
(§'s in thousands) Assets	Level 1	Decemb Level 2	er 31, 2022 Leve		_	Total
(S's in thousands)		Level 2	,		\$	Total 32,003

Notes to Financial Statements

NOTE 8. RELATED PARTY AGREEMENTS

Shared Services Agreement with Fortress

On November 12, 2021, the Company and Fortress entered into an arrangement to share the cost of certain legal, finance, regulatory, and research and development employees (the "Shared Services Agreement"). Fortress' Executive Chairman and Chief Executive Officer is the Executive Chairman of the Company. Under the terms of the Shared Services Agreement, the Company will reimburse Fortress for the salary and benefit costs associated with these employees based upon actual hours worked on Journey-related projects following the completion of the Company's initial public offering, which occurred in November 2021. In addition, the Company reimburses Fortress for various payroll-related costs and selling, general and administrative costs incurred by Fortress for the benefit of the Company.

For the year ended December 31, 2023 and 2022, the Company recorded related party expenses to Fortress of approximately \$1.1 million and \$0.1 million, respectively. The due to related party liability at December 31, 2023 and 2022, were \$0.2 million and \$0.4 million, respectively, and primarily relate to reimbursable expenses incurred by Fortress on behalf of the Company. The Company would have incurred these costs irrespective of the relationship with Fortress.

NOTE 9. ACCRUED EXPENSES

Accrued expenses for the years ended December 31, 2023 and 2022 consisted of the following:

	December 31,			
's in thousands)		2023		2022
Accrued expenses and other short-term liabilities:				
Accrued coupons and rebates	\$	9,987	\$	7,604
Accrued compensation		3,374		2,586
Accrued royalties payable		2,015		2,627
Return reserve		4,077		3,689
Accrued inventory		352		112
Accrued research and development		20		1,404
Accrued legal, accounting and tax		185		334
Accrued iPledge program		174		447
Other		166		585
Total accrued expenses	\$	20,350	\$	19,388

NOTE 10. INSTALLMENT PAYMENTS — LICENSES

The following tables show the details of the Company's installment payments – licenses for the years ended December 31, 2023 and 2022:

	December 31, 2023					
(\$'s in thousands)	Short-Term Long-Term					Total
Installment payments - licenses	\$	3,000	\$	_	\$	3,000
Less: imputed interest						
Sub-total installment payments - licenses	\$	3,000	\$	_	\$	3,000
	December 31, 2022					
(\$'s in thousands)	Sh	ort-Term	Lo	ng-Term		Total
Installment payments - licenses	\$	2,500	\$	1,500	\$	4,000
Less: imputed interest		(256)		(88)		(344)
Sub-total installment payments - licenses	\$	2,244	\$	1,412	\$	3,656

Notes to Financial Statements

NOTE 11. OPERATING LEASE OBLIGATIONS

The Company leases 3,681 square feet of office space in Scottsdale, Arizona. The lease was set to expire on December 31, 2022. In September 2022, the Company amended the lease to extend the lease term for an additional 25 months at an annual rate of approximately \$0.1 million. The amended lease will expire on January 31, 2025.

The Company recorded rent expense as follows (dollars in thousands):

	For th	For the Years Ended December 31,				
	20	23		2022		
Operating lease cost	\$	97	\$	93		
Variable lease cost		5		4		
Total lease cost	\$	102	\$	97		

The following table summarizes quantitative information about the Company's operating leases (dollars in thousands):

	For t	For the Years Ended December 31,			
	2023			2022	
Cash paid for amounts included in the measurement of lease liabilities	\$	92	\$	100	
Right-of-use assets exchanged for new operating lease liabilities	\$		\$	188	
Weighted-average remaining lease term - operating leases		1.1		2.1	
Weighted-average discount rate - operating leases		6.25 %		6.25 %	

As of December 31, 2023, future payments of operating lease liabilities are as follows:

For the year ended December 31,	(\$'s in thousands)	
2024	\$ 10	02
2025		9
Total lease payments	1	11
Less: present value discount		(3)
Total operating lease liabilities	\$ 1	08

Notes to Financial Statements

NOTE 12. DEBT

The Company's Debt obligations at December 31, 2023 and 2022 were as follows:

(\$'s in thousands)	Dec	cember 31, 2023
Principal balance	\$	15,000
Plus: Exit fee		750
Less: Debt discount and fees		(1,128)
Net carry amount (Long-term)	\$	14,622

December 31, 2022

(\$'s in thousands)	rincipal Salance	 mortized int & Fees	Net Carry Amount		
Deferred cash payment	\$ 5,000	\$ 9	\$	4,991	
EWB Revolving LOC	2,948	_		2,948	
Total Short-Term Debt	\$ 7,948	\$ 9	\$	7,939	
EWB Term Loan (Long-term)	\$ 20,000	\$ 174	\$	19,826	
Total Debt & Obligations	\$ 27,948	\$ 183	\$	27,765	

SWK Long-Term Debt

On December 27, 2023, the Company entered into a Credit Agreement with SWK. The Credit Agreement provides for a term loan Credit Facility in the original principal amount of up to \$20.0 million. On the Closing Date, the Company drew \$15.0 million. The remaining \$5.0 million may be drawn upon request by the Company within 12 months after the Closing Date. Term Loans under the Credit Facility mature on December 27, 2027. The Term Loans accrue interest which is payable quarterly in arrears. The Term Loans bear interest at a rate per annum equal to the three-month term SOFR (subject to a SOFR floor of 5%) plus 7.75%. The interest rate resets quarterly.

Beginning in February 2026, the Company is required to repay a portion of the outstanding principal of the Term Loans quarterly in an amount equal to 7.5% of the principal amount of funded Term Loans, with any remaining principal balance due on the maturity date. If the total revenue of the Company, measured on a trailing twelve-month basis, is greater than \$70.0 million as of December 31, 2025, the principal repayment start date is extended from February 2026 to February 2027, at which point the Company is required to repay a portion of the outstanding principal of the Term Loans quarterly in an amount equal to 15% of the principal amount of funded Term Loans, with any remaining principal balance due on the maturity date.

The Company may at any time prepay the outstanding principal balance of the Term Loans in whole or in part. Prepayment of the Term Loans is subject to payment of a prepayment premium equal to (i) 2% of the Term Loans prepaid plus the amount of interest that would have been due through the first anniversary of the Closing Date if the Term Loans are prepaid prior to the first anniversary of the Closing Date, (ii) 1% of the Term Loans prepaid if the Term Loans are prepaid on or after the first anniversary of the Closing Date but prior to the second anniversary of the Closing Date, or (iii) 0% if prepaid thereafter.

Upon repayment in full of the Term Loans, the Company will pay an exit fee equal to5% of the original principal amount of the Term Loans. Additionally, the Company paid an origination fee of \$0.2 million on the Closing Date and incurred issuance costs of \$0.2 million, both of which have been recorded as a debt discount. The Company is accreting the carrying value of the SWK Term Loan to the original principal balance plus the exit fee over the term of the loan using the effective interest method. The amortization of the discount is accounted for as interest expense. The effective interest rate on the SWK Term Loan for the fiscal year ended December 31, 2023 was 15.1%. The fair value of the debt approximates its carrying value.



Notes to Financial Statements

The SWK Credit Facility also includes both revenue and liquidity covenants, restrictions as to payment of dividends, and is secured by substantially all assets of the Company. As of December 31, 2023, the Company was in compliance with the financial covenants under the SWK Credit Facility.

As of December 31, 2023, the contractual maturities of the long-term debt, including the payment of the exit fee, are as follows (dollars in thousands):

Years ending December 31,	Т	erm Loan
2024	\$	_
2025		
2026		4,500
2027		11,250
Total		15,750
Debt discount		(1,128)
Total, net		14,622
Current portion		—
Term-loan (long-term)	\$	14,622

East West Bank Line of Credit and Long-Term Debt

The Company was previously party to a Loan and Security Agreement, dated March 31, 2021 (as amended, the "EWB Facility"), with East West Bank ("EWB"), under which EWB made a \$20.0 million term loan and a \$10.0 million revolving line of credit available to the Company. During 2023, the Company voluntarily repaid the entire \$20.0 million outstanding term loan principal balance under the EWB Facility. The repayment satisfied all of the Company's outstanding debt obligations under the EWB Facility. The Company has no further obligations to EWB.

NOTE 13. INTEREST EXPENSE AND FINANCING FEES

Interest expense and financing fees for the years ended December 31, 2023 and 2022 consisted of the following:

	,	Year Ended December 31,				
		2023	2022			
Interest payments on term loans and LOC	\$	989	\$	1,153		
Amortization/accretion		356		347		
Imputed interest on acquired intangible assets		353		519		
Total interest expense and financing fees	\$	1,698	\$	2,019		

NOTE 14. COMMITMENTS AND CONTINGENCIES

License Agreements

The Company has undertaken to make contingent milestone payments to the licensors of its portfolio of drug products and candidates. In addition, the Company is required to pay royalties to such licensors based on a percentage of net sales of each drug candidate following regulatory marketing approval. For additional information on future milestone payments and royalties, see Note 4 and Note 6.

NOTE 15. STOCKHOLDERS' EQUITY

Common Stock

The Company's Certificate of Incorporation, as amended, authorizes the Company to issue 50,000,000 shares of \$0.0001 par value Common Stock of which 6,000,000 shares are designated and authorized as Class A Common Stock.



Notes to Financial Statements

Voting Rights

Each holder of Common Stock is entitled to one vote per share of Common Stock held on all matters submitted to a vote of the stockholders, including the election of directors. The Company's Certificate of Incorporation and bylaws do not provide for cumulative voting rights.

Each holder of Class A Common Stock is entitled to a number of votes that is equal to 1.1 times a fraction, the numerator of which is the sum of the shares of outstanding Common Stock, including the Class A Common Stock, and the denominator of which is the number of outstanding shares of Class A Common Stock. Thus, the holders of the Class A Common Stock will at all times constitute a voting majority.

Dividends

The holders of the Company's outstanding shares of Common Stock and Class A Common Stock are entitled to receive dividends, if any, as may be declared from time to time by the Company's Board of Directors out of legally available funds.

Liquidation

In the event of the Company's liquidation, dissolution or winding up, holders of Common Stock and Class A Common Stock will be entitled to share ratably in the net assets legally available for distribution to stockholders after the payment of all of the Company's debts and other liabilities, subject to the satisfaction of any liquidation preference granted to the holders of any outstanding shares of Preferred Stock.

Rights and Preference

Holders of the Company's Common Stock and Class A Common Stock have no preemptive, conversion or subscription rights, and there is no redemption or sinking fund provisions applicable to either the Common Stock or the Class A Common Stock. The rights, preferences and privileges of the holders of Common Stock and Class A Common Stock are subject to, and may be adversely affected by, the rights of the holders of shares of any series of the Company's Preferred Stock that are or may be issued.

NOTE 16. SHARE-BASED COMPENSATION

In 2015, the Company's Board of Directors adopted, and stockholders approved, the Journey Medical Corporation 2015 Stock Plan (the "Plan") authorizing the Company to grant up to 4,642,857 shares of Common Stock to eligible employees, directors, and consultants in the form of restricted stock, restricted stock units ("RSUs"), stock options and other types of grants. The amount, terms, and exercisability provisions of grants are determined by the Board of Directors. At the Company's 2022 Annual Meeting, held on June 21, 2022, the Company's stockholders approved, among other matters, an amendment to the Plan to increase the number of shares of Common Stock issuable under the Plan by 3,000,000 to 7,642,857. At December 31, 2023 there were 1,487,994 shares available for issuance under the Plan.

The Company grants stock options to employees, non-employees and Directors with exercise prices equal to the closing price of the underlying shares of the Company's common stock on the Nasdaq Capital Market on the date that the options are granted. Options granted have a term of ten years from the grant date. Options granted generally vest over four-year period. Compensation cost for stock options is charged against operations on a straight-line basis over the vesting period. The Company estimates the fair value of stock options on the grant date by applying the Black-Scholes option pricing valuation model.



Notes to Financial Statements

In 2023, the Company's Board of Directors adopted, and stockholders approved, the Journey Medical Corporation 2023 Employee Stock Purchase Plan (the "2023 ESPP"). The Company initially reserved 300,000 shares of common stock for future issuance under the 2023 ESPP. As of December 31, 2023, 300,000 shares were available for issuance under the 2023 ESPP.

The following table summarizes the components of share-based compensation expense in the consolidated statements of operations for the years ended December 31, 2023 and 2022:

	Year Ended December 31			er 31,
(\$'s in thousands)		2023		2022
Research and development	\$	110	\$	73
Selling, general and administrative		2,496		4,352
Total non-cash compensation expense related to share-based compensation included in operating expense	\$	2,606	\$	4,425

Stock Options

The weighted-average key assumptions used in determining the fair value of options granted for the year ended December 31, 2023 are as follows:

	2023
Risk-free interest rate	3.45% - 4.44%
Expected volatility	89.09% - 101.75%
Weighted average expected volatility	91.08%
Expected term (years)	5.37 - 6.25
Expected dividend yield	0%

The weighted average grant-date fair value of stock options issued during the year ended December 31, 2023 was \$.27 per share. The weighted average grant-date fair value of stock options issued during the year ended December 31, 2022 was \$2.67 per share.

The following table summarizes the Company's stock option activity for the year ended December 31, 2023:

	Number of Shares	Weighted average exercise price	Aggregate intrinsic value	Weighted average remaining contractual life (years)
Outstanding options at December 31, 2022	2,960,000	\$ 1.76	\$ 2,217,815	5.65
Granted	430,756	1.64		_
Exercised	(82,300)	1.46	_	_
Forfeited	(504,687)	3.06		_
Expired	(33,900)	3.58	_	_
Outstanding options at December 31, 2023	2,769,869	\$ 1.49	\$ 3,441,146	4.53
Options vested and exercisable at December 31, 2023	1,984,475	\$ 0.96	\$ 1,758,134	2.83

For the years ended December 31, 2023 and 2022, the Company issued82,300 and 155,649 shares, respectively, of Common Stock upon the exercise of outstanding stock options and received proceeds of \$120,555 and \$142,330, respectively. For the years ended December 31, 2023 and 2022, approximately \$0.5 million and \$0.8 million, respectively, of stock option compensation cost was charged against operations. At December 31, 2023, the Company had unrecognized share-based compensation expense related to all unvested options of \$0.9 million, which the Company expects to recognize over a weighted-average period of approximately 1.9 years.



Notes to Financial Statements

Restricted Stock Units

The following table summarizes the Company's RSU activity for the year ended December 31, 2023:

	Number of units	av gra	eighted verage ant date ir value
Unvested balance at December 31, 2022	2,261,048	\$	4.05
Granted	119,888		1.82
Vested	(727,249)		3.96
Forfeited	(346,764)		4.10
Unvested balance at December 31, 2023	1,306,923	\$	3.88

For the years ended December 31, 2023 and 2022 the Company issued727,249 and 293,707 shares of Common Stock, respectively, upon the vesting of RSU's amounting to \$2.9 million and \$0.9 million, respectively, in total aggregate fair market value. For the years ended December 31, 2023 and 2022, approximately \$2.0 million and \$3.6 million, respectively, of RSU compensation cost was charged against operations. At December 31, 2023 approximately 1,306,923 of RSU's remained unvested and there was approximately \$1.6 million of unrecognized compensation cost related to RSUs, which the Company expects to recognize over a weighted-average period of approximately 1.5 years.

Employee Stock Purchase Plan

The 2023 ESPP provides that eligible employees may contribute up to 10% of their eligible earnings toward a semi-annual purchase of the Company's common stock. The 2023 ESPP is qualified under Section 423 of the Internal Revenue Code. The employee's purchase price is derived from a formula based on the closing price of the common stock on the first day of the offering period versus the closing price on the last date of purchase (or, if not a trading day, on the immediately preceding trading day). The offering period under the 2023 ESPP has a duration of six months, and the purchase price with respect to each offering period beginning on or after such date is, until otherwise amended, equal to 85% of the lesser of (i) the fair market value of the Company's common stock at the commencement of the applicable six-month offering period or (ii) the fair market value of the Company's common stock at the company estimates the fair value of the common stock under the 2023 ESPP using a Black-Scholes valuation model. The fair value of the offering period beginning August 1, 2023 using the Black-Scholes option valuation model and the straight-line attribution approach with the following assumptions: risk-free interest rate (5.5%); expected term (0.5 years); expected volatility (129%); and an expected dividend yield (0%). The Company recorded \$46,700 of stock-based compensation under the 2023 ESPP for the year ended December 31, 2023. As of December 31, 2023, there was unrecognized stock-based compensation expense of \$9,524 related to the current ESPP offering period, which ends January 31, 2024.

NOTE 17. REVENUES FROM CONTRACTS WITH CUSTOMERS

Disaggregation of Net Revenues

The Company has the following actively marketed products, Qbrexza®, Amzeeq®, Zilxi®, Accutane®, Exelderm®, Targadox®, and Luxamend®. All of the Company's product revenues are recorded in the U.S.



Notes to Financial Statements

Revenues by product are summarized as follows:

	Year Ended December 31,		ber 31,	
(\$ in thousands)		2023		2022
Qbrexza®	\$	25,410	\$	26,715
Accutane®		20,168		18,373
Amzeeq		6,201		7,242
Targadox®		3,204		7,972
Exelderm®		2,395		3,463
Zilxi		1,962		2,273
Ximino®		287		4,957
Luxamend®		35		_
Total product revenues	\$	59,662	\$	70,995

The Company recognized other revenue as follows:

	Year Ended December 31,			ıber 31,
(\$in thousands)		2023		2022
Non-refundable upfront payment from Maruho	\$	19,000	\$	_
Net milestone payment from Maruho		_		2,500
Royalties on sales of Rapifort® Wipes2.5%	\$	519		174
Total other revenue	\$	19,519	\$	2,674

Other revenue reflects royalties on sales of Rapifort® Wipes 2.5% in Japan, from Maruho, the Company's exclusive out-licensing partner in Japan. Other revenue for the year ended December 31, 2023 also reflects a net \$19.0 million payment from Maruho under the New License Agreement. Other revenue for the year ended December 31, 2022 also reflects a net \$2.5 million milestone payment from Maruho. In January 2022, Maruho received manufacturing and marketing approval in Japan for Rapifort Wipes 2.5% (Japanese equivalent to U.S. FDA approved Qbrexza[®]), for the treatment of primary axillary hyperhidrosis, triggering the net payment.

Maruho License Agreement

On August 31, 2023, the Company entered into the New License Agreement with Maruho. Under the terms of the New License Agreement, the Company granted an exclusive license to develop and commercialize Qbrexza for the treatment of primary axillary hyperhidrosis in the Territory. Prior to the date of the New License Agreement, the Company and Maruho were party to an existing exclusive amended and restated license agreement (the "First A&R License Agreement"), under which Maruho acquired exclusive license rights to Qbrexza in Japan.

In connection with Journey's entry into the New License Agreement, Journey and Maruho also entered into the Second Amended and Restated Exclusive License Agreement (the "Second A&R License Agreement"), which supersedes the First A&R License Agreement. The Second A&R License Agreement contains modifications that remove Maruho's obligation to pay Journey royalties on its net sales of Rapifort (the Japanese equivalent of Qbrexza) in Japan for sales occurring after October 1, 2023 and removes Maruho's obligation to pay \$10.0 million to Journey in the event that Maruho achieves net sales of at least ¥4 billion (yen) of Rapifort during a single fiscal year. All other remaining potential milestone payment obligations, which aggregate to \$45.0 million, remain in full force and effect.

Under the terms of the New License Agreement, in exchange for the exclusive rights to Qbrexza in the Territory, Maruho paid the Company a \$19.0 million non-refundable upfront payment. Maruho is also obligated to pay royalties to the Company related to sales of the product in the Territory equal to the corresponding rate payable by the Company to Dermira under the asset purchase agreement between Journey and Dermira.

The New License Agreement may be terminated by Maruho in its entirety or on a region-by-region basis for convenience upon30 days' notice to the Company.

Notes to Financial Statements

The Company does not have any obligation to assist in the regulatory approval efforts of Maruho under the New License Agreement in the Territory. The arrangement with Maruho provides for the transfer of the following: (i) an exclusive license of Qbrexza from Journey to Maruho, including all related patents and know-how, and (ii) a non-exclusive license from Journey to Maruho to manufacture or have manufactured drug substance and products outside of the Territory, but exclusively for the sale of products in the Territory.

Significant Customers

As of December 31, 2023, one of the Company's customers accounted for more than 10.0% of its total accounts receivable balance at13.0%. As of December 31, 2022, two of the Company's customers accounted for more than 10.0% of its total accounts receivable balance at16.7% and 10.4%.

For the year ended December 31, 2023 and 2022, none of the Company's customers accounted for more than 10.0% of its total gross product revenue.

NOTE 18. INCOME TAXES

The components of the income tax provision are as follows:

	Years Ended December 31,			ıber 31,
(\$'s in thousands)		2023	2022	
Current:				
Federal	\$	34	\$	_
State		187		63
Total current		221		63
Deferred:				
Federal		(753)		(6,701)
State		(94)		(1,737)
Total deferred		(847)		(8,438)
Valuation allowance		847		8,438
Total income tax expense	\$	221	\$	63

Notes to Financial Statements

Deferred income taxes reflect the net tax effects of (a) temporary differences between the carrying amounts of assets and liabilities for financial reporting purposes and the amounts used for income tax purposes, and (b) operating losses and tax credit carryforwards.

The significant components of the Company's deferred tax assets consisted of the following:

		December 31,			
(\$'s in thousands)	2	023		2022	
Deferred tax assets:					
Net operating loss carryforwards	\$	4,376	\$	6,553	
Amortization of license fees		5,603		4,951	
R&D capitalization		3,312		2,462	
Stock compensation		1,081		1,293	
Lease liability		27		48	
Reserve on sales return, discount and bad debt		3,566		2,988	
Accruals and reserves		723		574	
Tax credits		1,491		1,152	
Business interest expense deduction limit				322	
State taxes				13	
Total deferred tax assets		20,179		20,356	
Less: valuation allowance		(20,154)		(19,307)	
Deferred tax assets, net	\$	25	\$	1,049	
Deferred tax liability:					
Section 481(a) adjustment on reserve on sales return, discount and bad debt				(1,001)	
Right-of-use asset		(25)		(48)	
Deferred tax assets, net	\$		\$		

A reconciliation of the statutory tax rates and the effective tax rates is as follows:

	Years Ended Dece	mber 31,
	2023	2022
Percentage of pre-tax income:		
U.S. federal statutory income tax rate	21 %	21 %
State taxes, net of federal benefit	(1)%	4 %
Non-deductible items	(4)%	(0)%
Provision to return	(1)%	0 %
State tax adjustments	(1)%	0 %
Change in valuation allowance	(23)%	(28)%
Share-based compensation	(10)%	0 %
Tax credits	13 %	3 %
Effective income tax rate	(6)%	(0)%

As required by ASC 740, the Company has evaluated the evidence bearing upon the realizability of its deferred tax assets. Based on the weight of available evidence, both positive and negative, the Company has determined that it is more likely than not that it will not realize the benefits of these assets. Accordingly, the Company recorded a valuation allowance of \$20.2 million at December 31, 2023. The valuation allowance increased by \$0.8 million during the year ended December 31, 2023, primarily as a result of the increase in NOL carryforwards generated in the current period.

As of December 31, 2023, the Company had federal and state NOL carryforwards of approximately \$15.6 million and \$22.7 million, respectively. The Federal NOL carryforwards do not expire, but \$19.5 million of the state NOL carryforwards expire if not utilized prior to 2042.

Notes to Financial Statements

Utilization of the U.S. federal and state NOL carryforwards may be subject to a substantial annual limitation under Sections 382 and 383 of the Internal Revenue Code of 1986, as amended, and corresponding provisions of state law, due to ownership changes that have occurred previously or that could occur in the future. These ownership changes may limit the amount of NOL carryforwards that can be utilized annually to offset future taxable income and tax liabilities, respectively. The Company has performed calculations to support that its NOL carryovers are subject to limitations under section 382 ("382 Limitations"). Based on the analysis of the NOL carryovers subject to the 382 Limitations, the Company has concluded that the 382 Limitations would not prevent the Company from utilizing all of its NOL carryovers prior to expiration.

The Company is subject to U.S. federal and state taxes. As of December 31, 2023, the earliest federal tax year open for the assessment of income taxes under the applicable statutes of limitations is its 2020 tax year. The expiration of the statute of limitations related to the various state income and franchise tax returns varies by state.

NOTE 19. NET (LOSS) INCOME PER COMMON SHARE

The Company accounts for and discloses net earnings (loss) per share using the treasury stock method. Net earnings (loss) per common share, or basic earnings (loss) per share, is computed by dividing net earnings (loss) by the weighted-average number of common shares outstanding. Net earnings (loss) per common share assuming dilutions, or diluted earnings (loss) per share, is computed by reflecting the potential dilution from the exercise of in-the-money stock options, and non-vested restricted stock units.

The Company's basic and diluted weighted-average number of common shares outstanding for years ended December 31, 2023 and 2022 were as follows:

	Year ended D	ecember 31,
	2023	2022
Basic and diluted	18,232,422	17,531,274
Potentially dilutive securities:		
Unvested restricted stock units	1,306,923	2,261,048
Stock options	1,345,193	1,566,131
Total potentially dilutive securities	20,884,538	21,358,453

The Company's Common Stock equivalents, including unvested restricted stock and options have been excluded from the computation of diluted loss per share for the years ended December 31, 2023 and 2022, as the effect would be to reduce the loss per share. Therefore, the weighted average Common Stock outstanding used to calculate both basic and diluted income loss per share is the same for the years ended December 31, 2023 and 2022.

NOTE 20. SUBSEQUENT EVENTS

The Company evaluates events that occur after the period's end date through the date the financial statements are available to be issued. Accordingly, management has evaluated subsequent events through the date these financial statements are issued and has determined that no subsequent events require disclosure in these financial statements.



SIGNATURES

Pursuant to the requirements of Section 12 of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Journey Medical Corporation

By: /s/ Claude Maraoui Name: Claude Maraoui Title: President, Chief Executive Officer, and Director March 28, 2024

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of the registrant and in the capacities and on the dates indicated.

Signature	Title	Date
/s/ Claude Maraoui	President, Chief Executive Officer and Director	March 28, 2024
Claude Maraoui	(Principal Executive Officer)	
/s/ Lindsay A. Rosenwald, M.D. Lindsay A. Rosenwald, M.D.	Executive Chairman	March 28, 2024
/s/ Joseph Benesch Joseph Benesch	Interim Chief Financial Officer (Principal Financial Officer)	March 28, 2024
/s/ Neil Herskowitz Neil Herskowitz	Director	March 28, 2024
/s/ Justin Smith Justin Smith	Director	March 28, 2024

78

Execution Version

CREDIT AGREEMENT

among

JOURNEY MEDICAL CORPORATION, as Borrower,

SWK FUNDING LLC, as Agent, Sole Lead Arranger and Sole Bookrunner,

and

the financial institutions party hereto from time to time as Lenders

Dated as of December 27, 2023

FOR PURPOSES OF SECTIONS 1272, 1273, AND 1275 OF THE UNITED STATES INTERNAL REVENUE CODE, THIS NOTE IS BEING ISSUED WITH "ORIGINAL ISSUE DISCOUNT." PLEASE CONTACT MICHAEL MINER, VICE PRESIDENT, 5956 SHERRY LANE, SUITE 650, DALLAS, TEXAS 75225, TELEPHONE: TO OBTAIN INFORMATION REGARDING THE ISSUE PRICE, THE ISSUE DATE, THE AMOUNT OF ORIGINAL ISSUE DISCOUNT, AND THE YIELD TO MATURITY.

SECTION 1.	DEFINITIONS; INTERPRETATION.	1
1.1	Definitions	1
1.2	Interpretation	13
SECTION 2.	CREDIT FACILITY.	14
2.1	Term Loan Commitments	14
2.2	Loan Procedures	14
	2.2.1 Initial Advance	14
	2.2.2 Subsequent Term Loan	14
2.3	Commitments Several	15
2.4	Indebtedness Absolute; No Offset; Waiver	15
2.5	Loan Accounting	16
	2.5.1 Recordkeeping	16
	2.5.2 Notes	16
2.6	Payment of Interest	16
	2.6.1 Interest Rates	16
	2.6.2 Payments of Interest and Principal	17
2.7	Fees	17
2.8	Prepayment	17
	2.8.1 Mandatory Prepayment	17
	2.8.2 Voluntary Prepayment	18
2.9	Repayment of Term Loan	18
	2.9.1 Revenue-Based Payment Amount	18
	2.9.2 Principal	20
2.10	Payment	20
	2.10.1 Making of Payments	20
	2.10.2 Application of Payments and Proceeds Following an Event of Default	20
	2.10.3 Set-off	20
	2.10.4 Proration of Payments	20
SECTION 3.	YIELD PROTECTION.	21
3.1	Taxes	21
3.2	Increased Cost	23
3.3	[Reserved]	25
3.4	Manner of Funding; Alternate Funding Offices	25
3.5	Conclusiveness of Statements; Survival	25
SECTION 4.	CONDITIONS PRECEDENT.	25
4.1	Prior Debt	25
4.2	General	25
4.3	Fees	26
4.4	Representations, Warranties, Defaults	27
4.5	Diligence	27
4.6	Corporate Matters	27
4.7	No Material Adverse Effect	27
SECTION 5.	REPRESENTATIONS AND WARRANTIES.	27
5.1	Organization	27

- i -

5.2	Authorization; No Conflict	28
5.3	Validity; Binding Nature	28
5.4	Financial Condition	28
5.5	No Material Adverse Effect	28
5.6	Litigation	28
5.7	Ownership of Properties; Liens	29
5.8	Capitalization	29
5.9	Pension Plans	29
5.10	Investment Company Act	29
5.11	No Default	29
5.12	Margin Stock	29
5.13	Taxes	29
5.14	Solvency	30
5.15	Environmental Matters	30
5.16	Insurance	30
5.17	Information	30
5.18	Intellectual Property; Products and Services	30
5.19	Restrictive Provisions	31
5.20	Labor Matters	31
5.21	Material Contracts	31
5.22	Compliance with Laws; Health Care Laws	32
5.23	Existing Indebtedness; Investments, Guarantees and Certain Contracts	33
5.24	Affiliated Agreements	33
5.25	Names; Locations of Offices, Records and Collateral; Deposit Accounts	33
5.26	Non-Subordination	33
5.27	Broker's or Finder's Commissions	33
5.28	Anti-Terrorism; OFAC	33
5.29	Security Interest	34
5.30	Survival	34
SECTION 6.	AFFIRMATIVE COVENANTS.	34
6.1	Information	34
	6.1.1 Annual Report	34
	6.1.2 Interim Reports	35
	6.1.3 Quarterly Review Meeting	35
	6.1.4 [Reserved.]	35
	6.1.5 Compliance Certificate	35
	6.1.6 Reports to Governmental Authorities and Shareholders	35
	6.1.7 Notice of Default; Litigation	36
	6.1.8 Projections	37
	6.1.9 Updated Schedules to Guarantee and Collateral Agreement	37
	6.1.10 Other Information	37
6.2	Books; Records, Inspections	38
0.2	6.2.1 Maintain Books and Records	38
	6.2.2 Access by the Agent etc	38
6.3	Conduct of Business; Maintenance of Property; Insurance	38
6.4	Compliance with Laws; Payment of Taxes and Liabilities	39
6.5	Maintenance of Existence	40
6.6	Employee Benefit Plans	40
0.0	6.6.1 Pension Plans	40
6.7	Environmental Matters	40
0.7		-10 10

- ii -

6.8	Further Assurances	40
6.9	Compliance with Health Care Laws	41
6.10	Cure of Violations	41
6.11	Corporate Compliance Program	42
6.12	Payment of Debt	42
6.13	Additional Subsidiaries	42
6.14	Post-Closing Obligations	43
SECTION 7.	NEGATIVE COVENANTS.	43
7.1	Debt	43
7.2	Liens	44
7.3	Dividends; Redemption of Equity Interests	45
7.4	Mergers; Consolidations; Asset Sales	46
7.5	Modification of Organizational Documents	46
7.6	Use of Proceeds	47
7.7	Transactions with Affiliates	47
7.8	Inconsistent Agreements	47
7.9	Business Activities	47
7.10	Investments	47
7.11	Restriction of Amendments to Certain Documents	49
7.12	Fiscal Year	49
7.12	Financial Covenants	49
/110	7.13.1 Minimum Consolidated Unencumbered Liquid Assets	49
	7.13.2 Minimum Total Revenue	50
	7.13.3 Treatment of Amounts Raised In Connection With Cure Rights	51
7.14	Deposit Accounts	51
7.14	Subsidiaries	51
7.16	Regulatory Matters	51
7.17	Name; Permits; Dissolution; Insurance Policies; Disposition of Collateral; Taxes; Trade Names; Location of Assets; Change	51
/.1/	of Chief Executive Office	51
7.18	Truth of Statements	52
/.10		02
SECTION 8.	EVENTS OF DEFAULT; REMEDIES.	52
8.1	Events of Default	52
0.1	8.1.1 Non-Payment of Credit	52
	8.1.2 Default Under Other Debt	52
	8.1.3 Bankruptcy; Insolvency	52
	8.1.4 Non-Compliance with Loan Documents	53
	8.1.5 Representations; Warranties	53
	8.1.6 Pension Plans	53
	8.1.7 Judgments	53
	8.1.8 Invalidity of Loan Documents or Liens	54
	8.1.9 Invalidity of Subordination Provisions	54
	8.1.10 Change of Control	54
	8.1.11 Certificate Withdrawals, Adverse Test or Audit Results, and Other Matters	54
	8.1.12 Material Adverse Effect	54
8.2	Remedies	54 54
0.2	Kenedits	54
SECTION 9.	AGENT.	55
9.1	Appointment; Authorization	55
9.2	Delegation of Duties	56
×		20

- iii -

9.3	Limited Liability	56
9.4	Reliance	56
9.5	Notice of Default	56
9.6	Credit Decision	57
9.7	Indemnification	57
9.8	Agent Individually	57
9.9	Successor Agent	58
9.10	Collateral and Guarantee Matters	58
9.11	Intercreditor and Subordination Agreements	59
9.12	Actions in Concert	59
SECTION 10.	MISCELLANEOUS.	59
10.1	Waiver; Amendments	59
10.2	Notices	60
10.3	Computations	60
10.4	Costs; Expenses	60
10.5	Indemnification by Borrower	61
10.6	Marshaling; Payments Set Aside	61
10.7	Non-liability of Lenders	61
10.8	Assignments	62
	10.8.1 Assignments	62
10.9	Participations	63
10.10	Confidentiality	64
10.11	Captions	64
10.12	Nature of Remedies	65
10.13	Counterparts; Electronic Signatures	65
10.14	Severability	65
10.15	Entire Agreement	65
10.16	Successors; Assigns	65
10.17	Governing Law	65
10.18	Forum Selection; Consent to Jurisdiction	66
10.19	Waiver of Jury Trial	66
10.20	Patriot Act	66
10.21	Independent Nature of Relationship	67

- iv -

Annexes Annex I	Commitments and Pro Rata Term Loan Shares
Exhibits Exhibit A Exhibit B Exhibit C	Form of Assignment Agreement Form of Compliance Certificate Form of Note
<u>Schedules</u> Schedule 4.1	Prior Debt
Schedule 5.1	Jurisdictions of Qualification
Schedule 5.6	Litigation
Schedule 5.7	Ownership of Properties; Liens
Schedule 5.8	Capitalization
Schedule 5.16	Insurance
Schedule 5.18(a)	Borrower's Registered Intellectual Property
Schedule 5.18(b)	Products and Required Permits
Schedule 5.21	Material Contracts
Schedule 5.25A	Names
Schedule 5.25B	Places of Business
Schedule 5.27	Broker's Commissions
Schedule 7.1	Permitted Indebtedness
Schedule 7.2	Permitted Liens
Schedule 7.7	Transactions with Affiliates
Schedule 7.10	Permitted Investments
Schedule 7.14	Deposit Accounts

- v -

CREDIT AGREEMENT

This Credit Agreement (as may be amended, restated, supplemented, or otherwise modified from time to time, this 'Agreement') dated as of December 27, 2023 (the "Closing Date"), among JOURNEY MEDICAL CORPORATION, a Delaware corporation ("Borrower"), the financial institutions party hereto from time to time as lenders (each a "Lender" and collectively, the "Lenders") and SWK Funding LLC, a Delaware limited liability company (in its individual capacity, "SWK"), as Agent for all Lenders.

In consideration of the mutual agreements herein contained, the parties hereto agree as follows:

Section 1. Definitions; Interpretation.

1.1 Definitions.

When used herein the following terms shall have the following meanings:

Account Control Agreement means, individually and collectively, any account control agreement, account bank agreement or similar agreement(s) entered into from time to time at Agent's request, among a Loan Party, Agent and any third party bank or financial institution at which such Loan Party maintains a Deposit Account.

Acquisition means any transaction or series of related transactions for the purpose of or resulting, directly or indirectly, in (a) the acquisition of all or substantially all of any business or division of a Person, (b) the acquisition of in excess of fifty percent (50%) of the capital stock, share capital partnership interests, membership interests or equity of any Person, or otherwise causing any Person to become a Subsidiary, (c) the acquisition of a product license or a product line, or (d) a merger, amalgamation or consolidation or any other combination (other than a merger, amalgamation, consolidation or combination that effects a Disposition) with another Person (other than a Person that is already a Subsidiary).

<u>Affiliate</u> of any Person means (a) any other Person which, directly or indirectly, controls or is controlled by or is under common control with such Person, (b) any managing member, manager, officer or director of such Person and (c) with respect to any Lender, any entity administered or managed by such Lender or an Affiliate or investment advisor thereof which is engaged in making, purchasing, holding or otherwise investing in commercial loans. Unless expressly stated otherwise herein, neither Agent nor any Lender shall be deemed an Affiliate of Borrower, any Loan Party or any Affiliate thereof.

Agent means SWK in its capacity as administrative and collateral agent for all Lenders hereunder and any successor thereto in such capacity.

Agreement shall have the meaning set forth in the Preamble.

<u>Approved Fund</u> means (a) any fund, trust or similar entity that invests in commercial loans in the ordinary course of business and is advised or managed by (i) a Lender, (ii) an Affiliate of a Lender, (iii) the same investment advisor that manages a Lender or (iv) an Affiliate of an investment advisor that manages a Lender or (b) any finance company, insurance company or other financial institution which temporarily warehouses loans for any Lender or any Person described in <u>clause (a)</u> above.

Assignment Agreement means an agreement substantially in the form of Exhibit A.

Authorization shall have the meaning set forth in Section 5.22(b).

Board means Borrower's board of directors or such similar governing body.



Borrower shall have the meaning set forth in the Preamble.

Business Day means any day on which commercial banks are open for commercial banking business in Dallas, Texas; provided that, with respect to any determination of the Term SOFR Reference Rate, Business Day shall exclude any day on which the Securities Industry and Financial Markets Association recommends that the fixed income departments of its members be closed for the entire day for purposes of trading in United States government securities.

<u>Capital Lease</u> means, with respect to any Person, any lease of (or other agreement conveying the right to use) any real or personal property by such Person that, in conformity with GAAP in respect of any Person incorporated in the United States of America, is accounted for as a capital lease and as a liability on the balance sheet of such Person.

Cash Equivalent Investment means, at any time, (a) any evidence of Debt, maturing not more than one year after such time, issued or guaranteed by the United States Government or any agency thereof,

(a) commercial paper, or corporate demand notes, in each case (unless issued by a Lender or its holding company) rated at least "A-I" by Standard & Poor's Ratings Group or "P-I" by Moody's Investors Service, Inc., (c) any certificate of deposit (or time deposit represented by a certificate of deposit) or banker's acceptance maturing not more than one year after such time, or any overnight Federal funds transaction that is issued or sold by any Lender (or by a commercial banking institution that is a member of the Federal Reserve System or is a U.S. branch of a foreign banking institution and has a combined capital and surplus and undivided profits of not less than \$500,000,000), (d) any repurchase agreement entered into with any Lender (or commercial banking institution of the nature referred to in clause (c) above) which (i) is secured by a fully perfected security interest in any obligation of the type described in any of clauses (a) through

(b) above and (ii) has a market value at the time such repurchase agreement is entered into of not less than one-hundred percent (100%) of the repurchase obligation of such Lender (or other commercial banking institution) thereunder, (e) money market accounts or mutual funds which invest exclusively or substantially in assets satisfying the foregoing requirements, (f) cash, and (g) other short term liquid investments approved in writing by Agent.

Change of Control means the occurrence of any of the following, unless such action has been consented to in advance in writing by Agent in its sole discretion:

(i) any Person (other than Fortress Biotech, Inc. and its Affiliates) acquires the direct or indirect ownership of more than fifty percent (50%) of the issued and outstanding total combined voting Equity Interests of Borrower;

(ii) Borrower shall at any time fail to own, directly or indirectly, one hundred percent (100%) of the Equity Interests of each of its Subsidiaries except as otherwise explicitly permitted by this Agreement; or (iii) the sale of all or substantially all of the assets of Borrower, or any merger, amalgamation, consolidation or acquisition by Borrower which does not result in such Person being the sole surviving entity.

<u>CLIA</u> means (a) the Clinical Laboratory Improvement Act of 1967, as the same may be amended, modified or supplemented from time to time, including without limitation the Clinical Laboratory Improvement Amendments, 42 U.S.C. § 263a et seq. ("<u>CLIA 88</u>"), and any successor statute thereto, and any and all rules or regulations promulgated from time to time thereunder, or (b) any equivalent state statute (and any and all rules or regulations promulgated from time to time thereunder) recognized by the relevant Governmental Authority as (x) having an "Equivalency" (as defined by CLIA) to CLIA, and (y) offering a compliance and regulatory framework that is applicable to a Person in such state in lieu of CLIA.

- 2 -

Closing Date shall have the meaning set forth in the Preamble.

CMS means the Centers for Medicare and Medicaid Services of the United States of America.

Collateral has the meaning set forth in the Guarantee and Collateral Agreement.

<u>Collateral Access Agreement</u> means an agreement in form and substance reasonably satisfactory to Agent pursuant to which a mortgagee or lessor of real property on which Collateral (or any books and records) is stored or otherwise located, or a warehouseman, processor or other bailee of Inventory or other property owned by any Loan Party, acknowledges the Liens of Agent and waives (or, if approved by Agent, subordinates) any Liens held by such Person on such property, and, in the case of any such agreement with a mortgagee or lessor, permits Agent reasonable access to any Collateral stored or otherwise located thereon.

<u>Collateral Documents</u> means, collectively, the Guarantee and Collateral Agreement, IP Security Agreement, each Collateral Access Agreement, any mortgage delivered in connection with the Loan from time to time, each Account Control Agreement, if any, and each other agreement or instrument pursuant to or in connection with which any Loan Party or any other Person grants a Lien in any Collateral to Agent for the benefit of Agent and Lenders, each as amended, restated or otherwise modified from time to time.

Commitment means, as to any Lender, such Lender's Pro Rata Term Loan Share. Compliance Certificate means a certificate substantially in the form of Exhibit B.

Consolidated Unencumbered Liquid Assets means as of any date of determination, the aggregate amount of unrestricted Cash Equivalent Investments owned by Loan Parties and their Subsidiaries, on a consolidated basis.

<u>Contingent Obligation</u> means any agreement, undertaking or arrangement by which any Person guarantees, endorses or otherwise becomes or is contingently liable upon (by direct or indirect agreement, contingent or otherwise, to provide funds for payment, to supply funds to or otherwise to invest in a debtor, or otherwise to assure a creditor against loss) any indebtedness, obligation or other liability of any other Person (other than by endorsements of instruments in the course of collection), or guarantees the payment of dividends or other distributions upon the shares of any other Person. The amount of any Person's obligation in respect of any Contingent Obligation shall be deemed to be the amount for which the Person obligated thereon is reasonably expected to be liable or responsible.

Contract Rate means a rate per annum equal to (x) the Term SOFR Rate, plus (y) seven and three-quarters of one percent (7.75%).

<u>Controlled Group</u> means all members of a controlled group of corporations and all members of a controlled group of trades or businesses (whether or not incorporated) under common control which, together with a Loan Party, are treated as a single employer under Section 414 of the IRC or Section 4001 of ERISA.

Controlled Substances Act means the Drug Abuse Prevention and Control Act; Title 21 of the United States Code, 13 U.S.C, as amended from time to time.

Copyrights has the meaning set forth in the Guarantee and Collateral Agreement.

DEA means the Federal Drug Enforcement Administration of the United States of America.

Debt of any Person means, without duplication, (a) all indebtedness of such Person for borrowed money, (b) all indebtedness evidenced by bonds, debentures, notes or similar instruments, (c) all obligations of such Person as lessee under Capital Leases which have been or should be recorded as liabilities on a balance sheet of such Person in accordance with GAAP, (d) all obligations of such Person to pay the deferred

purchase price of property or services (excluding (i) trade accounts payable in the ordinary course of business, (ii) royalty payments or cash milestone payments made or to be made by such Person from time to time in connection with an Acquisition or a licensing or sublicensing transaction, (iii) any earnout obligation unless either such obligation is not paid after becoming due and payable or such obligation is required to be reflected on the Issuer's balance sheet in accordance with GAAP, and (iv) accruals for payroll and deferred compensation arrangements), (e) all indebtedness secured by a Lien on the property of such Person, whether or not such indebtedness shall have been assumed by such Person (with the amount thereof being measured as the lesser of (x) the aggregate unpaid amount of such indebtedness and (y) the fair market value of such property), (f) all reimbursement obligations, contingent or otherwise, with respect to letters of credit (whether or not drawn), banker's acceptances and surety bonds issued for the account of such Person, (h) all Contingent Obligations of such Person in respect of Debt of others, (i) all indebtedness of any partnership of which such Person is a general partner except to the extent such obligations are considered borrowed money indebtedness for tax purposes but the transaction is classified as an operating lease in accordance with GAAP.

Debtor Relief Law means, collectively: (a) Title 11 of the United States Code, 11 U.S.C. § 101 et. seq., as amended from time to time, and (b) all other United States or foreign applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, receivership, insolvency, administration, reorganization or similar debtor relief laws from time to time in effect affecting the rights of creditors generally, in each case as amended from time to time.

Default means any event that, if it continues uncured, will, with the lapse of time or the giving of notice or both, constitute an Event of Default.

Default Rate means a rate per annum equal to the lesser of (i) three percent (3%) over the Contract Rate, or (ii) the maximum rate of interest permitted to be charged by applicable laws, directives or regulations governing this Agreement until paid.

Deposit Account means, individually and collectively, any bank or other depository accounts of a Loan Party.

Disposition has the meaning set forth in Section 7.4(b).

Division means, with respect to any Person which is an entity, the division of such Person into two (2) or more separate such Persons, with the dividing Person either continuing or terminating its existence as part of such division, including as contemplated under Section 18-217 of the Delaware Limited Liability Act for limited liability companies formed under Delaware law, or any analogous action taken pursuant to any other applicable law with respect to any corporation, limited liability company, partnership or other entity. The word "Divide," when capitalized, shall have a correlative meaning.

Dollar and \$ mean lawful money of the United States of America.

Drug Application means a new drug application, an abbreviated drug application, or a product license application for any Product, as appropriate, as those terms are defined in the FDA Law and Regulation.

Elapsed Period has the meaning set forth in Section 2.9.1(a).

Environmental Claims means all claims, however asserted, by any Governmental Authority or other Person alleging potential liability or responsibility for violation of any Environmental Law, or for release or injury to the environment or any Person or property.

- 4 -

Environmental Laws means all present or future foreign, federal, state or local laws, statutes, common law duties, rules, regulations, directives, ordinances and codes, together with all administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case relating to any matter arising out of or relating to the effect of the environment on health and safety, or pollution or protection of the environment or workplace, including any of the foregoing relating to the presence, use, production, generation, handling, transport, treatment, storage, disposal, distribution, discharge, release, control or cleanup of any Hazardous Substance.

Equity Cure has the meaning set forth in Section 8.4.1.

Equity Interests means, with respect to any Person, its equity ownership interests, its common stock and any other capital stock or other equivalents ownership units of such Person authorized from time to time, its share capital, and any other shares, options, interests, participations or other equivalents (however designated) of or in such Person, whether voting or nonvoting, including, without limitation, common stock, options, warrants, preferred stock, phantom stock, membership units (common or preferred), stock appreciation rights, membership unit appreciation rights, convertible notes or debentures, SAFE's or similar instruments, stock purchase rights, membership unit purchase rights and all securities convertible, exercisable or exchangeable, in whole or in part, into any one or more of the foregoing, but excluding any debt securities convertible into any of the foregoing to the extent not converted.

ERISA means the Employee Retirement Income Security Act of 1974, as amended from time to time.

Event of Default means any of the events described in Section 8.1.

Excluded Taxes has the meaning set forth in Section 3.1(a).

Exempt Accounts means any Deposit Accounts, securities accounts or other similar accounts (i) into which there are deposited no funds other than those intended solely to cover compensation or salary to employees of the Loan Parties (and related contributions to be made on behalf of such employees to health and benefit plans) plus balances for outstanding checks for compensation or salary and such contributions from prior periods; (ii) constituting employee withholding accounts and contain only funds deducted from pay otherwise due to employees for services rendered to be applied toward the Tax obligations of such Person or its employees, or (iii) into which there are deposited no funds other than those received in trust or in escrow, or as cash collateral to secure performance or for Permitted Liens.

Exit Fee has the meaning set forth in Section 2.7(b).

<u>FATCA</u> means Sections 1471 through 1474 of the IRC, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof, any agreements entered into pursuant to Section 1471(b)(1) of the IRC, any fiscal, Tax or regulatory legislation, rules or official practices adopted pursuant to any intergovernmental agreement entered into in connection with the implementation of Sections 1471 through 1474 of the IRC and any current or future regulations promulgated thereunder and any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement, treaty or convention among Governmental Authorities implementing such Section of the IRC.

FD&C Act means the Federal Food, Drug, and Cosmetic Act, 21 U.S.C. §§ 301 et seq., as amended, and all applicable regulations or guidance promulgated by the FDA.

FDA means the Food and Drug Administration of the United States of America.

FDA Law and Regulation means the provisions of the FD&C Act and all applicable regulations or guidance promulgated by the FDA.

FDA Products means any finished products sold by Borrower or any of the other Loan Parties for itself or for a third party that are subject to applicable Health Care Laws.

<u>Federal Funds Effective Rate</u> means, for any day, the greater of (a) the rate calculated by the Federal Reserve Bank of New York based on such day's Federal funds transactions by depositary institutions (as determined in such manner as the Federal Reserve Bank of New York shall set forth on its public website from time to time) and published on the next succeeding day on which commercial banks are open for commercial banking business in New York, New York, by the Federal Reserve Bank of New York as the Federal funds effective rate and (b) 1.00%.

Fiscal Quarter means a calendar quarter of a Fiscal Year.

Fiscal Year means the fiscal year of Borrower, which period shall be the twelve (12) month period ending on December 31 of each year.

Foreign Lender means any Lender that is not a "United States person" within the meaning of Section 7701(a)(30) of the IRC.

Fortress Note Documents means that certain Amended and Restated Future Advance Promissory Note between Fortress Biotech Inc. and the Borrower, and the documents, instruments, and agreements executed in conjunction therewith, as in place on the Closing Date.

FRB means the Board of Governors of the Federal Reserve System or any successor thereto.

GAAP means generally accepted accounting principles in effect in the United States of America set forth from time to time in the opinions and pronouncements of the Accounting Principles Board and the American Institute of Certified Public Accountants and statements and pronouncements of the Financial Accounting Standards Board (or agencies with similar functions of comparable stature and authority within the U.S. accounting profession), which are applicable to the circumstances as of the date of determination.

<u>Governmental Authority</u> means any nation or government, any state or other political subdivision thereof, and any agency, branch of government, department or Person exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government and any corporation or other Person owned or controlled (through stock or capital ownership or otherwise) by any of the foregoing, whether domestic or foreign. Governmental Authority shall include any agency, branch or other governmental body charged with the responsibility and/or vested with the authority to administer and/or enforce any Health Care Laws.

Guarantee and Collateral Agreement means the Guarantee and Collateral Agreement dated as of the Closing Date executed by each Loan Party signatory thereto in favor of Agent for the benefit of Lenders.

Hazardous Substances means hazardous waste, pollutant, contaminant, toxic substance, oil, hazardous material, chemical or other substance regulated by any Environmental Law.

Health Care Laws mean all foreign, federal and state fraud and abuse laws relating to the regulation of healthcare products, pharmaceutical products, laboratory facilities and services, healthcare providers, healthcare professionals, healthcare facilities, clinical research facilities or healthcare payors, including but not limited to (i) the federal Anti-Kickback Statute (42 U.S.C. (§1320a-7b(b))), the Stark Law (42 U.S.C. §1395nn and §1395(q)), the civil False Claims Act (31 U.S.C. §3729 et seq.), TRICARE (10 U.S.C. Section 1071 et seq.), Section 1320a-7 and 1320a-7a of Title 42 of the United States Code and the regulations promulgated pursuant to such statues; (ii) the Health Insurance Portability and Accountability Act of 1996 (Pub. L. No. 104-191), as amended by the Health Information, Technology for Economic and Clinical Health Act of 2009, and the regulations promulgated thereunder, (iii) Medicare (Title XVIII of the Social Security

- 6 -

Act) and the regulations promulgated thereunder; (iv) Medicaid (Title XIX of the Social Security Act) and the regulations promulgated thereunder; (v) the FD&C Act and all applicable requirements, regulations and guidances issued thereunder by the FDA (including FDA Law and Regulation); (vi) the Controlled Substances Act, as amended, and all applicable requirements, regulations and guidances issued thereunder by the DEA; (vii) [reserved]; (viii) quality, safety and accreditation standards and requirements of all applicable foreign and domestic federal, provincial or state laws, directives, regulations or regulatory bodies; (ix) all applicable licensure laws, directives and regulations; (x) all applicable professional standards regulating healthcare providers, healthcare professionals, healthcare facilities, clinical research facilities or healthcare payors; and (xi) any and all other applicable health care laws (whether foreign or domestic), regulations, directives, manual provisions, policies and administrative guidance, including those related to the corporate practice of medicine, fee-splitting, state anti-kickback or self-referral prohibitions, each of <u>clauses (i)</u> through (xi) as may be amended from time to time.

<u>Hedging Obligation</u> means, with respect to any Person, any liability of such Person under any interest rate, currency or commodity swap agreement, cap agreement or collar agreement, and any other agreement or arrangement designed to protect a Person against fluctuations in interest rates, currency exchange rates or commodity prices. The amount of any Person's obligation in respect of any Hedging Obligation shall be deemed to be the incremental obligation that would be reflected in the financial statements of such Person in accordance with GAAP.

Indemnified Taxes has the meaning set forth in Section 3.1(a).

Intellectual Property has the meaning set forth in the Guarantee and Collateral Agreement.

Inventory has the meaning set forth in the Guarantee and Collateral Agreement.

Investment means, with respect to any Person, (a) the purchase of any debt or equity security of any other Person, (b) the making of any loan or advance to any other Person, (c) becoming obligated with respect to a Contingent Obligation in respect of obligations of any other Person (other than travel and similar advances to employees in the ordinary course of business) or (d) the making of an Acquisition.

<u>IP Security Agreement</u> means the Intellectual Property Security Agreement dated on or about the Closing Date by each Loan Party signatory thereto in favor of Agent for the benefit of Lenders.

IRC means the Internal Revenue Code of 1986, as amended.

IRS means the United States Internal Revenue Service.

Legal Costs means, with respect to any Person, all reasonable, duly documented, out-of-pocket fees and charges of any counsel, accountants, auditors, appraisers, consultants and other professionals to such Person, and all court costs and similar legal expenses.

Lenders has the meaning set forth in the Preamble.

Lien means, with respect to any Person, any interest granted by such Person in any real or personal property, asset or other right owned or being purchased or acquired by such Person which secures payment or performance of any obligation and shall include any mortgage (whether legal or equitable), lien, encumbrance, charge, pledge, assignment by way of security or other security interest of any kind, whether arising by contract, as a matter of law, by judicial process or otherwise.

Loan or Loans means, individually and collectively the Term Loan and any other advances made by Agent and Lenders in accordance with the Loan Documents.

- 7 -

Loan Documents means this Agreement, any Notes, any intercreditor agreements, any Subordination Agreement, the Collateral Documents and all documents, instruments and agreements delivered in connection with the foregoing.

Loan Party means Borrower and each of its Subsidiaries.

Margin Stock means any "margin stock" as defined in Regulation T, U or X of the FRB.

<u>Material Adverse Effect</u> means (a) a material adverse change in, or a material and adverse effect upon, the financial condition, operations, assets, or business of Loan Parties and their Subsidiaries taken as a whole, (b) a material impairment of the ability of any Loan Party to perform any of its payment Obligations under any Loan Document or (c) a material and adverse effect upon any material portion of the Collateral under the Collateral Documents or upon the legality, validity, binding effect or enforceability against any Loan Party of any material Loan Document.

<u>Material Contract</u> means each "material definitive agreement" identified as such in Borrower's public filings with the U.S. Securities and Exchange Commission to the extent the loss or termination of any such contract would reasonably be expected to result in a Material Adverse Effect.

Multiemployer Pension Plan means a multiemployer plan, as defined in Section 4001(a)(3) of ERISA, to which Borrower or any member of the Controlled Group may have any liability.

Net Cash Proceeds means, with respect to any Disposition, the aggregate cash proceeds (including cash proceeds received pursuant to policies of insurance and by way of deferred payment of principal pursuant to a note, installment receivable or otherwise, but only as and when received) received by any Loan Party pursuant to such Disposition net of (i) the reasonable direct costs relating to such Disposition (including sales commissions and legal, accounting and investment banking fees, commissions and expenses), (ii) any portion of such proceeds deposited in an escrow account pursuant to the documentation relating to such Disposition (provided that such amounts shall be treated as Net Cash Proceeds upon their release from such escrow account to and receipt by the applicable Loan Party), (iii) Taxes and other governmental costs and expenses paid or reasonably estimated by a Loan Party to be payable as a result thereof (after taking into account any available tax credits or deductions and any tax sharing arrangements), (iv) amounts required to be applied to the repayment of any Debt (together with any interest thereon, premium or penalty and any other amount payable with respect thereto) secured by a Lien that has priority over the Lien, if any, of Agent on the asset subject to such Disposition, (v) reserves for purchase price adjustments and retained liabilities reasonably expected to be payable by the Loan Parties in connection therewith established in accordance with GAAP (provided that upon the final determination of the amount paid in respect of such purchase price adjustments and retained liabilities, the actual amount of purchase price adjustments and retained liabilities, paid is less than such reserves, the difference shall, at such time, constitute Net Cash Proceeds) and (vi) with respect to any Disposition, all money actually applied within one hundred eighty (180) days to purchase assets used or useful in the business of the Loan Parties and their Subsidiaries.

Note means a promissory note substantially in the form of Exhibit C.

<u>Obligations</u> means all liabilities, indebtedness and obligations (monetary (including post-petition interest, allowed or not) or otherwise) of any Loan Party under this Agreement, any other Loan Document or any other document or instrument executed in connection herewith or therewith which are owed to any Lender or Affiliate of a Lender, in each case howsoever created, arising or evidenced, whether direct or indirect, absolute or contingent, now or hereafter existing, or due or to become due. For the avoidance of doubt, "Obligations" shall include Borrower's obligation to pay any amounts due under <u>Sections 2.7</u> and <u>2.8.2</u> and payable on such date of determination.

OFAC means the U.S. Department of Treasury's Office of Foreign Asset Control.



Operating Burn means, for any period being measured, the product of (x) -1 and (y) the sum of (i) aggregate net cash used in operating activities from operations of Loan Parties, plus (ii) Unfinanced Capital Expenditures, less (iii) DFD-29 FDA submission related expenses including the approximate \$4,048,695 user fee under the Prescription Drug User Fee Act to FDA for a new drug application ("NDA") submission; \$3,000,000 NDA acceptance milestone payable to Dr. Reddy's Laboratories, Ltd. ("DRL"); and \$15,000,000 NDA approval milestone to DRL, pursuant to, and in accordance with, that certain Assignment, License, and Collaboration Agreement, between DRL and Borrower, dated June 29, 2021, less (iv) any other one-time or extraordinary expenses made within such measurement period, as mutually agreed upon by Agent and Borrower, in each case as determined from the cash flow statement provided by Borrower and in accordance with GAAP.

Origination Fee shall have the meaning set forth in Section 2.7(a).

Paid in Full, Pay in Full or Payment in Full means, with respect to any Obligations, the payment in full in cash of all such Obligations (other than contingent indemnification obligations, yield protection and expense reimbursement to the extent no claim giving rise thereto has been asserted in respect of contingent indemnification obligations, and to the extent no amounts therefor have been asserted, in the case of yield protection and expense reimbursement obligations, which Obligations shall survive the Payment in Full of the Obligations).

<u>Patents</u> has the meaning set forth in the Guarantee and Collateral Agreement.

Payment Date means the fifteenth (15th) day of each of February, May, August and November (or the next succeeding Business Day to the extent such 15th day is not a Business Day), commencing with February 15, 2024.

PBGC means the Pension Benefit Guaranty Corporation and any entity succeeding to any or all of its material functions under ERISA.

<u>Pension Plan</u> means a "pension plan", as such term is defined in Section 3(2) of ERISA, which is subject to Title IV of ERISA (other than a Multiemployer Pension Plan), and to which Borrower or any member of the Controlled Group may have any liability, including any liability by reason of having been a substantial employer within the meaning of Section 4063 of ERISA at any time during the preceding five years, or by reason of being deemed to be a contributing sponsor under Section 4069 of ERISA.

Permit means, with respect to any Person, any permit, approval, clearance, authorization, license, registration, certificate, concession, grant, franchise, variance or permission from, and any other contractual obligations with, any Governmental Authority, including without limitation all registrations with Governmental Authorities.

Permitted Acquisition means any Acquisition so long as:

(c) both immediately before and immediately after the consummation of such Acquisition, no Default, Event of Default or Material Adverse Effect shall have occurred and be continuing or result therefrom;

(d) the Acquisition shall be with respect to an operating company or division or line of business that engages in, and that substantially all of the sales and operating profits generated by such company or division or line of business are in, a line of business substantially similar, reasonably related, ancillary or incidental to the principal business in which the Borrower is engaged;

- 9 -

(e) the board of directors (or other comparable governing body) of the Person to be acquired or owning such assets or Equity Interests (and, if required, the holders of any Equity Interests in such Person) shall have duly approved such Acquisition;

(f) Agent shall have received written notice not less than ten (10) Business Days' prior to the closing of the proposed Acquisition and such information with respect thereto as Agent may reasonably request and which is then readily available, including (i) the proposed date and amount of the Acquisition, (ii) a list and description of the assets or Equity Interests to be acquired and (iii) the total purchase price for the assets or Equity Interests to be purchased (and the terms of payment of such purchase price); and

(g) the total cash consideration (excluding the proceeds of concurrent equity issuances) shall not exceed \$5,000,000 per

year for all such Acquisitions.

Permitted Liens means Liens permitted by Section 7.2.

Person means any natural person, corporation, partnership, trust, limited liability company, association, Governmental Authority or unit, or any other entity, whether acting in an individual, fiduciary or other capacity.

Prior Debt means the Debt listed on Schedule 4.1.

<u>Pro Rata Term Loan Share</u> means, with respect to any Lender, the applicable percentage (as adjusted from time to time in accordance with the terms hereof) specified opposite such Lender's name on <u>Annex I</u> which percentage represents the aggregate percentage of the Term Loan Commitment held by such Lender, which percentage shall be with respect to the outstanding balance of the Term Loan as of any date of determination after the Term Loan Commitment has terminated.

<u>Product</u> means any products manufactured, sold, developed, tested or marketed by Borrower or any of its Subsidiaries, including, without limitation, those products set forth on <u>Schedule 5.18(b)</u> (as updated from time to time in accordance with <u>Section 6.1.2</u>); *provided, however*, that if Borrower shall fail to comply with the obligations under <u>Section 6.1.2</u> to give notice to Agent and update <u>Schedule 5.18(b)</u> prior to manufacturing, selling, developing, testing or marketing any new Product, any such improperly undisclosed Product shall be deemed to be included in this definition; and *provided, further*, that products manufactured by Borrower for unaffiliated third parties shall not be deemed "Products" hereunder.

<u>Registered Intellectual Property</u> means all applications, registrations and recordings for or of Patents, Trademarks or Copyrights filed by a Loan Party with any Governmental Authority, all internet domain name registrations owned by a Loan Party, and all proprietary software owned by a Loan Party.

Required Lenders means Lenders having an aggregate Pro Rata Term Loan Share in excess of fifty percent (50%), collectively.

Required Permit means a Permit (a) required under applicable law for the business of Borrower or any of its Subsidiaries or necessary in the manufacturing, importing, exporting, possession, ownership, warehousing, marketing, promoting, sale, labeling, furnishing, distribution or delivery of goods or services under any laws applicable to the business of Borrower or any of its Subsidiaries (including, without limitation, any applicable Health Care Laws) or any Drug Application (including without limitation, at any point in time, all licenses, approvals and permits issued by the FDA, CMS, or any other applicable Governmental Authority necessary for the testing, manufacture, marketing or sale of any Product by Borrower or any of its Subsidiaries as such activities are being conducted by such Person with respect to such Product at such time), and (b) required by any Person from which Borrower or any of its Subsidiaries have received an accreditation.

- 10 -

<u>Responsible Officer</u> means (a) the chief executive officer, chief operating officer, or chief financial officer a Person, and (b) in respect of any other Person, the president, vice president or secretary of such Person, or any other officer having substantially the same authority and responsibility; or, with respect to compliance with financial covenants or delivery of financial information, the chief financial officer, the treasurer or the controller of such Person, or any other officer having substantially the same authority and responsibility, and in all cases such person shall be listed on an incumbency certificate delivered to Agent, in form and substance acceptable to Agent in its sole discretion.

Revenue-Based Payment Amount has the meaning set forth in Section 2.9.1(a).

<u>Royalties</u> means the amount of any and all royalties, license fees and any other payments or income of any type recognized as revenue in accordance with GAAP by the Loan Parties with respect to the sale of Products or the provision of services by independent licensees or sublicensees of Borrower and/or its Subsidiaries, including any such payments characterized as a share of net profits, any up-front or lump sum payments, any milestone payments, commissions, fees or any other similar amounts, less deductions for amounts deducted, repaid or credited by reason of adjustments to the sales upon which royalty amounts are based, regardless of the reason for such adjustment to such sales. For the purposes of calculating Royalties, Lenders and Agent understand and agree that Affiliates of Borrower shall not be regarded as independent licensees.

Services means services provided by Borrower or any Subsidiary of Borrower to un-Affiliated Persons, including without limitation any sales, laboratory analysis, testing, consulting, marketing, commercialization and any other healthcare-related services.

SOFR shall mean a rate equal to the secured overnight financing rate as administered by the SOFR Administrator.

SOFR Administrator shall mean the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

Solvent means, as to any Person at any time, that (a) the fair value of the property of such Person is greater than the amount of such Person's liabilities (including disputed, contingent, prospective, unmatured and unliquidated liabilities); (b) the present fair saleable value of the property of such Person is not less than the amount that will be required to pay the probable liability of such Person on its debts as they become absolute and matured; (c) such Person is able to pay its debts and other liabilities (including subordinated, disputed, contingent, unmatured and unliquidated liabilities) as they mature in the normal course of business; (d) such Person does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's ability to pay as such debts and liabilities mature; (e) such Person is not engaged in business or a transaction, and is not about to engage in business or a transaction, for which such Person's property would constitute unreasonably small capital; (f) such Person has not admitted in writing its inability generally to pay its debts as they become due or suspended or threatened to suspend making payments on any of its debts; and (g) such Person, by reason of actual or anticipated financial difficulties, has not commenced negotiations with one or more of its creditors (excluding the Agent or any Lender in its capacity as such) with a view to rescheduling any of its indebtedness.

Spot Rate means the exchange rate, as determined by Agent, that is applicable to conversion of one currency into another currency, which is (a) the exchange rate reported by Bloomberg (or other commercially available source designated by Agent) as of the end of the preceding Business Day in the financial market for the first currency; or (b) if such report is unavailable for any reason, the spot rate for the purchase of the first currency with the second currency as in effect during the preceding Business Day in Agent's principal foreign exchange trading office for the first currency.

- 11 -

Subordinated Debt means any Debt incurred by Borrower and/or any other Loan Party that is subordinated to the Obligations pursuant to a subordination agreement entered into between Agent, any applicable Loan Party and the subordinated creditor(s) upon terms acceptable to Agent in its sole discretion.

Subordination Agreement means any subordination agreement that may be executed from time to time in connection with any Subordinated Debt.

Subsequent Term Loan means the Term Loan, if any, made to the Borrower pursuant to Section 2.2.2.

<u>Subsidiary</u> means, with respect to any Person, a corporation, partnership, limited liability company or other entity of which such Person owns, directly or indirectly, such number of outstanding shares or other equity interests as to have more than fifty percent (50%) of the ordinary voting power for the election of directors or other managers of such corporation, partnership, limited liability company or other entity. Unless the context otherwise requires, each reference to Subsidiaries herein shall be a reference to direct and indirect Subsidiaries of Borrower.

SWK has the meaning set forth in the Preamble.

Tax or Taxes has the meaning set forth in Section 3.1(a).

Term Loan has the meaning set forth in Section 2.1.

Term Loan Commitment means \$20,000,000.

Term Loan Maturity Date means December 27, 2027.

Term SOFR Administrator means the CME Group Benchmark Administration Limited (CBA) (or a successor administrator of the Term SOFR Reference Rate selected by Agent in its reasonable discretion after consultation with Borrower).

<u>Term SOFR Rate</u> means the Term SOFR Reference Rate for a three (3) month period that is ten (10) Business Days prior to each Payment Date (such day, the "<u>Periodic Term SOFR Determination Day</u>"), and effective on the Payment Date immediately following such determination date and continuing to but not including the next succeeding Payment Date, as such rate is published by the Term SOFR Administrator; provided, however, that if as of 5:00 p.m. (New York City time) on any Periodic Term SOFR Determination Day the Term SOFR Reference Rate has not been published by the Term SOFR Administrator, then Term SOFR will be the Term SOFR Reference Rate for such three (3) month period, as published by the Term SOFR Administrator on the first preceding Business Day for which such Term SOFR Reference Rate was published by the Term SOFR Administrator. Notwithstanding the foregoing, (i) if at any time Agent determines (which determination shall be conclusive absent manifest error) that the Term SOFR Rate is no longer available for determining interest rates for loans or notes similar to the Loans, then Agent shall, in consultation with Borrower, endeavor to establish an alternate rate of interest to the Term SOFR Rate that gives due consideration to the then prevailing market convention for determining a rate of interest for loans or notes similar to the Loans in the United States at such time, and, if requested by Agent, Agent and Lenders at such time party hereto and the Borrower shall enter into an amendment to this Agreement to reflect such alternate rate of interest at estage to doubt, any amendments to the definition of "Contract Rate" to ensure that the interest rate payable by Borrower hereunder is substantially similar to the interest rate that would otherwise be paid prior to the selection of such alternate rate of interest, and (ii) in no event shall the "Term SOFR Rate" or any such alternate rate of interest to the Term soft doubt, any amendments to the Grenterest and such other related changes to this Agreement as

Term SOFR Reference Rate means the forward-looking term rate based on SOFR.



<u>Termination Date</u> means the earlier to occur of (a) the Term Loan Maturity Date, or (b) the date upon which the Loan and all other Obligations are Paid in Full, whether as a result of (i) the prepayment of the Term Loan and all Obligations through any other mandatory or voluntary prepayment of the Term Loan in full, (ii) the contractual acceleration of the Loan hereunder, (iii) the acceleration of the Loan by Agent in accordance with this Agreement, or (iv) otherwise.

Total Revenue shall mean revenue of the Borrower and its Subsidiaries on a consolidated basis as determined in accordance with GAAP excluding the upfront payment in the amount of \$19,000,000 payable pursuant to that certain Exclusive License Agreement, dated as of August 31, 2023, by and between Maruho Co., Ltd. and Borrower.

Trademarks has the meaning set forth in the Guarantee and Collateral Agreement.

<u>Unfinanced Capital Expenditures</u> means capital expenditures (i) not financed with the proceeds of any incurrence of Debt, the proceeds of any sale or issuance of Equity Interests or equity contributions, the proceeds of any asset sale (other than the sale of Inventory in the ordinary course of business) or any insurance proceeds, and (ii) that are not reimbursed by a third person (excluding any Loan Party) in the period such expenditures are made pursuant to a written agreement.

<u>Uniform Commercial Code</u> means the Uniform Commercial Code as in effect in the State of New York; *provided* that if perfection or the effect of perfection or non-perfection or the priority of any security interest in any Collateral is governed by the Uniform Commercial Code as in effect in a jurisdiction other than the State of New York, "<u>Uniform Commercial Code</u>" means the Uniform Commercial Code as in effect from time to time in such other jurisdiction for purposes of the provisions hereof relating to such perfection, effect of perfection or non-perfection or priority.

U.S. Lender means any Lender that is a "United States person" within the meaning of Section 7701(a)(30) of the IRC.

Wholly-Owned Subsidiary means, as to any Person, another Person all of the Equity Interests of which (except directors' qualifying shares) are at the time directly or indirectly owned by such Person and/or another Wholly-Owned Subsidiary of such Person.

1.2 Interpretation.

(a) In the case of this Agreement and each other Loan Document, (i) the meanings of defined terms are equally applicable to the singular and plural forms of the defined terms; (ii) Annex, Exhibit, Schedule and Section references are to such Loan Document unless otherwise specified; (iii) the term "including" is not limiting and means "including but not limited to"; (iv) in the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including"; the words "to" and "until" each mean "to but excluding", and the word "through" means "to and including"; (v) unless otherwise expressly provided in such Loan Document, (A) references to agreements and other contractual instruments shall be deemed to include all subsequent amendments, restatements and other modifications thereto, but only to the extent such amendments, restatements and other modifications thereto, but only to the extent such amendments, restatements and other modifications are not prohibited by the terms of any Loan Document, and (B) references to ary statute, directive or regulation shall be construed as including all statutory and regulatory provisions amending, replacing, supplementing or interpreting such statute, directive or regulation; (vi) this Agreement and the other Loan Documents may use several different limitations, tests or measurements to regulate the same or similar matters, all of which are cumulative and each shall be performed in accordance with its terms and (vii) this Agreement and the other Loan Documents are the reviewed by counsel to Agent, Borrower, Lenders merely because of Borrower's, Agent's or Lenders' involvement in their preparation. Except where otherwise expressly provided in the Loan

- 13 -

Documents, in any instance where the approval, consent or the exercise of Agent's judgment is required, the granting or denial of such approval or consent and the exercise of such judgment shall be (x) within the sole and absolute discretion of Agent and/or Lenders; and (x) deemed to have been given only by a specific writing intended for such purpose executed by Agent.

(b) For purposes of converting any amount denominated in any currency other than Dollars to Dollars under or in connection with the Loan Documents, Agent shall calculate such currency conversion using the current Spot Rate.

(c) If at any time any change in GAAP would affect the computation of any financial ratio or requirement set forth in any Loan Document, and either the Borrower or the Required Lenders shall so request, the Agent, the Lenders and the Borrower shall negotiate in good faith to amend such ratio or requirement to preserve the original intent thereof in light of such change in GAAP (subject to the approval of the Required Lenders); provided that, until so amended, (i) such ratio or requirement shall continue to be computed in accordance with GAAP prior to such change therein and (ii) the Borrower shall provide to the Agent and the Lenders financial statements and other documents required under this Agreement or as reasonably requested hereunder setting forth a reconciliation between calculations of such ratio or requirement made before and after giving effect to such change in GAAP.

(d) Notwithstanding anything to the contrary contained in this Agreement, all obligations that are or would have been treated as operating leases for purposes of GAAP prior to the issuance by the Financial Accounting Standards Board on February 25, 2016 of an Accounting Standards Update (the "ASU") shall continue to be accounted for as operating leases for purposes of all financial definitions and calculations for purposes of the Loan Documents (whether or not such operating lease obligations were in effect on such date) notwithstanding the fact that such obligations are required in accordance with the ASU (on a prospective or retroactive basis or otherwise) to be treated as capitalized lease obligations in the financial statements to be delivered pursuant to the Loan Documents.

Section 2. Credit Facility.

2.1 <u>Term Loan Commitments.</u>

On and subject to the terms and conditions of this Agreement, each Lender, severally and for itself alone, agrees to make a multi-draw term loan to Borrower (each such loan, individually and collectively, a "**Term Loan**") in an amount equal to such Lender's applicable Pro Rata Term Loan Share of the Term Loan Commitment. The Commitments of Lenders to make any portion of the Term Loan shall terminate concurrently with the making of such portion of the Term Loan, such portion terminated to equal (i) on the Closing Date, the amount of the Term Loan set forth in <u>Section 2.2.1</u>, and (ii) on the date of the making of the Subsequent Term Loan, the amount of the Subsequent Term Loan set forth in <u>Section 2.2.2</u>. The Loan is not a revolving credit facility, and therefore any amount thereof that is repaid or prepaid by Borrower, in whole or in part, may not be re-borrowed.

2.2 Loan Procedures.

2.2.1 Initial Advance.

On the Closing Date, each Lender shall advance to Borrower an amount equal to its Pro Rata Share of Fifteen Million and No/100 Dollars (\$15,000,000), upon Borrower's satisfaction of the conditions to closing described in Section 4 of this Agreement.

2.2.2 <u>Subsequent Term Loan.</u>

- 14 -

Borrower may, no later than the date occurring on the twelve (12) month anniversary of the Closing Date, request, in writing, a subsequent advance of the Term Loan and, so long as no Material Adverse Effect, Default or Event of Default has occurred and is continuing or would be caused thereby, each Lender shall make one (1) additional advance (within five (5) Business Days of receipt by Agent of such written request for advance) to Borrower in the amount equal to, but not less than, such lender's Pro Rata Share of Five Million and No/100 Dollars (\$5,000,000).

2.3 <u>Commitments Several.</u>

The failure of any Lender to make the initial Term Loan on the Closing Date or the Subsequent Term Loan in accordance with <u>Section</u> 2.2.2 above shall not relieve any other Lender of its obligation (if any) to make its Loan on the applicable date, but no Lender shall be responsible for the failure of any other Lender to make any Term Loan to be made by such other Lender.

2.4 Indebtedness Absolute; No Offset; Waiver.

The payment obligations of Borrower hereunder are absolute and unconditional, without any right of rescission, set-off, counterclaim or defense for any reason against Agent and Lenders to the maximum extent permitted by applicable law. As of the Closing Date, the Loan has not been compromised, adjusted, extended, satisfied, rescinded, set-off or modified, and the Loan Documents are not subject to any litigation, dispute, refund, claims of rescission, set-off, netting, counterclaim or defense whatsoever, including but not limited to, claims by or against any Loan Party or any other Person. Payment of the Obligations by Borrower, shall be made only by wire transfer, in Dollars, and in immediately available funds when due and payable pursuant to the terms of this Agreement and the other Loan Documents, is not subject to compromise, adjustment, extension, satisfaction, rescission, setoff, counterclaim, defense, abatement, suspension, deferment, deductible, reduction, termination or modification, whether arising out of transactions concerning the Loan, or otherwise. Without limitation to the foregoing, to the fullest extent permitted under applicable law and notwithstanding any other term or provision contained in this Agreement or any other Loan Document, Borrower hereby waives (and shall cause each Loan Party to waive) (a) presentment, protest and demand, notice of default (except as expressly required in the Loan Documents), notice of intent to accelerate, notice of acceleration, notice of protest, notice of demand and of dishonor and non-payment of the Obligations, (b) any requirement of diligence or promptness on Agent's part in the enforcement of its rights under the provisions of this Agreement and any other Loan Document, (c) any rights, legal or equitable, to require any marshalling of assets or to require foreclosure sales in a particular order, (d) all notices of every kind and description which may be required to be given by any statute or rule of law except as specifically required hereunder, (e) the benefit of all laws now existing or that may hereafter be enacted providing for any appraisement before sale or any portion of the Collateral, (f) all rights of homestead, exemption, redemption, valuation, appraisement, stay of execution, notice of election to mature or declare due the whole of the Obligations in the event of foreclosure of the Liens created by the Loan Documents, (g) the pleading of any statute of limitations as a defense to any demand under any Loan Document and (h) any defense to the obligation to make any payments required under the Loan Documents, including the obligation to pay taxes based on any damage to, defects in or destruction of the Collateral or any other event, including obsolescence of any of the Collateral, it being agreed and acknowledged that such payment obligations are unconditional and irrevocable. Borrower further acknowledges and agrees (i) to any substitution, subordination, exchange or release of any security or the release of any party primarily or secondarily liable for the payment of the Loan; (ii) that Agent shall not be required to first institute suit or exhaust its remedies hereon against others liable for repayment of all or any part of the Loan, whether primarily or secondarily (collectively, the "Obligors"), or to perfect or enforce its rights against any Obligor or any security for the Loan; and (iii) that its liability for payment of the Loan shall not be affected or impaired by any determination that any security interest or lien taken by Agent for the benefit of Agent and Lenders to secure the Loan is invalid or unperfected. Borrower acknowledges, warrants and represents in connection with each waiver of any right or remedy of Borrower contained in any Loan Document, that it has been fully informed with respect to, and represented by counsel of its choice in connection with, such rights and remedies, and all such

- 15 -

waivers, and after such advice and consultation, has presently and actually intended, with full knowledge of its rights and remedies otherwise available at law or in equity, to waive or relinquish such rights and remedies to the full extent specified in each such waiver.

2.5 Loan Accounting.

2.5.1 <u>Recordkeeping.</u>

Agent, on behalf of each Lender, shall record in its records the date and amount of the Loan made by each Lender, each prepayment and repayment thereof. The aggregate unpaid principal amount so recorded shall be final, binding and conclusive absent manifest error. The failure to so record any such amount or any error in so recording any such amount shall not, however, limit or otherwise affect the Obligations of Borrower hereunder or under any Note to repay the principal amount of the Loans hereunder, together with all interest accruing thereon.

2.5.2 <u>Notes.</u>

At the request of any Lender, the Loan of such Lender shall be evidenced by a Note, with appropriate insertions, payable to such Lender in a face principal amount equal to such Lender's Pro Rata Term Loan Share and payable in such amounts and on such dates as are set forth herein.

2.6 Payment of Interest.

2.6.1 Interest Rates.

The outstanding principal balance under the Loan shall bear interest at a per annum rate of interest equal to the (a) Contract Rate (as may be adjusted from time to time in accordance with this Section 2.6.1). The Contract Rate applicable to the period beginning on the Closing Date through the date that is one (1) day immediately prior to the initial Payment Date shall be calculated based on the Term SOFR Rate as of the Closing Date. Whenever, on or subsequent to the initial Payment Date, the Term SOFR Rate is increased or decreased (as determined on the date that is ten(10) Business Days prior to each Payment Date), the Contract Rate, as set forth herein, shall be similarly changed effective as of such subsequent Payment Date, without notice or demand of any kind by an amount equal to the amount of such change in the Term SOFR Rate on the date that is ten (10) Business Days prior to each such Payment Date. The interest due on the principal balance of the Loan outstanding as of any Payment Date shall be computed for the actual number of days elapsed during the period in question on the basis of a year consisting of three hundred sixty (360) days and shall be calculated by determining the daily principal balance outstanding for each day of such period in question. The daily rate shall be equal to 1/360th times the Contract Rate. If any statement furnished by Agent for the amount of a payment due exceeded the actual amount that should have been paid because the Term SOFR Rate decreased and such decrease was not reflected in such statement, Borrower shall make the payment specified in such statement from Agent and Borrower shall receive a credit for the overpayment, which credit shall be applied towards the next subsequent payment due hereunder. If any statement furnished by Agent for the amount of a payment due was less than the actual amount that should have been paid because the Term SOFR Rate increased and such increase was not reflected in such statement, Borrower shall make the payment specified in such statement from Agent and Borrower shall be required to pay any resulting underpayment with the next subsequent payment due hereunder.

(b) Borrower recognizes and acknowledges that any default on any payment, or portion thereof, due hereunder or to be made under any of the other Loan Documents, will result in losses and additional expenses to Agent in servicing the Loan, and in losses due to Lenders' loss of the use of funds not timely received. Borrower further acknowledges and agrees that in the event of any such Event of Default, Lenders would be entitled to damages for the detriment proximately caused thereby, but that it would be extremely difficult and impracticable to ascertain the extent of or compute such damages. Therefore, upon

the Term Loan Maturity Date and/or upon the occurrence and during the existence of an Event of Default (or upon any acceleration), interest shall automatically accrue hereunder, without notice to Borrower, at the Default Rate. The Default Rate shall be calculated and due from the date that the Event of Default occurred and shall be payable upon demand.

(c) Notwithstanding anything herein to the contrary, if at any time the interest rate for any Loan (if applicable), together with all fees, charges and other amounts that are treated as interest on such Loan under applicable law (collectively, "charges"), shall exceed the maximum lawful rate (the "**Maximum Rate**") that may be contracted for, charged, taken, received or reserved by the Lender holding such Loan in accordance with applicable law, the rate of interest payable in respect of such Loan hereunder (if applicable), together with all charges payable in respect of the Loan, shall be limited to the Maximum Rate. To the extent lawful, the interest and charges that would have been paid in respect of such Loan but were not paid as a result of the operation of this Section shall be cumulated and the interest (if any) and charges payable to such Lender in respect of other Loans or periods shall be increased (but not above the amount collectible at the Maximum Rate therefor) until such cumulated amount, together with interest thereon at the Federal Funds Effective Rate for each day to the date of repayment, shall have been received by such Lender. Any amount collectible at the Maximum Rate shall be applied to the reduction of the principal balance of such Loan or refunded to the Borrower so that at no time shall the interest (if any) and charges paid or payable in respect of such Loan or refunded to the Maximum Rate.

2.6.2 Payments of Interest and Principal.

Borrower shall pay to Lenders all accrued interest on the Loan in arrears on each Payment Date, upon a prepayment of such Loan in accordance with <u>Section 2.8</u> and at maturity in cash. Any partial prepayment of the Loan shall be applied pursuant to <u>Section 2.9.1</u> (but this shall not be construed as permitting any partial prepayment other than as may be expressly permitted elsewhere in this Agreement).

2.7 <u>Fees.</u>

(a) <u>Origination Fee</u>. Borrower shall pay to Agent, for the benefit of Lenders, a fee (the "**Origination Fee**") in the amount of \$200,000, which Origination Fee shall be deemed fully earned and non-refundable on the Closing Date.

(b) <u>Exit Fee.</u> Upon the Termination Date, Borrower shall pay an exit fee (the "**Exit Fee**") to Agent, for the benefit of Lenders, in an amount equal to five percent (5.00%) <u>multiplied by</u> the aggregate amount of the Term Loan funded hereunder on or prior to such date, which Exit Fee shall be deemed fully earned and non-refundable on the Termination Date.

2.8 <u>Prepayment.</u>

2.8.1 Mandatory Prepayment.

(a) Borrower shall prepay the Obligations, or any portion thereof, as applicable, (which shall include the amounts due and payable under <u>Section 2.7(b)</u> hereof to the extent such prepayment results in a prepayment in full of the Term Loan) until paid in full within ten (10) Business Days after the receipt by a Loan Party of any Net Cash Proceeds in excess of \$5,000,000 in the aggregate during any calendar year from one (1) or more Dispositions made pursuant to <u>Section 7.4(b)(iii)</u>, in an amount equal to such excess Net Cash Proceeds.

(b) In connection with any prepayment of the Term Loan made pursuant to this <u>Section 2.8.1</u>, Borrower shall pay to Agent, for the benefit of Lenders, any amounts that would otherwise be

- 17 -

due and payable on such date had Borrower voluntarily prepaid the Obligations pursuant to <u>Section 2.8.2</u> (in addition to any such prepayment of the Term Loan and related Obligations).

2.8.2 <u>Voluntary Prepayment.</u>

(a) Subject to <u>clause (b)</u> below, Borrower may, on at least five (5) Business Days' written notice or telephonic notice (followed on the same Business Day by written confirmation thereof) to Agent (which shall promptly advise each Lender thereof) not later than 12:00 noon Dallas time on such day, prepay the Term Loan and all related Obligations in whole or in part at any time prior to the Term Loan Maturity Date. Such notice to Agent shall specify the amount and proposed date of such prepayment, and the application of such amounts to be prepaid shall be applied in accordance with <u>Section 2.9.1(b)</u> or <u>2.10.2</u> (as applicable).

(b) If Borrower makes a prepayment of the Term Loan under <u>Section 2.8.2(a)</u>, it shall pay to Agent, for the benefit of Lenders, the following amounts (in addition to any such prepayment of the Term Loan and related Obligations) on the date of such prepayment: (i) if such prepayment is made prior to the first anniversary of the Closing Date, an amount equal to (A) two percent (2.0%) of the aggregate amount of the Term Loan so prepaid <u>plus</u> (B) an amount equal to the aggregate interest that would have accrued pursuant to this Agreement in relation to the aggregate amount of the Term Loan so prepaid from the date of such prepayment through the first anniversary of the Closing Date assuming a static Contract Rate equal to the Contract Rate in effect on such date of prepayment, (ii) if such prepayment is made on or after the first anniversary of the Closing Date, one percent (1.0%) of the aggregate amount of the Term Loan so prepaid, and (iii) if such prepayment is made on or after the second anniversary of the Closing Date, zero percent (0%) of the aggregate amount of the Term Loan so prepaid.

(c) For the avoidance of doubt, a permitted payment under this <u>Section 2.8.2</u> is independent of and in addition to Revenue-Based Payment Amounts that are credited toward the principal of the Loans under <u>Section 2.9.1(b)</u>. Notwithstanding anything set forth herein or in any other Loan Documents to the contrary, any prepayment of the Loans other than via the application of Revenue-Based Payment Amounts made pursuant to <u>Section 2.9.1</u> or <u>Section 2.10.2</u>, as applicable, shall be limited and governed by this <u>Section 2.8.2</u>.

2.9 <u>Repayment of Term Loan.</u>

2.9.1 <u>Revenue-Based Payment Amount.</u>

(a) During the period commencing January 1, 2024 until the Obligations are Paid in Full, Borrower promises to pay to Agent, for the account of each Lender according to its Pro Rata Term Loan Share, an amount based on a percentage of Total Revenue in each Fiscal Quarter (the **"Revenue-Based Payment Amount**"), which will be applied to the Obligations as provided inclause (b) below. The Revenue-Based Payment Amount with respect to each Fiscal Quarter shall be applied by Borrower on the Payment Date next following the end of such Fiscal Quarter in accordance with clause (b) below. The Revenue-Based Payment Amount with respect to each Fiscal Quarter shall be equal to the aggregate Revenue-Based Payment Amount spayable during the period commencing as of January 1 of the Fiscal Year of which such Fiscal Quarter is part, through the end of such Fiscal Quarter (such elapsed portion of the Fiscal Year, the "Elapsed Period"), calculated as the sum of:

- (i) One hundred percent (100%) of Total Revenue during the Elapsed Period up to and including \$10,000,000;plus
- (ii) Seventy-five percent (75%) of Total Revenue during the Elapsed Period greater than \$10,000,000 minus

- 18 -

(iii) the aggregate amount of Revenue-Based Payment Amounts, if any, paid in cash to Agent, for the benefit of Lenders, pursuant to this Section 2.9.1, with respect to each prior Fiscal Quarter in such Fiscal Year; *provided* that the Revenue-Based Payment Amount is payable solely upon Total Revenue in a given Fiscal Year, and will not be calculated on a cumulative, year-over-year basis; <u>minus</u>

applicable Payment Date.

(iv) the aggregate amount of mandatory and voluntary prepayments made pursuant to Section 2.8 prior to the

(b) So long as no Event of Default has occurred and is continuing and until the Obligations have been Paid in Full, on each Payment Date the applicable Revenue Based Payment Amount will be applied in the following priority:

(i) FIRST, to the payment of all fees, costs, expenses and indemnities due and owing to Agent pursuant to <u>Sections</u> <u>2.7, 3.1, 3.2, 10.4</u> and/or <u>10.5</u> under this Agreement or otherwise pursuant to the Collateral Documents, and any other Obligations owing to Agent in respect of sums advanced by Agent to preserve or protect the Collateral or to preserve or protect its security interest in the Collateral;

(ii) SECOND, to the payment of all fees, costs, expenses and indemnities due and owing to Lenders in respect of the Loans and Commitments pursuant to <u>Sections 2.7, 3.1, 3.2, 10.4</u> and/or <u>10.5</u> under this Agreement or otherwise pursuant to the Collateral Documents, pro rata based on each Lender's Pro Rata Term Loan Share, until Paid in Full;

(iii) THIRD, to the payment of all accrued but unpaid interest in respect of the Loans as of such Payment Date pursuant to Section 2.6 under this Agreement, pro rata based on each Lender's Pro Rata Term Loan Share, until Paid in Full

(iv) FOURTH,

(A) if the Total Revenue of the Loan Parties (on a consolidated basis) for the consecutive twelve (12) month period ended on December 31, 2025 is less than or equal to \$70,000,000, as it relates to each Payment Date on or after the Payment Date occurring in February 2026, to the payment of outstanding principal of the Loans, pro rata based on each Lender's Pro Rata Term Loan Share, in an amount equal to seven and one-half of one percent (7.5%) multiplied by the aggregate amount of the Term Loan funded hereunder as of such date of determination minus the aggregate amount of mandatory and voluntary prepayments made pursuant to Section 2.8 prior to the applicable Payment Date, or

(B) if the Total Revenue of the Loan Parties (on a consolidated basis) for the consecutive twelve (12) month period ended on December 31, 2025 is greater than \$70,000,000, as it relates to each Payment Date on or after the Payment Date occurring in February 2027, to the payment of outstanding principal of the Loans, pro rata based on each Lender's Pro Rata Term Loan Share, in an amount equal to fifteen percent (15.0%) multiplied by the aggregate amount of the Term Loan funded hereunder as of such date of determination minus the aggregate amount of mandatory and voluntary prepayments made pursuant to Section 2.8 prior to the applicable Payment Date; and

(v) FIFTH, all remaining amounts to be retained by Borrower.

For the avoidance of doubt, on each Payment Date the Borrower shall not be required to pay more than the amounts set forth in clauses (b)(i) through (b) (iv) above.

- 19 -

In the event that the Revenue-Based Payment Amount in relation to any Payment Date is insufficient for payment of the amounts set forth in<u>clauses (b)(i)</u> through $(\underline{b})(\underline{iv})$ above for such Payment Date, Borrower shall pay an amount equal to the extent of such insufficiency, in immediately available funds, within two (2) Business Days of request by Agent.

(c) In the event that Borrower makes any adjustment to Total Revenue after it has been reported to Agent, and such adjustment results in an adjustment to the Revenue-Based Payment Amount due to the Lenders pursuant to this <u>Section 2.9.1</u>, Borrower shall so notify Agent and such adjustment shall be captured, reported and reconciled with the next scheduled report and payment of Revenue-Based Payment Amount hereunder. Notwithstanding the foregoing, Agent and Borrower shall discuss and agree on the amount of any such adjustment prior to it being given effect with respect to future Revenue-Based Payment Amounts.

2.9.2 Principal.

Notwithstanding the foregoing, the outstanding principal balance of the Term Loan and all other Obligations then due and owing shall be Paid in Full on the Termination Date.

2.10 Payment.

2.10.1 Making of Payments.

All payments of principal, interest, fees and other amounts, shall be made in immediately-available funds, via wire transfer as directed by Agent in writing, not later than 1:00 p.m. Dallas time on the date due, and funds received after that hour shall be deemed to have been received by Agent on the following Business Day. Not later than two (2) Business Days prior to each Payment Date, Agent shall provide to Borrower and each Lender a quarterly statement with the amounts payable by Borrower to Agent on such Payment Date in accordance with Section 2.9.1(b) hereof, which shall include, for additional clarity, Agent's calculation of the Revenue-Based Payment Amount for the prior Fiscal Quarter, which statement shall be binding on Borrower absent manifest error, and Borrower shall be entitled to rely on such quarterly statement in relation to its payment obligations on such Payment Date.

2.10.2 Application of Payments and Proceeds Following an Event of Default.

Following the occurrence and during the continuance of an Event of Default, or if the Obligations have otherwise become or have been declared to become immediately due and payable in accordance with this Agreement, then notwithstanding anything herein or in any other Loan Document to the contrary, Agent shall apply all or any part of payments in respect of the Obligations and proceeds of Collateral, in each case as received by Agent, to the payment of the Obligations in the order and priority as determined by Agent in its sole discretion.

2.10.3 Set-off.

Borrower agrees that Agent and each Lender and its Affiliates have all rights of set-off and bankers' lien provided by applicable law, and in addition thereto, Borrower agrees that at any time an Event of Default exists, Agent and each Lender may, to the fullest extent permitted by applicable law, apply to the payment of any Obligations of Borrower hereunder then due, any and all balances, credits, deposits, accounts or moneys of Borrower then or thereafter with Agent or such Lender. Notwithstanding the foregoing, no Lender shall exercise any rights described in the preceding sentence without the prior written consent of Agent.

2.10.4 Proration of Payments.

If any Lender shall obtain any payment or other recovery (whether voluntary, involuntary, by application of set-off or otherwise, on account of principal of, interest on or fees in relation to any Loan, but excluding any payment pursuant to <u>Section 3.1, 3.2</u>, <u>10.5</u> or <u>10.8</u>) in excess of its applicable Pro Rata Term Loan Share of payments and other recoveries obtained by all Lenders on account of principal of, interest on or fees in relation to such Term Loan then held by them, then such Lender shall purchase from the other Lenders such participations in the Loans held by them as shall be necessary to cause such purchasing Lender to share the excess payment or other recovery ratably with each of them; *provided* that if all or any portion of the extent of such recovery.

Section 3. <u>Yield Protection.</u>

3.1 <u>Taxes.</u>

All payments of principal and interest on the Loans and all other amounts payable hereunder by or on behalf of (a) Borrower to or for the account of Agent or any Lender shall be made free and clear of and without deduction for any present or future income, excise, stamp, documentary, property or franchise taxes and other taxes, fees, duties, levies, withholdings or other similar charges imposed by any Governmental Authority that is a taxing authority ("Tax" or "Taxes"), except as required by applicable law. If any withholding or deduction from any payment to be made by Borrower hereunder is required in respect of any Taxes pursuant to any applicable law, rule or regulation, then Borrower shall: (w) be entitled to make such withholding or deduction; (x) pay directly to the relevant Governmental Authority the full amount so withheld or deducted; (y) as promptly as practicable forward to Agent the original or a certified copy of an official receipt or other documentation reasonably satisfactory to Agent evidencing such payment to such Governmental Authority; and (z) if the withholding or deduction is with respect to Indemnified Taxes, pay to Agent for the account of Lenders such additional amount or amounts as is necessary to ensure that the net amount actually received by each Lender will equal the full amount such Lender would have received had no such withholding or deduction of Indemnified Taxes been required. For purposes of this Agreement, "Indemnified Taxes" mean any Taxes excluding (i) Taxes imposed on or measured by Agent's or any Lender's net income (however denominated) or gross profits, and franchise Taxes, imposed by any jurisdiction (or subdivision thereof) under the laws of which Agent or such Lender is organized or in which Agent or such Lender conducts business or, in the case of any Lender, in which its applicable lending office is located at the time such Lender acquires its initial interest in any Term Loan Commitment, (ii) any branch profit Taxes imposed by the United States of America or any similar tax imposed by any other jurisdiction in which Agent or a Lender is located or conducts business; (iii) in the case of any Lender, any withholding Tax that is imposed on amounts payable to such Lender at the time such Lender becomes a party to this Agreement or designates a new lending office; (iv) in the case of any U.S. Lender, any United States federal backup withholding Tax; and (v) Taxes imposed under FATCA; (vi) Taxes attributable to a Foreign Lender's failure to comply with Section 3.1(c) or inability to provide the applicable IRS Form set forth in Section 3.1(c) to Borrower and Agent; (vii) with respect to Agent or any Lender, Taxes imposed as a result of a present or former connection between such Agent or Lender and the jurisdiction imposing such Tax (other than connections arising from such Agent or Lender having executed, delivered, become a party to, performed its obligations under, received payments under, received or perfected a security interest under, engaged in any other transaction pursuant to or enforced any Loan Document, or sold or assigned an interest in any Loan or Loan Document); and (viii) in the case of a Lender, U.S. federal withholding Taxes, if any and not otherwise included in clauses (i) through (vii), imposed on amounts payable to or for the account of such Lender with respect to an applicable interest in a Loan or Commitment pursuant to a law in effect on the date on which such Lender acquires such interest in the Loan or Commitment or changes its lending office (items in clauses (i) through (viii), "Excluded Taxes"). To the extent that any amounts shall ever be paid by Borrower in respect of Indemnified Taxes, such amounts shall, for greater certainty, be considered to have accrued and to have been paid by Borrower as interest on the Loans.

- 21 -

(b) Borrower shall indemnify Agent and each Lender for any Indemnified Taxes paid by Agent or such Lender, as applicable, on or with respect to any payment by or on account of any obligation of Borrower hereunder, and any additions to Tax, penalties and interest paid by Agent or such Lender with respect to such Indemnified Taxes; *provided* that Borrower shall not have any obligation to indemnify any party hereunder for any Indemnified Taxes or additions to Tax, penalties or interest with respect thereto that result from or are attributable to such party's own fraud, gross negligence or willful misconduct. Payment under this <u>Section 3.1(b)</u> shall be made within thirty (30) days after the date Agent or the Lender, as applicable, makes written demand therefor; *provided*, *however*, that if such written demand is made more than one-hundred eighty (180) days after the earlier of (i) the date on which Agent or the Lender, as applicable, pays such Indemnified Taxes or additions to Tax, penalties or interest with respect thereto, then Borrower shall not be obligated to indemnify Agent or such Lender for such Lender, as applicable, Taxes or additions to Tax, penalties or interest with respect thereto.

(c) Any Lender that is entitled to an exemption from or reduction of withholding Tax with respect to payments made under any Loan Document shall deliver to Borrower and Agent, at the time or times reasonably requested by Borrower or Agent, such properly completed and executed documentation reasonably requested by Borrower or Agent as will permit such payments to be made without withholding or at a reduced rate of withholding. In addition, any Lender, if reasonably requested by Borrower or Agent, shall deliver such other documentation prescribed by applicable law or reasonably requested by Borrower or Agent to determine whether or not such Lender is subject to backup withholding or information reporting requirements. Without limiting the generality of the foregoing:

(i) Each Foreign Lender shall deliver to Borrower and Agent on or prior to the date on which such Foreign Lender becomes a party to this Agreement:

- Two duly completed and executed originals of IRS Form W-8BEN (or IRS Form W-8BENE) claiming exemption from withholding of Taxes under an income tax treaty to which the United States of America is a party;
- (2) two duly completed and executed originals of IRS Form W-8ECI;
- (3) a certificate in form and substance reasonably satisfactory to Agent and Borrower claiming entitlement to the portfolio interest exemption under Section 881(c) of the IRC and certifying that such Foreign Lender is not (w) a conduit entity participating in a conduit financing arrangement as defined in Treasury Regulation 1.881-3, (x) a "bank" within the meaning of Section 881(c)(3)(A) of the IRC, (y) a "10 percent shareholder" of Borrower within the meaning of Section 881(c)(3)(B) of the IRC, or (z) a "controlled foreign corporation" described in Sections 881(c)(3)(C) of the IRC, together with two duly completed and executed originals of IRS Form W-8BEN (or IRS Form W-8BENE); or
- (4) if the Foreign Lender is not the beneficial owner of amounts paid to it hereunder, two duly completed and executed originals of IRS Form W-8IMY, each accompanied by a duly completed and executed IRS Form W-8ECI, IRS

- 22 -

Form W-8BEN (or IRS Form W-8BEN-E), IRS Form W-9 and/or other certification documents from each beneficial owner of such amounts claiming entitlement to exemption from withholding or backup withholding of Taxes.

(ii) Each Foreign Lender shall (to the extent legally entitled to do so) provide updated forms to Borrower and Agent on or prior to the date any prior form previously provided under this <u>clause (c)</u> becomes obsolete or expires, after the occurrence of an event requiring a change in the most recent form or certification previously delivered by it pursuant to this <u>clause (c)</u> or from time to time if requested by Borrower or Agent.

(iii) Each U.S. Lender shall deliver to Agent and Borrower on or prior to the date on which such Lender becomes a party to this Agreement (and from time to time thereafter upon the request of Borrower or Agent) properly completed and executed originals of IRS Form W-9 certifying that such Lender is exempt from backup withholding.

Notwithstanding anything to the contrary contained in this Agreement, Borrower shall not be required to pay additional amounts to or indemnify any Lender pursuant to this <u>Section 3.1</u> with respect to any Taxes required to be deducted or withheld (or any additions to Tax, penalties or interest with respect thereto) (A) on the basis of the information, certificates or statements of exemption provided by a Lender pursuant to this <u>clause (c)</u> or (B) if such Lender shall fail to comply with the certification requirements of this <u>clause (c)</u>. For the avoidance of doubt, all references to IRS Forms in this<u>clause (c)</u> shall include, in each case, any successor form.

(d) Without limiting the foregoing, each Lender shall timely comply with any certification, documentation, information or other reporting necessary to establish an exemption from withholding under FATCA and shall provide any documentation reasonably requested by Borrower or Agent sufficient for Borrower and Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such applicable reporting requirements. Solely for purposes of this paragraph (d), "FATCA" shall include any amendments to FATCA after the date of this Agreement.

(e) If Agent or a Lender determines that it is entitled to or has received a refund or credit of any Taxes for which it has been indemnified by Borrower (or another Loan Party) or with respect to which Borrower (or another Loan Party) shall have paid additional amounts pursuant to this <u>Section 3.1</u>, it shall promptly notify Borrower of such refund or credit, and promptly make an appropriate claim to the relevant Governmental Authority for such refund or credit (if it has not previously done so). If Agent or a Lender receives a refund or credit (whether or not pursuant to such claim) of such Taxes, it shall promptly pay over such refund or credit to Borrower (but only to the extent of indemnity payments made, or additional amounts paid, by Loan Parties under this <u>Section 3.1</u> with respect to the Taxes giving rise to such refund or credit), net of all reasonable out-of-pocket and documented third-party expenses of the Agent or such Lender and without interest (other than any interest paid by the relevant Governmental Authority with respect to such refund or credit); *provided* that Borrower, upon the request of Agent or such Lender, agrees to repay to Agent or such Lender the amount paid over to Borrower in the event Agent or such Lender is required to repay such refund to such Governmental Authority. This <u>Section 3.1(e)</u> shall not be construed to require Agent or any Lender to make available its Tax returns (or any other information relating to its Taxes which it deems confidential) to Borrower or any other Person or to alter its internal practices or procedures with respect to the administration of Taxes.

3.2 Increased Cost.

(a) If, after the Closing Date, the adoption of, or any change in, any applicable law, rule, directive or regulation, or any change in the interpretation or administration of any applicable law, rule, directive or regulation by any Governmental Authority, central bank or comparable agency charged

- 23 -

with the interpretation or administration thereof *(provided* that notwithstanding anything herein to the contrary, the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith shall be considered a change in applicable law, regardless of the date enacted, adopted or issued), or compliance by any Lender with any request or directive (whether or not having the force of law) issued after the Closing Date of any such authority, central bank or comparable agency: (i) shall impose, modify or deem applicable any reserve (including any reserve imposed by the FRB), special deposit or similar requirement against assets of, deposits with or for the account of, or credit extended by any Lender; or (ii) shall impose on any Lender any other condition affecting its ability to make loans based on the Term SOFR Rate; and the result of anything described in <u>clauses (i)</u> and <u>(ii)</u> above is to increase the cost to (or to impose a cost on) such Lender of making or maintaining any loan based on the Term SOFR Rate, or to reduce the amount of any sum received or receivable by such Lender under this Agreement or under its Note with respect thereto, then upon demand by such Lender (which demand shall be furnished to Agent), and without duplication of other payment obligations of Borrower hereunder (including pursuant to <u>Section 3.1</u>). Borrower shall pay directly to such Lender such additional amount as will compensate such Lender for such Lender first made demand therefor; *provided* that if the event giving rise to such costs or reductions has retroactive effect, such one-hundred eighty (180) day period shall be extended to include the period of retroactive effect. For the avoidance of doubt, this <u>clause (a)</u> will not apply to any such increased costs or reductions resulting from Taxes, as to which<u>Section 3.1</u> shall govern.

(b) If any Lender shall reasonably determine that any change after the Closing Date in, or the adoption or phase-in after the Closing Date of, any applicable law, rule, directive or regulation regarding capital adequacy, or any change after the Closing Date in the interpretation or administration thereof by any Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof, or the compliance by any Lender or any Person controlling such Lender with any request or directive issued after the Closing Date regarding capital adequacy (whether or not having the force of law) of any such authority, central bank or comparable agency (provided that notwithstanding anything herein to the contrary, the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines or directives thereunder or issued in connection therewith shall be considered a change in applicable law, regardless of the date enacted, adopted or issued), has or would have the effect of reducing the rate of return on such Lender's or such controlling Person's capital as a consequence of such Lender's obligations hereunder to a level below that which such Lender or such controlling Person could have achieved but for such change, adoption, phase-in or compliance (taking into consideration such Lender's or such controlling Person's policies with respect to capital adequacy) by an amount deemed by such Lender or such controlling Person to be material, then from time to time, within five (5) Business Days of demand by such Lender (which demand shall be accompanied by a statement setting forth the basis for such demand and a calculation of the amount thereof in reasonable detail, a copy of which shall be furnished to Agent), Borrower shall pay to such Lender such additional amount as will compensate such Lender or such controlling Person for such reduction, so long as such amounts have accrued on or after the day which is one-hundred eighty (180) days prior to the date on which such Lender first made demand therefor; provided that if the event giving rise to such costs or reductions has retroactive effect, such one-hundred eighty (180) day period shall be extended to include the period of retroactive effect.

(c) Each Lender agrees that, as promptly as practicable after the officer of such Lender responsible for administering its Loans, becomes aware of the occurrence of an event or the existence of a condition that would entitle such Lender to receive payments under this <u>Section</u> <u>3.2</u>, it will, to the extent not inconsistent with the internal policies of such Lender and any applicable legal or regulatory restrictions, use reasonable efforts to (i) make, issue, fund or maintain its Loans through another office of such Lender, or (ii) take such other measures as such Lender may deem reasonable, if as a result thereof the additional amounts which would otherwise be required to be paid to such Lender pursuant to this <u>Section 3.2</u> would be

- 24 -

materially reduced and if, as determined by such Lender in its sole discretion, the making, issuing, funding or maintaining of such Loans through such other office or in accordance with such other measures, as the case may be, would not otherwise adversely affect such Loans or the interests of such Lender; *provided* that such Lender will not be obligated to utilize such other office pursuant to this<u>clause (c)</u> unless Borrower agrees to pay all incremental expenses incurred by such Lender as a result of utilizing such other office as described above. A certificate as to the amount of any such expenses payable by Borrower pursuant to this <u>clause (c)</u> (setting forth in reasonable detail the basis for requesting such amount) submitted by such Lender to Borrower (with a copy to Agent) shall be conclusive absent manifest error.

3.3 [Reserved].

3.4 Manner of Funding; Alternate Funding Offices.

Notwithstanding any provision of this Agreement to the contrary, each Lender shall be entitled to fund and maintain its funding of all or any part of its Loans in any manner it may determine at its sole discretion. Each Lender may, if it so elects, fulfill its commitment to make the Term Loan by causing any branch or Affiliate of such Lender to make such Loan; *provided* that in such event for the purposes of this Agreement (other than<u>Section</u> <u>3.1</u>) such Loan shall be deemed to have been made by such Lender and the obligation of Borrower to repay such Loan shall nevertheless be to such Lender and shall be deemed held by it, to the extent of such Loan, for the account of such branch or Affiliate.

3.5 Conclusiveness of Statements; Survival.

Determinations and statements of any Lender pursuant to <u>Section 3.1, 3.2, 3.3</u> or <u>3.4</u> shall be conclusive absent demonstrable error. Lenders may use reasonable averaging and attribution methods in determining compensation under <u>Sections 3.1</u> or <u>3.2</u>, and the provisions of such Sections shall survive repayment of the Loans, cancellation of the Notes and termination of this Agreement.

Section 4. <u>Conditions Precedent.</u>

The obligation of each Lender to make its Loan hereunder is subject to the following conditions precedent, each of which shall be reasonably satisfactory in all respects to Agent.

4.1 <u>Prior Debt.</u>

The Prior Debt other than the Subordinated Debt, if any, (i) has been (or substantially concurrently with the funding of the initial borrowing on the Closing Date will be) paid in full and (ii) Agent shall have received evidence that arrangements satisfactory to Agent have been made for the termination and release of all related Liens, if any, granted in connection with such Prior Debt.

4.2 <u>General.</u>

Borrower shall have delivered the following documents in form and substance acceptable to Agent in its sole discretion (and, as applicable, duly executed):

(a) Loan Documents. The Loan Documents to which any Loan Party is a party, each duly executed by a Responsible Officer of each Loan Party and the other parties thereto (except Agent and the Lenders), and each other Person (except Agent and the Lenders) shall have delivered to Agent and Lenders the Loan Documents to which it is a party, each duly executed and delivered by such Person and the other parties thereto (except Agent and the Lenders).

(b) <u>Financing Statements</u>. Properly completed Uniform Commercial Code financing statements and other filings and documents required by law or the Loan Documents to provide Agent, for the benefit of Lenders, perfected first priority Liens in the Collateral.

(c) <u>Lien Searches</u>. Copies of Uniform Commercial Code, foreign, state and county search reports listing all effective financing statements filed and other Liens of record against any Loan Party, with copies of any financing statements and applicable searches of the records of the U.S. Patent and Trademark Office and the U.S. Copyright Office performed with respect to each Loan Party, all in each jurisdiction reasonably determined by Agent.

(d) <u>Payoff; Release</u>. Payoff letters with respect to the repayment in full of all Prior Debt other then the Subordinated Debt, termination of all agreements relating thereto and the release of all Liens granted in connection therewith, with Uniform Commercial Code or other appropriate termination statements and documents effective to evidence the foregoing or authorization to file the same.

(e) <u>Authorization Documents</u>. For each Loan Party, such Person's (i) charter, certificate of incorporation (or similar formation document), and (if any) certificate of incorporation on change of name, certified by the appropriate Governmental Authority, as applicable, (ii) good standing certificates in its jurisdiction of incorporation (or formation), as applicable, and in each other jurisdiction reasonably requested by Agent, (iii) bylaws or memorandum and articles of association (or similar governing document), (iv) resolutions of its board of directors (or similar governing body) and shareholders, in each case approving and authorizing such Person's execution, delivery and performance of the Loan Documents to which it is party and the transactions contemplated thereby, and (v) specimen signature and incumbency certificates of its Responsible Officers executing any of the Loan Documents, all certified by its director, secretary or an assistant secretary (or similar officer) as being in full force and effect without modification, in form and substance reasonably satisfactory to Agent.

(f) <u>Opinions of Counsel</u>. Opinions of counsel for each Loan Party in form and substance acceptable to Agent regarding certain closing matters, and Borrower hereby requests such counsel to deliver such opinions and authorizes Agent and Lenders to rely thereon.

(g) Insurance. Certificates or other evidence of insurance (including, in respect of the policies listed in<u>Schedule 5.16</u>, a letter from the relevant insurance brokers addressed to Agent and Lenders listing the insurance policies of the Loan Parties and confirming that they are on risk and covering appropriate risks for the business carried out by the Loan Parties) in effect as required by Section 6.3(c) and (d), in respect of policies issued by any insurance company in the United States.

(h) <u>Financials</u>. The financial statements, projections and pro forma balance sheet described in<u>Section 5.4</u>.

(i) <u>Consents.</u> Evidence that all necessary consents, permits and approvals (governmental or otherwise) required for the execution, delivery and performance by each Loan Party of the Loan Documents have been duly obtained and are in full force and effect.

(i) <u>Borrowing Limits</u>. A certificate of Borrower confirming that borrowing or guaranteeing or securing, as appropriate, the Term Loan Commitment would not cause any borrowing, guarantee, security or similar limit binding on any Loan Party to be exceeded.

(j) <u>Other Documents</u>. Such other certificates, documents and agreements as Agent or any Lender may reasonably request.

4.3 <u>Fees.</u>

- 26 -

The Lenders and Agent shall have received all fees required to be paid, and all expenses for which invoices have been presented (including Legal Costs), required to be paid under the Loan Documents on or before the Closing Date. All such amounts will be paid with proceeds of the initial advance of the Term Loan and any previous expense deposits made with Agent on or before the Closing Date and will be reflected in the funding instructions given by Borrower to Agent on or before the Closing Date.

4.4 <u>Representations, Warranties, Defaults.</u>

As of the Closing Date, after giving effect to the making of the Loans, (a) all representations and warranties of Borrower set forth in any Loan Document shall be true and correct in all material respects as if made on and as of the Closing Date (except to the extent already qualified by materiality, in which case it shall be true and correct in all respects and shall not be false or misleading in any respect and except for representations and warranties that specifically refer to an earlier date, which shall be true and correct in all material respects as of such earlier date) and (b) no Default or Event of Default shall exist. The acceptance of the Term Loan by Borrower shall be deemed to be a certification by Borrower that the conditions set forth in this Section 4.4 have been satisfied.

4.5 <u>Diligence.</u>

Agent and Lenders shall have completed their due diligence review of the Loan Parties and their Subsidiaries, their assets, business, obligations and the transactions contemplated herein, the results of which shall be reasonably satisfactory in form and substance to Lenders, including, without limitation, (i) an examination of (A) Borrower's projected Total Revenue for such periods as required by Lenders, (B) such valuations of Borrower and its assets as Lenders shall require (C) the terms and conditions of all obligations owed by Borrower deemed material by Lenders, the results of which shall be satisfactory in form and substance to Lenders and (D) background checks with respect to the managers, officers and owners of Borrower required by Agent; (ii) an examination of the Collateral, the financial statements and the books, records, business, obligations, financial condition and operational state of Borrower, and Borrower shall have demonstrated to Agent's satisfaction, in its reasonable discretion, that no operations of Borrower are the subject of any governmental investigation, evaluation or any remedial action which could reasonably be expected to result in a Material Adverse Effect.

4.6 <u>Corporate Matters.</u>

All corporate and other proceedings, documents, instruments and other legal matters in connection with the transactions contemplated by the Loan Documents (including, but not limited to, those relating to corporate and capital structures of Borrower) shall be satisfactory to Lenders in their reasonable discretion.

4.7 <u>No Material Adverse Effect.</u>

No Material Adverse Effect shall have occurred and be continuing.

Section 5. <u>Representations and Warranties.</u>

To induce Agent and Lenders to enter into this Agreement and to induce Lenders to make the Loan hereunder, Borrower represents and warrants to Agent and Lenders, as of the Closing Date and the date of the Subsequent Term Loan (if any) made by Lenders pursuant to <u>Section 2.2.2</u>, that:

5.1 Organization.

Each Loan Party is duly incorporated, validly existing and (if applicable) in good standing under the laws of its state or country of jurisdiction as set forth on <u>Schedule 5.1</u>, and is duly qualified to carry

- 27 -

on its business in each jurisdiction set forth on <u>Schedule 5.1</u>, which are all of the jurisdictions in which failure to so qualify would reasonably be likely to have or result in a Material Adverse Effect. Each Loan Party has the power to own its assets and carry on its business as it is being conducted.

5.2 Authorization; No Conflict.

Each Loan Party is duly authorized to execute and deliver each Loan Document to which it is a party, to borrow or guarantee monies thereunder, as applicable, and to perform its Obligations under each Loan Document to which it is a party. The execution, delivery and performance by each Loan Party of this Agreement and the other Loan Documents to which it is a party, as applicable, and the transactions contemplated therein, do not and will not (a) require any consent or approval of any Governmental Authority (other than any consent or approval which has been obtained and is in full force and effect), (b) conflict with (i) any provision of applicable law (including any Health Care Law), (ii) the charter, certificate of incorporation, by-laws, or other organizational documents of such Loan Party or (iii) (except as it relates to the documents governing the Prior Debt, each of which will be terminated and/or paid on the Closing Date) any Material Contract, or any judgment, order or decree, which is binding upon any Loan Party or any of its properties or (c) require, or result in, the creation or imposition of any Lien on any asset of any Loan Party (other than Liens in favor of Agent created pursuant to the Collateral Documents). No limit on any Loan Party's powers will be exceeded as a result of the borrowing, grant of security or giving of guarantees or indemnities contemplated by the Loan Documents to which it is a party.

5.3 Validity; Binding Nature.

Each of this Agreement and each other Loan Document to which any Loan Party is a party, as applicable, is the legal, valid and binding obligation of such Loan Party, enforceable against such Loan Party in accordance with its terms, subject to bankruptcy, insolvency and similar laws affecting the enforceability of creditors' rights generally and to general principles of equity and concepts of reasonableness.

5.4 Financial Condition.

(a) The audited financial statements of Borrower for the Fiscal Year 2022 and the unaudited financial statements of Borrower for the Fiscal Quarter ended September 2023, copies of each of which have been delivered pursuant hereto, were prepared in accordance with GAAP and present fairly in all material respects the consolidated financial condition of Borrower as at such dates and the results of its operations for the periods then ended, subject, in the case of unaudited financial statements, to the absence of footnotes and normal year end audit adjustments.

(b) The consolidated financial projections (including an operating budget and a cash flow budget) of Borrower delivered to Agent and Lenders on or prior to the Closing Date (i) were prepared by Borrower in good faith and (ii) were prepared in accordance with assumptions for which Borrower believes it has a reasonable basis, and the accompanying consolidated and consolidating pro forma unaudited balance sheet of Borrower as at the Closing Date, adjusted to give effect to the financings contemplated hereby as if such transactions had occurred on such date, is consistent in all material respects with such projections (it being understood that the projections are not a guaranty of future performance and that actual results during the period covered by the projections may materially differ from the projected results therein).

5.5 <u>No Material Adverse Effect.</u>

Since December 31, 2022, there has been no Material Adverse Effect.

5.6 <u>Litigation.</u>

- 28 -

Except as set forth on <u>Schedule 5.6</u>, no litigation (including derivative actions), arbitration proceeding, administrative proceeding or governmental investigation or proceeding is pending or, to Borrower's knowledge, threatened against any Loan Party that would reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect. As of the Closing Date, other than any liability incidental to such litigation or proceedings, no Loan Party has any material Contingent Obligations not disclosed in the financial statements specified in <u>Section 5.4(a)</u> and other Contingent Obligations permitted by <u>Section 7.1</u>.

5.7 Ownership of Properties; Liens.

Borrower and each other Loan Party owns, or leases or licenses, as applicable, all of its material properties and assets, tangible and intangible, of any nature whatsoever that it purports to own, or lease, as applicable (including Intellectual Property), free and clear of all Liens and charges and claims (including infringement claims with respect to Intellectual Property), except Permitted Liens and as set forth on <u>Schedule 5.7</u>.

5.8 Capitalization.

All issued and outstanding Equity Interests of Loan Parties are duly authorized, validly issued, fully paid, non-assessable, and such securities were issued in compliance in all material respects with all applicable laws concerning the issuance of securities. <u>Schedule 5.8</u> sets forth the authorized Equity Interests of each Loan Party (other than Borrower) as of the Closing Date as well as all Persons owning more than ten percent (10%) of the outstanding Equity Interests in each such Loan Party (other than Borrower) as of the Closing Date.

5.9 Pension Plans.

No Loan Party has a Pension Plan except in compliance with Section 6.6.1.

5.10 Investment Company Act.

No Loan Party is an "investment company" or a company "controlled" by an "investment company" or a "subsidiary" of an "investment company", within the meaning of the Investment Company Act of 1940.

5.11 <u>No Default.</u>

No Event of Default or Default exists or would result from the incurrence by Borrower of any Debt hereunder or under any other Loan Document or as a result of any Loan Party entering into the Loan Documents to which it is a party.

5.12 Margin Stock.

No Loan Party is engaged principally, or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying Margin Stock. As of the Closing Date, no portion of the Obligations is secured directly or indirectly by Margin Stock.

5.13 <u>Taxes.</u>

Each Loan Party has filed, or caused to be filed, all material federal, state, foreign and other tax returns and reports required by law to have been filed by it and has paid all federal, state, foreign and other taxes and governmental charges thereby shown to be owing, except any such taxes or charges (a) that are not delinquent, (b) that are being diligently contested in good faith by appropriate proceedings and for

which adequate reserves in accordance with GAAP have been set aside on its books, or (c) that do not exceed \$500,000.

5.14 Solvency.

On the Closing Date, and immediately prior to and after giving effect to the borrowing hereunder and the use of the proceeds hereof, Borrower and its Subsidiaries, on a consolidated basis, are, and will be, Solvent.

5.15 Environmental Matters.

The on-going operations of Loan Parties comply in all respects with all applicable Environmental Laws, except for non-compliance which could not (if enforced in accordance with applicable law) reasonably be expected to result in a Material Adverse Effect. Each Loan Party has obtained, and maintained in good standing, all licenses, permits, authorizations and registrations required under any Environmental Law and necessary for its respective ordinary course operations, and each Loan Party is in compliance with all material terms and conditions thereof, except where the failure to do so would not reasonably be expected to result in a Material Adverse Effect. No Loan Party nor any of their respective properties or operations is subject to any outstanding written order from or agreement with any federal, state, or local Governmental Authority, nor subject to any judicial or docketed administrative proceeding, respecting any Environmental Law, Environmental Claim or Hazardous Substances, in each case, except as could not reasonably be expected to result in a Material Adverse Effect. There are no Hazardous Substances or other conditions or circumstances existing with respect to any property, or arising from operations prior to the Closing Date, of any Loan Party that would reasonably be expected to result in a Material Adverse Effect. No Loan Party that would reasonably be expected to result in a Material Adverse Effect.

5.16 Insurance.

Loan Parties and their respective properties are insured with financially sound and reputable insurance companies which are not Affiliates of any Loan Party, in such amounts, with such deductibles and covering such risks as are customarily carried by companies engaged in similar businesses and owning similar properties in localities where such Loan Parties operate, as applicable. A true and complete listing of such insurance as of the Closing Date, including issuers, coverages and deductibles, is set forth on <u>Schedule 5.16</u>.

5.17 <u>Information.</u>

All written information heretofore or contemporaneously herewith furnished in writing by Borrower to Agent or any Lender for purposes of or in connection with this Agreement and the transactions contemplated hereby, taken as a whole, is, and all written information hereafter furnished by or on behalf of Borrower to Agent or any Lender pursuant hereto or in connection herewith, taken as a whole, will be true and accurate in all material respects on the date as of which such information, taken as a whole, is dated or certified, and none of such information is or will be incomplete by omitting to state any material fact necessary to make such information not misleading in any material respect in light of the circumstances under which made (it being recognized by Agent and Lenders that any projections and forecasts provided by Borrower are based on good faith estimates and assumptions believed by Borrower to be reasonable as of the date of the applicable projections or assumptions and that actual results during the period or periods covered by any such projections and forecasts may materially differ from projected or forecasted results).

5.18 Intellectual Property; Products and Services.

(a) <u>Schedule 5.18(a)</u> (as updated from time to time in accordance with<u>Section 6.1.2</u> hereof) accurately and completely lists all of Loan Parties' Registered Intellectual Property. Each Loan

- 30 -

Party owns and possesses or has a license or other right to use all Intellectual Property as is necessary for the conduct of the business of such Loan Party, and, to the knowledge of such Loan Party, without any infringement upon the intellectual property rights of others, except as otherwise set forth on <u>Schedule 5.18(a)</u> hereto.

(b) <u>Schedule 5.18(b)</u> (as updated from time to time in accordance with<u>Section 6.1.2</u> hereof) accurately and completely lists all material Products and Services and all Required Permits in relation thereto.

(c) With respect to any Product or Service being tested, manufactured, marketed, sold, and/or delivered by Loan Parties, the applicable Loan Party has received (or the applicable, authorized third parties have received), and such Product or Service is the subject of, all Required Permits needed in connection with the testing, manufacture, marketing, sale, and/or delivery of such Product or Service by or on behalf of Loan Parties as currently conducted. No Loan Party has received any notice from any applicable Governmental Authority, specifically including the FDA and/or CMS, that such Governmental Authority is conducting an investigation or review (other than a normal routine scheduled inspection) of any Loan Party's (*x*) manufacturing facilities, laboratory facilities, the processes for such Product, or any related sales or marketing activities and/or the Required Permits related to such Product, and (*y*) laboratory facilities, the processes for such Services, or any related sales or marketing activities and/or the Required Permits related to such Services. There are no material deficiencies or violations of applicable laws in relation to the manufacturing, processes, sales, marketing, or delivery of such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services and/or the Required Permits related to such Product or Services, no Required Permit has been revoked or withdrawn, nor, to the best of Borrower's knowledge, has any such Governmental Authority issued any order or recommendation stating that the development, testing, manufacturing, sales and/or marketing of such Product or Services by or on behalf o

(d) Except as set forth on <u>Schedule 5.18(b)</u>, (A) there have been no materially adverse clinical trial results in respect of any Product since the date on which the applicable Loan Party acquired rights to such Product, and (B) there have been no product recalls or voluntary product withdrawals from any market in respect of any material Product since the date on which the applicable Loan Party acquired rights to such Product.

(e) No Loan Party has experienced any significant failures in its manufacturing of any Product which caused any reduction in material Products sold.

5.19 <u>Restrictive Provisions.</u>

No Loan Party is a party to any agreement or contract or subject to any restriction contained in its operative documents which would reasonably be expected to have a Material Adverse Effect.

5.20 Labor Matters.

No Loan Party is subject to any labor or collective bargaining agreement. There are no existing or threatened strikes, lockouts or other labor disputes involving any Loan Party that singly or in the aggregate would reasonably be expected to have a Material Adverse Effect. Hours worked by and payment made to employees of each Loan Party are not in violation in any material respect of the Fair Labor Standards Act or any other applicable law, rule, directive or regulation dealing with such matters. Each Loan Party has fully and timely made any and all social benefits and pension contributions and payments required to be made by such Loan Party according to any applicable law or agreement.

5.21 <u>Material Contracts.</u>

- 31 -

Schedule 5.21 sets forth, with respect to each real estate lease agreement to which any Loan Party is a party as of the Closing Date, the address of the subject property. The consummation of the transactions contemplated by the Loan Documents will not give rise to a right of termination in favor of any party to any Material Contract (other than a Loan Party) which could reasonably be expected to have, either individually or in the aggregate, a Material Adverse Effect.

5.22 <u>Compliance with Laws; Health Care Laws.</u>

(a) <u>Laws Generally</u>. Each Loan Party is in compliance with, and is conducting and has conducted its business and operations in material compliance with the requirements of all applicable laws, rules, regulations, directives, decrees, orders, judgments and Permits, in each case, except where the failure to be in compliance would not reasonably be expected to have a Material Adverse Effect.

(b) <u>Health Care Laws</u>. Without limiting the generality of <u>clause (a)</u> above:

(i) No Loan Party is in violation of any applicable Health Care Laws, except for any such violation which would not reasonably be expected (either individually and taken as a whole with any other violations) to have a Material Adverse Effect.

(ii) Each Loan Party (either directly or through one or more authorized third parties) has (i) all licenses, consents, accreditations, certificates, permits, authorizations, approvals, franchises, registrations, qualifications and other rights from, and has made all applicable declarations and filings with, all applicable Governmental Authorities and self-regulatory authorities (each, an "<u>Authorization</u>") necessary to engage in the business conducted by it, except for such Authorizations with respect to which the failure to obtain would not reasonably be expected to have a Material Adverse Effect, and (ii) no knowledge that any Governmental Authorization would not reasonably be expected to have a Material Adverse Effect. All such Authorizations, suspension or revocation of such Authorization would not reasonably be expected to have a Material Adverse Effect. All such Authorizations are valid and in full force and effect and such Loan Party is in material compliance with the terms and conditions of all such Authorizations and with the rules, guidance documents, directives and regulations of the applicable regulatory authorities having jurisdiction with respect to such Authorizations, except where failure to be in such compliance or for an Authorization to be valid and in full force and effect.

(iii) Each Loan Party has received and maintains accreditation in good standing and without limitation or impairment by all applicable accrediting organizations, to the extent required by applicable law or regulation (including any foreign law or equivalent directive or regulation), except where the failure to be so accredited and in good standing without limitation would not reasonably be expected to have a Material Adverse Effect.

(iv) Except where any of the following would not reasonably be expected to have a Material Adverse Effect, no Loan Party has been, or has been threatened to be, (i) excluded from U.S. health care programs pursuant to 42 U.S.C. §1320(a)7 or any related regulations, (ii) "suspended" or "debarred" from selling products to the U.S. government or its agencies pursuant to the Federal Acquisition Regulation, relating to debarment and suspension applicable to federal government agencies generally (48 C.F.R. Subpart 9.4), or other applicable laws, directives or regulations, or (iii) made a party to any other action by any Governmental Authority that may prohibit it from selling products to any governmental or other purchaser pursuant to any federal, state, local or foreign laws, directives or regulations.

(v) No Loan Party has received any written notice from the FDA, CMS, or any other Governmental Authority with respect to, nor to Borrower's best knowledge is there, any

- 32 -

actual or threatened investigation, inquiry, or administrative or judicial action, hearing, or enforcement proceeding by the FDA, CMS, or any other Governmental Authority against any Loan Party regarding any violation of applicable law, except for such investigations, inquiries, or administrative or judicial actions, hearings, or enforcement proceedings which, individually and in the aggregate, could not reasonably be expected to result in a Material Adverse Effect.

5.23 Existing Indebtedness; Investments, Guarantees and Certain Contracts.

Except as otherwise permitted pursuant to <u>Section 7.10</u> or <u>Section 7.10</u>, no Loan Party (a) has any outstanding Debt, except Debt under the Loan Documents, or (b) owns or holds any equity or longterm debt investments in, or has any outstanding advances to or any outstanding guarantees for the obligations of, or any outstanding borrowings from, any other Person.

5.24 Affiliated Agreements.

Except as permitted by <u>Section 7.7</u> and employment agreements entered into with employees, managers, officers and directors from time to time in the ordinary course of business, there are no existing or proposed agreements, arrangements, understandings or transactions between any Loan Party, on the one hand, and such Loan Party's members, managers, managing members, investors, officers, directors, stockholders, other equity holders, employees, or Affiliates or any members of their respective families, on the other hand.

5.25 Names; Locations of Offices, Records and Collateral; Deposit Accounts.

No Loan Party has conducted business under or used any name (whether corporate, partnership or assumed) other than such names set forth on <u>Schedule 5.25A</u>. Each Loan Party is the sole owner(s) of all of its respective names listed on<u>Schedule 5.25A</u>, and any and all business conducted and invoices issued in such names are such Loan Party's sales, business and invoices. Each Loan Party maintains respective places of business only at the locations set forth on <u>Schedule 5.25B</u>, and all books and records of Loan Parties relating to or evidencing the Collateral are located in and at such locations (other than (i) Deposit Accounts, (ii) Collateral in the possession of Agent, for the benefit of Agent and Lenders, and (iii) other locations disclosed to Agent from time to time in writing). <u>Schedule 7.14</u> lists all of Loan Parties' Deposit Accounts as of the Closing Date. All of the material tangible Collateral is located exclusively within the United States.

5.26 Non-Subordination.

The payment and performance of the Obligations by Loan Parties are not subordinated in any way to any other obligations of such Loan Parties or to the rights of any other Person.

5.27 Broker's or Finder's Commissions.

Except as set forth in <u>Schedule 5.27</u>, no broker's, finder's or placement fee or commission will be payable to any broker or agent engaged by any Loan Party or any of its officers, directors or agents with respect to the Loan or the transactions contemplated by this Agreement except for fees payable to Agent and Lenders. Borrower agrees to indemnify Agent and each Lender and hold each harmless from and against any claim, demand or liability for broker's, finder's or placement fees or similar commissions, whether or not payable by Borrower, alleged to have been incurred in connection with such transactions, other than any broker's fees payable to Persons engaged by Agent and/or Lenders.

5.28 Anti-Terrorism; OFAC.

- 33 -

(a) No Loan Party nor any Person controlling or controlled by a Loan Party, nor, to Borrower's knowledge, any Person having a beneficial interest in a Loan Party, nor any Person for whom a Loan Party is acting as agent or nominee in connection with this transaction (1) is a Person whose property or interest in property is blocked or subject to blocking pursuant to Section 1 of Executive Order 13224 of September 23, 2001, Blocking Property and Prohibiting Transactions With Persons Who Commit, Threaten to Commit, or Support Terrorism (66 Fed. Reg. 49079 (2001)), (2) engages in any dealings or transactions prohibited by Section 2 of such executive order, or is otherwise associated with any such Person in any manner that violates of Section 2 of such executive order, or (3) is a Person on the list of Specially Designated Nationals and Blocked Persons or is in violation of the limitations or prohibitions under any other OFAC regulation or executive order.

(b) No part of the proceeds of the Loan will be used, directly or indirectly, for any payments to any governmental official or employee, political party, official of a political party, candidate for political office, or anyone else acting in an official capacity, in order to obtain, retain or direct business or obtain any improper advantage, in violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

5.29 Security Interest.

Each Loan Party has full right and power to grant to Agent, for the benefit of itself and the other Lenders, a perfected, first priority (subject to Permitted Liens) security interest and Lien on the Collateral pursuant to this Agreement and the other Loan Documents, as applicable, subject to the following sentence. Upon the execution and delivery of this Agreement and the other Loan Documents, and upon the filing of the necessary financing statements and/or appropriate filings and/or delivery of the necessary certificates evidencing any equity interest, control and/or possession, as applicable, without any further action, Agent will have a good, valid and first priority (subject to Permitted Liens) perfected Lien and security interest in the Collateral, for the benefit of Agent and Lenders. Borrower is not party to any agreement, document or instrument that conflicts with this Section 5.29.

5.30 Survival.

Borrower hereby makes the representations and warranties contained herein with the knowledge and intention that Agent and Lenders are relying and will rely thereon. All such representations and warranties will survive the execution and delivery of this Agreement, the closing and the making of the Loan.

Section 6. <u>Affirmative Covenants.</u>

Until all Obligations have been Paid in Full, Borrower agrees that, unless at any time Agent shall otherwise expressly consent in writing,

it will:

6.1 Information.

Furnish to Agent (which shall furnish to each Lender):

6.1.1 <u>Annual Report.</u>

Within one hundred fifty (150) days after the close of each Fiscal Year a copy of the annual audited report of Borrower and its Subsidiaries for such Fiscal Year, including therein (i) a consolidated balance sheet and statement of earnings and cash flows of Borrower and its Subsidiaries as at the end of and for such Fiscal Year, certified without qualification (except for qualifications relating to changes in accounting principles or practices reflecting changes in GAAP and required or approved by Borrower's independent certified public accountants) by independent auditors of recognized standing

- 34 -

selected by Borrower and reasonably acceptable to Agent; it being understood that Borrowers auditor as of the Closing Date and any other auditor of nationally recognized standing are acceptable to Agent, and (ii) a comparison with the previous Fiscal Year.

6.1.2 Interim Reports.

(a) Within five (5) days after the quarterly filing thereof with the U.S. Securities and Exchange Commission, unaudited consolidated balance sheets of Loan Parties as of the end of such Fiscal Quarter, together with consolidated statements of earnings and cash flows for such Fiscal Quarter and for the period beginning with the first day of such Fiscal Year and ending on the last day of such Fiscal Quarter, together with a comparison with the corresponding period of the previous Fiscal Year and a comparison with the budget for such period of the current Fiscal Year (which may be in preliminary form), certified by the chief financial officer or other executive officer of Borrower.

(b) Together with each such quarterly report to be delivered pursuant to <u>clause (a)</u> above, Borrower shall provide to Agent (i) a written statement of Borrower's management in setting forth a summary discussion of Borrower's financial condition, changes in financial condition and results of operations, and (ii) updated Schedules to this Agreement, as applicable, setting forth any material changes to the disclosures set forth in such schedules as most recently provided to Agent or, as applicable, a written statement of Borrower's management stating that there have been no changes to such disclosures as most recently provided to Agent.

(c) Within fifteen (15) days after the end of each calendar month, monthly unaudited consolidated balance sheets of Loan Parties as of the end of such month, together with consolidated statements of earnings and cash flows for such month, utilizing the reporting format then in use by Borrower's management, certified by a Responsible Officer of Borrower, in form and substance reasonably acceptable to Agent.

(d) Within five (5) days after the end of each calendar month, a statement of the Consolidated Unencumbered Liquid Assets of Loan Parties as determined on the last day of the prior calendar month, in form and substance reasonably acceptable to Agent.

6.1.3 Quarterly Review Meeting.

Borrower and any other Loan Parties as requested by Agent shall be available in person or via teleconference as and when reasonably requested by Agent and no less frequently than quarterly for a review meeting regarding the status of Borrower, the Collateral and performance of the same.

6.1.4 [Reserved.]

6.1.5 <u>Compliance Certificate.</u>

Contemporaneously with the furnishing of a copy of each annual audit report pursuant to <u>Section 6.1.1</u> and each set of quarterly statements pursuant to <u>Section 6.1.2</u> (including, for the avoidance of doubt the quarterly statements delivered for the Fiscal Quarter ending December 3^{kt} of each year), a duly completed Compliance Certificate, with appropriate insertions, dated the date of delivery and corresponding to such annual report or such quarterly statements, and signed by a Responsible Officer of Borrower, containing computations, if applicable, showing compliance with <u>Section 7.13</u> and a statement to the effect that such officer has not become aware of any Event of Default or Default that exists or, if there is any such event, describing it and the steps, if any, being taken to cure it.

6.1.6 Reports to Governmental Authorities and Shareholders.

- 35 -

Promptly upon the filing or sending thereof, copies of (a) all regular, periodic or special material reports of each Loan Party filed with any Governmental Authority (excluding all regular and periodic filings related to Taxes (other than annual income tax filings)), (b) all material registration statements (or such equivalent documents) of each Loan Party filed with any Governmental Authority and (c) all proxy statements or other communications made to the holders of Borrower's Equity Interests generally.

6.1.7 Notice of Default; Litigation.

Promptly upon becoming aware of any of the following, written notice describing the same and summarizing the steps being taken by Borrower or the applicable Loan Party affected thereby with respect thereto:

(a) the occurrence of an Event of Default;

(b) any litigation, arbitration or administrative or governmental investigation or proceeding not previously disclosed by Borrower to Lenders which has been instituted or, to the knowledge of Borrower, is threatened in writing against Borrower or any other Loan Party or to which any of the properties of any thereof is subject, which in each case would reasonably be expected to have a Material Adverse Effect;

(c) the institution of any steps by any member of the Controlled Group or any other Person to terminate any Pension Plan, or the failure of any member of the Controlled Group to make a required contribution to any Pension Plan (if such failure is sufficient to give rise to a Lien under Section 303(k) of ERISA) or to any Multiemployer Pension Plan, or the taking of any action with respect to a Pension Plan which could result in the requirement that Borrower or any other Loan Party furnish a bond or other security to the PBGC or such Pension Plan, or the occurrence of any event with respect to any Pension Plan or Multiemployer Pension Plan which could result in the incurrence by any member of the Controlled Group of any material liability, fine or penalty (including any claim or demand for withdrawal liability or partial withdrawal from any Multiemployer Pension Plan), or any material increase in the contingent liability of Borrower or any other Loan Party with respect to any post-retirement welfare plan benefit, or any notice that any Multiemployer Pension, that increased contributions may be required to avoid a reduction in plan benefits or the imposition of an excise Tax, that any such plan is or has been funded at a rate less than that required under Section 412 of the IRC, that any such plan is or may become insolvent, in each case, to the extent such action or event could reasonably be expected to result in a Material Adverse Effect;

(d) any cancellation or material adverse change in any insurance maintained by Borrower or any other Loan Party;

(e) any other event (including (i) any violation of any law, including any Environmental Law, or the assertion of any Environmental Claim or (ii) the enactment or effectiveness of any law, rule, directive or regulation) which would reasonably be expected to have a Material Adverse Effect; or

(f) (i) any suspension, revocation, cancellation or withdrawal of an Authorization required for Borrower or any other Loan Party, is threatened or there is any basis for believing that such Authorization will not be renewable upon expiration or will be suspended, revoked, cancelled or withdrawn, (ii) Borrower or any other Loan Party enters into any consent decree or order pursuant to any Health Care Law and Regulation, or becomes a party to any judgment, decree or judicial or administrative order pursuant to any Health Care Law, (iii) receipt of any written notice or other written communication from the FDA, CMS, or any other applicable Governmental Authority alleging non-compliance with CLIA or any other applicable Health Care Law, (iv) the occurrence of any violation of any Health Care Law by Borrower or any of the other Loan Parties in the development or provision of Services, and record keeping

- 36 -

and reporting to the FDA or CMS that would reasonably be expected to require or lead to an investigation, corrective action or enforcement, regulatory or administrative action, (v) the occurrence of any civil or criminal proceedings relating to Borrower or any of the other Loan Parties or any of their respective employees, which involve a matter within or related to the FDA's or CMS' jurisdiction, (vi) any officer, employee or agent of Borrower or any of the other Loan Parties is convicted of any crime or has engaged in any conduct for which debarment is mandated or permitted by 21 U.S.C. § 335a, or (vii) any officer, employee or agent of Borrower or any of the other Loan Parties has been convicted of any crime or engaged in any conduct for which debarment is mandated or permitted by 21 U.S.C. § 335a, or (vii) any officer, employee or agent of Borrower or any of the other Loan Parties has been convicted of any crime or engaged in any conduct for which such Person could be excluded from participating in any federal, provincial, state or local health care programs under Section 1128 of the Social Security Act or any similar law or regulation.

6.1.8 Projections.

Within sixty (60) days after the commencement of each Fiscal Year, financial projections on a quarterly basis for the Loan Parties for such Fiscal Year prepared in a manner consistent with the projections delivered by Borrower to Agent prior to the Closing Date or otherwise in a manner reasonably satisfactory to Agent, accompanied by a certificate of a Responsible Officer of Borrower on behalf of Borrower to the effect that (a) such projections were prepared by them in good faith, (b) Borrower believes that it has a reasonable basis for the assumptions contained in such projections, (c) such projections have been prepared in accordance with such assumptions and (d) such projections have been approved in writing by the Board as the operating plan for the subsequent Fiscal Year.

6.1.9 Updated Schedules to Guarantee and Collateral Agreement.

Contemporaneously with the furnishing of each annual audit report pursuant to <u>Section 6.1.1</u>, updated versions of the Schedules to the Guarantee and Collateral Agreement showing information as of the date of such audit report (it being agreed and understood that this requirement shall be in addition to the notice and delivery requirements set forth in the Guarantee and Collateral Agreement).

6.1.10 Other Information.

Promptly, from time to time as Agent reasonably requests, Borrower shall deliver or shall cause to be delivered to Agent:

(a) copies of any reports, statements or written materials (other than routine communications (electronic or otherwise) between Borrower or its Subsidiaries and such entities that are not material in nature) in relation to any Material Contract;

(b) such other information concerning Borrower and any other Loan Party as Agent may reasonably request;

(c) copies of all material communication as well as other material documents received by Loan Parties or any of their Subsidiaries from the FDA, CMS, DEA, or any other Governmental Authority; and

(d) copies of (x) any notices or other communications relating to any breach, default, or event of default with respect to any Subordinated Debt and (y) any other modifications or amendments entered into in relation to any Subordinated Debt.

Documents required to be delivered pursuant to the terms of this Section 6.1 (to the extent any such documents are included in materials otherwise filed with the U.S. Securities and Exchange Commission) may be delivered electronically and if so delivered, shall be deemed to have been delivered on the date on which Borrower or any of its Subsidiaries posts such documents, or provides a link thereto, on Borrower's

- 37 -

or any of its Subsidiaries' website on the internet at Borrower's or any of its Subsidiaries' website address or when such documents are filed with EDGAR.

6.2 Books; Records; Inspections.

6.2.1 Maintain Books and Records

Borrower shall keep, and cause each other Loan Party to keep, its books and records in accordance with sound business practices sufficient to allow the preparation of financial statements in accordance with GAAP.

6.2.2 Access by the Agent etc.

Borrower shall

(a) permit, and cause each other Loan Party to permit (at any reasonable time and with reasonable notice), Agent or any representative thereof to inspect the properties and operations of Borrower or any other Loan Party;

(b) permit, and cause each other Loan Party to permit, at any reasonable time and with reasonable notice (or at any time without notice if an Event of Default exists), Agent (accompanied by any Lender) or any representative thereof to visit any or all of its offices, to discuss its financial matters with its officers and its independent auditors (and Borrower hereby authorizes such independent auditors to discuss such financial matters with any Lender or Agent or any representative thereof), and to examine (and, at the expense of Borrower or the applicable Loan Party, photocopy extracts from) any of its books or other records; and

(c) permit, and cause each other Loan Party to permit, (at any reasonable time and with reasonable notice) Agent and its representatives to inspect the Collateral and other tangible assets of Borrower or Loan Party, to perform appraisals of the equipment of Borrower or Loan Party, and to inspect, audit, check and make copies of and extracts from the books, records, computer data, computer programs, journals, orders, receipts, correspondence and other data relating to any Collateral.

Notwithstanding the foregoing, audits and inspections shall not be conducted more than once per year absent the continuance of an Event of Default.

6.3 <u>Conduct of Business; Maintenance of Property; Insurance.</u>

(a) Borrower shall, and shall cause each other Loan Party to, (i) conduct its business substantially in accordance with its current business practices, (ii) engage principally in the same or similar lines of business substantially as heretofore conducted and lines of business ancillary, supplemental or reasonably related thereto, (iii) collect the Royalties in the ordinary course of business, (iv) maintain all of its Collateral used or useful in its business in good repair, working order and condition (normal wear and tear excepted and except as may be disposed of in the ordinary course of business and in accordance with the terms of the Loan Documents), (v) from time to time to make all necessary repairs, renewals and replacements to the Collateral; (vi) maintain and keep in full force and effect all material Permits and qualifications to do business makes such Permits or qualification necessary and in which failure to maintain such Permits or qualification could reasonably be expected to be, have or result in a Material Adverse Effect; (vii) remain operations would not reasonably be expected to be, have or result in a Material Adverse Effect; and (viii)

- 38 -

maintain, comply with and keep in full force and effect all Intellectual Property and Permits necessary to conduct its business.

(b) Borrower shall keep, and cause each other Loan Party to keep, all property necessary in the business of Borrower or each other Loan Party in good working order and condition, ordinary wear and tear excepted.

(c) Borrower shall maintain, and cause each other Loan Party to maintain, with responsible insurance companies, such insurance coverage as shall be required by all laws, directives, governmental regulations and court decrees and orders applicable to it and such other insurance, to such extent and against such hazards and liabilities, as is customarily maintained by Persons operating in the same geographical region as Borrower that are (A) subject to CLIA and other applicable Health Care Laws, or (B) otherwise delivering to customers products or services similar to the Services (in each case, as determined by Agent acting in its reasonable discretion). Upon request of Agent or any Lender, Borrower shall furnish to Agent or such Lender a certificate setting forth in reasonable detail the nature and extent of all insurance maintained by Borrower and each other Loan Party. Borrower shall cause each issuer of an insurance policy to provide Agent with an endorsement (x) showing Agent as a lender's loss payee with respect to each policy of property or casualty insurance and naming Agent as an additional insured with respect to each policy of liability insurance promptly upon request by Agent, (y) providing that the insurance carrier will endeavor to give at least thirty (30) days' prior written notice to Borrower and Agent (or ten (10) days' prior written notice if the Agent consents to such shorter notice) before the termination or cancellation of the policy prior to the expiration thereof and (z) reasonably acceptable in all other respects to Agent.

(d) Unless Borrower provides Agent with evidence of the continuing insurance coverage required by this Agreement, Agent (upon reasonable advance notice to Borrower) may purchase insurance at Borrower's expense to protect Agent's and Lenders' interests in the Collateral. This insurance shall protect Borrower's and each other Loan Party's interests. The coverage that Agent purchases shall pay any claim that is made against Borrower or any other Loan Party in connection with the Collateral. Borrower may later cancel any insurance purchased by Agent, but only after providing Agent with evidence that Borrower has obtained the insurance coverage required by this Agreement. If Agent purchases insurance for the Collateral, as set forth above, Borrower will be responsible for the reasonable costs of that insurance, including interest and any other charges that may be imposed with the placement of the insurance, until the effective date of the cancellation or expiration of the insurance, and such costs of the insurance may be added to the principal amount of the Loans owing hereunder.

6.4 <u>Compliance with Laws; Payment of Taxes and Liabilities.</u>

(a) Comply, and cause each other Loan Party to comply, in all material respects with all applicable laws, rules, regulations, directives, decrees, orders, judgments, licenses and permits, except where failure to comply would not reasonably be expected to have a Material Adverse Effect; (b) without limiting clause (a) above, ensure, and cause each other Loan Party to ensure, that no person who Controls a Loan Party is (i) listed on the Specially Designated Nationals and Blocked Person List maintained by OFAC, and/or any other similar lists maintained by OFAC pursuant to any authorizing statute, Executive Order or regulation or (ii) a Person designated under Section 1(b), (c) or (d) or Executive Order No. 13224 (September 23, 2001), any related enabling legislation or any other similar Executive Orders; (c) without limiting clause (a) above, comply and cause each other Loan Party to comply, with all applicable Bank Secrecy Act and anti-money laundering laws, directives and regulations, (d) file, or cause to be filed, all material federal, state, foreign and other Tax returns and reports required by law to be filed by any Loan Party, and (e) pay, and cause each other Loan Party to pay, prior to delinquency, all material foreign, federal, state and other Taxes and other material governmental charges against it or any of its property, as well as material claims of any kind which, if unpaid, could become a Lien (other than a Permitted Lien) on any of its property; provided that the foregoing shall not require Borrower or any other Loan Party to pay any such tax, charge or claim (i) so long as it shall contest the validity thereof in good faith by appropriate proceedings

- 39 -

and shall set aside on its books adequate reserves with respect thereto in accordance with GAAP or (ii) to the extent it is less than \$500,000. For purposes of this Section 6.4, "Control" shall mean, when used with respect to any Person, (x) the direct or indirect beneficial ownership of fifty-one percent (51%) or more of the outstanding Equity Interests of such Person or (y) the power to direct or cause the direction of the management and policies of such Person whether by contract or otherwise.

6.5 <u>Maintenance of Existence.</u>

Maintain and preserve, and (subject to <u>Section 7.4</u>) cause each other Loan Party to maintain and preserve, (a) its existence and good standing in the jurisdiction of its organization and (b) its qualification to do business and good standing in each jurisdiction where the nature of its business makes such qualification necessary, other than any such jurisdiction where the failure to be qualified or in good standing would not reasonably be expected to have a Material Adverse Effect.

6.6 Employee Benefit Plans.

6.6.1 <u>Pension Plans</u>

Except to the extent that failure to do so would not be reasonably expected to result in (a) a Material Adverse Effect or (b) liability in excess of \$500,000 of any Loan Party, maintain, and cause each other Loan Party to maintain, each Pension Plan (if any) in substantial compliance with all applicable requirements of law, directives and regulations.

6.7 Environmental Matters.

Except to the extent the failure to do so would not be reasonably expected to result in a Material Adverse Effect, if any release or disposal of Hazardous Substances shall occur or shall have occurred on any real property or any other assets of Borrower or any other Loan Party, cause, or direct the applicable Loan Party to cause, the prompt containment and removal of such Hazardous Substances and the remediation of such real property or other assets as is necessary to comply in all material respects with all Environmental Laws and to preserve the value of such real property or other assets. Without limiting the generality of the foregoing, except to the extent the failure to do so would not be reasonably expected to result in a Material Adverse Effect, Borrower shall, and shall cause each other Loan Party to, comply with each valid federal or state judicial or administrative order requiring the performance at any real property by Borrower or any other Loan Party of activities in response to the release or threatened release of a Hazardous Substance.

6.8 <u>Further Assurances.</u>

Take, and cause each other Loan Party to take, such actions as are necessary or as Agent or the Required Lenders may reasonably request from time to time to ensure that the Obligations of Borrower and each other Loan Party under the Loan Documents are secured by a perfected Lien in favor of Agent (subject only to the Permitted Liens) on substantially all of the assets of Borrower and each Loan Party (as well as all equity interests of each Loan Party) and guaranteed by each Loan Party (including, promptly upon the acquisition or creation thereof, any Subsidiary of Borrower acquired or created after the Closing Date), in each case including (a) the execution and delivery of guaranties, security agreements, pledge agreements, mortgages, deeds of trust, financing statements and other documents, and the filing or recording of any of the foregoing; (b) the delivery of certificated securities (if any) and other Collateral with respect to which perfection is obtained by possession but excluding (i) the requirement for the Loan Parties to execute and deliver leasehold mortgages, and (ii) any other Excluded Collateral as defined in the Guarantee and Collateral Agreement; and (c) using commercially reasonable efforts to obtain and deliver executed Collateral Access Agreements in relation to any foreign and domestic location where a material portion of the Collateral is held or otherwise stored from time to time.

- 40 -

6.9 <u>Compliance with Health Care Laws.</u>

(a) Without limiting or qualifying <u>Section 6.4</u> or any other provision of this Agreement, Borrower will comply, and will cause each other Loan Party to comply, in all material respects with all applicable Health Care Laws relating to the operation of such Person's business, except where failure to comply would not reasonably be expected to have a Material Adverse Effect.

(b) Borrower will, and will cause each other Loan Party to:

(i) Keep in full force and effect all Authorizations required to operate such Person's business under applicable Health Care Laws and maintain any other qualifications necessary to conduct, arrange for, administer, provide services in connection with or receive payment for all applicable Services, except to the extent such failure to keep in full force and effect or maintain would not reasonably be expected to have a Material Adverse Effect.

(ii) Promptly furnish or cause to be furnished to the Agent, (w) copies of all material reports of investigational/inspectional observations issued to and received by the Loan Parties or any of their Subsidiaries, and issued by any Governmental Authority relating to such Person's business, (x) copies of all material establishment investigation/inspection reports (including, but not limited to, FDA Form 483's) issued to and received by Loan Parties or any of their Subsidiaries and issued by any Governmental Authority, (y) copies of all material warnings and material untitled letters as well as other material documents received by Loan Parties or any of their Subsidiaries from the FDA, CMS, DEA, or any other Governmental Authority relating to or arising out of the conduct applicable to the business of the Loan Parties or any of their Subsidiaries that asserts past or ongoing lack of compliance with any Health Care Law or any other applicable foreign, federal, state or local law, directive or regulation of similar import and (z) notice of any material investigation or material audit or similar proceeding by the FDA, DEA, CMS, or any other Governmental Authority.

(iii) Promptly furnish or cause to be furnished to the Agent, (in such form as may be reasonably required by Agent) copies of all non-privileged, reports, correspondence, pleadings and other communications relating to any matter that could lead to the loss, revocation or suspension (or threatened loss, revocation or suspension) of any material Authorization or of any material qualification of any Loan Party or Subsidiary; *provided* that any internal reports to a Person's compliance "hot line" which are promptly investigated and determined to be without merit need not be reported.

(iv) Promptly furnish or cause to be furnished to the Agent notice of all material fines or penalties imposed by any Governmental Authority under any Health Care Law against any Loan Party or any of its Subsidiaries.

(v) Promptly furnish or cause to be furnished to the Agent notice of all material allegations by any Governmental Authority (or any agent thereof) of fraudulent activities of any Loan Party or any of its Subsidiaries in relation to the provision of clinical research or related services.

Notwithstanding anything to the contrary in any Loan Document, no Loan Party or any of its Subsidiaries shall be required to furnish to Agent or any Lender patient-related or other information, the disclosure of which to Agent or such Lender is prohibited by any applicable law.

6.10 <u>Cure of Violations.</u>

- 41 -

If there shall occur any breach of <u>Section 6.9</u>, Borrower shall take such commercially reasonable action as is necessary to validly challenge or otherwise appropriately respond to such fact, event or circumstance within any timeframe required by applicable Health Care Laws, and shall thereafter use commercially reasonable efforts to diligently pursue the same.

6.11 Corporate Compliance Program.

Maintain, and will cause each other Loan Party to maintain on its behalf, a corporate compliance program reasonably acceptable to Agent to ensure continuing compliance in all material respects with all applicable Health Care Laws. Borrower will permit Agent and/or any of its outside consultants to review such corporate compliance programs from time to time upon reasonable notice and during normal business hours of Borrower.

6.12 Payment of Debt.

Except as otherwise prescribed in the Loan Documents, Borrower shall pay, discharge or otherwise satisfy when due and payable (subject to applicable grace periods and, in the case of trade payables, to ordinary course of past payment practices) all of its material obligations and liabilities, except (i) when the amount or validity thereof is being contested in good faith by appropriate proceedings and appropriate reserves shall have been made in accordance with GAAP consistently applied, or (ii) where the failure to make any such payments could not reasonably be expected to result in a Material Adverse Effect.

6.13 Additional Subsidiaries.

(a) Additional Subsidiaries. Promptly after the creation or acquisition of any Subsidiary (and, in any event, within thirty (30) days after such creation or acquisition, as such time period may be extended by Agent in its sole discretion), cause such Person to (i) become a Loan Party by delivering to Agent a duly executed supplement to the Guarantee and Collateral Agreement or such other document as Agent shall approve for such purpose, (ii) grant a security interest in all Collateral (but not any Excluded Collateral as defined in the Guarantee and Collateral Agreement) owned by such Subsidiary by delivering to Agent a duly executed supplement to each applicable Collateral Document or such other document as Agent shall reasonably deem appropriate for such purpose and comply with the terms of each applicable Collateral Document, (iii) deliver to Agent such customary opinions, documents and certificates referred to in Section 4.2 as may be reasonably requested by Agent, (iv) deliver to Agent such original certificated Equity Interests or other cortificates and stock or other transfer powers evidencing the Equity Interests in such Person, (v) deliver to Agent such updated Schedules to the Loan Documents as reasonably requested by Agent with respect to such Person, (vi) using commercially reasonable efforts to obtain and deliver executed Collateral Access Agreements in relation to any foreign and domestic location where a material portion of the Collateral is held or otherwise stored from time to time, and (vii) deliver to Agent such other documents as may be reasonably requested by Agent in order to comply with this <u>Section 6.13</u>, all in form, content and scope reasonably satisfactory to Agent.

(b) <u>Merger Subsidiaries</u>. Notwithstanding the foregoing, to the extent any new Subsidiary is created solely for the purpose of consummating a merger transaction pursuant to an Acquisition permitted hereby, and such new Subsidiary at no time holds any material assets or liabilities other than any merger consideration contributed to it contemporaneously with the closing of such merger transaction (<u>provided</u>, however, that such merger consideration shall not be held by such new Subsidiary for more than five (5) Business Days without the approval of Agent in its reasonable discretion), such new Subsidiary shall not be required to take the actions set forth in <u>Section 6.13(a)</u> until the consummation of such Acquisition (at which time, the surviving entity of the respective merger transaction shall be required to so comply with <u>Section 6.13(a)</u> within thirty (30) days of the consummation of such Acquisition, as such time period may be extended by Agent in its sole discretion).

- 42 -

6.14 Post-Closing Obligations.

On or before January 31, 2024 (or such longer period as permitted by Agent in its reasonably discretion), Borrower shall deliver endorsements of insurance in effect as required by <u>Section 6.3(c)</u>, naming Agent as lender's loss payee and/or additional insured, as applicable, and providing that insurance carrier will endeavor to give at least thirty (30) days' prior written notice to Borrower and Agent (or ten (10) days' prior written notice if the Agent consents to such shorter notice) before the termination or cancellation of the applicable policy prior to the expiration thereof, in each case in form and substance reasonably satisfactory to Agent.

Section 7. <u>Negative Covenants.</u>

Until all Obligations have been Paid in Full, Borrower agrees that, unless at any time Agent shall otherwise expressly consent in writing, in its sole discretion, it will:

7.1 <u>Debt.</u>

Not, and not permit any other Loan Party to, create, incur, assume or suffer to exist any Debt, except:

- (a) Obligations under this Agreement and the other Loan Documents;
- (b) Subordinated Debt;

(c) Debt secured by Liens permitted by Section 7.2(b), Section 7.2(c) or Section 7.2(m) and extensions, renewals and refinancings thereof; *provided* that the aggregate amount of all such Debt permitted under Section 7.2(c) at any time outstanding shall not exceed \$500,000;

(d) Debt with respect to any Hedging Obligations incurred for bona fide hedging purposes and not for speculation;

(e) Debt (i) arising from customary agreements for indemnification related to sales of goods, licensing of intellectual property or adjustment of purchase price or similar obligations in any case incurred in connection with the acquisition or disposition of any business, assets or Subsidiary of Borrower otherwise permitted hereunder, (ii) representing deferred compensation to employees of any Loan Party incurred in the ordinary course of business, or (iii) representing trade payables incurred with suppliers in the ordinary course of business and customer deposits and advance payments received in the ordinary course of business from customers for goods purchased in the ordinary course of business;

(f) Debt with respect to cash management obligations and other Debt in respect of automatic clearing house arrangements, netting services, overdraft protection and similar arrangements, in each case incurred in the ordinary course of business;

(g) Debt incurred in connection with surety bonds, performance bonds or letters of credit for worker's compensation, unemployment compensation and other types of social security and otherwise in the ordinary course of business or referred to in <u>Section 7.2(e)</u>;

(h) unsecured Debt (which for further clarity shall exclude accounts payable and other current liabilities incurred by Loan Parties in the ordinary course of business), in addition to the Debt listed above, in an aggregate outstanding amount not at any time exceeding \$500,000;

(i) unsecured Debt among the Loan Parties;

(j) royalties, milestones, installment payments and notes payable incurred in connection with licenses and sublicenses;

(k) Debt consisting of the financing of insurance premiums in the ordinary course of business;

(1) Debt incurred in the ordinary course of business in connection with corporate credit cards, not to exceed Five Hundred Thousand Dollars (\$500,000.00) in the aggregate outstanding at any time;

(m) Indebtedness in the form of purchase price adjustments, earn outs, deferred compensation, or other arrangements representing acquisition consideration or deferred payments of a similar nature incurred in connection with Investments permitted by <u>Section 7.10</u>;

- (n) to the extent constituting Debt, Investments permitted by Section 7.10; and
- (o) Debt existing on the Closing Date and set forth on<u>Schedule 7.1;</u> and

(p) Unsecured Debt owing under the Fortress Note Documents; provided that prior to Borrower incurring any such Debt, Borrower shall cause such Debt to be subject to a Subordination Agreement.

7.2 <u>Liens.</u>

Not, and not permit any other Loan Party to, create or permit to exist any Lien on any of its real or personal properties, assets or rights of whatsoever nature (whether now owned or hereafter acquired), except:

(a) Liens for taxes or other governmental charges not at the time delinquent or thereafter payable without penalty, or being diligently contested in good faith by appropriate proceedings and for which it maintains adequate reserves in accordance with GAAP;

(b) Liens arising in the ordinary course of business (including without limitation (i) Liens of carriers, warehousemen, mechanics, landlords and materialmen and other similar Liens imposed by law and (ii) Liens incurred in connection with worker's compensation, unemployment compensation and other types of social security or in connection with surety bonds, bids, tenders, performance bonds, trade contracts not for borrowed money, licenses, statutory obligations and similar obligations) for sums not overdue or being diligently contested in good faith by appropriate proceedings and not involving any deposits or advances or borrowed money or the deferred purchase price of property or services and, in each case, for which it maintains adequate reserves in accordance with GAAP and with respect to which no execution or other enforcement of which is effectively stayed;

(c) (i) Liens arising in connection with Capital Leases (and attaching only to the property being leased and the proceeds thereof), (ii) Liens on any property securing debt incurred for the purpose of financing all or any part of the cost of acquiring or improving such property; *provided* that any such Lien attaches to such property within ninety (90) days of the acquisition or improvement thereof and attaches solely to the property so acquired or improved and the proceeds thereof, and (iii) the replacement, extension or renewal of a Lien permitted by one of the foregoing <u>clauses (i)</u> or (<u>ii)</u> in the same property subject thereto arising out of the extension, renewal or replacement of the Debt secured thereby (without increase in the amount thereof);

- 44 -

(d) Liens relating to litigation bonds and attachments, appeal bonds, judgments and other similar Liens arising in connection with any judgment or award that is not an Event of Default hereunder;

(e) easements, rights of way, restrictions, minor defects or irregularities in title and other similar Liens not interfering in any material respect with the ordinary conduct of the business of Borrower or any other Loan Party;

(f) Liens arising under the Loan Documents;

(g) any interest or title of a licensor, sublicensor, lessor or sublessor under any license, lease, sublicense or sublease agreement entered into in the normal course of business, only to the extent limited to the item licensed or leased;

(h) (i) Liens of a collection bank arising under Section 4-210 of the Uniform Commercial Code on items in the course of collection and (ii) customary set off rights of deposit banks with respect to deposit accounts maintained at such deposit banks or which are contained in standard agreements for the opening of an account with a bank;

(i) Liens arising from precautionary filings of financing statements under the Uniform Commercial Code or similar legislation of any applicable jurisdiction in respect of operating leases permitted hereunder and entered into by a Loan Party in the ordinary course of business;

(j) Liens attaching to cash earnest money deposits in connection with any letter of intent or purchase agreement permitted hereunder or indemnification other post-closing escrows or holdbacks;

(k) Liens incurred with respect to Hedging Obligations incurred for bona fide hedging purposes and not for speculation;

(l) Liens to secure obligations of a Loan Party to another Loan Party;

(m) Liens arising out of conditional sale, title retention, consignment or similar arrangements for the sale of goods in the ordinary course of business;

(n) Licenses permitted by <u>Section 7.4(c)</u>;

(o) Liens on insurance proceeds securing the payment of financed insurance premiums that are promptly paid on or before the date they become due (provided that such Liens extend only to such insurance proceeds and not to any other property or assets);

(p) Utility, lease, contract and similar deposits in the ordinary course of business;

(q) Liens in favor of customs and revenue authorities arising as a matter of law to secure payment of custom duties that are promptly paid on or before the date they become due;

(r) Liens on Subordinated Debt to the extent permitted by the Subordination Agreement applicable thereto; and

(s) Liens existing on the Closing Date and set forth on<u>Schedule 7.2</u>.

7.3 Dividends; Redemption of Equity Interests.

Not (a) declare, pay or make any dividend or distribution on any Equity Interests or other securities or ownership interests, other than dividends or distributions declared, paid or made to a Loan Party or in the form of Equity Interests, (b) apply any of its funds, property or assets to the acquisition, redemption or other retirement of any Equity Interests or other securities or interests or of any options to purchase or acquire any of the foregoing, (c) otherwise make any payments, dividends or distributions to any member, manager, managing member, stockholder, shareholder, director or other equity owner in such Person's capacity as such other than in compliance with Section 7.7 hereof, or (d) make any payment of any management, service or related or similar fee to any Affiliate or holder of Equity Interests of Borrower other than in compliance with Section 7.7 hereof.

7.4 Mergers; Consolidations; Asset Sales.

(a) Except for Permitted Acquisitions, not be a party to any amalgamation or any other form of Division, demerger, merger or consolidation, unless agreed to by Agent in its sole discretion, nor permit any other Loan Party to be a party to any Division, demerger, amalgamation or any other form of merger or consolidation, unless agreed to by Agent in its reasonable discretion; provided that any Subsidiary may merge with and into a Loan party.

(b) Not, and not permit any other Loan Party to, sell, transfer, dispose of, convey, lease or license any of its real or personal property assets or Equity Interests (each, a "**Disposition**"), except for (i) sales of Inventory in the ordinary course of business for at least fair market value, (ii) transfers, destruction or other disposition of obsolete, surplus or worn-out assets in the ordinary course of business and (iii) at all times subject to <u>Section 2.8.1</u>, any other sales and dispositions of assets (excluding (A) any Equity Interests of Borrower or any Subsidiary or (B) sales of Inventory described in <u>clause (i)</u> above) for at least fair market value (as determined by the Board, (iv) sales and dispositions to Loan Parties, (v) leases, subleases and sublicenses entered into in the ordinary course of business, including licensing transactions permitted by <u>Section 7.4(c)</u>, (vi) sales and exchanges of Cash Equivalent Investments to the extent otherwise permitted hereunder, (vii) Liens expressly permitted under <u>Section 7.2</u> and transactions expressly permitted by <u>clause (a)</u> or <u>Section 7.10</u>, (viii) sales or issuances of Equity Interests by Borrower, (ix) issuances of Equity Interests by any Loan Party to any other Loan Party, (x) dispositions in the ordinary course of business consisting of the abandonment of intellectual property rights which, in the reasonable good faith determination of Borrower, are not material to the conduct of the business of the Loan Parties, (xi) a cancellation of any intercompany Debt among the Loan Parties, (xii) a disposition which constitutes an insured event or pursuant to a condemnation, expropriation, "eminent domain" or similar proceeding, (xiii) sales and dispositions which constitutes an insured event or pursuant to a condemnation, expropriation, "eminent domain" or similar proceeding, (xiii) sales and dispositions anong Subsidiaries of Borrower, (xiv) exchanges of existing equipment for new equipment that is substantially similar to the equipment being exchanged an

(c) Notwithstanding any provision in this Agreement or any other Loan Documents to the contrary, the prior consent of Agent shall not be required in connection with the licensing or sublicensing (whether in-licensing or out-licensing) of Intellectual Property pursuant to collaborations, licenses or other strategic transactions with third parties executed (i) in the ordinary course of a Loan Party's business, (ii) on an arms-length basis and (iii) as long as no Event of Default has occurred and is continuing.

7.5 Modification of Organizational Documents.

Not permit the charter, by-laws or other organizational documents or constitutional documents of Borrower or any other Loan Party to be amended or modified in any way which could reasonably be expected to materially and adversely affect the interests of Agent or any Lender. An amendment to Borrower's certificate of incorporation to increase Borrower's authorized share capital shall not be deemed to adversely affect the interests of Agent or any Lender.

- 46 -

7.6 <u>Use of Proceeds.</u>

Use the proceeds of the Loans solely to refinance the Prior Debt, if any, and otherwise for working capital, for fees and expenses related to the negotiation, execution, delivery and closing of this Agreement and the other Loan Documents and the transactions contemplated hereby and thereby, and for other general business purposes of Borrower and its Subsidiaries, and not use any proceeds of any Loan or permit any proceeds of any Loan to be used, either directly or indirectly, for the purpose, whether immediate, incidental or ultimate, of "purchasing or carrying" any Margin Stock.

7.7 Transactions with Affiliates.

Not, and not permit any other Loan Party to, enter into, or cause, suffer or permit to exist any transaction, arrangement or contract with any of its other Affiliates, which is on terms which are less favorable than are obtainable from any Person which is not one of its Affiliates, other than (i) reasonable compensation and indemnities to, benefits for, reimbursement of expenses of, and employment arrangements with, officers, employees and directors in the ordinary course of business, (ii) transactions among Loan Parties, (iii) transactions pursuant to agreements in existence on the Closing Date and set forth on <u>Schedule 7.7</u>, (iv) the Shared Services Agreement with Fortress Biotech, Inc., dated as of November 12, 2021, (v) Investments permitted by <u>Section 7.10</u> and transactions permitted by <u>Section 7.3</u>, (vi) the Subordinated Debt in existence as of the Closing Date, and (vii) any tax sharing arrangements entered into in the ordinary course of business.

7.8 Inconsistent Agreements.

Not, and not permit any other Loan Party to, enter into any agreement containing any provision which would (a) be violated or breached by any borrowing by Borrower hereunder or by the performance by Borrower or any other Loan Party of any of its Obligations hereunder or under any other Loan Document, (b) prohibit Borrower or any other Loan Party from granting to Agent and Lenders a Lien on the Collateral or (c) create or permit to exist or become effective any encumbrance or restriction on the ability of any other Loan Party to (i) pay dividends or make other distributions to Borrower or any other Loan Party, or pay any Debt owed to Borrower or any other Loan Party, (ii) make loans or advances to Borrower or any other Loan Party or (iii) transfer any of its assets or properties to Borrower or any other Loan Party, other than, in the cases of <u>clauses (b)</u> and (c). (A) restrictions or conditions imposed by any agreement relating to purchase money Debt, Capital Leases and other secured Debt or to leases and licenses permitted by this Agreement if such restrictions or conditions apply only to the property or assets securing such Debt or the property leased or licensed, (B) customary provisions in leases, licenses and other contracts restricting the assignment thereof, (C) restrictions and conditions imposed by law, (D) those arising under any Loan Document or any loan documents governing any Subordinated Debt, and (E) customary provisions in contracts for the disposition of any assets; *provided* that the restrictions in any such contract shall apply only to the assets or Subsidiary that is to be disposed of and such disposition is permitted hereunder.

7.9 Business Activities.

Not, and not permit any other Loan Party to, engage in any line of business other than the businesses engaged in on the Closing Date and businesses reasonably related, ancillary or supplemental thereto or extensions thereof.

7.10 <u>Investments.</u>

Not, and not permit any other Loan Party to, make or permit to exist any Investment in any other Person, except the following:

- 47 -

(a) The creation of any Wholly-Owned Subsidiary and contributions by Borrower to the capital of any Wholly-Owned Subsidiary of Borrower, so long as the recipient of any such contribution has guaranteed the Obligations and such guaranty is secured by a pledge of all of its equity interests and substantially all of its real and personal property, in each case in accordance with Section 6.14;

- (b) Cash Equivalent Investments;
- (c) bank deposits in the ordinary course of business;

(d) any purchase or other acquisition by Borrower or any Wholly-Owned Subsidiary of Borrower of the assets or equity interests of any Subsidiary of Borrower;

- (e) transactions among Loan Parties;
- (f) Hedging Obligations permitted under <u>Section 7.1(d)</u>;
- (g) lease, utility and other similar deposits made in the ordinary course of business and trade credit extended in the ordinary

course of business;

hereunder;

(h) Investments consisting of the non-cash portion of the consideration received in respect of Dispositions permitted

(i) Investments permitted by Borrower or any Loan Party as a result of the receipt of insurance and/or condemnation or expropriation proceeds in accordance with the Loan Documents;

(j) Investments (i) received as a result of the bankruptcy or reorganization of any Person or taken in settlement of or other resolution of claims or disputes or (ii) in securities of customers and suppliers received in connection with the bankruptcy or reorganization of, or settlement of delinquent accounts and bona fide disputes with, customers and suppliers, and, in each case, extensions, modifications and renewals thereof;

- (k) Permitted Acquisitions;
- (l) licensing transactions permitted by <u>Section 7.4(c)</u>;

(m) Investments held by any Person as of the date such Person is acquired in connection with a Permitted Acquisition, provided that such Investments were not made, in any case, by such Person in connection with, or in contemplation of, such Permitted Acquisition;

(n) Investments received in connection with dispositions permitted by Section 7.4;

(o) Investments consisting of notes receivable of, or prepaid royalties and other credit extensions, to customers and suppliers who are not Affiliates, in the ordinary course of business;

(p) Investments consisting of travel advances in the ordinary course of business;

(q) joint ventures, strategic alliances, collaboration arrangements or non-exclusive licensing arrangements in the ordinary course of a Borrower's business consisting of the nonexclusive licensing of technology, the development of technology or the providing of technical support;

(r) Investments consisting of loans not involving the net transfer on a substantially contemporaneous basis of cash proceeds to employees, officers or directors relating to the

- 48 -

purchase of Equity Interests of a Borrower pursuant to employee stock purchase plans or other similar agreements approved by such Borrower's Board of Directors; and

(s) Investments existing on the Closing Date and set forth on<u>Schedule 7.10</u>.

7.11 Restriction of Amendments to Certain Documents.

Not, nor permit any Loan Party to, amend or otherwise modify in any material manner, or waive any material rights under, any provisions of any of (i) any loan documents governing any Subordinated Debt (except that the terms of any document governing any Subordinated Debt be amended, modified or otherwise waived to the extent permitted under the applicable subordination agreement that Agent is a party to in connection therewith), or (ii) any Material Contracts (or any replacements thereof) following the occurrence and continuance of an Event of Default; in either case without the written approval of Agent.

7.12 Fiscal Year.

Not change its Fiscal Year.

- 7.13 Financial Covenants.
 - 7.13.1 Minimum Consolidated Unencumbered Liquid Assets.

(a) Not permit the Consolidated Unencumbered Liquid Assets, as determined on the last day of each calendar month, to be less than the greater of (i) \$2,000,000, or (ii) the Operating Burn for the most recently-completed Fiscal Quarter.

Notwithstanding anything to the contrary contained in this Agreement, in the event that Borrower fails, as determined (b) on the date of receipt by Agent of the interim reports to be delivered pursuant to Section 6.1.2(d) (the "Liquidity Default Date"), to comply with the financial covenant set forth in Section 7.13.1(a) above (a "Liquidity Covenant Default"), Borrower shall have the right to effect a "cure" of such Liquidity Covenant Default (the "Liquidity Cure Right"), subject to the terms and conditions of this Section 7.13.1(b). So long as no other Event of Default has occurred and is continuing, Borrower may exercise the Liquidity Cure Right by (i) notifying Agent in writing of its intent to exercise its Liquidity Cure Right within five (5) Business Days of such Liquidity Default Date (the "Liquidity Cure Right Notice") and (ii) providing evidence, acceptable to Agent in its commercially-reasonable discretion, that Borrower has received net cash proceeds in an aggregate amount that is equal to or greater than the amount required to bring Borrower into compliance with the covenant set forth in Section 7.13.1(a) above (the "Liquidity Cure Amount") within thirty (30) Business Days after delivery of the Liquidity Cure Right Notice (each such period, the "Liquidity Cure Right Exercise Period") pursuant to (A) the issuance by Borrower of Subordinated Debt, on terms and conditions satisfactory to Agent in its commercially-reasonable discretion, (B) the issuance by Borrower of additional Equity Interests, on terms and conditions satisfactory to Agent in its commercially-reasonable discretion, (C) the receipt by Loan Parties of cash flow from operations and/or cash realized on Investments held by Loan Parties or (D) some combination of (A)-(C). Upon Borrower's satisfaction of the requirements set forth in the prior sentence prior to the expiration of the Liquidity Cure Right Exercise Period, the Liquidity Covenant Default shall be deemed cured and no longer continuing. For the avoidance of doubt, Agent and Lenders shall automatically be deemed to reserve all rights and remedies available to them during the occurrence and continuance of an Event of Default (including the right to charge interest on the Obligations at the Default Rate from the Liquidity Default Date) during any Liquidity Cure Right Exercise Period. For the avoidance of doubt, any amounts received during a Liquidity Cure Right Exercise Period in connection with a Revenue Cure Right (as defined below) under Section 7.13.2(b) may be included in determining whether the Liquidity Cure Amount was received under this Section 7.13.1(b) during such Liquidity Cure Right Exercise Period.

. 49 .
. 49 .

7.13.2 Minimum Total Revenue.

(a) Not permit the Total Revenue of the Loan Parties (on a consolidated basis) for the consecutive twelve (12) month period ending on the last Business Day of any Fiscal Quarter set forth in the table below (designated by "Q" in the table below) to be less than the applicable amount set forth in the table below for such period.

Minimum Total Revenue as of the end of:		
Q4 2023	\$49,500,000	
Q1 2024	\$51,750,000	
Q2 2024	\$53,000,000	
Q3 2024	\$54,000,000	
Q4 2024 and each Fiscal Quarter thereafter	\$55,000,000	

Notwithstanding anything to the contrary contained in the Agreement, in the event that Borrower fails, as determined (b) on the date of receipt by Agent of the interim reports to be delivered pursuant to Section 6.1.2(a) (the "Revenue Default Date"), to comply with the financial covenant set forth in Section 7.13.2(a) above (a "Revenue Covenant Default"), Borrower shall have the right to effect a "cure" of such Revenue Covenant Default (the "Revenue Cure Right"), subject to the terms and conditions of this Section 7.13.2(b). So long as no other Event of Default has occurred and is continuing, Borrower may exercise the Revenue Cure Right by (i) notifying Agent in writing of its intent to exercise its Revenue Cure Right within five (5) Business Days of the Revenue Default Date (the "Revenue Cure Right Notice") and (ii) providing evidence, acceptable to Agent in its commercially-reasonable discretion, that Borrower has received net cash proceeds in an aggregate amount that is equal to or greater than one hundred percent (100%) of the amount by which Borrower's actual Total Revenue for the applicable reporting period was less than the minimum Total Revenue required pursuant to Section 7.13.2(a) above (the "Revenue Cure Amount") within thirty (30) Business Days after delivery of the Revenue Cure Right Notice (each such period, the "Revenue Cure Right Exercise Period") pursuant to the issuance by Borrower of (A) Subordinated Debt, (B) additional Equity Interests or (C) some combination of (A) and (B), in each case on terms and conditions satisfactory to Agent in its commercially-reasonable discretion. Notwithstanding the forgoing, Borrower shall be permitted to exercise the Revenue Cure Right (i) on no more than two (2) occasions during any period of twelve (12) consecutive months, and (ii) a maximum of three (3) occasions during the term of the Loan. Upon Borrower's satisfaction of the requirements set forth in the prior sentence prior to the expiration of the Revenue Cure Right Exercise Period, the Revenue Covenant Default shall be deemed cured and no longer continuing. For the avoidance of doubt, Agent and Lenders shall automatically be deemed to reserve all rights and remedies available to them during the occurrence and continuance of an Event of Default (including the right to charge interest on the Obligations at the Default Rate from the Liquidity Default Date) during any Revenue Cure Right Exercise Period. Upon any "cure" of a Revenue Covenant Default in accordance with this Section 7.13.2(b), the Revenue Cure Amount shall be deemed to be included in the Total Revenue as of the last Business Day of the Fiscal Quarter giving rise to such Revenue Covenant Default for purposes of calculating Borrower's Total Revenue for subsequent Fiscal Quarters where the Fiscal Quarter giving Rise to such Revenue Covenant Default would be included in such calculation. For the avoidance of doubt, any amounts received during a Revenue Cure Right Exercise Period in connection with any Liquidity Cure Right under Section 7.13.1(b) may be included in determining whether the Revenue Cure Amount was received under this Section 7.13.2(b) during such Revenue Cure Right Exercise Period, provided that any amounts received in connection with any Liquidity Cure Right during any period other

- 50 -

than a Revenue Cure Right Exercise Period shall not be deemed to be included in calculating Borrower's Total Revenue for any period.

7.13.3 Treatment of Amounts Raised In Connection With Cure Rights.

For the avoidance of doubt, any amounts received by Borrower in connection with the issuance of Subordinated Debt and/or Equity Interests pursuant to Section 7.13.1(b) and Section 7.13.2(b) shall be excluded from the calculation of Total Revenue and the Revenue-Based Payment Amount.

7.14 Deposit Accounts.

Not, and not permit any other Loan Party, to maintain or establish any new Deposit Accounts other than (a) Exempt Accounts and (b) the Deposit Accounts set forth on <u>Schedule 7.14</u> (which Deposit Accounts constitute all of the Deposit Accounts, securities accounts or other similar accounts maintained by the Loan Parties as of the Closing Date) without prior written notice to Agent. Upon the request of Agent at any time following the occurrence of a Material Adverse Effect, Default or Event of Default, Borrower or such other applicable Loan Party shall promptly enter into an Account Control Agreement, in form and substance reasonably satisfactory to Agent, in relation to the Deposit Account(s) selected by Agent.

7.15 Subsidiaries.

Not, and not permit any other Loan Party to, in each case without the prior written consent of Agent in its sole discretion, establish or acquire any Subsidiary unless (i) no Default or Event of Default has occurred and is continuing or would result therefrom, (ii) such Subsidiary shall have assumed and joined each Loan Document as a Loan Party pursuant to documentation acceptable to Agent in its sole discretion and (iii) all other Loan Parties shall have reaffirmed all Obligations as well as all representations and warranties under the Loan Documents (except to the extent such representations and warranties specifically relate to a prior date only).

7.16 <u>Regulatory Matters.</u>

Not, and not permit any other Loan Party to, (i) make, and use commercially reasonable efforts to not permit any officer, employee or agent of any Loan Party to make, any untrue statement of material fact or fraudulent statement to the FDA or any Governmental Authority; fail to disclose a material fact required to be disclosed to the FDA or any Governmental Authority; or commit a material act, make a material statement, or fail to make a statement in breach of CLIA or that could otherwise reasonably be expected to provide the basis for CMS or any Governmental Authority to undertake action against such Loan Party, (ii) introduce into commercial distribution any FDA Products which are, upon their shipment, adulterated or misbranded in violation of 21 U.S.C. § 331, (iii) make, and use commercially reasonable efforts to not permit any officer, employee or agent of any Loan Party to make, any untrue statement to the FDA or any other Governmental Authority; fail to disclose a material fact required to be disclosed to the FDA or any other Governmental Authority; fail to make a statement in breach of the FDA or any the governmental to the basis for the FDA or any other Governmental fact required to be disclose a material fact or fraudulent statement to the FDA or any other Governmental Authority; fail to make a statement in breach of the FD& C Act or that could otherwise reasonably be expected to provide the basis for the FDA or any other Governmental Authority to invoke its policy respecting "Fraud, Untrue Statements of Material Facts, Bribery, and Illegal Gratuities," as set forth in 56 Fed. Reg. 46191 (September 10, 1991), or (iv) otherwise incur any material liability (whether actual or contingent) for failure to comply with Health Care Laws.

7.17 Name: Permits: Dissolution: Insurance Policies: Disposition of Collateral: Taxes: Trade Names: Location of Assets: Change of Chief Executive Office.

Borrower shall not, nor shall it permit any Loan Party to, (a) change its jurisdiction of organization, change the jurisdiction in which its chief executive office is located or change its corporate



name without thirty (30) calendar days prior written notice to Agent, (b) amend, alter, suspend, terminate or make provisional in any material way, any Permit, the suspension, amendment, alteration or termination of which would reasonably be expected to be, have or result in a Material Adverse Effect without the prior written consent of Agent, which consent shall not be unreasonably withheld, (c) wind up, liquidate or dissolve (voluntarily or involuntarily) or commence or suffer any proceedings seeking or that would result in any of the foregoing, (d) amend, modify, restate or change any insurance policy in a manner adverse to Agent or Lenders or otherwise allow its aggregate products liability insurance coverage to be less than an amount that is commercially reasonable and consistent with customary industry practices, (e) change its federal tax employer identification number or similar tax identification number under the relevant jurisdiction or establish new or additional trade names without providing not less than thirty (30) days advance written notice to Agent, (f) revoke, alter or amend any Tax Information Authorization (on IRS Form 8821 or otherwise) or other similar authorization mandered by the relevant Governmental Authority given to any Lender, or (g) permit any of its material tangible personal property to be located in or relocated to any jurisdiction in which Agent has not registered or perfected its security interest without thirty (30) calendar days prior written notice to Agent.

7.18 <u>Truth of Statements.</u>

Borrower shall not knowingly furnish to Agent or any Lender any certificate or other document that contains any untrue statement of a material fact or that omits to state a material fact necessary to make it not misleading in light of the circumstances under which it was furnished.

Section 8. Events of Default; Remedies.

8.1 Events of Default.

Each of the following shall constitute an Event of Default under this Agreement:

8.1.1 Non-Payment of Credit.

(a) Default in the payment when due of all outstanding Obligations on the Termination Date; (b) default in the payment of any Revenue-Based Payment Amount on or before the applicable Payment Date; or (c) without duplication of clause (b) hereof, default, and continuance thereof for five (5) Business Days, in the payment when due of any interest, fee, or other amount payable by any Loan Party hereunder or under any other Loan Document.

8.1.2 Default Under Other Debt.

Any "Event of Default" (or such similar defined term) shall occur under the terms applicable to any Debt of any Loan Party (excluding the Obligations) in an aggregate principal amount (for all such Debt so affected and including undrawn committed or available amounts and amounts owing to all creditors under any combined or syndicated credit arrangement) exceeding \$500,000.

8.1.3 Bankruptcy; Insolvency.

(a) Any Loan Party shall (i) be unable to pay its debts generally as they become due, (ii) file a petition under any insolvency statute, (iii) make a general assignment for the benefit of its creditors, (iv) commence a proceeding for the appointment of a receiver, trustee, interim receiver, receiver and manager, liquidator or conservator of itself or of the whole or any substantial part of its property or shall otherwise be dissolved or liquidated, or (v) make an application or commence a proceeding seeking reorganization or liquidation or similar relief under any Debtor Relief Law or any other applicable law; or

(b) (i) a court of competent jurisdiction shall (A) enter an order, judgment or decree appointing a custodian, receiver, trustee, , interim receiver, receiver and manager, liquidator or conservator of any Loan Party or the whole or any substantial part of any of Loan Party's properties, which shall continue unstayed and in effect for a period of sixty (60) calendar days, (B) approve a petition or claim filed against any Loan Party seeking reorganization, liquidation, appointment of a receiver, interim receiver, liquidator, conservator, trustee or special manager or similar relief under the any Debtor Relief Law or any other applicable law, which is not dismissed within sixty (60) calendar days or, (C) under the provisions of any Debtor Relief Law or other applicable law or statute, assume custody or control of any Loan Party or of the whole or any substantial part of any of Loan Party's properties, which is not irrevocably relinquished within sixty (60) calendar days, or (ii) there is commenced against any Loan Party any proceeding or petition seeking reorganization, liquidation or similar relief under any Debtor Relief Law or any other applicable law or statute, which (A) is not unconditionally dismissed within sixty (60) calendar days after the date of commencement, or (B) is with respect to which Borrower takes any action to indicate its approval of or consent.

8.1.4 Non-Compliance with Loan Documents.

(a) Any failure by Borrower to comply with or to perform any covenant set forth in<u>Section 7</u>; or (b) failure by any Loan Party to comply with or to perform any other provision of this Agreement or any other Loan Document applicable to it (and not constituting an Event of Default under any other provision of this <u>Section 8</u>) and continuance of such failure described in this <u>clause (b)</u> for thirty (30) days after the earlier of any Loan Party becoming aware of such failure or notice thereof to Borrower from Agent or any Lender.

8.1.5 <u>Representations; Warranties.</u>

Any representation or warranty made by any Loan Party herein or any other Loan Document is false or misleading in any material respect when made, or any schedule, certificate, financial statement, report, notice or other writing furnished by any Loan Party to Agent or any Lender in connection herewith is false or misleading in any material respect on the date as of which the facts therein set forth are stated or certified and, if such representation or warranty is capable of being cured, remains incorrect for a period of 30 days after the making of such representation or warranty.

8.1.6 Pension Plans.

(a) Institution of any steps by any Person to terminate a Pension Plan if as a result of such termination any Loan Party or any member of the Controlled Group could be required to make a contribution to such Pension Plan, or could incur a liability or obligation to such Pension Plan, in excess of \$500,000; (b) a contribution failure occurs with respect to any Pension Plan sufficient to give rise to a Lien under Section 303(k) of ERISA securing obligations in excess of \$500,000; or (c) there shall occur any withdrawal or partial withdrawal from a Multiemployer Pension Plan and the withdrawal liability (without un-accrued interest) to Multiemployer Pension Plans as a result of such withdrawal (including any outstanding withdrawal liability that Borrower or any other Loan Party or any member of the Controlled Group have incurred on the date of such withdrawal) exceeds \$500,000.

8.1.7 Judgments.

Final judgments which exceed an aggregate of \$500,000 (to the extent not adequately covered by insurance as to which the insurance company has not disclaimed liability (provided that customary "reservation of rights" letters shall not be deemed to be disclaimers of liability)) shall be rendered against any Loan Party and shall not have been paid, discharged or vacated or had execution thereof stayed pending appeal within thirty (30) calendar days after entry or filing of such judgments.

- 53 -

8.1.8 Invalidity of Loan Documents or Liens.

(a) Any Loan Document shall cease to be in full force and effect otherwise in accordance with its express terms that results in a material diminution of the rights and remedies afforded to Agent and/or Lenders or any other secured parties thereunder; (b) any Loan Party (or any Person by, through or on behalf of any Loan Party) shall contest in any manner the validity, binding nature or enforceability of any Loan Document; or (c) any Lien created pursuant to any Loan Document ceases to constitute a valid first priority perfected Lien (subject to Permitted Liens) on any material portion of the Collateral in accordance with the terms thereof, or Agent ceases to have a valid perfected first priority security interest (subject to Permitted Liens) in any material portion of the Collateral pledged to Agent, for the benefit of Agent and Lenders, pursuant to the Collateral Documents.

8.1.9 Invalidity of Subordination Provisions.

Any subordination provision in any document or instrument governing any Subordinated Debt and any subordination provision in any intercreditor agreement or Subordination Agreement in relation thereto shall cease to be in full force and effect, or any Loan Party shall contest in any manner the validity, binding nature or enforceability of any such provision.

8.1.10 Change of Control.

A Change of Control shall occur that does not result in the Payment In Full in accordance with Section 2.8.

8.1.11 Certificate Withdrawals, Adverse Test or Audit Results, and Other Matters.

(a) The institution of any proceeding by FDA, CMS, or any other Governmental Authority to order the withdrawal of any Product or Product category or Service or Service category from the market or to enjoin Borrower or any of its Subsidiaries from manufacturing, marketing, selling, distributing, or otherwise providing any Product or Product category or Service or Service category that would reasonably be expected to have a Material Adverse Effect, (b) the institution of any action or proceeding by DEA, FDA, CMS, or any other Governmental Authority to revoke, suspend, reject, withdraw, limit, or restrict any Required Permit held by Borrower or any of its Subsidiaries or any of their representatives, which, in each case, would reasonably be expected to have a Material Adverse Effect, (c) the commencement of any enforcement action against Borrower or any of its Subsidiaries by DEA, FDA, CMS, or any other Governmental Authority the recall of any Products or Service from the market, the voluntary withdrawal of any Products or Service from the market, the voluntary withdrawal of any Products or Service of adverse test, audit, or inspection results in connection with a Product or Service which would reasonably be expected to have a Material Adverse Effect, (e) the occurrence of any event described in clauses (a) through (e) above that would otherwise cause Borrower to be excluded from participating in any federal, provincial, state or local health care programs under Section 1128 of the Social Security Act or any similar law or regulation.

8.1.12 Material Adverse Effect.

Any Material Adverse Effect shall occur that is not otherwise provided for in this Section 8.1.

8.2 <u>Remedies.</u>

(a) If any Event of Default described in <u>Section 8.1.3</u> shall occur, the Loan and all other Obligations shall become immediately due and payable without presentment, demand, protest or

notice of any kind; and, if any other Event of Default shall occur and be continuing, Agent may, and upon the written request of Required Lenders shall, declare all or any part of the Loans and other Obligations to be due and payable, whereupon the Loans and other Obligations (including without limitation the Exit Fee and any amounts due pursuant to <u>Section 2.8</u> hereof, payable with respect thereto) shall become immediately due and payable (in whole or in part, as applicable), all without presentment, demand, protest or notice of any kind. Agent shall use commercially reasonable efforts to promptly advise Borrower of any such declaration, but failure to do so shall not impair the effect of such declaration.

(b) In addition to the acceleration provisions set forth in Section 8.2(a) above, upon the occurrence and continuation of an Event of Default, Agent may (or shall at the request of Required Lenders) exercise any and all rights, options and remedies provided for in any Loan Document, under the Uniform Commercial Code, any other applicable foreign or domestic laws or otherwise at law or in equity, including, without limitation, the right to (i) apply any property of Borrower held by Agent to reduce the Obligations, (ii) foreclose the Liens created under the Loan Documents, (iii) realize upon, take possession of and/or sell any Collateral or securities pledged, with or without judicial process, (iv) exercise all rights and powers with respect to the Collateral as Borrower might exercise, (v) collect and send notices regarding the Collateral, with or without judicial process, (vi) by its own means or with judicial assistance, enter any premises at which Collateral and/or pledged securities, or other sums, and Borrower shall not resist or interfere with such action, (vii) at Borrower's expense, require that all or any part of the Collateral be assembled and made available to Agent, for the benefit of Lenders, or Required Lenders at any place reasonably designated by Agent, in its sole discretion, and/or relinquish or abandon any Collateral or securities pledged or any Lien thereon.

(c) The enumeration of any rights and remedies in any Loan Document is not intended to be exhaustive, and all rights and remedies of Agent and Lenders described in any Loan Document are cumulative and are not alternative to or exclusive of any other rights or remedies which Agent and Lenders otherwise may have. The partial or complete exercise of any right or remedy shall not preclude any other further exercise of such or any other right or remedy.

(d) Notwithstanding any provision of any Loan Document, Agent, in its sole discretion shall have the right, but not any obligation, at any time that Loan Parties fail to do so, subject to any applicable cure periods permitted by or otherwise set forth in the Loan Documents, and from time to time, without prior notice, to: (i) discharge (at Borrower's expense) taxes or Liens affecting any of the Collateral that have not been paid in violation of any Loan Document or that jeopardize Agent's Lien priority in the Collateral; or (ii) make any other payment (at Borrower's expense) for the administration, servicing, maintenance, preservation or protection of the Collateral (each such advance or payment set forth in clauses (i) and (ii) herein, a "Protective Advance"). Agent shall be reimbursed for all Protective Advances pursuant to <u>Section 2.9.1(b)</u> and/or <u>Section 2.10</u>, as applicable, and any Protective Advance is paid by Agent until it is repaid. No Protective Advance by Agent, shall be construed as a waiver by Agent, or any Lender of any Default, Event of Default or any of the rights or remedies of Agent or any Lender under any Loan Document.

Section 9. Agent.

9.1 Appointment; Authorization.

Each Lender hereby irrevocably appoints, designates and authorizes Agent to take such action on its behalf under the provisions of this Agreement and each other Loan Document and to exercise such powers and perform such duties as are expressly delegated to it by the terms of this Agreement or any other Loan Document, together with such powers as are reasonably incidental thereto. Notwithstanding any provision to the contrary contained elsewhere in this Agreement or in any other Loan Document, Agent shall not have any duty or responsibility except those expressly set forth herein, nor shall Agent have or be deemed

- 55 -

to have any fiduciary relationship with any Lender, and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or any other Loan Document or otherwise exist against Agent.

9.2 <u>Delegation of Duties.</u>

Agent may execute any of its duties under this Agreement or any other Loan Document by or through agents, employees or attorneys-infact and shall be entitled to advice of counsel concerning all matters pertaining to such duties. Agent shall not be responsible for the negligence or misconduct of any agent or attorney-in-fact that it selects with reasonable care.

9.3 Limited Liability.

None of Agent or any of its Affiliates, directors, officers, employees or agents shall (a) be liable for any action taken or omitted to be taken by any of them under or in connection with this Agreement or any other Loan Document or the transactions contemplated hereby (except to the extent resulting from its own gross negligence or willful misconduct as determined by a court of competent jurisdiction), or (b) be responsible in any manner to any Lender for any recital, statement, representation or warranty made by any Loan Party or Affiliate of any Loan Party, or any officer thereof, contained in this Agreement or in any other Loan Document, or the validity, effectiveness, genuineness, enforceability or sufficiency of this Agreement or any other Loan Document (or the creation, perfection or priority of any Lien or security interest therein), or for any failure of any Loan Party or any other party to any Loan Document to perform its Obligations hereunder or thereunder. Agent shall not be under any obligation to any Lender to ascertain or to inquire as to the observance or performance of any of the agreements contained in, or conditions of, this Agreement or any other Loan Document, or the agreements contained in, or conditions of, this Agreement or any other Loan Document to perform act of any Loan Party or Affiliate of any Loan Party.

9.4 <u>Reliance.</u>

Agent shall be entitled to rely, and shall be fully protected in relying, upon any writing, resolution, notice, consent, certificate, affidavit, letter, telegram, facsimile, telex or telephone message, statement or other document believed by it to be genuine and correct and to have been signed, sent or made by the proper Person or Persons, and upon advice and statements of legal counsel (including counsel to any Loan Party), independent accountants and other experts selected by Agent. Agent shall be fully justified in failing or refusing to take any action under this Agreement or any other Loan Document unless it shall first receive such advice or concurrence of Required Lenders (or all Lenders if expressly required hereunder) as it deems appropriate and, if it so requests, confirmation from Lenders of their obligation to indemnify Agent against any and all liability and expense which may be incurred by it by reason of taking or continuing to take any such action. Agent shall in all cases be fully protected in acting, or in refraining from acting, under this Agreement or any other Loan Document in accordance with a request or consent of Required Lenders (or all Lenders if expressly required hereunder) and such request and any action taken or failure to act pursuant thereto shall be binding upon each Lender.

9.5 <u>Notice of Default.</u>

Agent shall not be deemed to have knowledge or notice of the occurrence of any Event of Default or Default except with respect to defaults in the payment of principal, interest and fees required to be paid to Agent for the account of Lenders, unless Agent shall have received written notice from a Lender or Borrower referring to this Agreement, describing such Event of Default or Default and stating that such notice is a "notice of default". Agent will notify Lenders of its receipt of any such notice or any such default in the payment of principal, interest and fees required to be paid to Agent for the account of Lenders. Agent shall take such action with respect to such Event of Default or Default as may be requested by Required



Lenders in accordance with <u>Section 8.2</u>; provided that unless and until Agent has received any such request, Agent may (but shall not be obligated to) take such action, or refrain from taking such action, with respect to such Event of Default or Default as it shall deem advisable or in the best interest of Lenders.

9.6 Credit Decision.

Each Lender acknowledges that Agent has not made any representation or warranty to it, and that no act by Agent hereafter taken, including any review of the affairs of Borrower and the other Loan Parties, shall be deemed to constitute any representation or warranty by Agent to any Lender. Each Lender represents to Agent that it has, independently and without reliance upon Agent and based on such documents and information as it has deemed appropriate, made its own appraisal of and investigation into the business, prospects, operations, property, financial and other condition and creditworthiness of Borrower, and made its own decision to enter into this Agreement and to extend credit to Borrower hereunder. Each Lender also represents that it will, independently and without reliance upon Agent and based on such documents as it shall deem appropriate at the time, continue to make its own credit analysis, appraisals and decisions in taking or not taking action under this Agreement and the other Loan Documents, and to make such investigations as it deems necessary to inform itself as to the business, prospects, operations, property, financial and other condition and creditworthiness of the Loan Parties. Except for notices, reports and other documents expressly herein required to be furnished to Lenders by Agent, Agent shall not have any duty or responsibility to provide any Lender with any credit or other information concerning the business, prospects, operations, property, financial or other condition or creditworthiness of any Loan Party which may come into the possession of Agent.

9.7 Indemnification.

Whether or not the transactions contemplated hereby are consummated, each Lender shall indemnify upon demand Agent and its Affiliates, directors, officers, employees and agents (to the extent not reimbursed by or on behalf of Borrower and without limiting the obligation of Borrower to do so), based on such Lender's Pro Rata Term Loan Share, from and against any and all actions, causes of action, suits, losses, liabilities, damages and out-of-pocket expenses, including Legal Costs, except to the extent any thereof result from the applicable Person's own gross negligence or willful misconduct, as determined by a court of competent jurisdiction. Without limitation of the foregoing, each Lender shall reimburse Agent upon demand for its ratable share of any costs or out-of-pocket expenses (including Legal Costs) incurred by Agent in connection with the preparation, execution, delivery, administration, modification, amendment or enforcement (whether through negotiations, legal proceedings or otherwise) of, or legal advice in respect of rights or responsibilities under, this Agreement, any other Loan Document, or any document contemplated by or referred to herein, to the extent that Agent is not reimbursed for such expenses by or on behalf of Borrower. The undertaking in this <u>Section 9.7</u> shall survive repayment of the Loans, cancellation of the Notes, any foreclosure under, or modification, release or discharge of, any or all of the Collateral Documents, termination of this Agreement and the resignation or replacement of Agent.

9.8 Agent Individually.

SWK and its Affiliates may make loans to, issue letters of credit for the account of, accept deposits from, acquire equity interests in and generally engage in any kind of banking, trust, financial advisory, underwriting or other business with any Loan Party and any Affiliate of any Loan Party as though SWK were not Agent hereunder and without notice to or consent of any Lender. Each Lender acknowledges that, pursuant to such activities, SWK or its Affiliates may receive information regarding Loan Parties or their Affiliates (including information that may be subject to confidentiality obligations in favor of any such Loan Party or such Affiliates shall have the same rights and powers under this Agreement as any other Lender and may exercise the same as

- 57 -

though SWK were not Agent, and the terms "Lender" and "Lenders" include SWK and its Affiliates, to the extent applicable, in their individual capacities.

9.9 Successor Agent.

Agent may resign as Agent at any time upon 30 days' prior notice to Lenders and Borrower (unless during the existence of an Event of Default such notice is waived by Required Lenders). If Agent resigns under this Agreement, Required Lenders shall, with (so long as no Event of Default exists) the consent of Borrower (which shall not be unreasonably withheld or delayed), appoint from among Lenders a successor agent for Lenders. If no successor agent is appointed prior to the effective date of the resignation of Agent, Agent may appoint, on behalf of, and after consulting with Lenders and (so long as no Event of Default exists) Borrower, a successor agent. Upon the acceptance of its appointment as successor agent hereunder, such successor agent shall succeed to all the rights, powers and duties of the retiring Agent and the term "Agent" shall mean such successor agent, and the retriving Agent's resignation hereunder as Agent becomes effective, the provisions of this <u>Section 9</u> and <u>Sections 10.4</u> and <u>10.5</u> shall continue to inure to its benefit as to any actions taken or omitted to be taken by it while it was Agent under this Agreement. If no successor agent has accepted appointment as Agent by the date which is thirty (30) days following a retiring Agent's notice of resignation, the retiring Agent's resignation shall nevertheless thereupon become effective and Lenders shall perform all of the duties of Agent hereunder until such time, if any, as Required Lenders appoint a successor agent as provided for above; *provided* that in the case of any collateral security held by Agent for the benefit of Lenders under any of the Loan Documents, the retiring Agent shall continue to intre to its benefit so long as retiring Agent shall sections 9 and <u>Section 9</u> and <u>Sec</u>

9.10 Collateral and Guarantee Matters.

Lenders irrevocably authorize Agent, at its option and in its discretion, (a) to release any Lien granted to or held by Agent under any Collateral Document (i) when all Obligations have been Paid in Full; (ii) constituting property sold or to be sold or disposed of as part of or in connection with any sale or other disposition permitted hereunder (including by consent, waiver or amendment and it being agreed and understood that Agent may conclusively rely without further inquiry on a certificate of an officer of Borrower as to the sale or other disposition of property being made in compliance with this Agreement); or (iii) subject to Section 10.1, if approved, authorized or ratified in writing by Required Lenders; (b) notwithstanding Section 10.1(a) (iii) hereof to release any party from its guaranty under the Guarantee and Collateral Agreement (i) when all Obligations have been Paid in Full or (ii) if such party was sold or is to be sold or disposed of as part of or in connection with any disposition permitted hereunder (including by consent, waiver or amendment and it being agreed and understood that Agent may conclusively rely without further inquiry on a certificate of Borrower as to the sale or other disposition being made in compliance with this Agreement); or (c) to subordinate its interest in any Collateral to any holder of a Lien on such Collateral which is permitted by Section 7.2(d) (it being understood that Agent may conclusively rely on a certificate from Borrower in determining whether the Debt secured by any such Lien is permitted by Section 7.1). Upon request by Agent at any time, Lenders will confirm in writing Agent's authority to release, or subordinate its interest in, particular types or items of Collateral pursuant to this <u>Section 9.10</u>.

Agent shall release any Lien granted to or held by Agent under any Collateral Document (i) when all Obligations have been Paid in Full, (ii) in respect of property sold or to be sold or disposed of as part of or in connection with any sale or other disposition permitted hereunder (it being agreed and understood that Agent may conclusively rely without further inquiry on a certificate of an officer of Borrower as to the

- 58 -

sale or other disposition of property being made in compliance with this Agreement) or (iii) subject to Section 10.1, if directed to do so in writing by Required Lenders.

In furtherance of the foregoing, Agent agrees to execute and deliver to Borrower, at Borrower's expense, such termination and release documentation as Borrower may reasonably request to evidence a Lien release that occurs pursuant to terms of this <u>Section 9.10</u>.

9.11 Intercreditor and Subordination Agreements.

Each Lender hereby irrevocably appoints, designates and authorizes Agent to enter into one or more intercreditor agreements and/or Subordination Agreements in relation to any other Debt of Borrower entered into in accordance with this Agreement or as otherwise approved by Required Lenders, on its behalf and to take such action on its behalf under the provisions of any such agreement (subject to the last sentence of this <u>Section 9.11</u>). Each Lender further agrees to be bound by the terms and conditions of any such intercreditor agreement and Subordination Agreement. Each Lender hereby authorizes Agent to issue blockages notices in connection with any such Debt of Borrower and such intercreditor agreement and Subordination Agreement, or any replacement intercreditor agreement and/or Subordination Agreement, in its discretion or, at the direction of Required Lenders.

9.12 Actions in Concert.

For the sake of clarity, each Lender hereby agrees with each other Lender that no Lender shall take any action to protect or enforce its rights arising out of this Agreement, the Notes or any other Loan Document (including exercising any rights of set-off) without first obtaining the prior written consent of Agent and Required Lenders, it being the intent of Lenders that any such action to protect or enforce rights under this Agreement, the Notes and the other Loan Documents shall be taken in concert and at the direction or with the consent of Agent or Required Lenders.

Section 10. <u>Miscellaneous.</u>

10.1 Waiver; Amendments.

(a) Except as otherwise expressly provided in this Agreement, no amendment, modification or waiver of, or consent with respect to, any provision of this Agreement or any of the other Loan Documents shall in any event be effective unless the same shall be in writing and signed by Borrower (with respect to Loan Documents to which Borrower is a party), by Lenders having aggregate Pro Rata Term Loan Shares of not less than the aggregate Pro Rata Term Loan Shares expressly designated herein with respect thereto or, in the absence of such express designation herein, by Required Lenders, and then any such amendment, modification, waiver or consent shall be effective only in the specific instance and for the specific purpose for which given; *provided, however*, that:

(i) no such amendment, modification, waiver or consent shall, unless in writing and signed by all of the Lenders directly affected thereby, in addition to Required Lenders and Borrower, do any of the following: (A) increase any of the Commitments (*provided* that only the Lenders participating in any such increase of the Commitments shall be considered directly affected by such increase), (B) extend the date scheduled for payment of any principal of (except as otherwise expressly set forth below in <u>clause (C)</u>), or interest on, the Loans or any fees or other amounts payable hereunder or under the other Loan Documents, or (C) reduce the principal amount of any Loan, the amount or rate of interest thereon, or any fees or other amounts payable hereunder or under the other Loan Documents; and (ii) no such amendment, modification, waiver or consent shall, unless in writing and signed by all of the Lenders in addition to Borrower (with respect to Loan Documents to which Borrower is a party), do any of the following: (A) release any material guaranty under the Guarantee and Collateral Agreement or release all or substantially all of the

- 59 -

Collateral granted under the Collateral Documents, except as otherwise specifically provided in this Agreement or the other Loan Documents, (B) change the definition of Required Lenders, (C) change any provision of this <u>Section 10.1</u>, (D) amend the provisions of <u>Section 2.10.2</u> or <u>Section 2.10.4</u>, or (E) reduce the aggregate Pro Rata Term Loan Shares required to effect any amendment, modification, waiver or consent under the Loan Documents.

(b) No amendment, modification, waiver or consent shall, unless in writing and signed by Agent, in addition to Borrower and Required Lenders (or all Lenders directly affected thereby or all of the Lenders, as the case may be, in accordance with the provisions above), affect the rights, privileges, duties or obligations of Agent (including without limitation under the provisions of <u>Section 9</u>), under this Agreement or any other Loan Document.

(c) No delay on the part of Agent or any Lender in the exercise of any right, power or remedy shall operate as a waiver thereof, nor shall any single or partial exercise by any of them of any right, power or remedy preclude other or further exercise thereof, or the exercise of any other right, power or remedy.

10.2 Notices.

All notices hereunder shall be in writing (including via electronic mail) and shall be sent to the applicable party at such party's address set forth beneath its signature on the signature page to this Agreement or at such other address as such party may, by written notice received by the other parties, have designated as its address for such purpose. Notices sent by electronic mail transmission shall be deemed to have been given when sent if sent during regular business hours on a Business Day, otherwise, such deemed delivery will be effective as of the next Business Day; notices sent by mail shall be deemed to have been given five (5) Business Days after the date when sent by registered or certified mail, first class postage prepaid; and notices sent by hand delivery or overnight courier service shall be deemed to have been given when received. Borrower, Agent and Lenders each hereby acknowledge that, from time to time, Agent, Lenders and Borrower may deliver information and notices using electronic mail.

10.3 Computations.

Unless otherwise specifically provided herein, any accounting term used in this Agreement shall have the meaning customarily given such term in accordance with GAAP, and all financial computations hereunder shall be computed in accordance with GAAP consistently applied. The explicit qualification of terms or computations by the phrase "in accordance with GAAP" shall in no way be construed to limit the foregoing. Notwithstanding any other provision contained herein, all terms of an accounting or financial nature used herein shall be construed, and all computations of amounts and ratios referred to herein shall be made, without giving effect to any election under Statement of Financial Accounting Standards 159 (Codification of Accounting Standards 825-10) to value any Debt or other liabilities of any Loan Party at "fair value", as defined therein.

10.4 Costs; Expenses.

Borrower agrees to pay on demand the reasonable, out-of-pocket costs and expenses of (a) Agent (including Legal Costs) in connection with (i) the preparation, execution, syndication and delivery (including perfection and protection of Collateral) of this Agreement, the other Loan Documents and all other documents provided for herein or delivered or to be delivered hereunder or in connection herewith, (ii) the administration of the Loans and the Loan Documents, and (iii) any proposed or actual amendment, supplement or waiver to any Loan Document, and (b) Agent and Lenders (including Legal Costs) in connection with the collection of the Obligations and enforcement of this Agreement, the other Loan Documents or any such other documents. In addition, Borrower agrees to pay and to save Agent and Lenders harmless from all liability for, any fees of Borrower's auditors in connection with any reasonable exercise

- 60 -

by Agent and Lenders of their rights pursuant to and to the extent provided in <u>Section 6.2</u>. All Obligations provided for in this <u>Section 10.4</u> shall survive repayment of the Loans, cancellation of the Notes, and termination of this Agreement.

10.5 Indemnification by Borrower.

In consideration of the execution and delivery of this Agreement by Agent and Lenders and the agreement to extend the Commitments provided hereunder, Borrower hereby agrees to indemnify, exonerate and hold Agent, each Lender and each of the officers, directors, employees, Affiliates and agents of Agent and each Lender (each a "Lender Party") free and harmless from and against any and all actions, causes of action, suits, losses, liabilities, damages and expenses, including Legal Costs (collectively, the "Indemnified Liabilities"), incurred by Lender Parties or any of them as a result of, or arising out of, or relating to any Loan Party or any of their respective officers, directors or agents, including, without limitation, (a) any tender offer, merger, amalgamation, purchase of equity interests, purchase of assets or other similar transaction financed or proposed to be financed in whole or in part, directly or indirectly, with the proceeds of any of the Loans, (b) the use, handling, release, emission, discharge, transportation, storage, treatment or disposal of any Hazardous Substance at any property owned or leased by Borrower or any other Loan Party, (c) any violation of any applicable Environmental Laws with respect to conditions at any property owned or leased by any Loan Party or the operations conducted thereon, (d) the investigation, cleanup or remediation of offsite locations at which any Loan Party or their respective predecessors are alleged to have directly or indirectly disposed of Hazardous Substances, (e) the execution, delivery, performance or enforcement of this Agreement or any other Loan Document by any Lender Party, except to the extent any such Indemnified Liabilities result solely from the applicable Lender Party's own gross negligence or willful misconduct as finally determined by a court of competent jurisdiction in a non-appealable judgment, or (f) such Person's general operation of its business including all product liability out of or in connection with such Person's or any of its Affiliates or licensees manufacture use or sale of a Product or the provision of a Service. If and to the extent that the foregoing undertaking may be unenforceable for any reason, Borrower hereby agrees to make the maximum contribution to the payment and satisfaction of each of the Indemnified Liabilities which is permissible under applicable law. All Obligations provided for in this Section 10.5 shall survive repayment of the Loans, cancellation of the Notes, any foreclosure under, or any modification, release or discharge of, any or all of the Collateral Documents and termination of this Agreement. Notwithstanding the foregoing, this Section 10.5 shall not apply with respect to Taxes other than any Taxes that represent losses, claims, damages, etc. arising from any non-Tax claim.

10.6 Marshaling; Payments Set Aside.

Neither Agent nor any Lender shall be under any obligation to marshal any assets in favor of Borrower or any other Person or against or in payment of any or all of the Obligations. To the extent that Borrower makes a payment or payments to Agent or any Lender, or Agent or any Lender enforces its Liens or exercises its rights of set-off, and such payment or payments or the proceeds of such enforcement or set-off or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside or required (including pursuant to any settlement entered into by Agent or any Lender in its discretion) to be repaid to a trustee, receiver, interim receiver, receiver and manager, or any other party in connection with any bankruptcy, insolvency or similar proceeding, or otherwise, then (a) to the fullest extent permitted by applicable law, to the extent of such recovery, the obligation hereunder or part thereof originally intended to be satisfied shall be revived and continued in full force and effect as if such payment had not been made or such enforcement or set-off had not occurred and (b) each Lender severally agrees to pay to Agent upon demand its ratable share of the total amount so recovered from or repaid by Agent to the extent paid to such Lender.

10.7 Non-liability of Lenders.

- 61 -

The relationship between Borrower on the one hand and Lenders and Agent on the other hand shall be solely that of borrower and lender. Neither Agent nor any Lender shall have any fiduciary responsibility to Borrower. Neither Agent nor any Lender undertakes any responsibility to Borrower to review or inform Borrower of any matter in connection with any phase of Borrower's business or operations. To the fullest extent permitted under applicable law, execution of this Agreement by Borrower constitutes a full, complete and irrevocable release of any and all claims which Borrower may have at law or in equity in respect of all prior discussions and understandings, oral or written, relating to the subject matter of this Agreement and the other Loan Documents. Neither Agent nor any Lender shall have any liability with respect to, and Borrower hereby, to the fullest extent permitted under applicable law, waives, releases and agrees not to sue for, any special, indirect, punitive or consequential damages or liabilities.

10.8 Assignments.

10.8.1 Assignments.

(a) Any Lender may at any time assign to one or more Persons (other than a Loan Party and their respective Affiliates) (any such Person, an "Assignee") all or any portion of such Lender's Loans and Commitments, with the prior written consent of Agent, and, so long as no Event of Default has occurred and is continuing, Borrower (which consents shall not be unreasonably withheld or delayed), provided, however, that no such consent(s) shall be required:

(i) <u>from Borrower</u> for an assignment by a Lender to another Lender, an Affiliate of a Lender, an Approved Fund of a Lender, or any other financial institution that invests in commercial loans in the ordinary course of its business, but such Lender will give written notice to Borrower of any such assignment;

(ii) from Agent for an assignment by a Lender to an Affiliate of a Lender or an Approved Fund of a Lender;

(iii) <u>from Borrower or Agent</u> for an assignment by SWK, as a Lender, to any Person for which SWK Advisors LLC acts as an investment advisor (or any similar type of representation or agency) pursuant to a written agreement, but SWK will give written notice to Borrower of any such assignment;

(iv) <u>from Borrower or Agent</u> for an assignment by a Lender of its Loans and its Note as collateral security to a Federal Reserve Bank or, as applicable, to such Lender's trustee for the benefit of its investors (but no such assignment shall release any Lender from any of its obligations hereunder); or

(v) from Borrower, Agent or any Lender for (A) the assignment of SWK's Loans and Commitments to a Permitted Assignee (as defined below) or (B) a collateral assignment by SWK of, and the grant by SWK of a security interest in, all of SWK's right, title and interest in, to and under each of the Loan Documents, including, without limitation, all of SWK's rights and interests in, to and under this Agreement, the Obligations and the Collateral (collectively, the "Assigned Rights"), to a Permitted Assignee, provided that no such collateral assignment shall release SWK from any of its obligations under any of the Loan Documents. In connection with any enforcement of or foreclosure upon its security interests in any of the Assigned Rights, a Permitted Assignee, upon notice to Borrower, SWK and the other Lenders, shall be entitled to substitute itself, or its designee, for SWK as a Lender under this Agreement. For purposes hereof, the term "Permitted Assignee" shall mean any lender to or funding source of SWK or its Affiliate, together with its successors, assigns or designees (including, without limitation, any purchaser or other assignee of the Assigned Rights from such Person). Effective immediately upon the replacement of SWK as a Lender under this Agreement by a Permitted Assignee with this <u>clause (v)</u>.

- 62 -

SWK shall automatically be deemed to have resigned as Agent pursuant to Section 9.9 of this Agreement (without the need for Agent giving advance written notice of such resignation as required pursuant to such Section 9.9), and Required Lenders shall appoint a successor Agent in accordance with Section 9.9 of this Agreement.

(b) From and after the date on which the conditions described above have been met, (i) such Assignee shall be deemed automatically to have become a party hereto and, to the extent that rights and obligations hereunder have been assigned to such Assignee pursuant to such Assignment Agreement, shall have the rights and obligations of a Lender hereunder and (ii) the assigning Lender, to the extent that rights and obligations hereunder have been assigned by it pursuant to such Assignment Agreement, shall be released from its rights (other than its indemnification rights) and obligations hereunder. Upon the request of the Assignee (and, as applicable, the assigning Lender) pursuant to an effective Assignment Agreement, Borrower shall execute and deliver to Agent for delivery to the Assignee (and, as applicable, the assigning Lender) a Note in the principal amount of the Assignee's Pro Rata Term Loan Share (and, as applicable, a Note in the principal amount of the Pro Rata Term Loan Share retained by the assigning Lender). Each such Note shall be date of such assignment. Upon receipt by the assigning Lender of such Note, the assigning Lender shall return to Borrower any prior Note held by it.

(c) Agent, acting solely for this purpose as an agent of Borrower, shall maintain at one of its offices in the United States a copy of each Assignment Agreement delivered to it and a register for the recordation of the names and addresses of each Lender, and the Commitments of, and principal amount of the Loans owing to, such Lender pursuant to the terms hereof. The entries in such register shall be, in the absence of manifest error, conclusive, and Borrower, Agent and Lenders may treat each Person whose name is recorded therein pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. Such register shall be available for inspection by Borrower and any Lender, at any reasonable time upon reasonable prior notice to Agent.

(d) Notwithstanding the foregoing provisions of this Section 10.8.1 or any other provision of this Agreement, any Lender may at any time assign all or any portion of its Loans and its Note (i) as collateral security to a Federal Reserve Bank or, as applicable, to such Lender's trustee for the benefit of its investors (but no such assignment shall release any Lender from any of its obligations hereunder) and (ii) to (w) an Affiliate of such Lender which is at least fifty percent (50%) owned (directly or indirectly) by such Lender or by its direct or indirect parent company, (x) its direct or indirect parent company, (y) to one or more other Lenders or (z) to an Approved Fund.

10.9 Participations.

Any Lender may at any time sell to one or more Persons participating interests in its Loans, Commitments or other interests hereunder (any such Person, a "Participant"). In the event of a sale by a Lender of a participating interest to a Participant, (a) such Lender's obligations hereunder shall remain unchanged for all purposes, (b) Borrower and Agent shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations hereunder and (c) all amounts payable by Borrower shall be determined as if such Lender had not sold such participation and shall be paid directly to such Lender. No Participant shall have any direct or indirect voting rights hereunder except with respect to any event described in Section 10.1 expressly requiring the unanimous vote of all Lenders or, as applicable, all affected Lenders. Each Lender agrees to incorporate the requirements of the preceding sentence into each participation agreement which such Lender enters into with any Participant. Borrower agrees, to the fullest extent permitted by applicable law, that if amounts outstanding under this Agreement are due and payable (as a result of acceleration or otherwise), each Participant shall be deemed to have the right of set-off in respect of its participating interest in amounts owing under this Agreement to the same extent as if the amount of its participating interest were owing directly to it as a Lender under this Agreement; *provided* that such right of set-off shall be subject to the obligation of each Participant to share with Lenders, and Lenders agree to share with each Participant, as provided in <u>Section 2.10.4</u>. Borrower also agrees that each Participant shall

- 63 -

be entitled to the benefits of Section 3 as if it were a Lender (provided that a Participant shall not be entitled to such benefits unless such Participant agrees, for the benefit of Borrower, to comply with the documentation requirements of Section 3.1(c) as if it were a Lender and complies with such requirements, and provided, further, that no Participant shall receive any greater compensation pursuant to Section 3 than would have been paid to the participating Lender if no participation had been sold). Any such Lender transferring a participation shall, as an agent for Borrower, maintain in the United States a register to record the names, address, and interest, principal and other amounts owing to, each Participant. The entries in such register shall be, in the absence of manifest error, conclusive, and Borrower, Agent and the Lenders may treat each Person whose name is recorded therein pursuant to the terms hereof as a Participant hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. Such participation register shall be available for inspection by the Agent or Borrower, at any reasonable time upon reasonable prior written notice from Agent or Borrower.

10.10 Confidentiality.

Borrower, Agent and each Lender agree to use commercially reasonable efforts (equivalent to the efforts Borrower, Agent or such Lender applies to maintain the confidentiality of its own confidential information) to maintain as confidential all information (including, without limitation, any information provided by Borrower pursuant to Sections 6.1, 6.2 and 6.9) provided to them by any other party hereto and/or any other Loan Party, as applicable, except that Agent and each Lender may disclose such information (a) to Persons employed or engaged by Agent or such Lender or any of their Affiliates (including collateral managers of Lenders) in evaluating, approving, structuring or administering the Loans and the Commitments (provided that such Persons have been informed of the covenants contained in this Section 10.10); (b) to any assignee, funding source of Agent or any Lender, or participant or potential assignee or participant that has agreed to comply with the covenants contained in this Section 10.10 (and any such assignee or participant or potential assignee or participant may disclose such information to Persons employed or engaged by them as described in clause (a) above); (c) as required or requested by any federal or state regulatory authority or examiner, or any insurance industry association, or as reasonably believed by Agent or such Lender to be compelled by any court decree, subpoena or legal or administrative order or process; (d) as, on the advice of Agent's or such Lender's counsel, is required by law; (e) in connection with the exercise of any right or remedy under the Loan Documents or in connection with any litigation to which Agent or such Lender is a party; (f) to any nationally recognized rating agency or investor of a Lender that requires access to information about a Lender's investment portfolio in connection with ratings issued or investment decisions with respect to such Lender; (g) that ceases to be confidential through no fault of Agent or any Lender; (h) to a Person that is an investor or prospective investor in a Securitization that agrees that its access to information regarding Borrower and the Loans and Commitments is solely for purposes of evaluating an investment in such Securitization and who agrees to treat such information as confidential; or (i) to a Person that is a trustee, collateral manager, servicer, noteholder or secured party in a Securitization in connection with the administration, servicing and reporting on the assets serving as collateral for such Securitization. For purposes of this Section, "Securitization" means a public or private offering by a Lender or any of its Affiliates or their respective successors and assigns, of securities which represent an interest in, or which are collateralized, in whole or in part, by the Loans or the Commitments. In each case described in clauses (c), (d) and (e) (as such disclosure in clause (e) pertains to litigation only), where the Agent or Lender, as applicable, is compelled to disclose a Loan Party's confidential information, promptly after such disclosure the Agent or such Lender, as applicable, shall notify Borrower of such disclosure provided, however, that neither the Agent nor any Lender shall be required to notify Borrower of any such disclosure (i) to any federal or state banking regulatory authority conducting an examination of the Agent or such Lender, or (ii) to the extent that it is legally prohibited from so notifying Borrower. Notwithstanding the foregoing, Agent reserves the right to provide to industry trade organizations information necessary and customary for inclusion in league table measurements.

10.11 Captions.

- 64 -

Captions used in this Agreement are for convenience only and shall not affect the construction of this Agreement.

10.12 Nature of Remedies.

All Obligations of Borrower and rights of Agent and Lenders expressed herein or in any other Loan Document shall be in addition to and not in limitation of those provided by applicable law. No failure to exercise and no delay in exercising, on the part of Agent or any Lender, any right, remedy, power or privilege hereunder, shall operate as a waiver thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

10.13 Counterparts; Electronic Signatures.

This Agreement and the other Loan Documents may be executed in counterparts with the same effect as if all parties had executed the same document. All counterparts shall be construed together and shall constitute a single agreement. Further, the parties hereto consent and agree that this Agreement and the other Loan Documents may be signed and/or transmitted by e-mail of any .pdf file, .jpeg file, or any other electronic or image file, or any "electronic signature" as defined under the U.S. Electronic Signatures in Global and National Commerce Act or the New York Electronic Signatures and Records Act, which includes any electronic signature provided using Orbit, Adobe Sign, DocuSign, or any other similar platform identified by the parties hereto and reasonably available at no undue burden or expense to the Agent), except to the extent the Agent requires otherwise. Any such electronic signatures shall be valid, effective and legally binding as if such electronic signatures were handwritten signatures and shall be deemed to have been duly and validly delivered for all purposes hereunder. No party hereto shall raise the use of e-mail or other electronic transmission as a defense to the formation or enforceability of a contract and each such party forever waives any such defense.

10.14 <u>Severability.</u>

The illegality or unenforceability of any provision of this Agreement or any instrument or agreement required hereunder shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Agreement or any instrument or agreement required hereunder.

10.15 Entire Agreement.

This Agreement, together with the other Loan Documents, embodies the entire agreement and understanding among the parties hereto and supersedes all prior or contemporaneous agreements and understandings of such Persons, verbal or written, relating to the subject matter hereof and thereof.

10.16 Successors; Assigns.

This Agreement shall be binding upon Borrower, Lenders and Agent and their respective successors and assigns, and shall inure to the benefit of Borrower, Lenders and Agent and the successors and assigns of Lenders and Agent. No other Person shall be a direct or indirect legal beneficiary of, or have any direct or indirect cause of action or claim in connection with, this Agreement or any of the other Loan Documents. Borrower may not assign or transfer any of its rights or Obligations under this Agreement without the prior written consent of Agent and each Lender.

10.17 Governing Law.

- 65 -

THIS AGREEMENT AND EACH NOTE SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE.

10.18 Forum Selection; Consent to Jurisdiction.

ANY LITIGATION BASED HEREON, OR ARISING OUT OF, UNDER, OR IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER LOAN DOCUMENT, SHALL BE BROUGHT AND MAINTAINED EXCLUSIVELY IN NEW YORK; PROVIDED THAT ANY SUIT SEEKING ENFORCEMENT AGAINST ANY COLLATERAL OR OTHER PROPERTY MAY BE BROUGHT, AT AGENT'S OPTION, IN THE COURTS OF ANY JURISDICTION WHERE SUCH COLLATERAL OR OTHER PROPERTY MAY BE FOUND. EACH PARTY HEREBY EXPRESSLY, VOLUNTARILY, AND IRREVOCABLY SUBMITS ITSELF EXCLUSIVELY TO PERSONAL JURISDICTION AND VENUE IN THE DISTRICT COURT OF DALLAS COUNTY, TEXAS AND IN THE UNITED STATES DISTRICT COURT FOR THE NORTHERN DISTRICT OF TEXAS, DALLAS DIVISION, FOR THE PURPOSE OF ANY SUCH LITIGATION AS SET FORTH ABOVE; SUCH COURTS SHALL BE THE EXCLUSIVE PROPER VENUE FOR THE PURPOSE OF ANY SUCH LITIGATION AS SET FORTH ABOVE. EACH PARTY HEREBY EXPRESSLY, VOLUNTARILY, AND IRREVOCABLY WAIVES, TO THE FULLEST EXTENT PERMITTED UNDER APPLICABLE LAW, ANY OBJECTION WHICH IT MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY SUCH LITIGATION BROUGHT IN ANY SUCH COURT REFERRED TO ABOVE AND ANY CLAIM THAT ANY SUCH LITIGATION HAS BEEN BROUGHT IN AN INCONVENIENT FORUM. EACH PARTY FURTHER EXPRESSLY, VOLUNTARILY, AND IRREVOCABLY CONSENTS TO SERVICE OF PROCESS RELATED TO ANY SUCH LITIGATION AS SET FORTH ABOVE BY FEDERAL EXPRESS OR REGISTERED/CERTIFIED MAIL SENT TO THE APPLICABLE PARTY AT such party's address set forth beneath its signature on the signature page to this Agreement OR AT SUCH OTHER ADDRESS AS SUCH PARTY MAY, BY WRITTEN NOTICE RECEIVED BY THE OTHER PARTIES, HAVE DESIGNATED AS ITS ADDRESS. THE PARTIES AGREE THAT THESE METHODS FOR SERVICE OF PROCESS ARE VALID FOR PURPOSES OF EFFECTING SERVICE OF PROCESS, AS THEY ARE EFFICIENT AND COST-EFFECTIVE ALTERNATIVES TO FORMAL SERVICE OF PROCESS (THE PARTIES MAY EFFECT SERVICE OF PROCESS IN ANY OTHER METHOD ALLOWED UNDER THE LAW IF THEY SO CHOOSE).

10.19 Waiver of Jury Trial.

EACH OF BORROWER, AGENT AND EACH LENDER, TO THE FULLEST EXTENT PERMITTED UNDER APPLICABLE LAW, HEREBY EXPRESSLY, VOLUNTARILY, AND IRREVOCABLY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHTS UNDER THIS AGREEMENT, ANY NOTE, ANY OTHER LOAN DOCUMENT AND ANY AMENDMENT, INSTRUMENT, DOCUMENT OR AGREEMENT DELIVERED OR WHICH MAY IN THE FUTURE BE DELIVERED IN CONNECTION HEREWITH OR THEREWITH OR ARISING FROM ANY LENDING RELATIONSHIP EXISTING IN CONNECTION WITH ANY OF THE FOREGOING, AND AGREES THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

10.20 Patriot Act.

Each Lender that is subject to the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the 'Patriot Act"), and Agent (for itself and not on behalf of any Lender), hereby notifies each Loan Party that, pursuant to the requirements of the Patriot Act, such Lender and Agent are required to obtain, verify and record information that identifies each Loan Party, which information includes the name and address of each Loan Party and other information that will allow such Lender or Agent, as applicable, to identify each Loan Party in accordance with the Patriot Act.

- 66 -

10.21 Independent Nature of Relationship.

Nothing herein contained shall constitute any Loan Party and SWK as a partnership, an association, a joint venture or any other kind of entity or legal form or constitute any party the agent of the other. No party shall hold itself out contrary to the terms of this <u>Section 10.21</u> and no party shall become liable by any representation, act or omission of the other contrary to the provisions hereof. No Loan Party, Lender, nor SWK has any fiduciary or other special relationship with the other party hereto or any of its Affiliates. The Loan Parties and SWK agree that SWK is not involved in or responsible for the manufacture, marketing or sale of any Product or the provision of any Service.

[Remainder of page intentionally blank; signature pages follow]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered by their duly authorized officers as of the date first set forth above.

BORROWER:

JOURNEY MEDICAL CORPORATION, a Delaware corporation

By: <u>/s/ Claude Maraoui</u> Name: Claude Maraoui Title: Chief Executive Officer

Address: 9237 E. Via de Ventura Blvd. Suite 105 Scottsdale, AZ 85258 Email: [***] With a copy to:

Cooley LLP 1299 Pennsylvania Ave NW Suite 700 Washington, DC 20004 Attention: Mike Tollini Email: [***]

AGENT AND LENDER:

SWK FUNDING LLC, a Delaware limited liability company, as Agent and a Lender

By: SWK Holdings Corporation, a Delaware corporation, its sole Manager

By: /s/ Joe D. Staggs Name: Joe D. Staggs Title: Chief Executive Officer

Address:

SWK Funding LLC 5956 Sherry Lane, Suite 650 Dallas, Texas 75225 Email: [***]

With a copy to:

Holland & Knight LLP One Arts Plaza 1722 Routh Street, Suite 1500 Dallas, Texas 75201 Attention: Ryan Magee Email: [***]

ANNEX I

Commitments and Pro Rata Term Loan Shares

[omitted]

EXHIBIT A

Form of Assignment Agreement

[omitted]

EXHIBIT B

Form of Compliance Certificate

[omitted]

EXHIBIT C

Form of Note

PROMISSORY NOTE

\$[●]

[●], 20[●]

FOR VALUE RECEIVED and pursuant to the terms of this PROMISSORY NOTE (as amended, restated, supplemented, or otherwise modified from time to time, this "Note"), the undersigned, JOURNEY MEDICAL CORPORATION, a Delaware corporation ("Borrower"), promises to pay to [•] (together with all subsequent registered holders of this Note being hereinafter referred to collectively, as "Holder"), at the offices of SWK FUNDING LLC, a Delaware limited liability company, as agent (in such capacity, together with its successors and assigns, the "Agent"), on behalf of Holder and the other Lenders (defined below), having an address at 5956 Sherry Lane, Suite 650, Dallas, Texas 75225, or at such other place as Holder hereof may designate in writing, the principal sum of up to [•] **DOLLARS (§[•])**, or such lesser amount as may be advanced by Holder pursuant to that certain Credit Agreement, dated as of December 27, 2023 (as amended, restated, supplemented, or otherwise modified from time to time, the "Credit Agreement"), among Borrower, the financial institutions party thereto from time to time (each a "Lender" and collectively, the "Lenders"), and Agent, together with all applicable interest and fees, the Loans under the Credit Agreement made by Lenders to Borrower pursuant to the Credit Agreement.

DEFINITIONS

Capitalized terms not otherwise defined herein shall have the meanings set forth in the Credit Agreement.

PRINCIPAL AND INTEREST

Principal. Borrower shall make payments on the principal balance of this Note and accrued interest on the principal balance of this Note in accordance with the provisions of the Credit Agreement. If not sooner paid, the entire unpaid principal balance of this Note and all interest thereon shall be paid on the Term Loan Maturity Date.

Interest. Interest on the unpaid balance of this Note will accrue from the date of this Note until final payment thereof in accordance with the applicable provisions of the Credit Agreement.

Prepayments. Borrower may prepay the principal sum outstanding from time to time hereunder as provided in the Credit Agreement, subject to any prepayment premium set forth in the Credit Agreement.

INCORPORATION OF CREDIT AGREEMENT

This Note has been issued pursuant to the Credit Agreement, and all of the terms, covenants and conditions of the Credit Agreement (including all Exhibits and Schedules thereto) and all other instruments evidencing or securing the indebtedness hereunder are hereby made a part of this Note and are deemed incorporated herein in full.

EVENTS OF DEFAULT

Upon the occurrence and during the continuance of an Event of Default, the Holder shall have the rights and remedies set forth in the Credit Agreement and the other Loan Documents, in addition to any other remedies to which the Holder may be entitled.

LAWFUL LIMITS

All agreements between Borrower and Holder are expressly limited so that in no contingency or event whatsoever, whether by reason of advancement of the proceeds hereof, acceleration of maturity of the unpaid principal balance hereof, or otherwise, shall the amount paid or agreed to be paid to Holder for the use, forbearance or detention of the money to be advanced hereunder exceed the highest lawful rate permissible under applicable usury laws. If, from any circumstances whatsoever, fulfillment of any provision hereof, of the Credit Agreement or of any other Loan Documents shall involve transcending the limit of validity prescribed by any law which a court of competent jurisdiction may deem applicable hereto, then, ipso facto, the obligation to be fulfilled shall be reduced to the limit of such validity, and, if from any circumstance Holder shall ever receive as interest an amount which would exceed the highest lawful rate, such amount which would be excessive interest shall be applied to the reduction of the unpaid principal balance due hereunder and not to the payment of interest. This provision shall control every other provision of all agreements between Borrower and Holder.

To the extent that either Chapter 303 or 306, or both, of the Texas Finance Code, as amended from time to time, apply in determining the Maximum Lawful Rate notwithstanding that the parties have chosen the laws of the State of New York (or applicable United States federal law to the extent that it permits Holder to contract for, charge, take, receive or reserve a greater amount of interest than the laws of the State of New York) to govern and control in the enforcement, interpretation and construction of the Loan Documents generally, Holder hereby elects to determine the applicable rate ceiling by using the weekly ceiling from time to time in effect, subject to Holder's right from time to time to change such method in accordance with applicable law, as the same may be amended or modified from time to time, to utilize any other method of establishing the Maximum Lawful Rate under the Texas Finance Code or under other applicable law by giving notice, if required, to Borrower as provided by applicable law now or hereafter in effect. To the extent United States federal law instead of applicable state law for the purpose of determining the Maximum Lawful Rate. As used herein, (*x*) the term "**Maximum Lawful Rate**" shall mean the maximum lawful rate of interest which may be contracted for, charged, taken, received or reserved by Holder in accordance with the applicable law (or applicable United States federal law (or applicable state law), taking into account all Charges made in connection with the transaction evidenced by the Note and the other Loan Documents, and (*y*) the term "**Charges**" shall mean all fees, charges and/or any other things of value, if any, contracted for, charged, taken or reserved by Holder in connection with the transactions relating to the Loan Agreement, the Note and the other Loan Documents, which are treated as interest under applicable law.

MISCELLANEOUS

WAIVERS. PRESENTMENT FOR PAYMENT, NOTICE OF NONPAYMENT OR DISHONOR, PROTEST, NOTICE OF PROTEST, DEMAND, NOTICE OF ACCELERATION OR INTENT TO ACCELERATE AND ALL OTHER NOTICES IN

Exhibit C-2

CONNECTION WITH THE DELIVERY, ACCEPTANCE, PERFORMANCE, DEFAULT OR ENFORCEMENT OF THIS NOTE ARE HEREBY IRREVOCABLY WAIVED BY BORROWER.

Exercise of Remedies. No delay on the part of Agent or Holder in the exercise of any right, power or remedy hereunder, under the Credit Agreement or under any other Loan Document shall operate as a waiver thereof, nor shall any single or partial exercise by Agent or Holder of any right, power or remedy hereunder, under the Credit Agreement or under any other Loan Document preclude other or further exercise thereof, or the exercise of any other right, power or remedy. Upon the occurrence and continuance of an Event of Default, Agent and Holder shall at all times have the right to proceed against any portion of the Collateral in such order and in such manner as Agent and Holder may deem fit, subject to and in accordance with the Credit Agreement, Guarantee and Collateral Agreement and IP Security Agreement without waiving any rights with respect to any other security.

Invalid Provisions. The illegality or unenforceability of any provision of this Note shall not in any way affect or impair the legality or enforceability of the remaining provisions of this Note.

Governing Law. THIS NOTE SHALL BE A CONTRACT MADE UNDER AND GOVERNED BY THE INTERNAL LAWS OF THE STATE OF NEW YORK APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY WITHIN SUCH STATE, WITHOUT REGARD TO CONFLICT OF LAWS PRINCIPLES (OTHER THAN SECTION 5-1401 AND SECTION 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS CODE).

Definition of Note. All references to "Note" or "Notes" in the Loan Documents shall also include this Note, to the extent not returned to Borrower for cancellation, as the same may be amended, supplemented, modified, divided and/or restated and in effect from time to time.

New Notes. Upon Agent's written request (on behalf of Holder) Borrower shall execute and deliver to Agent new Notes and/or split or divide the Notes, or any of them, in exchange for the then existing Notes, in such smaller amounts or denominations as Agent shall specify; provided, that the aggregate principal amount of such new, split or divided Notes shall not exceed the aggregate principal amount of the Notes outstanding at the time such request is made; and provided, further, that such Notes that are replaced shall then be deemed no longer outstanding under the Credit Agreement and replaced by such new Notes and returned to Borrower within a reasonable period of time after Agent's receipt of the replacement Notes.

Replacement Notes. Upon receipt of evidence reasonably satisfactory to Borrower of the mutilation, destruction, loss or theft of any Notes and the ownership thereof, Borrower shall, upon the written request of the holder of such Notes, execute and deliver in replacement thereof new Notes in the same form, in the same original principal amount and dated the same date as the Notes so mutilated, destroyed, lost or stolen; and such Notes so mutilated, they shall be surrendered to Borrower; and if such replaced Notes have been destroyed, lost or stolen, such holder shall furnish Borrower with an indemnity in writing to indemnify, defend and save them harmless in respect of such replaced Notes.

Original Issue Discount. THE FOLLOWING INFORMATION IS SUPPLIED SOLELY FOR U.S. FEDERAL INCOME TAX PURPOSES. THIS NOTE WAS ISSUED WITH ORIGINAL ISSUE DISCOUNT ("OID") WITHIN THE MEANING OF SECTION 1273 OF THE INTERNAL REVENUE CODE OF 1986, AS AMENDED (THE "CODE"), AND THIS LEGEND IS REQUIRED BY SECTION 1275(C) OF THE CODE. HOLDERS MAY OBTAIN INFORMATION REGARDING THE AMOUNT OF OID, THE ISSUE PRICE, THE ISSUE DATE AND THE YIELD TO MATURITY RELATING TO THE NOTES BY CONTACTING THE ISSUER AT [.]

[Remainder of page intentionally blank; signature page follows].

IN WITNESS WHEREOF, the undersigned has caused this Promissory Note to be executed as of the day and year first written above.

BORROWER:

JOURNEY MEDICAL CORPORATION,

a Delaware corporation, as Borrower

By:	
Name:	
Title: []	

JOURNEY MEDICAL CORPORATION

List of Subsidiaries

Subsidiaries of Journey Medical Corporation at December 31, 2023, with jurisdiction of incorporation or formation:

• JG Pharma Inc. (Delaware)

Consent of Independent Registered Public Accounting Firm

We consent to the incorporation by reference in the registration statements (No. 333-263888, No. 333-266125, and No. 333-276080) on Form S-8 and (No. 333-269079) on Form S-3 of our report dated March 28, 2024, with respect to the consolidated financial statements of Journey Medical Corporation.

/s/ KPMG LLP

Short Hills, New Jersey

March 28, 2024

CERTIFICATION PURSUANT TO

SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Claude Maraoui certify that:

(1) I have reviewed this Annual Report on Form 10-K for the year ended December 31, 2023 of Journey Medical Corporation. (the registrant);

(2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

(3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects, the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

(4) The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

(5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: March 28, 2024

By: /s/ Claude Maraoui

Claude Maraoui President, Chief Executive Officer and Director Principal Executive Officer

CERTIFICATION PURSUANT TO

SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Joseph Benesch, certify that:

(1) I have reviewed this Annual Report on Form 10-K for the year ended December 31, 2023 of Journey Medical Corporation (the registrant);

(2) Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;

(3) Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects, the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;

(4) The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:

(a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;

(b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;

(c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and

(d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

(5) The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

(a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and

(b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Dated: March 28, 2024

By: /s/ Joseph Benesch Joseph Benesch Interim Chief Financial O

Interim Chief Financial Officer Principal Financial Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report on Form 10-K of Journey Medical Corporation (the "Company") for the period ended December 31, 2023, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Claude Maraoui, President and Chief Executive Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company as of, and for, the periods presented in the Report.

Dated: March 28, 2024

By: /s/ Claude Maraoui Claude Maraoui

President, Chief Executive Officer and Director Principal Executive Officer

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350 AS ADOPTED PURSUANT TO

SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report on Form 10-K of Journey Medical Corporation (the "Company") for the period ended December 31, 2023, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Joseph Benesch, Principal Financial Officer of the Company, hereby certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to my knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company, as of, and for, the periods presented in the Report.

Dated: March 28, 2024

By: <u>/s/ Joseph Benesch</u> Joseph Benesch Interim Chief Financial Officer Principal Financial Officer

JOURNEY MEDICAL CORPORATION Clawback Policy

Effective as of October 1, 2023

The Board of Directors ("**Board**") of Journey Medical Corporation ("**Company**") believes that it is in the best interests of the Company and its shareholders to adopt this Clawback Policy ("**Policy**") which provides for the recoupment of certain executive compensation in the event of an Accounting Restatement (as defined below).

This Policy is designed to comply with, and shall be interpreted to be consistent with, Section 10D of the Securities Exchange Act of 1934, as amended ("**Exchange Act**"), and final rules and amendments adopted by the Securities and Exchange Commission ("**SEC**") to implement the aforementioned legislation, and Rule 5608 of the Nasdaq Stock Exchange's listing standards.

This policy shall be effective as of October 2, 2023, the Effective Date of Rule 5608 of the Nasdaq Stock Exchange's listing standards (the "**Effective Date**") and applies to all Covered Officers (as defined below) of Journey Medical Corporation.

Administration

This Policy shall be administered by the Compensation Committee of the Board (if composed entirely of independent directors) or if so designated by the Board, a separate committee of the Board, consisting of a majority of the independent directors serving on the board (as applicable, the "Administrator"). The Administrator is authorized to interpret and construe this Policy and to make all determinations necessary, appropriate or advisable for the administration of this Policy. Any determinations made by the Administrator shall be final and binding on all affected individuals and need not be uniform with respect to each individual covered by the Policy. In the administration of this Policy, the Administrator is authorized and directed to consult with the full Board or such other committees of the Board, such as the Audit Committee or the Compensation Committee, as may be necessary or appropriate as to matters within the scope of such other committee's responsibility and authority.

Subject to any limitation under applicable law, the Administrator may authorize and empower any officer or employee of the Company to take any and all actions necessary or appropriate to carry out the purpose and intent of this Policy (other than with respect to any recovery under this Policy involving such officer or employee).

Definitions

For purposes of this Policy, the following definitions will apply:

"Accounting Restatement" means an accounting restatement of the Company's financial statements due to the Company's material noncompliance with any financial reporting requirement under the securities laws, including those that either (a) correct an error in a previously issued financial statement that is material to such previously issued financial statement or (b) correct an error that is not material to a previously issued financial statement but would result in a material misstatement if left uncorrected in a current report or the error correction was not recognized in the current period.

"Administrator" has the meaning set forth in the "Administration" section above.

"Board" means the Company's Board of Directors.

"Clawback Exception" has the meaning ascribed to such term in the "Clawback Exceptions" section below.

"Covered Officer" means the Company's officers for purposes of Section 16 under the Exchange Act during any portion of the performance period of the Incentive-Based Compensation.

"Excess Compensation" means any amount of Incentive-Based Compensation Received by a Covered Officer that exceeds the amount of Incentive-Based Compensation that otherwise would have been received had it been determined based on the restated financial information or properly calculated financial measure. Excess Compensation shall be calculated on a pre-tax basis.

"Incentive-Based Compensation" means any non-equity incentive plan awards, bonuses paid from a bonus pool, cash awards, equity or equity-based awards, or proceeds received upon sale of shares acquired through an incentive plan; provided that such compensation is granted, earned, and/or vested based wholly or in part on the attainment of a financial performance measure, as determined in accordance with Section 10D of the Exchange Act and the Nasdaq Stock Exchange listing standards (the "Clawback Rules"). Incentive-Based Compensation does not include any salaries, discretionary bonuses, non-equity incentive plan awards earned upon satisfying a strategic measure or operational measure (e.g., completion of a project), or equity-based awards that are not contingent on achieving any financial reporting measure (e.g., time vested stock options, restricted stock or restricted stock units).

"Look-Back Period" means the three (3) completed fiscal years immediately preceding the earlier of the date on which (a) the Board or appropriate committee concludes, or reasonably should have concluded, that an Accounting Restatement is required or (b) a regulator directs an Accounting Restatement.

"**Received**" means any Incentive-Based Compensation that is received during the fiscal year in which the applicable financial reporting measure upon which the payment is based is achieved, even if payment or grant of the Incentive-Based Compensation occurs after the end of such period.

Clawback Due to Accounting Restatement

In the event the Company is required to prepare an Accounting Restatement, the Administrator shall require reimbursement or forfeiture ("clawback") of any Excess Compensation Received by any Covered Officer (current or former) during the applicable Look-Back Period, regardless of whether the Covered Officer engaged in misconduct or was otherwise directly or indirectly responsible, in whole or in part, for the Accounting Restatement.

In the event the Administrator cannot determine the Excess Compensation from the information in the Accounting Restatement or from the recalculated financial measure, then it will make its determination based on a reasonable estimate of the effect of the Accounting Restatement or recalculation. Such determination will be final and binding.

If a Clawback Exception applies with respect to a Covered Officer, the Company may forgo the recovery described in this Section from such Covered Officer.

Clawback Method

The Administrator may determine, in its sole discretion, the method for the clawback of any amounts due under this Policy, which may include without limitation direct payment from the Covered

Officer, recovery over time, the forfeiture or reduction of future pay or awards, or any other method that will provide for recovery within a reasonable manner and without undue delay. The Company may enter into deferred payment plans with Covered Officers to effectuate clawback to avoid unreasonable economic hardship. Any amounts due under this Policy may be deducted as an offset from amounts due to the Covered Officer from the Company, except to the extent such set-off is prohibited by law or would violate Section 409A of the Internal Revenue Code of 1986, as amended and the regulations thereunder.

Clawback Exceptions

The Company will be required, in the event of an Accounting Restatement, to recover all Excess Compensation received by a Covered Officer during the Look-Back Period unless: (i) one of the following conditions is met; and (ii) the Committee has made a determination that recovery would be impracticable in accordance with Rule 10D-1 of the Exchange Act:

- the direct expense paid to a third party to assist in enforcing this Policy would exceed the amount to be recovered (and the Company has already made a reasonable attempt to recover such erroneously awarded Excess Compensation from such Covered Officer, has documented such reasonable attempt(s) to recover, and has provided such documentation to the Nasdaq Stock Exchange);
- (ii) recovery would violate home country laws that existed at the time of adoption of the rule and the Company receives an opinion of counsel to that effect; or
- (iii) recovery would likely cause an otherwise tax-qualified retirement plan, under which benefits are broadly available to employees of the Company, to fail to meet the requirements of Section 401(a)(13) or Section 411(a) of the Internal Revenue Code and regulations thereunder. For purposes of clarity, this Clawback Exception only applies to taxqualified retirement plans and does not apply to other plans, including long term disability, life insurance, and supplemental executive retirement plans, or any other compensation that is based on Incentive-Based Compensation in such plans, such as earnings accrued on notional amounts of Incentive-Based Compensation contributed to such plans.

General

The Company shall not indemnify any Covered Officer against the loss of any covered compensation as a result of the application of this Policy.

This Policy is in addition to (and not in lieu of) any right of repayment, forfeiture or right of offset against any employees that is required pursuant to any statutory repayment requirement (regardless of whether implemented at any time prior to or following the adoption or amendment of this Policy), including Section 304 of the Sarbanes-Oxley Act of 2002. Any amounts paid to the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 shall be considered in determining any amounts recovered under this Policy.

The terms of this Policy shall be binding and enforceable against all Covered Officers subject to this Policy and their beneficiaries, heirs, executors, or other legal representatives. If any provision of this Policy or the application of such provision to any Covered Officer shall be adjudicated to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provisions of this Policy, and the invalid, illegal or unenforceable provisions shall be deemed amended to

the minimum extent necessary to render any such provision (or the application of such provision) valid, legal or enforceable.

Each Covered Officer shall sign and return to the Company, within the later of: (i) 60 calendar days following the Effective Date or (ii) 30 calendar days following the date the individual becomes a Covered Officer, the Acknowledgement and Agreement Form attached hereto as Exhibit A, pursuant to which the Covered Officer agrees to be bound by, and to comply with, the terms and conditions of this Policy.

To the extent the Clawback Rules require recovery of Incentive-Based Compensation in additional circumstances beyond those specified above, nothing in this Policy shall be deemed to limit or restrict the right or obligation of the Company to recover Incentive-Based Compensation to the fullest extent required by the Clawback Rules.

The Board may amend this Policy from time-to-time in its discretion and as necessary to comply with any rules or standards adopted by the SEC and the listings standards of any national securities exchange on which the Company's securities are listed.

<u>Exhibit A</u>

Journey Medical Corporation (the "Company") Clawback Policy

Acknowledgement and Agreement Form

I, the undersigned, acknowledge and agree that I have received and reviewed the Clawback Policy of Journey Medical Corporation (the "**Policy**"), effective as of October 1, 2023, as adopted by the Company's Board of Directors

Furthermore, I acknowledge and agree:

- that I am fully bound by, and subject to, all of the terms and conditions of the Policy, as may be amended, restated, supplemented or otherwise modified from time to time.
- that I have been designated as a "Covered Officer" as defined in the policy.
- that my execution of this Acknowledgement and Agreement Form is in consideration of, and is a condition to, my continued employment (if currently an employee) and my receipt of future awards from the Company, though nothing in this Acknowledgement and Agreement Form shall obligate the Company to make any particular award.

In the event of any inconsistency between the Policy and the terms of any employment agreement to which I am a party, or to the terms of any compensation plan, program, agreement or arrangement under which any incentive-based compensation covered by the Policy is payable, the terms of this Policy shall govern and shall be deemed incorporated into all such plans, programs, agreements (including any employment agreements) or arrangements, including and without limitation, those granted or awarded prior to the date hereof and those granted or awarded in the future.

In the event any Incentive-Based Compensation (as defined in the Policy) is subject to recoupment or recovery under the terms of the Policy, I will promptly take any action necessary to effectuate the recoupment or recovery of such compensation by the Company.

COVERED OFFICER

Signature	-
Print	-
Name	-
Date	-